UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended December 31, 2017

or

to

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File Number: 001-35198

Pandora Media, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 2101 Webster Street, Suite 1650 Oakland, CA

(Address of principal executive offices)

(510) 451-4100 (Registrant's telephone number, including area code)

Securities registered pursuant to section 12(g) of the Act:

Title of each class

Common stock, \$0.0001 par value

Name of each exchange on which registered

94-3352630

(I.R.S. Employer

Identification No.)

94612

(Zip Code)

The New York Stock Exchange

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes Z No 🗆

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No 🗷

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No \square

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§ 229.405 of this chapter) is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by a check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or emerging growth company. See definition of "large accelerated filer," "accelerated filer", "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer ⊠ Non-accelerated filer □ (Do not check if a smaller reporting company) Accelerated filer Smaller reporting company Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗷

The aggregate market value of the voting common stock held by non-affiliates of the registrant as offune 30, 2017 (the last business day of the registrant's most recently completed second quarter), based on the closing price of such stock on The New York Stock Exchange on such date was approximately \$1,189 million. This calculation excludes the shares of common stock held by executive officers, directors and stockholders whose ownership exceeds 5% outstanding at June 30, 2017. This calculation does not reflect a determination that such persons are affiliates for any other purposes.

On February 21, 2018 the registrant had 254,884,000 shares of common stock outstanding.

DOCUMENTS INCORPORATED BY REFERENCE

Portions of the Proxy Statement for the registrant's 2018 Annual Meeting of Stockholders (the "Proxy Statement"), to be filed within 120 days of the end of the fiscal year ended December 31, 2017, are incorporated by reference in Part III hereof. Except with respect to information specifically incorporated by reference in this Annual Report on Form 10-K, the Definitive Proxy Statement is not deemed to be filed as part of this Annual Report on Form 10-K.

Pandora Media, Inc. Form 10-K Table of Contents

Page No.

	PART I	
Item 1	Business	<u>2</u>
Item 1A	Risk Factors	<u>11</u>
Item 1B	Unresolved Staff Comments	<u>35</u>
Item 2	Properties	<u>35</u>
Item 3	Legal Proceedings	<u>35</u>
Item 4	Mine Safety Disclosures	<u>35</u>
	PART II	
Item 5	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	<u>36</u>
Item 6	Selected Financial Data	<u>38</u>
<u>Item 7</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>40</u>
Item 7A	Quantitative and Qualitative Disclosures About Market Risk	<u>61</u>
Item 8	Financial Statements and Supplementary Data	<u>62</u>
Item 9	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	<u>106</u>
Item 9A	Controls and Procedures	<u>106</u>
Item 9B	Other Information	<u>107</u>
	PART III	
<u>Item 10</u>	Directors, Executive Officers and Corporate Governance	<u>108</u>
<u>Item 11</u>	Executive Compensation	<u>108</u>
<u>Item 12</u>	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	<u>108</u>
<u>Item 13</u>	Certain Relationships and Related Transactions and Director Independence	<u>108</u>
<u>Item 14</u>	Principal Accountant Fees and Services	<u>108</u>
	PART IV	
<u>Item 15</u>	Exhibits, Financial Statement Schedules	<u>109</u>
Signatures 1 1		<u>110</u>

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS AND INDUSTRY DATA

This Annual Report on Form 10-K contains "forward-looking statements" that involve substantial risks and uncertainties. The statements contained in this Annual Report on Form 10-K that are not purely historical are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), including, but not limited to, statements regarding our expectations, beliefs, intentions, strategies, future operations, future financial position, future revenue, projected expenses and plans and objectives of management. In some cases, you can identify forward-looking statements by terms such as "anticipate," "believe," "estimate," "expect," "intend," "may," "might," "plan," "project," "will," "would," "should," "could," "can," "predict," "potential," "continue," "objective," or the negative of these terms, and similar expressions intended to identify forward-looking statements. However, not all forward-looking statements contain these identifying words. These forward-looking statements reflect our current views about future events and involve known risks, uncertainties and other factors that may cause our actual results, levels of activity, performance or achievement to be materially different from those expressed or implied by the forward-looking statements. Factors that could cause or contribute to such differences include, but are not limited to, those identified below, and those discussed in Item 1A—"Risk Factors" included in this Annual Report on Form 10-K. Furthermore, such forward-looking statements. We qualify all of our forward-looking statements by these cautionary statements. In addition, the industry in which we operate is subject to a high degree of uncertainty and risk due to a variety of factors including those described in the section entitled "Risk Factors." These and other factors could cause our results to differ materially from those expressed in this Annual Report

Some of the industry and market data contained in this Annual Report on Form 10-K are based on independent industry publications, including those generated by Triton Digital Media or "Triton," International Data Corporation or "IDC," eMarketer or other publicly available information. This information involves a number of assumptions and limitations. Although we believe that each source is reliable as of its respective date, we have not independently verified the accuracy or completeness of this information.

As used herein, "Pandora," the "Company," "we," "our," and similar terms refer to Pandora Media, Inc. and, where appropriate, its wholly owned subsidiaries, unless the context indicates otherwise.

"Pandora" and other trademarks of ours appearing in this report are our property. This report contains additional trade names and trademarks of other companies. We do not intend our use or display of other companies' trade names or trademarks to imply an endorsement or sponsorship of us by such companies, or any relationship with any of these companies.



ITEM 1. BUSINESS

Overview

Pandora—Streaming Radio and On-Demand Music Services

Pandora is the world's most powerful music discovery platform, offering a personalized experience for each of our listeners wherever and whenever they want to listen to music—whether through mobile devices, car speakers or connected devices in the home. Unlike traditional radio that broadcasts the same content at the same time to all of its listeners, we enable our listeners to create personalized stations and playlists, as well as search and play songs and albums on-demand. The Music Genome Project, our content programming algorithms and data collected from our listeners power our ability to predict listener music preferences, play music content suited to the tastes of each individual listener and introduce listeners to the music we think they will love. Founded by musicians, Pandora also empowers artists with valuable data and tools to help grow their audience and connect with their fans.

Pandora is available as an ad-supported radio service, a radio subscription service called Pandora Plus and an on-demand subscription service called Pandora Premium. The majority of our listener hours occur on mobile devices, with the majority of our revenue generated from advertising on our ad-supported radio service on these devices. With billions of data points that help us understand our users' preferences, we offer both local and national advertisers the opportunity to deliver targeted messages to our listeners using a combination of audio, display and video advertisements. We also generate increasing revenue from our subscription offerings.

For the year ended December 31, 2017, we streamed 20.61 billion hours of content, and as of December 31, 2017, we had 74.7 million active users during the trailing 30day period and 5.48 million paid subscribers. Since we first launched our ad-supported service in 2005, our listeners have created over12 billion stations.

Pandora—Ad-Supported Radio Service

Our ad-supported service allows listeners to access our catalog of music, comedy, livestreams and podcasts through our personalized playlist generating system for free across all of our delivery platforms. Our ad-supported service is valued by lean-back listeners, as it uses the Music Genome Project to instantly generate a station that plays music we think that listener will enjoy. Over time, this service has evolved by using data science to further tailor the listener experience based on listener reactions to the content we pick. Listeners also have the ability to add variety to and rename stations, which further allows for the personalization of the service. We also offer listeners on our ad-supported service the option to gain temporary access to on-demand listening experience, which includes some features of our Pandora Premium service, in exchange for viewing a video ad. We call this experience Premium Access.

Pandora Plus—Subscription Radio Service

Pandora Plus is an ad-free, paid subscription version of the Pandora radio service that also includes replays, additional skipping of songs, offline listening, higher quality audio on supported devices and longer timeout-free listening. This service is valued by listeners who want the ability to have limited interactive features such as controlling skips and replays. Similar to the ad-supported service, the more the listener interacts with the platform, the more we tailor the content we recommend to the listener. Premium Access is also available to Pandora Plus listeners.

Pandora Premium—On-Demand Subscription Service

Our on-demand subscription service, Pandora Premium, launched in the United States in April 2017. Pandora Premium combines the radio features of Pandora Plus with a unique, on-demand experience, providing users with the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button, listen to curated playlists and share playlists on social networks. Unique to Pandora, a listener can create partial playlists and have Pandora complete the playlist based on the user's listening activity using the Music Genome Project.

Beyond song delivery, listeners can discover more about the music they hear by reading the history of their favorite artists and viewing artist photos. Mobile listeners also have access to customized listener profiles which leverage data science to showcase a listener's musical identity by highlighting user data such as recent favorites, top artists of all time, playlists and thumbs. Our services also incorporate social networking features. Our music feed enables a real-time, centralized stream for

Table of Contents

listeners to view the music that their social connections are experiencing and to provide and receive recommendations for songs, albums and artists. Listeners can also share their stations, songs, albums or playlists through social media, messaging apps and email by using our share feature.

Ticketing Service

Prior to September 1, 2017, we operated a ticketing service through our former subsidiary Ticketfly, a leading live events technology company that provides ticketing and marketing software and services for clients, which are venues and event promoters, across North America. We completed the sale of Ticketfly on September 1, 2017. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

Our Strategy

Drive and Retain Active Users

It is critical that we drive and retain active users. We believe that our three-tier offering, which includes our ad-supported service, Pandora Plus and Pandora Premium, provides listeners multiple ways to enjoy Pandora. Continued product development, including more focus on interactivity and voice, more effective marketing and expanded partnerships with CE providers, content platforms and others, are important elements of our efforts to drive and retain active users. Additionally, we are actively focused on offering content beyond music, such as podcasts.

Increase Monetization Opportunities

Pandora is a significant provider of digital audio advertising and we believe there is a large opportunity to expand monetization. These improvements will primarily be enabled by new technology development and are intended to yield a combination of incremental advertising capacity, optimized pricing and increased sell through.

Focus on Execution and Efficiencies

Digital audio and music is a large and growing market, but highly competitive. It is critical that we focus on execution and efficiencies. This includes focusing resources on strategic growth areas, accelerating product development cycles and increasing marketing effectiveness as well as managing content costs where possible.

In January 2018, we announced an organizational restructuring designed to prioritize our strategic growth initiatives and optimize overall business performance. The redesign shifts resources to focus on ad-tech and audience development efforts, while positioning the company for improved operating leverage over time. It also simplifies the organization into a flatter structure for smarter, faster execution.

Our Technologies

Pandora-Streaming Radio and On-Demand Music Services

At the heart of our service are proprietary personalization technologies, including the Music Genome Project and our radio and playlist generating algorithms.

Music Genome Project and Playlisting Algorithms

The Music Genome Project is a database of over 1.5 million uniquely analyzed songs from over 250 thousand artists, spanning over 660 genres and sub-genres, which our team of trained musicologists has developed one song at a time by evaluating and cataloging each song's particular attributes. The Music Genome Project database is a subset of our full catalog available to be played. Once we select music to become part of the Music Genome Project, our music analysts genotype the music by examining up to 450 attributes including objectively observable metrics such as tone and tempo, as well as subjective characteristics, such as lyrics, vocal texture and emotional intensity. Over time, our service has evolved by using data science to develop playlisting algorithms that further tailor the listener experience based on individual listener and broader audience reactions to the recordings we pick. We have integrated this technology into our on-demand music service, Pandora Premium, giving listeners the ability to search and play any track or album as well as offering unique playlist features tailored to each listener's distinct preferences. Listeners have the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button and automatically generate playlists based on their listening activity and the power of our Music Genome Project. Our team of curators, music analysts and data scientists have sifted through tens of millions of tracks to help

Table of Contents

listeners quickly find what they are looking for to create their own unique playlists. We employ rigorous hiring and training standards for selecting our music analysts, who typically have four-year degrees in music theory, composition or performance, and we provide them with intensive training in the Music Genome Project's precise methodology.

Next Big Sound ("NBS")

NBS Platform. NBS is the leading provider of online music analytics and insights tracking hundreds of thousands of artists around the world. NBS was founded in 2009 and tracks social, streaming and video data in one centralized platform. Sources range from Facebook and Twitter to Wikipedia, YouTube, Vevo, Instagram and many others. The NBS platform complements Pandora's Artist Marketing Platform ("AMP") and expands the suite of data-driven products that Pandora offers music makers and content partners.

Distribution and Partnerships

Pandora—Streaming Radio and On-Demand Music Services

A key element of our strategy is to make the Pandora service available in any environment that has internet connectivity. To this end, we make the Pandora service available through a variety of distribution channels. In addition to streaming our service to computers, we have developed Pandora mobile device applications ("apps") for smartphones and mobile operating systems, such as the iPhone and Android, and for tablets, including the iPad and Android tablets. We distribute those mobile apps free to listeners via app stores.

Pandora is integrated with connected devices, including automobiles, automotive aftermarket devices and consumer electronic devices, including many voice-based devices. Currently, most automobile integrations rely on smartphones for internet connectivity, which has enabled Pandora to be available in the best-selling passenger vehicles in the United States. Some automobiles are now using embedded, built-in internet connectivity to power the Pandora experience. These native integrations allow drivers to control the service via in-dash entertainment systems. As part of this ongoing effort to extend our reach in the car, we also built support for Android Auto and Apple CarPlay into our mobile applications. We expect these platforms will develop and grow significantly and help broaden our reach and provide consumers with additional flexibility for accessing Pandora in the car. As of December 31, 2017, approximately 35.2 million unique users have activated Pandora through a native integration in27 major automobile brands and 8 automotive aftermarket manufacturers.

Additionally, Pandora is integrated with thousands of consumer electronic and voice-based devices, including Sonos, Fitbit, Roku, Google Home, Amazon Echo, Comcast Xfinity, Apple TV and Microsoft Xbox. These integrations have made it easier for listeners to personalize their music experience anywhere they listen to music.

Advertising Revenue

We derive the substantial majority of our revenue from the sale of audio, display and video advertising for delivery across our computer, mobile and other connected device platforms. We generate the majority of our revenue from mobile and other connected devices, which presents an opportunity for our current and potential advertisers to reach our audience anytime, anywhere that they enjoy music.

Our advertising products allow both local and national advertisers to target and connect with listeners based on attributes including age, gender, zip code and content preferences using multi-platform ad campaigns to target their advertising messages to listeners anytime and anywhere. In the years ended December 31, 2015, 2016 and 2017, advertising revenue accounted for approximately 80%, 77% and 73% of our total revenue, respectively. The year-over-year change in revenue mix is due to an increase in subscription revenue related to increased Pandora Plus and Pandora Premium subscribers.

Audio Advertising. Our audio advertising products allow custom audio messages to be delivered during content programming sessions via short ad interludes on our adsupported service. Audio ads are available across all of our delivery platforms. On supported delivery platforms, audio ads can be accompanied by display ads to further enhance advertisers' messages. We are a significant provider of digital audio advertising in the United States.

Display Advertising. Our display advertising products offer opportunities to maximize advertising exposure to listeners of our ad-supported service on computers and mobile platforms. The graphical interface for our services on these delivery platforms is divided between our tuner, which contains our player controls and "now playing" information, and the information space surrounding our tuner. Our display ads include industry-standard banner ads of various sizes and placements within the information space surrounding our tuner, depending on the delivery platform and listener interaction.

Video Advertising. Our video advertising products allow delivery of rich, branded messages to further engage listeners of our ad-supported service. We currently offer video ad products, all of which are designed to play only upon listener interaction or opt-in, ensuring that advertisers have 100% share of the listener's attention during natural breaks accepted by an engaged audience. Some video ad products operate across our mobile and computer platforms, while others operate only on our mobile platform. All are designed to ensure they are viewable while playing, using interactions with Pandora to determine when a listener is engaged with the screen. Video Plus, Pandora's newest mobile video product, enables listeners to elect to watch a video ad in return for limited access to features available in our subscription services, while brands pay only for 15 seconds of completed video view. Another video advertising product, Sponsored Listening, offers listeners one or four hours of uninterrupted music, in exchange for the listener's active brand interaction, such as watching a video advertisement, interacting with rich media or visiting the advertiser's landing page.

Advertising Technologies. We have developed a suite of proprietary advertising technologies including order management, advertising serving and timing, native advertising formats, targeting and reporting. Given the ongoing evolution in advertising technologies, we plan to continue to invest aggressively in our ad tech to make Pandora a more efficient advertising partner and to become more competitive with our digital peers.

Native Advertising. Our audio, display and video advertising products can be designed and modified by us and by advertisers to tailor advertising campaigns to fit specific advertiser needs. Our advertisers can create custom "branded" stations from our music library that can be accessed by listeners of our ad-supported and subscription services, as well as engage listeners by allowing them to personalize the branded stations through listener-controlled variables.

Audience Targeting. Our audio, display, video and native advertising products have access to a set of over 2,000 targeting segments across all of our platforms, ranging from Pandora's unique proprietary targeting segments to second- and third-party enabled segments. Examples include Pandora's inferred Spanish Speakers and Political Preference proprietary segments, direct customer CRM upload and Datalogix and Neustar third-party segments. Additionally, advertisers can benefit from our proprietary ad targeting capabilities, which leverage listener-submitted profile information, enabling advertisers to precisely reach sought-after consumers without needing third-party cookies.

Programmatic Advertising. We have a programmatic advertising solution that allows advertisers and agencies to purchase display advertising. We intend to continue investing in programmatic advertising solutions and are currently testing video and audio programmatic offerings. We believe this is an important element to improving monetization.

To enhance our ability to compete with local terrestrial radio stations for advertisers' spending, our advertising inventory is integrated into the leading radio media buying platforms, Mediaocean and STRATA, and we are continuing to enhance the ability of radio advertisers to purchase media on these platforms, which incorporate Triton measurements of our radio audience reach side-by-side with terrestrial radio metrics.

We provide in-car advertising solutions, offering advertisers the opportunity to reach in-car audiences through audio ads running on vehicle models and aftermarket automotive devices with native Pandora automotive integrations.

In addition, we have invested in building a local advertising sales force in major radio markets. As of December 31, 2017, Pandora has 130 local sellers in 37 markets in the United States.

Our integration into standard radio media-buying processes and measurement, our in-car advertising solutions, our local advertising sales force and our programmatic advertising buying solution are key elements of our strategy to expand our penetration of the radio advertising market. Our success in executing this strategy is subject to numerous risks and uncertainties, including those described in Item 1A—"Risk Factors".

Subscription and Other Revenue

Subscription and other revenue is generated primarily through the sale of monthly or annual paid subscriptions to Pandora Plus and Pandora Premium. Subscription revenue is recognized on a straight-line basis over the duration of the subscription period. For the years ended December 31, 2015, 2016 and 2017, subscription and other revenue accounted for 19%, 16% and 22% of our total revenue, respectively.

5

Ticketing Service Revenue

Prior to the sale of Ticketfly on September 1, 2017, ticketing service revenue was generated primarily from service and merchant processing fees generated on ticket sales through the Ticketfly platform. Ticketfly sells tickets to fans for events on behalf of clients and charges either a fee per ticket or a percentage of the total convenience charge and order processing fee for its services at the time the ticket for an event is sold. Ticketfly service revenue is recorded net of the face value of the ticket at the time of the sale, as Ticketfly generally acts as the agent in these transactions. For the years ended December 31, 2015, 2016 and 2017, ticketing service revenue accounted for approximately 1%, 6% and 5% of our total revenue. As Ticketfly was acquired on October 31, 2015, and subsequently sold on September 1, 2017, the consolidated statements of operations include ticketing service revenue for the two months ended December 31, 2015, the year ended December 31, 2016 and the period ended September 1, 2017. Ticketing service revenue does not include revenue subsequent to the disposition of Ticketfly. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

Content, Copyrights and Royalties

The content we stream to users of the Pandora service consists primarily of music. To stream music, we must obtain licenses from, and pay royalties to, copyright owners, or their agents, for each sound recording that we stream, as well as for each underlying musical work, subject to certain exclusions. These licensing and royalty arrangements strongly impact our business operations. Refer to our discussion of these matters in <u>Item 1A—"Risk Factors"</u>.

Sound Recordings

For sound recordings, we pay content acquisition costs based on the terms of direct license agreements with major and independent music labels and distributors for the significant majority of the sound recordings we stream on our ad-supported service, Pandora Plus and Pandora Premium. Depending on the applicable service, these license agreements generally require us to pay either a per-performance fee based on the number of sound recordings we transmit, a percentage of revenue associated with the service, or a per-subscriber minimum amount. Certain of these license agreements require minimum guarantee payments, some of which are paid in advance. Refer to Note 2, "Summary of Significant Accounting Policies" for our policy on prepaid content acquisition costs.

If we have not entered into a direct license agreement with the copyright owner of a particular sound recording that is streamed on our services, we stream that sound recording pursuant to the statutory license and pay the applicable rates set by the Copyright Royalty Board ("CRB") for the period from January 1, 2016 through December 31, 2020. The rates for non-subscription services, such as our ad-supported service, were set at \$0.0017 per play and the rates for subscription services, such as Pandora Plus, were set at \$0.0022, adjusted for inflation. Effective January 1, 2018, these rates were adjusted for inflation to \$0.0018 per play for non-subscription services and \$0.0023 per play for subscription services. Sound recordings streamed under the statutory license and paid at the CRB-set rates can only be played in radio mode on our services. These sound recordings cannot be played on-demand or offline and are not eligible for replay or additional skips.

Musical Works

Content acquisition costs for musical works are negotiated with and paid to performing rights organizations ("PROs") such as the American Society of Composers, Authors and Publishers ("ASCAP"), Broadcast Music, Inc. ("BMI"), SESAC, Inc. ("SESAC") and Global Music Rights ("GMR"), and directly to publishing companies. Content acquisition costs for the streaming of musical works on our ad-supported service are calculated such that each copyright holder receives its usage-based and ownershipbased share of a royalty pool equal to 20% of the content acquisition costs paid by us for sound recordings on our ad-supported service. Content acquisition costs for the streaming of musical works on our subscription services are equal to the rates determined in accordance with the statutory license set forth in 17 U.S.C. §115 ("Section 115").

Rate-Setting Proceedings for Reproduction Rights under Section 115

The rate structure for the statutory license for reproduction rights under Section 115 expired at the end of 2017. A new rate structure was set in January 2018 covering the period from January 1, 2018 through December 31, 2022 by a three judge CRB panel, and we were one of five commercial music service operators (along with Amazon, Apple, Google and Spotify) that participated in rate-setting proceedings that determined these rates (the "Phonorecords III Proceedings"). The Nashville Songwriters Association International, the National Association of Music Publishers and George Johnson Music Publishing are also participating in the Phonorecords III Proceedings. A trial before the CRB concluded in April 2017, and the CRB rendered a decision in January 2018 ("the Initial Determination"). If the Initial Determination is adopted and affirmed, the "all-in" rate that streaming services, including Pandora, will pay to music publishers and songwriters for the mechanical rights and performance rights needed in connection with interactive streaming will increase annually between 2018 and 2022: from 11.4% of revenues or 22.0% of label payments in 2018, to 15.1% of revenues or 26.2% of label payments by 2022. Certain per-subscriber

minimum royalty floors also apply depending on the type of service. The CRB may elect to modify the Initial Determination at the request of the participants in the Phonorecords III Proceedings, and thereafter the final CRB decision may be appealed for review by the U.S. Court of Appeals for the District of Columbia Circuit.

The Phonorecords III Proceedings are important to us because our direct licenses with music publishers use the Section 115 rates. As a result, increases in the Section 115 rates increase our content acquisition costs for our subscription services, which, if such increase were substantial, could materially harm our financial condition and hinder our ability to provide interactive features in our services, or cause one or more of our subscription services to not be economically viable.

Government Regulation

As a company conducting business on the internet, we are subject to a number of foreign and domestic laws and regulations relating to consumer protection, information security and data protection, among other things. Many of these laws and regulations are still evolving and could be interpreted in ways that could harm our business. In the area of information security and data protection, the laws in several states require companies to implement specific information security controls to protect certain types of information. Likewise, all but a few states have laws in place requiring companies to notify users if there is a security breach that compromises certain categories of their information. We also may be subject to certain foreign and domestic laws, such as the European General Data Protection Regulation, regarding privacy of listener data, among other things. Our privacy policy and terms of use describe our practices concerning the use, transmission and disclosure of listener information and are posted on our website.

Sales and Marketing

We organize our sales force into multiple geographically-based teams that are each focused on selling advertising across our computer, mobile and other connected device platforms. Teams are located in our Oakland, California headquarters and in regional and local sales offices throughout the United States.

Our marketing team is charged with using all forms of communications to reach our business and reputational goals. Prior to our reorganization, the marketing team was organized into four groups focused on campaign planning and analytics, consumer marketing, communications and our Music Makers Group. As part of the reorganization, the marketing team will increase their focus on listeners, drive marketing effectiveness and efficiency of spend and improve collaboration with advertisers. While we have historically relied on the success of viral marketing to expand consumer awareness of our service, in recent years we have launched marketing and advertising campaigns to increase consumer awareness of our brand and expand our listener base. We anticipate that we will continue to utilize these types of campaigns in the future.

Artist Relations

Industry Relations and Artist Marketing

Industry Relations and Artist Marketing, formerly the Music Makers Group, is a team that works directly with the music industry to help drive connections with fans across all channels at Pandora. Our vision is to ensure artists can promote and market their music to fans, drive engagement with experiences from live events to original content and audio messages to fans and understand all of the benefits of these interactions via our analytics tools.

Pandora Artist Marketing Platform

Pandora AMP consists of products and programs that enable music makers to leverage data to grow an audience, track progress and connect with fans on Pandora. Since its original launch focusing on artist data and insights in 2014, AMP has evolved into a set of powerful self-serve marketing tools including Artist Audio Messaging, AMPcast and Featured Tracks. Additional features of AMP, which include a dynamic feed of an artist's campaign, performance metrics and suggestions for new campaigns, were launched in October 2016 alongside a data integration with NBS, shifting artist insights and analytics to the NBS platform and enhancing the set of desktop and mobile marketing tools offered via AMP.

Derived from tens of billions of hours of personalized listening on Pandora, the marketing tools on AMP along with the analytics tools on NBS are available to all artists spinning on Pandora and their teams. Additionally, enterprise-level tools and marketing opportunities are available to artist and label teams who have opted into a direct deal with Pandora.

7

Competition

Competition for Listeners

We compete for the time and attention of our listeners with other content providers on the basis of a number of factors, including quality of experience, relevance, acceptance and perception of content quality, ease of use, price, accessibility, perceptions of ad load on our ad-supported service, brand awareness and reputation. We also compete for listeners on the basis of our presence, branding and visibility as compared with other providers that deliver content through the internet, mobile devices and consumer products. We believe that we compete favorably on these factors. For additional details on risks related to competition for listeners, please refer to Item 1A—"Risk Factors".

Many of our current and potential future competitors enjoy competitive advantages, such as greater name recognition, legacy operating histories and larger marketing budgets, as well as greater financial, technical and other resources. We compete with many forms of digital media for the time and attention of our listeners.

We also compete for listeners with broadcast radio providers, including terrestrial radio providers. Many broadcast radio conglomerates own large numbers of radio stations or other media properties. Many terrestrial radio stations broadcast digital signals, which provide high quality audio transmission.

We also face competition for listeners and listener hours from on-demand music streaming services such as Amazon Prime, Apple Music, Google Play Music, Spotify and YouTube. These services offer consumers the ability to choose the songs and artists they want to hear and create customized playlists. With the launch of Pandora Premium in April 2017, we are now able to compete more directly against these other services. Our ability to attract and retain listeners may be dependent on the success of this service and our ability to market the service effectively. This on-demand content is accessible in automobiles and homes, using portable players, mobile phones and other wireless and consumer electronic devices. The audio entertainment marketplace continues to rapidly evolve, providing our listeners with a growing number of alternatives and new media platforms. Certain technology companies that operate music streaming businesses as an ancillary revenue stream generally enjoy an advantage in that they do not rely on their music streaming services to generate profitability. In addition, they are able to favor their own music services within the devices they sell, helping to increase their listener base from existing consumers, making their music services fast growing segments. Apple, Google and Amazon are examples of competitors of this kind.

At a macro level, we compete for the time and attention of our listeners with providers of other forms of in-home and mobile entertainment. To the extent existing or potential listeners choose to watch cable television, stream video from on-demand services or play interactive video games on their home-entertainment system, computer or mobile phone rather than listen to the Pandora service, these content services pose an adjacent competitive threat for consumer time and mindshare.

Competition for Advertisers

We compete with traditional media products and services, and with other internet-based services, for a share of our advertising customers' overall marketing budgets. We compete on the basis of a number of factors, including perceived return on investment, effectiveness and relevance of our advertising products, pricing structure, the ability to deliver large volumes or precise types of ads to targeted demographics, transactional capabilities and reporting capabilities. We believe that our ability to deliver targeted and relevant ads across a wide range of platforms allows us to compete favorably on the basis of these factors and justify a long-term profitable pricing structure. However, the market for online advertising is intensely competitive and rapidly changing, and with the introduction of new technologies and market entrants, we expect competition to intensify in the future.

Advertisers are allocating increasing amounts of their overall marketing budgets to online advertising, yet the market for online advertising has also become increasingly competitive. We compete for online advertisers with other internet companies, including major internet portals, search engine companies and social media sites. Large internet companies with greater brand recognition have significant numbers of direct sales personnel, more advanced programmatic advertising capabilities and substantial proprietary advertising inventory and web traffic that provide a significant competitive advantage and have a significant impact on pricing for internet advertising and web traffic.

Terrestrial radio is a source of competition for advertising dollars. Broadcast radio providers deliver ads across a more familiar platform than the internet may be to traditional advertisers.

Our competitors include large scale online advertising platforms such as Amazon, Facebook, Google and traditional media companies such as large television broadcasters and national print outlets. We directly compete against Cumulus, Entercom, iHeartRadio, Spotify and other companies in the traditional broadcast radio market. For additional details on risks related to competition for advertisers, please refer to <u>Item 1A—"Risk Factors"</u>.

Seasonality

Pandora-Streaming Radio and On-Demand Music Services

Our results reflect the effects of seasonal trends in listener and advertising behavior. During the last quarter of each calendar year, and particularly during the holiday season, we expect to experience both higher advertising sales due to greater advertiser demand during the holiday season and increased usage of our service due to the popularity of holiday music. In addition, in the first quarter of each calendar year, we expect to experience lower advertising sales due to reduced advertiser demand, but sustain higher levels of usage by listeners due to increased use of media-streaming devices received as gifts during the holiday season. See the section entitled "Business Trends" in Item 7 of this Annual Report on Form 10-K for a more complete description of the seasonality of our financial results.

Intellectual Property

Our success depends in part upon our ability to protect our technologies and intellectual property. To accomplish this, we rely on a combination of intellectual property rights, including trade secrets, patents, copyrights, trademarks, contractual restrictions, technological measures and other methods. We enter into confidentiality and proprietary rights agreements with our employees, consultants and business partners, and we control access to and distribution of our technology and proprietary information.

We have filed and acquired dozens of active patent applications and issued patents in the United States and other countries. We continue to pursue additional patent protection, both in the United States and abroad where appropriate and cost effective. We intend to hold these patents as part of our strategy to protect and defend Pandora in patent-related litigation. We also acquire patents and patent applications from time to time as part of other transactions.

Our registered trademarks in the United States include "Pandora," "Ampcast," "Comedy Genome Project" and the "Music Genome Project," in addition to a number of Pandora logos and other marks. "Pandora" is also registered in Australia, Canada, Chile, the European Union, India, Israel, Mexico, New Zealand, Switzerland, Taiwan and other countries. "Music Genome Project" is also registered in Australia, Canada, China and New Zealand. We have pending trademark applications in the United States and other countries.

We are the registrant of the internet domain names for our primary website, pandora.com, as well as others related to our current and potential future businesses.

In addition to the forms of intellectual property listed above, we own rights to proprietary processes and trade secrets, including those underlying the Pandora service. We use contractual, policy and technological means to generally control access to, and the use and distribution of our proprietary software, trade secrets and other confidential information, both internally and externally, including contractual protections with employees, contractors, customers and partners.

Customer Concentration

For each of the years ended December 31, 2015, 2016 and 2017, we had no individual customers that accounted for 10% or more of total revenue.

Information About Geographic Revenue

Information about geographic revenue is set forth in Note 15 of the Notes to the Consolidated Financial Statements under Item 8 of this Annual Report on Form 10-K.

Employees

As of December 31, 2017, we had 1,938 employees. None of our employees are covered by collective bargaining agreements, and we consider our relations with our employees to be good.

Corporate and Available Information

We were incorporated as a California corporation in January 2000 and reincorporated as a Delaware corporation in December 2010. Our principal executive offices are located at 2101 Webster Street, Suite 1650, Oakland, California 94612 and



our telephone number is (510) 451-4100. Our website is located at www.pandora.com and our Investor Relations website is located at investor.pandora.com.

We changed our fiscal year to the calendar year ending December 31, effective beginning with the period ended December 31, 2013. As a result, the period ended December 31, 2013 was shortened from a year to an eleven-month transition period.

We file reports with the Securities and Exchange Commission ("SEC"), including Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and any other filings required by the SEC. Such reports and other information filed by us with the SEC are available free of charge on our website at investor.pandora.com when such reports are available on the SEC's website. The information on our website is not incorporated by reference into this Annual Report on Form 10-K or in any other report or document we file with the SEC.

The public may read and copy any materials we file with the SEC at the SEC's Public Reference Room at 100 F Street, NE, Washington, DC 20549. The public may obtain information on the operation of the Public Reference Room by calling the SEC at 1-800-SEC-0330. The SEC maintains an internet site that contains reports, proxy and information statements and other information regarding issuers that file electronically with the SEC at www.sec.gov.

Investors and others should note that we use our Investor Relations website, SEC filings, press releases, public conference calls and webcasts as means of disclosing material non-public information and for complying with our disclosure obligations under Regulation FD. We use these channels as well as social media to communicate with the public about the Company, our services and other issues. It is possible that the information we post on social media could be deemed to be material information. Therefore, we encourage investors, the media, and others interested in the Company to review the information we post on the social media channels listed on our Investor Relations website.

ITEM 1A. RISK FACTORS

The risks and uncertainties set forth below, as well as other factors described elsewhere in this Annual Report on Form 10-K or in other filings by us with the SEC, could adversely affect our business, financial condition, results of operations and the trading price of our common stock. Additional risks and uncertainties that are not currently known to us or that are not currently believed by us to be material may also harm our business operations and financial results. Because of the following factors, as well as other factors affecting our financial condition and operating results, past financial performance should not be considered to be a reliable indicator of future performance, and investors should not use historical trends to anticipate results or trends in future periods.

Risks Related to Our Business

If our efforts to attract prospective listeners and to retain existing listeners are not successful, our growth prospects and revenue will be adversely affected.

Our ability to grow our business and generate advertising revenue depends on retaining and expanding our listener base and increasing listener hours. We must convince prospective listeners of the benefits of our service and existing listeners of the continuing value of our service. The more listener hours we stream, the more ad inventory we have to sell. Further, growth in our listener base increases the size of demographic pools targeted by advertisers, which improves our ability to deliver advertising in a manner that maximizes our advertising customers' return on investment and, ultimately, to demonstrate the effectiveness of our advertising solutions and justify a pricing structure that is profitable for us. If we fail to grow our listener base and listener hours, particularly in key demographics such as young adults, we will be unable to grow advertising revenue, and our business will be materially and adversely affected.

Our ability to increase the number of our listeners and listener hours will depend on effectively addressing a number of challenges. Some of these challenges include:

- providing listeners with a consistent high quality, user-friendly and personalized experience;
- successfully expanding our share of listening in cars;
- continuing to build and maintain availability of catalogs of music and comedy and other content that our listeners enjoy;
- continuing to innovate and keep pace with changes in technology and our competitors;
- maintaining and building our relationships with makers of consumer products such as mobile devices and other consumer electronic products to make our service available through their products;
- maintaining positive listener perception of our service while managing ad-load to optimize inventory utilization; and
- minimizing listener churn and attracting lapsed listeners back to the service.

In addition, we have historically relied heavily on the success of viral marketing to expand consumer awareness of our service. We recently began supplementing our viral marketing strategy with larger, more costly marketing campaigns, and this increase in marketing expenses could fail to achieve expected returns and therefore have an adverse effect on our results of operations. We cannot guarantee that we will be successful in maintaining or expanding our listener base and failure to do so would materially and adversely affect our business, operating results and financial condition.

Further, although we use our number of active users as a key indicator of our brand awareness and the growth of our business, the number of active users exceeds the number of unique individuals who register for, or actively use, our service. We define active users as the number of distinct users that have requested audio from our servers within the trailing 30 days from the end of each calendar month. To establish an account, a person does not need to provide personally unique information. For this reason, a person may have multiple accounts. If the number of actual listeners does not result in an increase in listener hours, then our business may not grow as quickly as we expect, which may harm our business, operating results and financial condition.

If we are unable to maintain revenue growth from our advertising products, particularly in mobile advertising, our results of operations will be materially adversely affected.

Our number of listener hours on mobile devices comprised approximately88% of our total listener hours for theyear ended December 31, 2017, and we expect that mobile listener hours will continue to be a substantial majority of our total listener hours for the foreseeable future. We must therefore continue to convince advertisers of the capabilities and value of mobile digital advertising opportunities so that they direct an increasing portion of their advertising spend toward our ad-supported service.

We continue to build our sales capability to penetrate local advertising markets, which we view as a key challenge to monetizing our listener hours, including listener hours on mobile and other connected devices. Our audio advertising capability also places us in direct competition with terrestrial radio, as many advertisers that purchase audio ads focus their spending on terrestrial radio stations who traditionally have strong connections with local advertisers. We cannot foresee whether we will be able to continue to capture local and audio advertising revenue at the current rate of growth, which may have an adverse impact on future revenue and income.

We continue to work on initiatives that, if successfully implemented, would increase our number of listener hours on mobile and other connected devices, including efforts to expand the reach of our service by making it available on a variety of devices, such as smartphones and devices connected to or installed in automobiles. In order to effectively monetize such increased listener hours, we must, among other things, convince advertisers to migrate spending to nascent advertising markets, penetrate local advertising markets and develop compelling ad product solutions. We may not be able to effectively monetize inventory generated by listeners using mobile and connected devices, or do so in a time frame that supports our business plans.

Advertising spending is increasingly being placed through new data-driven channels, such as the programmatic buying ecosystem, where our competitors offer more advanced systems and services that in many cases exceed our capabilities. We must invest heavily in advertising technology to remain competitive in the digital advertising ecosystem, and if our investments and development efforts are not successful, our business will be adversely affected.

As new advertising buying technologies, such as programmatic buying, develop around data-driven technologies and advertising products, an increasing percentage of advertising spend is likely to shift to such channels and products. These data-driven advertising products and programmatic buying technologies allow publishers to use data to target advertising toward specific groups of consumers who are more likely to be interested in the advertising message delivered. Our programmatic ad-buying systems currently lag behind those offered by market leaders—such as Google and Facebook—in terms of functionality and ease of use. We are investing heavily in developing more advanced data-driven, programmatic advertising competitive. If we fail to develop competitive data-driven, programmatic ad-buying systems will be competitive. If we fail to develop competitive data-driven, programmatic ad-buying services, we may experience a decline in advertiser interest in our services, which may adversely affect our advertising revenue and our business results.

Emerging industry trends in digital advertising measurement and pricing may pose challenges for our ability to forecast or optimize our advertising inventory, which may adversely impact our advertising revenue.

The digital advertising marketplace is introducing new ways to measure and price advertising inventory. Specifically, the industry is in the process of moving from purchasing ad impressions based on the number of ads served by the applicable ad server to a new "viewable" impression standard (based on number of pixels in view and duration) for select products. In the absence of a uniform industry standard, agencies and advertisers have adopted several different billing standards and measurement methodologies. In addition, measurement services may require technological integrations which are still being evaluated by the industry. As these trends in the industry continue to evolve, our advertising revenue may be adversely affected by the availability, accuracy and utility of the available analytics and measurement technologies as well as our ability to successfully implement and operationalize such technologies and standards.

Our failure to convince advertisers of the benefits of our ad-supported service in the future could harm our business.

For the year ended December 31, 2017, we derived 73% of our revenue from the sale of advertising and expect to continue to derive a substantial majority of our revenue from the sale of advertising in the future. Our ability to attract and retain advertisers, and ultimately to sell our advertising inventory to generate advertising revenue, depends on a number of factors, including:

increasing the number of listener hours, particularly within desired demographics;

- keeping pace with changes in technology and our competitors;
- competing effectively for advertising dollars from other online marketing and media companies;
- penetrating the market for local radio advertising;
- demonstrating the value of advertisements to reach targeted audiences across all of our delivery platforms, including the value of mobile digital
 advertising;
- ensuring that new ad formats and ad product offerings are attractive to advertisers and that inventory management decisions (such as changes to ad load, frequency, prominence and quality of ads that we serve listeners) do not have a negative impact on listener hours;
- continuing to develop and diversify our advertising platform, which currently includes delivery of display, audio and video advertising products through multiple delivery channels, including computers, mobile and other connected devices; and
- coping with ad blocking technologies that have been developed and are likely to continue to be developed that can block the display of our ads.

Our agreements with advertisers are generally short-term or may be terminated at any time by the advertiser. Advertisers that are spending only a small amount of their overall advertising budget on our service may view advertising with us as experimental and unproven and may leave us for competing alternatives at any time. We may never succeed in capturing a greater share of our advertisers' core advertising spending, particularly if we are unable to achieve the scale and industry penetration necessary to demonstrate the effectiveness of our advertising platforms, or if our advertising model proves ineffective or not competitive when compared to alternatives. Failure to demonstrate the value of our service would result in reduced spending by, or loss of, existing or potential future advertisers, which would materially harm our revenue and business.

Unavailability of, or fluctuations in, third-party measurements of our audience may adversely affect our ability to grow advertising revenue.

Selling ads, locally and nationally, requires that we demonstrate to advertisers that our service has substantial reach and usage. Third-party measurement vendors' technology may not reflect our true listening audience and their underlying methodologies are subject to change at any time. In addition, the methodologies we apply to measure the key metrics that we use to monitor and manage our business may differ from the methodologies used by third-party measurement service providers. For example, we calculate listener hours based on the total bytes served for each track that is requested and served from our servers, as measured by our internal analytics systems, whether or not a listener listens to the entire track. By contrast, certain third-party measurement service providers may calculate and report the number of listener hours using a client-based approach, which measures time elapsed during listening sessions. Measurement technologies for mobile and consumer electronic devices may be even less reliable in quantifying the reach, usage and location of our service, and it is not clear whether such technologies will integrate with our systems or uniformly and comprehensively reflect the reach, usage and location of our service. While we have been working with third-party measurement service providers and certain of their measurements have earned Media Ratings Council accreditation, some providers have not yet developed uniform measurement systems that comprehensively measure the reach, usage and location of our services have not yet developed uniform measurement systems that comprehensively measurement have and of our service. In order to demonstrate to potential advertisers the benefits of our service, we supplement third-party measurement data with our internal reporting, which may be perceived as less valuable than third-party numbers. If third-party measurement providers report lower metrics than we do, or if there is wide variance among reported metrics, our ability to attract advertisers to our service co

The lack of accurate cross-platform measurements for internet radio and broadcast radio may adversely affect our ability to grow advertising revenue.

We have invested substantial resources to create accurate cross-platform measurements for internet radio and broadcast radio in the major automated media-buying platforms, attempting to create a one-stop shop that enables media buyers to compare internet radio audience reach with terrestrial radio audience reach using traditional broadcast radio metrics.

Media buying agencies receive measurement metrics from third parties, such as Triton for internet radio and Nielsen for more traditional media like terrestrial radio and television. Media buying agencies may choose not to show, or may be prohibited by contract from showing, internet radio metrics alongside traditional terrestrial metrics. Despite our efforts to



achieve parity within the tools available to media buying agencies, a lack of comparable internet radio metrics in these buying tools could have a materially negative effect on our ability to sell advertising on our service and achieve our revenue goals.

If we are unable to continue to make our technology compatible with the technologies of third-party distribution partners who make our service available to our listeners through mobile devices, consumer electronic products and automobiles, we may not remain competitive and our business may fail to grow or decline.

In order to deliver music everywhere our listeners want to hear it, our service must be compatible with mobile, consumer electronic, automobile and website technologies. Our service is accessible in part through both Pandora-developed and third-party developed apps that hardware manufacturers embed in, and distribute through, their devices. Most of our agreements with makers of mobile operating systems and devices through which our service may be accessed, including Apple and Google, are short-term or can be canceled at any time with little or no prior notice or penalty. The loss of these agreements, or the renegotiation of these agreements on less favorable economic or other terms, could limit the reach of our service and its attractiveness to advertisers. Some of these mobile device makers and operating system providers, including Apple, Amazon, Samsung and Google, are now, or may in the future become, competitors of ours, and could stop allowing or supporting access to our service through their products for competitive reasons.

Connected devices and their underlying technologies are constantly evolving. As internet connectivity of automobiles, mobile devices and other consumer electronic products expands and as new internet-connected products are introduced, we must constantly adapt our technology. It is challenging to keep pace with the continual release of new devices and technological advances in digital media delivery. If manufacturers fail to make products that are interoperable with our technology or we fail to adapt our technology to their evolving requirements, our ability to grow or sustain the reach of our service, increase listener hours and sell advertising could be adversely affected.

Consumer tastes and preferences can change in rapid and unpredictable ways and consumer acceptance of these products depends on the marketing, technical and other efforts of third-party manufacturers, which is beyond our control. If consumers fail to accept the products of the companies with whom we partner or if we fail to establish relationships with makers of leading consumer products, our business could be adversely affected.

If we fail to accurately predict and play music, comedy or other content that our listeners enjoy, we may fail to retain existing and attract new listeners.

We believe that a key differentiating factor between the Pandora service and other music content providers is our ability to predict music that our listeners will enjoy. Our personalized playlist generating system, based on the Music Genome Project and our proprietary algorithms, is designed to enable us to predict listener music preferences and select music content tailored to our listeners' individual music tastes. We have invested, and will continue to invest, significant resources in refining these technologies; however, we cannot guarantee that such investments will produce the intended results. The effectiveness of our personalized playlist generating system depends in part on our ability to gather and effectively analyze large amounts of listener data and listener feedback and we have no assurance that we will continue to be successful in enticing listeners to give a thumbs-up or thumbs-down to enough songs for our database to effectively predict and select new and existing songs. In addition, our ability to offer listeners songs that they have not previously heard and impart a sense of discovery depends on our ability to acquire and appropriately categorize additional tracks that will appeal to our listeners' diverse and changing tastes. While we have over 1.5 million analyzed songs in our library, we must continuously identify and analyze additional tracks that our listeners will enjoy and we may not effectively do so. Further, many of our competitors currently have larger catalogs than we offer and they may be more effective in providing their listeners with a more appealing listener experience.

We also provide comedy content on Pandora, and for that content we also try to predict what our listeners will enjoy, using technology similar to the technology that we use to generate personalized playlists for music. The risks that apply to predicting our listeners' musical tastes apply to comedy and other content to an even greater extent, particularly as we lack experience with content other than music, do not yet have as large a data set on listener preferences for comedy and other content, and have a much smaller catalog as compared to music. Our ability to predict and select music, comedy and other content that our listeners enjoy is critical to the perceived value of our service among listeners and failure to make accurate predictions would adversely affect our ability to attract and retain listeners, increase listener hours and sell advertising.

We face, and will continue to face, competition with other content providers for listener hours and advertising spending.

We compete for the time and attention of our listeners with other content providers on the basis of a number of factors, including quality of experience, relevance, acceptance and perception of content quality, ease of use, price, accessibility,

perception of ad load, brand awareness and reputation. Such competition affects the amount of quality advertising inventory available which we can offer to advertisers on our ad-supported service.

Many of our competitors may leverage their existing infrastructure, brand recognition and content collections to augment their services by offering competing internet radio features within a more comprehensive digital music streaming service. We face increasing competition for listeners from a growing variety of music services that deliver music content through mobile phones and other wireless devices. Our competitors include traditional broadcast and internet radio services and other terrestrial radio companies. We also directly compete with the non-interactive, internet radio offerings from providers such as Amazon Prime, Apple Music, Google Play Music and Spotify.

Our competitors also include terrestrial radio services, many of which also broadcast on the internet. Terrestrial radio providers offer their content for free, are well established and accessible to listeners and offer content, such as news, sports, traffic, weather and talk that we currently do not offer. In addition, many terrestrial radio stations have begun broadcasting digital signals, which provide high-quality audio transmission. We also compete directly with other emerging non-interactive internet radio providers, which may offer more extensive content libraries than we offer and some of which may be accessed internationally.

We compete for the time and attention of our listeners with providers of other forms of in-home and mobile entertainment. To the extent existing or potential listeners choose to watch cable television, stream video from on-demand services or play interactive video games on their home-entertainment system, computer or mobile phone rather than listen to the Pandora service, these content services pose a competitive threat. We also compete with many other forms of media and services for the time and attention of our listeners, including large scale online advertising platforms such as Amazon, Facebook and Google and traditional media companies such as large television broadcasters and national print outlets.

We believe that companies with a combination of financial resources, technical expertise and digital media experience also pose a significant threat. For example, Apple, Amazon and Google operate competing services. These and other competitors may devote greater resources than we have available, have a more accelerated time frame for deployment, be willing to absorb significant costs to acquire customers through free trials or other initiatives, operate their music services at a loss in order to drive their other profitable businesses, and leverage their existing user base and proprietary technologies to provide products and services that our listeners and advertisers may view as superior or more cost effective. Our current and future competitors may have more well established brand recognition, more established relationships with music content companies and consumer product manufacturers, greater financial, technical and other resources, more sophisticated technologies or more experience in the markets, both domestic and international, in which we compete.

We also compete for listeners on the basis of the presence and visibility of our app, which is distributed via the largest app stores operated by Apple, Google and Amazon. Such distribution is subject to an application developer license agreement in each case. We face significant competition for listeners from these companies, who are also promoting their own digital music and content online through their app stores. Search engines and app stores rank responses to search queries based on the popularity of a website or mobile application, as well as other factors that are outside of our control. Additionally, app stores often offer users the ability to browse applications by various criteria, such as the number of downloads in a given time period, the length of time since a mobile app was released or updated, or the category in which the application is placed. The websites and mobile applications of our competitors may rank higher than our website and our Pandora app, and our app may be difficult to locate in app stores, which could draw potential listeners away from our service and toward those of our competitors. In addition, our competitors' products may be pre-loaded or integrated into consumer electronics products or automobiles, creating an initial visibility advantage. If we are unable to compete successfully for listener hours may fail to increase as by maintaining and increasing our presence and visibility online, in app stores and in consumer electronics products and automobiles, our listener hours may fail to increase as expected or decline and our business may suffer. Additionally, should any of these parties reject our app from their app store or amend the terms of their license in such a way that inhibits our ability to distribute our apps, or negatively affects our economics in such distribution, our ability to increase ad-supported listener hours and sell advertising would be adversely affected, which would reduce our revenue and harm our operating results.

To compete effectively, we must continue to invest significant resources in the development of our service to enhance the user experience of our listeners.

Additionally, in order to compete successfully for advertisers against new and existing competitors, we must continue to invest resources in developing and diversifying our advertisement platform, harnessing listener data and ultimately proving the effectiveness and relevance of our advertising products. There can be no assurance that we will be able to compete successfully for listeners and advertisers in the future against existing or new competitors, and failure to do so could result in loss of existing



Table of Contents

or potential listeners, loss of current or potential advertisers or a reduced share of our advertisers' overall marketing budget, which could adversely affect our pricing and margins, lower our revenue, increase our research and development and marketing expenses, diminish our brand strength and prevent us from achieving or maintaining profitability.

Our ability to increase the number of our listeners will depend in part on our ability to establish and maintain relationships with automakers, automotive suppliers and consumer electronics manufacturers with products that integrate our service.

A key element of our strategy to expand the reach of our service and increase the number of our listeners and listener hours is to establish and maintain relationships with automakers, automotive suppliers and consumer electronics manufacturers that integrate our service into and with their products. Working with certain third-party distribution partners, we currently offer listeners the ability to access our service through a variety of consumer electronics products used in the home and devices connected to or installed in automobiles. We intend to broaden our ability to reach additional listeners, and increase current listener hours, through other platforms and partners over time, including through direct integration into connected cars. However, product design cycles in consumer products and automotive manufacturing are lengthy, and we may not be able to achieve our goals in our desired timeframe, which could adversely impact our ability to grow our business.

Our existing agreements with partners in the automobile and consumer electronics industries generally do not obligate those partners to offer our service in their products. In addition, some automobile manufacturers or their supplier partners may terminate their agreements with us for convenience. Our business could be adversely affected if our automobile partners and consumer electronics partners do not continue to provide access to our service or are unwilling to do so on terms acceptable to us. If we are forced to amend the business terms of our distribution agreements as a result of competitive pressure, our ability to maintain and expand the reach of our service and increase listener hours would be adversely affected, which would reduce our revenue and harm our operating results.

If our efforts to attract and retain subscribers or convert ad-supported listeners into subscribers of our subscription offerings are not successful, our business will be adversely affected.

Our ability to continue to attract and retain users of our paid subscription services will depend in part on our ability to consistently provide our subscribers with a quality experience through Pandora Plus and Pandora Premium. If Pandora Plus and Pandora Premium subscribers do not perceive these offerings to be of value, or if we introduce new or adjust existing features or pricing in a manner that is not favorably received by them, we may not be able to attract and retain subscribers or be able to convince listeners to become subscribers of such additional service offerings. Subscribers may cancel their subscription to our service for many reasons, including a perception that they do not use the service sufficiently, the need to cut household expenses, competitive services that provide a better value or experience or as a result in changes in pricing. Further, in a number of cases, the rates that we pay pursuant to our direct license agreements with record labels are significantly affected by the number of subscribers we are able to attract and retain subscribers are not successful, our business, operating results and financial condition may be adversely affected.

Our ability to offer interactive features in our services depends upon maintaining commercially viable direct licenses with copyright owners of the music we play. If we are not able to maintain these direct licenses, we could lose the right to provide interactive features in our ad-supported service and Pandora Plus and the right to operate Pandora Premium. If we are not able to renew these direct licenses on similar terms when they expire, our profitability may be negatively affected.

We have direct license agreements with dozens of music sound recording copyright owners, commonly known as "record labels", with thousands of musical work copyright owners, commonly known as "publishers", and with ASCAP, BMI and SESAC, the three largest performing rights organizations, commonly known as "PROS". In total, these agreements give us the right to add interactive features such as replays, additional skips and offline play to our ad-supported service and Pandora Plus, and they also give us the right to operate Pandora Premium. The direct licenses we have entered into with record labels and publishers are complex and require significant on-going efforts to operationalize, and there is risk that we may not be able to comply with the terms of these licenses, which could result in the loss of some or all of these licenses and some or all of the rights they convey. Similarly, many of these licenses provide that if the licensor loses rights in a portion of the content licensed under the agreement, that content may be removed from the license going-forward. In addition, if we are acquired, certain terms of our direct licenses, or if rights to certain music were no longer available under these licenses, then we may have to remove the affected music from our services, or discontinue certain interactive features for such music, or it might become commercially impractical for us to operate Pandora Premium. Any of these occurrences could have a material adverse effect on our business, financial condition and results of operations.



Several of these direct licenses also include provisions related to the royalty payments and structures of those agreements relative to other content licensing arrangements, which, if breached, could cause our payments under those agreements to escalate substantially. In addition, record labels, publishers and PROs with whom we have entered into direct licenses have the right to audit our content acquisition payments, and any such audit could result in disputes over whether we have paid the proper content acquisition costs. If such a dispute were to occur, we could be required to pay additional content acquisition costs, audit fees and interest or penalties, and the amounts involved could materially and adversely affect our business, financial condition and results of operations.

Further, there is no guarantee that the direct licenses we have now will be renewed in the future or that such licenses will be available at the rates for content acquisition costs associated with the current licenses. If we are unable to secure and maintain direct licenses for the rights to provide music on our services at rates that are similar to those under our current direct licenses, or other commercially viable rates, our content costs could rise and materially adversely affect our business, financial condition and results of operations.

Minimum guarantees required under certain of our music license agreements may limit our operating flexibility and could adversely affect our liquidity and results of operations.

As described in "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Note 11, Commitments and Contingencies", certain of our music license agreements contain minimum guarantees and require that we make upfront minimum guarantee payments. As of December 31, 2017, we have future minimum guarantee commitments of \$405.2 million. Such minimum guarantees related to our content acquisition costs are not tied to our number of subscribers, active users or the number of sound recordings used on our services. As such, our ability to achieve and sustain profitability and operating leverage on our services depends on our ability to increase our revenue through increased paid subscriptions and advertising sales on terms that maintain an adequate gross margin. Given the multiple-year duration and largely fixed cost nature of these minimum guarantees, if subscriber acquisition are retention or advertising sales do not meet our expectations, our liquidity and results of operations could also be adversely affected as a result of such minimum guarantees. In addition, the long-term and fixed cost nature of these minimum guarantees may limit our flexibility in planning for, or reacting to, changes in our business and the market segments in which we operate.

We rely on estimates of the market share of licensable content controlled by each content provider to forecast whether such minimum guarantees will be recoupable against our actual content acquisition costs incurred over the duration of the license agreement. To the extent that these market share estimates are incorrect and our actual content acquisition costs for a particular content provider is less than the amount of the minimum guarantee, our margins may be materially and adversely affected.

The rates we must pay for "mechanical rights" to use musical works on our services are set by the Copyright Royalty Board, which is currently in the process of determining these rates for the five-year period beginning January 1, 2018. If these rates increase significantly, it will adversely affect our business.

Our direct licenses with thousands of music publishers provide that the content acquisition payments for "reproduction rights" or "mechanical rights"—which are implicated in the interactive features of our Pandora Plus service, as well as in the operation of Pandora Premium—are determined in accordance with the rate formula set by the Copyright Royalty Board for the compulsory license made available by Section 115 of the Copyright Act. Further, these rates are also applicable to our use of musical works for which we do not have a direct license with the copyright owners. The current rate structure for the Section 115 compulsory license expired at the end of 2017. The Copyright Royalty Board has rendered its initial determination of the new rate formula that will be applicable from January 1, 2018 through December 31, 2022 (the "Initial Determination"). If the Copyright Royalty Board adopts and affirms its Initial Determination, the "all-in" rate that streaming services, including Pandora, will pay to music publishers and songwriters for the mechanical rights and performance rights needed in connection with interactive streaming will increase annually between 2018 and 2022, with a total increase over that period of 32%. Certain per-subscriber minimums also apply depending on the type of service. If the Initial Determination is adopted and affirmed, and is not revised through an appeals process, our content acquisition costs will significantly increase, which could materially harm our financial condition and hinder our ability to provide interactive features in our services, or cause one or more of our subscription services to not be commercially viable.

If we are unsuccessful at operating Pandora Premium, our on-demand subscription offering, our business may be adversely affected.

The development and launch of our Pandora Plus and Pandora Premium service offerings has required and will continue to require significant engineering as well as marketing and other resources. In addition, to support the launch of these services we have entered into direct license agreements with the major record labels and publishers, several of which entail substantial minimum guaranteed content acquisition cost payments by us. There is no assurance that we will be able to successfully operate Pandora Premium, or obtain and maintain the content license rights to enable the continued offering of Pandora Plus and Pandora Premium. Further, in a number of cases, the rates for content acquisition costs payable under certain of our direct license agreements with record labels are significantly affected by the number of subscribers we are able to attract and retain to our subscription services, including Pandora Premium. If we fail to successfully operate Pandora Plus and/or Pandora Premium, or if we are not able to attract sufficient paying subscribers to those services, we will not realize the benefits of the substantial investment made in the development of these product offerings.

Our inability to obtain accurate and comprehensive information necessary to identify the ownership of sound recordings and musical works used on our services may impact our ability to perform our obligations under our licenses from the copyright holders, may require us to remove or decrease the number of performances of certain music on our services, and may subject us to potential copyright infringement claims and difficulties in controlling content acquisition costs.

We currently rely on the assistance of third parties to determine comprehensive and accurate rightsholder information for the sound recordings and the musical works underlying the sound recordings that we use on our services. If the information provided to us does not comprehensively or accurately identify which labels, artists, composers, songwriters or publishers own or administer sound recordings or musical works, or if we are unable to determine which musical works correspond to specific sound recordings, it may be difficult to identify the appropriate rightsholders to pay under our direct licenses, and it may also be difficult to comply with other obligations under those agreements. Further, our inability to accurately identify rights holders may prevent us from obtaining necessary additional licenses, which could lead to a reduction in the music available to stream on our service, adversely impacting our ability to retain and expand our listener base. Such a lack of ownership data may also make it difficult to identify the sound recordings that we should remove from our service, which may subject us to significant liability for copyright infringement.

We also rely on the assistance of third parties to provide notices of intent ("NOIs") for a compulsory license under Section 115 of the Copyright Act to those copyright owners with whom we do not have a direct license agreement or whose contact information is unavailable or unknown. If the third parties on which we rely do not provide NOIs to the correct parties (including the United States Copyright Office for unknown copyright owners), or to all parties who should receive an NOI, or do not serve the NOI in a timely manner, we may be subject to significant liability for copyright infringement.

Expansion of our operations into content beyond pre-recorded music, including comedy and podcasts, subjects us to additional business, legal, financial and competitive risks.

Expansion of our operations into delivery of content beyond pre-recorded music involves numerous risks and challenges, including increased capital requirements, new competitors and the need to develop new strategic relationships. Growth into these new areas may require changes to our existing business model and cost structure, modifications to our infrastructure and exposure to new regulatory and legal risks, including infringement liability, any of which may require additional expertise that we currently do not have. There is no guarantee that we will be able to generate sufficient revenue from advertising sales associated with comedy, podcasts or other non-precorded-music content to offset the costs of acquiring this content. Further, we have established a reputation as a music format internet radio provider and our ability to gain acceptance and listenership for comedy, podcasts or other non-music content, and thus our ability to attract advertisers to this content, is not certain. Failure to obtain or retain rights to comedy, podcasts or other non-acceptable terms, or at all, to successfully monetize and generate revenues from such content, or to effectively manage the numerous risks and challenges associated with such expansion could adversely affect our business and financial condition.

We are not able to obtain licenses for the underlying literary works for the sound recordings of spoken-word comedy content that we stream. Third parties could assert copyright claims against us as a result.

We stream spoken word comedy content, for which the underlying literary works are not currently entitled to eligibility for licensing by any performing rights organization in the United States. Rather, pursuant to industry-wide custom and practice, this content is performed absent a specific license from any such performing rights organization or individual rights owners, although content acquisition costs are paid to SoundExchange for the public performance of the sound recordings in which such literary works are embodied. There can be no assurance that this industry custom will not change or that we will not otherwise become subject to additional licensing costs for spoken word comedy content imposed by performing rights organizations or individual copyright owners in the future or be subject to claims of copyright infringement.

We may require additional capital to pursue our business objectives and respond to business opportunities, challenges or unforeseen circumstances. If capital is not available to us, our business, operating results and financial condition may be harmed.

Some of our current or future strategic initiatives may require substantial additional capital resources before they begin to generate revenue. Additional funds may not be available when we need them, on terms that are acceptable to us, or at all. For example, our current credit facility contains restrictive covenants relating to our capital raising activities and other financial and operational matters, and any debt financing secured by us in the future could involve further restrictive covenants, which may make it more difficult for us to obtain additional capital and to pursue business opportunities. In addition, volatility in the credit markets may have an adverse effect on our ability to obtain debt financing. If we do not have funds available to enhance the Pandora service, maintain the competitiveness of our technology and pursue business opportunities, we may not be able to service our existing listeners, acquire new listeners or attract or retain advertising customers, each of which could inhibit the implementation of our business plan and materially harm our operating results.

Our ability to expand our services into countries outside the United States is dependent on our ability to obtain necessary licenses from content owners, which we currently do not have.

We currently only operate the Pandora service in the United States. In order to launch any interactive or non-interactive music streaming service anywhere else in the world, we will need to negotiate and execute license agreements with the necessary rights holders, as there is no statutory licensing regime for digital streaming of music content available outside of the United States. We may not be able to obtain all of the necessary rights to expand our service offerings on commercially viable terms, or at all. If we are not able to obtain the necessary licenses on commercially viable terms, then we will not be able to expand our service internationally, and our growth could suffer materially. This could adversely affect our business, financial condition and results of operations.

We face many risks associated with our long-term plan to further expand our operations outside of the United States.

Expanding our operations into international markets is an element of our long-term strategy. However, offering our services outside of the United States involves numerous risks and challenges. Addressing content-licensing and other business issues in foreign jurisdictions may require a significant investments of time, capital and other resources by us, and there can be no assurance that we would succeed or achieve any return on these investments.

In addition, international expansion exposes us to other risks such as:

- the need to modify our technology and market our service in non-English speaking countries;
- the need to localize our service to foreign customers' preferences and customs;
- the need to conform our operations, and our marketing and advertising efforts, with the laws and regulations of foreign jurisdictions, including, but not limited to, the use of any personal information about our listeners;
- the need to amend existing agreements and to enter into new agreements with automakers, automotive suppliers, consumer electronics manufacturers with products that integrate our service, and others in order to provide that service in foreign countries;
- difficulties in managing operations due to language barriers, distance, staffing, cultural differences and business infrastructure constraints and domestic laws
 regulating corporations that operate internationally;
- our lack of experience in marketing, and encouraging viral marketing growth without incurring significant marketing expenses, in foreign countries;
- application of foreign laws and regulations to us;
- fluctuations in currency exchange rates;
- reduced or ineffective protection of our intellectual property rights in some countries; and
- potential adverse tax consequences associated with foreign operations and revenue.



Furthermore, in most international markets, we would not be the first entrant, and our competitors may be better positioned than we are to succeed. In addition, in jurisdictions where copyright protection has been insufficient to protect against widespread music piracy, achieving market acceptance of our service may prove difficult as we would need to convince listeners to stream our service when they could otherwise download the same music for free. As a result of these obstacles, we may find it impossible or prohibitively expensive to enter or sustain our presence in foreign markets, or entry into foreign markets could be delayed, which could hinder our ability to grow our business.

We have engaged and may in the future engage in the acquisition or disposition of other companies, technologies and businesses, which could divert our management's attention, result in additional dilution to our stockholders and otherwise disrupt our operations and harm our operating results.

We have recently acquired and may in the future seek to acquire or invest in businesses, products or technologies that we believe could complement or expand our service, enhance our technical capabilities or otherwise offer growth opportunities. For example, in 2015, we acquired Next Big Sound, Ticketfly and certain assets of Rdio. These acquisitions, and our pursuit of future potential acquisitions, may divert the attention of management and cause us to incur various expenses in identifying, investigating and pursuing suitable acquisitions, whether or not they are consummated. In addition, we have limited experience acquiring and integrating other businesses. We may be unsuccessful in integrating our acquired businesses or any additional business we may acquire in the future or may not otherwise realize anticipated benefits of past or future acquisitions. For example, in September 2017, we sold Ticketfly at a significant loss.

We also may not achieve the anticipated benefits from any acquired business due to a number of factors, including:

- unanticipated costs or liabilities associated with the acquisition;
- incurrence of acquisition-related costs;
- diversion of management's attention from other business concerns;
- regulatory uncertainties;
- harm to our existing business relationships with business partners and advertisers as a result of the acquisition;
- harm to our brand and reputation;
- the potential loss of key employees;
- use of resources that are needed in other parts of our business; and
- use of substantial portions of our available cash to consummate the acquisition.

In addition, a significant portion of the purchase price of companies we acquire may be allocated to acquired goodwill and other intangible assets, which must be assessed for impairment at least annually. In the future, if our acquisitions do not yield expected returns, we may be required to take charges to our operating results based on this impairment assessment process. Acquisitions could also result in dilutive issuances of equity securities or the incurrence of debt, which could adversely affect our operating results. In addition, if an acquired business fails to meet our expectations, our operating results, business and financial condition may suffer.

In 2017, we concluded that operating Ticketfly would no longer be part of our strategy, and as such we sold Ticketfly to a third party. The sale price in the disposition of Ticketfly was substantially less that we paid to acquire Ticketfly in 2015. The disposition also required significant attention by our management and board of directors and caused us to incur significant expenses related to the sale. While we entered into a commercial arrangement with the purchaser that we anticipate will afford us some of the benefits we had sought to obtain when we purchased Ticketfly, we may not realize the expected benefits of this commercial agreement. Further, we will be subject to potential liability for indemnities we have agreed to in the sale agreement.

We rely upon an agreement with DoubleClick, which is owned by Google, for delivering and monitoring most of our ads. Failure to renew the agreement on favorable terms, or termination of the agreement, could adversely affect our business.



We use DoubleClick's ad-serving platform to deliver and monitor most of the ads for our ad-supported service. There can be no assurance that our agreement with DoubleClick, which is owned by Google, will be extended or renewed upon expiration, that we will be able to extend or renew our agreement with DoubleClick on terms and conditions favorable to us or that we could identify another alternative vendor or develop our own technology to take its place. Our agreement with DoubleClick also allows DoubleClick to terminate our relationship before the expiration of the agreement on the occurrence of certain events, including material breach of the agreement by us, and to suspend provision of the services if DoubleClick determines that our use of its service violates certain security, technology or content standards.

We rely on third parties to provide software and related services necessary for the operation of our business.

We incorporate and include third-party software into and with our apps and service offerings and expect to continue to do so. The operation of our apps and service offerings could be impaired if errors occur in the third-party software that we use. It may be more difficult for us to correct any defects in third-party software because the development and maintenance of the software is not within our control. Accordingly, our business could be adversely affected in the event of any errors in this software. There can be no assurance that any third-party licensors will continue to make their software available to us on acceptable terms, to invest the appropriate levels of resources in their software to maintain and enhance its capabilities, or to remain in business. Any impairment in our relationship with these third-party licensors could harm our ability to maintain and expand the reach of our service, increase listener hours and sell advertising, each of which could harm our operating results, cash flow and financial condition.

Digital music streaming is an evolving industry, which makes it difficult to evaluate our near- and long-term business prospects.

Digital music streaming continues to develop as an industry and our near- and long-term business prospects are difficult to evaluate. The marketplace for digital music streaming is subject to significant challenges and new competitors. As a result, the future revenue, income and growth potential of our business is uncertain. Investors should consider our business and prospects in light of the risks and difficulties we encounter in this evolving business, which risks and difficulties include, among others, risks related to:

- our evolving business model and new licensing models for content as well as the potential need for additional types of content;
- our ability to develop additional products and services, or products and services in adjacent markets, in order to maintain revenue growth, and the resource requirements of doing so;
- our ability to retain current levels of active listeners, build our listener base and increase listener hours;
- our ability to effectively monetize listener hours by growing our sales of advertising inventory created from developing new and compelling ad product solutions
 that successfully deliver advertisers' messages across the range of our delivery platforms while maintaining our listener experience;
- our ability to attract new advertisers, retain existing advertisers and prove to advertisers that our advertising platform is effective enough to justify a pricing structure that is profitable for us;
- our ability to maintain relationships with platform providers, makers of mobile devices, consumer electronic products and automobiles; and
- our ability to continue to secure the rights to music and other content that attracts listeners to the service on fair and reasonable economic terms.

Failure to successfully address these risks and difficulties and other challenges associated with operating in an evolving marketplace could materially and adversely affect our business, financial condition and results of operations.

We have incurred significant operating losses in the past and may not be able to generate sufficient revenue to be profitable.

Since our inception in 2000, we have incurred significant net operating losses and, as of December 31, 2017, we had an accumulated deficit of \$1.3 billion. A key element of our strategy is to increase the number of listeners and listener hours to increase our industry penetration, including the number of listener hours on mobile and other connected devices. However, as our number of listener hours increases, our content acquisition costs also increase. In addition, we have adopted a strategy to

invest in our operations in advance of, and to drive, future revenue growth. This strategy includes recently completed acquisitions and other initiatives. As a result of these trends, we have not in the past generated, and may not in the future generate, sufficient revenue from the sale of advertising and subscriptions, or new revenue sources, to offset our expenses. In addition, we plan to continue to invest heavily in our operations to restart growth. As a result of these factors, we expect to incur annual net losses in the near term.

Our revenue increased rapidly in some historical periods, but the growth rate has slowed in recent periods. We do not expect to sustain our high revenue growth rates in the future as a result of a variety of factors, including increased competition and the maturation of our business, and we cannot guarantee that our revenue will continue to grow or will not decline. Investors should not consider our historical revenue growth or operating expenses as indicative of our future performance. If revenue growth is lower than our expectations, or our operating expenses exceed our expectations, our financial performance will be adversely affected. Further, if our future growth and operating performance fail to meet investor or analyst expectations, it could have a material adverse effect on our stock price.

If we fail to effectively manage our growth, our business and operating results may suffer.

Our rapid growth has placed, and will continue to place, significant demands on our management and our operational and financial infrastructure. In order to attain and maintain profitability, we will need to recruit, integrate and retain skilled and experienced sales personnel who can demonstrate our value proposition to advertisers and increase the monetization of listener hours, particularly on mobile devices, by developing relationships with both national and local advertisers to convince them to migrate advertising spending to online and mobile digital advertising markets and utilize our advertising product solutions. Continued growth could also strain our ability to maintain reliable service levels for our listeners, effectively monetize our listener hours, develop and improve our operational, financial and management controls and enhance our reporting systems and procedures. If our systems do not evolve to meet the increased demands placed on us by an increasing number of advertisers, we may also be unable to meet our obligations under advertising areements with respect to the timing of our delivery of advertising or other performance obligations. As our operations grow in size, scope and complexity, we will need to improve and upgrade our systems and infrastructure, which will require significant expenditures and allocation of valuable management resources. If we fail to maintain the necessary level of discipline and efficiency and allocate limited resources effectively in our organization as it grows, our business, operating results and financial condition may suffer.

Our business and prospects depend on the strength of our brands and failure to maintain and enhance our brands would harm our ability to expand our base of listeners, advertisers and other partners.

Maintaining and enhancing the "Pandora" and "Next Big Sound" brands is critical to expanding our base of listeners, advertisers, content owners and other partners. Maintaining and enhancing our brands will depend largely on our ability to continue to develop and provide an innovative and high quality experience for our listeners and attract advertisers, content owners, and automobile, mobile device and other consumer electronic product manufacturers to work with us, which we may not do successfully.

Our brands may be impaired by a number of other factors, including service outages, data privacy and security issues, listener perception of ad load and exploitation of our trademarks by others without permission. In addition, if our partners fail to maintain high standards for products that integrate our service, or if we partner with manufacturers of products that our listeners reject, the strength of our brand could be adversely affected.

Assertions by third parties of violations under state law with respect to the public performance and reproduction of pre-1972 sound recordings could result in significant costs and substantially harm our business and operating results.

Federal copyright protection does not apply to sound recordings created prior to February 15, 1972 ("pre-1972 sound recordings"). The protection of such recordings is instead governed by a patchwork of state statutory and common laws. Copyright owners of pre-1972 sound recordings have brought litigation against us in a number of states, alleging violations of state statutory and common laws arising from the reproduction and public performance of pre-1972 sound recordings. Despite settling one such suit with the major record labels in October 2015, and entering into direct license agreements for recorded music with major and independent labels, distributors and publishers during the first nine months of 2016 that also permit us to stream certain pre-1972 sound recordings, we still face a number of class-action suits brought by various plaintiffs who seek, among other things, restitution, disgorgement of profits, and punitive damages as well as injunctive relief prohibiting further violation of those copyright owners' alleged exclusive rights.

If we are found liable for the violation of the exclusive rights of any pre-1972 sound recording copyright owners, then we could be subject to liability, the amount of which could be significant. Similarly, any settlements of the remaining litigation



could require substantial payments. If we are required to obtain licenses from individual sound recording copyright owners for the reproduction and public performance of pre-1972 sound recordings, then the time, effort and cost of securing such licenses directly from all owners of sound recordings used on our service could be significant and could harm our business and operating results. If we are required to obtain licenses for pre-1972 sound recordings to avoid liability and are unable to secure such licenses, then we may have to remove pre-1972 sound recordings from our service, which could harm our ability to attract and retain users.

We could be adversely affected by regulatory restrictions on the use of mobile and other electronic devices in motor vehicles and legal claims arising from use of such devices while driving.

Regulatory and consumer agencies have increasingly focused on distraction to drivers that may be associated with use of mobile and other devices in motor vehicles. In 2010, the U.S. Department of Transportation identified driver distraction as a top priority, and in April 2013, the National Highway Traffic Safety Administration (the "NHTSA") released voluntary Phase 1 Driver Distraction Guidelines for visual-manual devices not related to the driving task that are integrated into motor vehicles. In November 2016, NHTSA released its proposed voluntary Phase 2 Driver Distraction Guidelines for portable and aftermarket devices that may be used in motor vehicles, which include recommendations such as automatic mobile device pairing with motor vehicle entertainment systems and lock-outs of mobile devices while being used by the driver of a motor vehicle. If NHTSA or other agencies implemented regulatory restrictions and took enforcement action related to how drivers and passengers in motor vehicles may engage with devices on which our service is broadcast, such restrictions or enforcement actions could inhibit our ability to increase listener hours and generate ad revenue, which would harm our operating results. In addition, concerns over driver distraction due to use of mobile and other electronic devices used to access our service in motor vehicles could result in product liability or personal injury litigation and negative publicity.

Federal, state and industry regulations as well as self-regulatory programs related to privacy and data security pose the threat of lawsuits and other liability, require us to expend significant resources for compliance, and may hinder our ability and our advertisers' ability to deliver certain types of advertising.

We collect and utilize demographic and other information from and about our listeners and artists as they interact with our service, including information which could fall under a definition of "personally identifiable information" under various state and federal laws. For example, to register for a Pandora account, our listeners self-report the following information: birth year, gender, zip code and e-mail address. Listeners must also provide their credit card or debit card numbers and other billing information if they choose to take advantage of additional service offerings, such as Pandora Plus and Pandora Premium. We also may collect location information from listeners who have enabled this feature on their mobile or other devices, information listeners provide on their profile page or in using other community or social networking features that are part of our service, or information listeners provide when they participate in polls or contests or sign up to receive e-mail newsletters. Further, we and third parties use tracking technologies, including "cookies" and related technologies, to help us manage and track our listeners' interactions with our service and deliver relevant advertising. We also collect information from and track artists' activity on our Pandora Artist Marketing Platform. Third parties may, either without our knowledge or consent, or in violation of contractual prohibitions, obtain, transmit or utilize our listeners' or artists' personally identifiable information, or data associated with particular users, devices or artists.

Various federal and state laws and regulations, as well as the laws of foreign jurisdictions in which we may choose to operate, govern the collection, use, retention, sharing and security of the data we receive from and about our listeners. Privacy groups and government authorities have increasingly scrutinized the ways in which companies link personal identities and data associated with particular users or devices with data collected through the internet, and we expect such scrutiny to continue to increase. Alleged violations of laws and regulations relating to privacy and data security, and any relevant claims, may expose us to potential liability and may require us to expend significant resources in responding to and defending such allegations and claims. Claims or allegations that we have violated laws and regulations relating to privacy and data security have resulted and could in the future result in negative publicity and a loss of confidence in us by our listeners and our advertisers.

Existing privacy-related laws and regulations, and new privacy-related regulations such as the European General Data Protection Regulation (GDPR), are evolving and subject to potentially differing interpretations, and various federal and state legislative and regulatory bodies, as well as foreign legislative and regulatory bodies, may expand current or enact new laws regarding privacy and data security-related matters. We may find it necessary or desirable to join self-regulatory bodies or other privacy-related organizations that require compliance with their rules pertaining to privacy and data security. We are also bound by our public-facing privacy statement, which sets forth the ways in which we collect, use and share information. We also may be bound by contractual obligations that limit our ability to collect, use, disclose and leverage listener data and to derive economic value from it. New laws, amendments to or re-interpretations of existing laws, rules of self-regulatory bodies, industry standards and contractual obligations, as well as changes in our listeners' expectations and demands regarding privacy



and data security, may limit our ability to collect, use and disclose, and to leverage and derive economic value from listener data. We may also be required to expend significant resources to adapt to these changes and to develop new ways to deliver relevant advertising or otherwise provide value to our advertisers. In particular, government regulators have proposed, and the GDPR will require, that services implement "do not track" mechanisms, and requirements that users affirmatively "opt-in" to certain types of data collection and use. Such mechanisms and requirements, to the extent required by government regulators or self-regulatory bodies, could significantly hinder our ability to collect and use data relating to listeners. Restrictions on our ability to collect, access and harness listener data, or to use or disclose listener data or any profiles that we develop using such data, could in turn limit our ability to stream personalized music content to our listeners and offer targeted advertising opportunities to our advertising customers, each of which are critical to the success of our business.

We have incurred, and will continue to incur, expenses to comply with privacy and security standards and protocols imposed by law, regulation, self-regulatory bodies, industry standards and contractual obligations. Increased regulation of data utilization and distribution practices, including self-regulation and industry standards, could increase our cost of operation, limit our ability to grow our operations or otherwise adversely affect our business.

Government regulation of the internet is evolving, and unfavorable developments could have an adverse effect on our operating results.

We are subject to general business regulations and laws, as well as regulations and laws specific to the internet. Such laws and regulations cover sales and other taxes and withholding of taxes, user privacy, data collection and protection, copyrights, electronic contracts, sales procedures, automatic subscription renewals, credit card processing procedures, consumer protections, broadband internet access and content restrictions. We cannot guarantee that we have been or will be fully compliant in every jurisdiction, as it is not entirely clear how existing laws and regulations governing issues such as privacy, taxation and consumer protection apply to the internet. Moreover, as internet commerce continues to evolve, increasing regulation by federal, state and foreign agencies becomes more likely. The adoption of any laws or regulations that adversely affect the popularity or growth in use of the internet, including laws limiting internet neutrality, could decrease listener demand for our service offerings and increase our cost of doing business. Future regulational historical or future liabilities for us, resulting in adverse impacts on our business and our operating results.

Our operating results may fluctuate, which makes our results difficult to predict and could cause our results to fall short of expectations.

Our revenue and operating results could vary significantly from quarter to quarter and year to year due to a variety of factors, many of which are outside our control. As a result, comparing our operating results on a period-to-period basis may not be meaningful. In addition to other risk factors discussed in this "Risk Factors" section, factors that may contribute to the variability of our quarterly and annual results include:

- costs associated with pursuing licenses or other commercial arrangements;
- costs associated with defending any litigation, including intellectual property infringement litigation, and any associated judgments or settlements;
- our ability to pursue, and the timing of, entry into new geographic or content markets or other strategic initiatives and, if pursued, our management of these
 initiatives;
- the impact of general economic and competitive conditions on our revenue and expenses; and
- changes in government regulation affecting our business.

Seasonal variations in listener and advertising behavior may also cause fluctuations in our financial results. We expect to experience some effects of seasonal trends in listener behavior due to higher advertising sales during the fourth quarter of each year due to greater advertiser demand during the holiday season and lower advertising sales in the first quarter of the following year. Expenditures by advertisers tend to be cyclical and discretionary in nature, reflecting overall economic conditions, the economic prospects of specific advertisers or industries, budgeting constraints and buying patterns and a variety of other factors, many of which are outside our control. In addition, we expect to experience increased usage during the fourth quarter of each year due to the holiday season, and in the first quarter of each year due to increased use of media-streaming devices received as gifts during the holiday season.



If we are unable to maintain effective internal control over financial reporting in the future, the accuracy and timeliness of our financial reporting may be adversely affected.

Pursuant to Section 404 of the Sarbanes-Oxley Act of 2002, we are required to furnish a report by our management on our internal control over financial reporting. The report contains, among other matters, an assessment of the effectiveness of our internal control over financial reporting as of year-end, including a statement as to whether or not our internal control over financial reporting is effective. This assessment must include disclosure of any material weaknesses in our internal control over financial reporting identified by management.

While we have determined that our internal control over financial reporting was effective as of December 31, 2017, as indicated in "Controls and Procedures-Evaluation of Disclosure Controls and Procedures", we must continue to monitor and assess our internal control over financial reporting. If our management identifies one or more material weaknesses in our internal control over financial reporting in 2018 and such weakness remains uncorrected at year-end, we will be unable to assert that such internal control is effective at year-end. If we are unable to assert that our internal control over financial reporting is effective at year-end, or if our independent registered public accounting firm is unable to express an opinion on the effectiveness of our internal controls or concludes that we have a material weakness in our internal controls, we could lose investor confidence in the accuracy and completeness of our financial reports, which could have a material adverse effect on our business and the price of our common stock.

Failure to protect our intellectual property could substantially harm our business and operating results.

The success of our business depends, in part, on our ability to protect and enforce our trade secrets, trademarks, copyrights and patents and all of our other intellectual property rights, including our intellectual property rights underlying the Pandora service. To establish and protect those proprietary rights, we rely on a combination of patents, patent applications, trademarks, copyrights, trade secrets (including know-how), license agreements, information security procedures, non-disclosure agreements with third parties, employee disclosure and invention assignment agreements, and other contractual obligations. These afford only limited protection. Despite our efforts to protect our intellectual property rights, unauthorized parties may copy or attempt to copy aspects of our technology. Moreover, policing our intellectual property rights is difficult, costly and may not always be effective.

We have filed, and may in the future file, patent applications and from time to time we have purchased patents and patent applications from third parties. It is possible, however, that these innovations may not be protectable. In addition, given the cost, effort, risks and downside of obtaining patent protection, including the requirement to ultimately disclose the invention to the public, we may choose not to seek patent protection for certain innovations. However, such patent protection could later prove to be important to our business. Furthermore, there is always the possibility that our patent applications may not issue as granted patents, that the scope of the protection gained will be insufficient or that an issued patent may be deemed invalid or unenforceable. Moreover, in certain circumstances there are additional risks, including:

- present or future patents or other intellectual property rights could lapse or be invalidated, circumvented, challenged or abandoned;
- our ability to assert our intellectual property rights against potential competitors or to settle current or future disputes may be limited by our relationships with third parties;
- our pending or future patent applications may not have coverage sufficient to provide the desired competitive advantage; and
- our intellectual property rights may not be enforced in jurisdictions where competition may be intense or where legal protection may be weak.

We have registered "Pandora," "Music Genome Project", "Next Big Sound", and other marks as trademarks in the United States and other countries. Nevertheless, competitors may adopt service names similar to ours, or purchase confusingly similar terms as keywords in internet search engine advertising programs, thereby impeding our ability to build brand identity and possibly leading to confusion among our listeners or advertising customers. In addition, there could be potential trade name or trademark infringement claims brought by owners of other registered trademarks or trademarks that incorporate variations of the term Pandora or our other trademarks. Any claims or customer confusion related to our trademarks could damage our reputation and brand and substantially harm our business and operating results.

We currently own the www.pandora.com internet domain name and various other domain names related to our business. The regulation of domain names in the United States and in foreign countries is subject to change. Regulatory bodies continue to establish additional top-level domains, appoint additional domain name registrars and modify the requirements for holding domain names. As a result, we may not be able to acquire or maintain the domain name sthat utilize our brand names in the United States or other countries in which we may conduct business in the future. If we lose or fail to acquire the right to use any domain name relevant to our current or future business in the United States or other countries, we may incur significant additional expense.

In order to protect our trade secrets and other confidential information, we rely in part on confidentiality agreements with our employees, consultants and third parties with whom we have relationships. These agreements may not effectively prevent disclosure of trade secrets and other confidential information and may not provide an adequate remedy in the event of misappropriation of trade secrets or any unauthorized disclosure of trade secrets and other confidential information. In addition, others may independently discover the same subject matter as that covered by our trade secrets and confidential information, and in some such cases we might not be able to assert any trade secret rights against such parties. Costly and time-consuming litigation could be necessary to enforce and determine the scope of our trade secret rights and related confidentiality and nondisclosure provisions, and failure to obtain or maintain trade secret protection, or our competitors' independent development of technology similar to ours for which we are unable to rely on other forms of intellectual property protection such as patents, could adversely affect our competitive business position.

Litigation or proceedings before the U.S. Patent and Trademark Office or other governmental authorities and administrative bodies in the United States and abroad may be necessary in the future to enforce our intellectual property rights, to protect our patent rights, trademarks, trade secrets and domain names and to determine the validity and scope of the proprietary rights of others. Our efforts to enforce or protect our proprietary rights may be ineffective and could result in substantial costs and diversion of resources and management time, each of which could substantially harm our operating results.

Assertions by third parties of infringement or other violation by us of their intellectual property rights could result in significant costs and substantially harm our business and operating results.

Internet, technology and media companies are frequently subject to litigation based on allegations of infringement, misappropriation or other violations of others' intellectual property rights. Some internet, technology and media companies, including some of our competitors, own large numbers of patents, copyrights, trademarks and trade secrets, which they may use to assert claims against us. In addition, we encourage third parties to submit content for our catalog and we cannot be assured that artist representations made in connection with such submissions accurately reflect the legal rights of the submitted content. Third parties have asserted, and may in the future assert, that we have infringed, misappropriated or otherwise violated their intellectual property rights. In addition, various federal and state laws and regulations govern the intellectual property and related rights associated with sound recordings and musical works. Existing laws and regulations are evolving and subject to different interpretations, and various federal and state legislative or regulatory bodies may expand current or enact new laws or regulations. We cannot guarantee that we are not infringing or violating any third-party rights.

We cannot predict whether assertions of third-party intellectual property rights or any infringement or misappropriation claims arising from such assertions will substantially harm our business and operating results. When we are forced to defend against any infringement or misappropriation claims, we may be required to expend significant time and financial resources on the defense of such claims, even if without merit, settled out of court, or adjudicated in our favor. Furthermore, an adverse outcome of a dispute may require us to: pay damages (potentially including treble damages and attorneys' fees if we are found to have willfully infringed a party's intellectual property); cease making, licensing or using products or services that are alleged to infringe or misappropriate the intellectual property of others; expend additional development resources to redesign our services to avoid infringement; enter into potentially unfavorable content acquisition or license agreements in order to obtain the right to use necessary technologies, content or materials; or to indemnify our partners and other third parties. We do not carry broadly applicable patent liability insurance and lawsuits regarding patent rights, regardless of their success, can be expensive to resolve and can divert the time and attention of our management and technical personnel.

Some of our services and technologies may use "open source" software, which may restrict how we use or distribute our service or require that we release the source code of certain services subject to those licenses.

Some of our services and technologies may incorporate software licensed under "open source" licenses. Such open source licenses often require that source code subject to the license be made available to the public and that any modifications or derivative works to open source software continue to be licensed under open source licenses. Few courts have interpreted open source licenses, and the manner in which these licenses may be interpreted and enforced is therefore subject to some

uncertainty. We rely on multiple employee and non-employee software programmers to design our proprietary technologies, and since we may not be able to exercise complete control over the development efforts of all such programmers we cannot be certain that they have not incorporated open source software into our products and services without our knowledge, or that they will not do so in the future. In the event that portions of our proprietary technology are determined to be subject to an open source license, we may be required to publicly release the affected portions of our source code, be forced to re-engineer all or a portion of our technologies, or otherwise be limited in the licensing of our technologies, each of which could reduce the value of our services and technologies and materially and adversely affect our ability to sustain and grow our business.

Interruptions or delays in service arising from our own systems or from our third-party vendors could impair the delivery of our service and harm our business.

We rely on systems housed at our own premises and at those of third-party vendors, including network service providers and data center facilities, to enable listeners to stream our content in a dependable and efficient manner. We have experienced and expect to continue to experience periodic service interruptions and delays involving our own systems and those of our third-party vendors. In the event of a service outage at our main site, we maintain a backup site that can function in read-only capacity. We do not currently maintain live fail-over capability that would allow us to instantaneously switch our streaming operations from one facility to another in the event of a service outage. In the event of an extended service outage at our main site, we do maintain and test fail-over capabilities that should allow us to switch our live streaming operations from one facility to another. Both our own facilities and those of our third-party vendors are vulnerable to damage or interruption from earthquakes, floods, fires, power loss, telecommunications failures and similar events. They also are subject to break-ins, hacking, denial of service attacks, sabotage, intentional acts of vandalism, terrorist acts, natural disasters, human error, the financial insolvency of our third-party vendors and other unanticipated problems or events. The occurrence of any of these events could result in interruptions in our service and to unauthorized access to, or alteration of, the content and data contained on our systems and that these third-party vendors store and deliver on our behalf.

We do not exercise complete control over our third-party vendors, which makes us vulnerable to any errors, interruptions, or delays in their operations. Any disruption in the services provided by these vendors could have significant adverse impacts on our business reputation, customer relations and operating results. Upon expiration or termination of any of our agreements with third-party vendors, we may not be able to replace the services provided to us in a timely manner or on terms and conditions, including service levels and cost, that are favorable to us, and a transition from one vendor to another vendor could subject us to operational delays and inefficiencies until the transition is complete.

If our security systems are breached, we may face civil liability and public perception of our security measures could be diminished, either of which would negatively affect our ability to attract and retain listeners and advertisers.

Techniques used to gain unauthorized access to corporate data systems are constantly evolving, and we may be unable to anticipate or prevent unauthorized access to data pertaining to our listeners, including credit card and debit card information and other personally identifiable information. Like all internet services, our service, which is supported by our own systems and those of third-party vendors, is vulnerable to computer malware, Trojans, viruses, worms, break-ins, phishing attacks, denial-of-service attacks, attempts to access our servers to acquire playlists or stream music in an unauthorized manner, or other attacks on and disruptions of our and third-party vendor computer systems, any of which could lead to system interruptions, delays, or shutdowns, causing loss of critical data or the unauthorized access to personally identifiable information. If an actual or perceived breach of security occurs on our systems or a vendor's systems, we may face civil liability and reputational damage, either of which would negatively affect our ability to attract and retain listeners, which in turn would harm our efforts to attract and retain advertisers. We also would be required to expend significant resources to mitigate the breach of security and to address related matters. Unauthorized access to music or playlists would potentially create additional content acquisition cost obligations with no corresponding revenue.

We may not be able to effectively control the unauthorized actions of third parties who may have access to the listener data we collect. The integration of the Pandora service with apps provided by third parties represents a significant growth opportunity for us, but we may not be able to control such third parties' use of listeners' data, ensure their compliance with the terms of our privacy policies, or prevent unauthorized access to, or use or disclosure of, listener information, any of which could hinder or prevent our efforts with respect to growth opportunities.

Any failure, or perceived failure, by us to maintain the security of data relating to our listeners and employees, to comply with our posted privacy policy, laws and regulations, rules of self-regulatory organizations, industry standards and contractual provisions to which we may be bound, could result in the loss of confidence in us, or result in actions against us by governmental entities or others, all of which could result in litigation and financial losses, and could potentially cause us to lose listeners, artists, advertisers, revenue and employees.

We are subject to a number of risks related to credit card and debit card payments we accept.

We accept subscription payments through credit and debit card transactions. For credit and debit card payments, we pay interchange and other fees, which may increase over time. An increase in those fees would require us to either increase the prices we charge for our products, which could cause us to lose subscribers and subscription revenue, or absorb an increase in our operating expenses, either of which could harm our operating results.

If we or any of our processing vendors have problems with our billing software, or the billing software malfunctions, it could have an adverse effect on our subscriber satisfaction and could cause one or more of the major credit card companies to disallow our continued use of their payment products. In addition, if our billing software fails to work properly and, as a result, we do not automatically charge our subscribers' credit cards on a timely basis or at all, or there are issues with financial insolvency of our third-party vendors or other unanticipated problems or events, we could lose subscription revenue, which would harm our operating results.

We are also subject to payment card association operating rules, certification requirements and rules governing electronic funds transfers, which could change or be reinterpreted to make it more difficult for us to comply. We are currently accredited against, and in compliance with, the Payment Card Industry Data Security Standard, or PCI DSS, the payment card industry's security standard for companies that collect, store or transmit certain data regarding credit and debit cards, credit and debit card holders and credit and debit card transactions. Currently we comply with PCI DSS version 3.2 as a Level 2 merchant. There is no guarantee that we will maintain PCI DSS compliance. Our failure to comply fully with PCI DSS in the future could violate payment card association operating rules, federal and state laws and regulations and the terms of our contracts with payment processors and merchant banks. Such failure to comply fully also could subject us to fines, penalties, damages and civil liability, and could result in the loss of our ability to accept credit and debit card payments. Further, there is no guarantee that PCI DSS compliance will prevent illegal or improper use of our payment systems or the theft, loss, or misuse of data pertaining to credit and debit cards, credit and debit card holders and credit and debit card transactions.

If we fail to adequately control fraudulent credit card transactions, we may face civil liability, diminished public perception of our security measures and significantly higher credit card-related costs, each of which could adversely affect our business, financial condition and results of operations. If we are unable to maintain our chargeback rate or refund rates at acceptable levels, credit card and debit card companies may increase our transaction fees or terminate their relationships with us. Any increases in our credit card and debit card fees could adversely affect our results of operations, particularly if we elect not to raise our rates for our service to offset the increase. The termination of our ability to process payments on any major credit or debit card would significantly impair our ability to operate our business.

Our ability to use our net operating loss carryforwards and certain other tax attributes may be limited.

At December 31, 2017, we had federal net operating loss carryforwards of approximately \$972.7 million and tax credit carryforwards of approximately \$20.5 million. At December 31, 2017, we had state net operating loss carryforwards of approximately \$640.2 million and tax credit carryforwards of approximately of \$28.2 million. Under Sections 382 and 383 of the Internal Revenue Code of 1986, as amended, ("the Code"), if a corporation undergoes an "ownership change," the corporation's ability to use its pre-change net operating loss carryforwards and other pre-change tax attributes, such as research tax credits, to offset its post-change income may be limited. In general, an "ownership change" will occur if there is a cumulative change in our ownership by "5-percent stockholders" that exceeds 50 percentage points over a rolling three-year period. Similar rules may apply under state tax laws. As a result of prior equity issuances and other transactions in our stock, we have previously experienced "ownership changes" under section 382 of the Code and comparable state tax laws. We may also experience ownership changes in the future as a result of other future transactions in our stock. As a result of we carn net taxable income, our ability to use our pre-change net operating loss carryforwards or other pre-change tax attributes to offset United States federal and state taxable income may be subject to limitations.

Changes in U.S. tax laws could have a material adverse effect on our business, cash flow, results of operations or financial conditions.

The Tax Cuts and Jobs Act ("the Act") was enacted on December 22, 2017 and contains many significant changes to U.S. Federal tax laws. The Act requires computations that were not previously provided for under U.S. tax law. The Company has provided for an estimated effect of the Act in its financial statements. The Act requires significant judgments to be made in interpretation of the law and significant estimates in the calculation of the provision for income taxes. However, additional guidance may be issued by the IRS, Department of the Treasury, or other governing body that may significantly



differ from the Company's interpretation of the law, which may result in a material adverse effect on our business, cash flow, results of operations or financial conditions.

We could be subject to additional income tax liabilities.

We are subject to income taxes in the United States and in certain foreign jurisdictions. If we expand our operations outside the United States, we would become subject to taxation based on the applicable foreign statutory rates and our effective tax rate could fluctuate accordingly. Significant judgment would be required in evaluating and estimating our worldwide provision for income taxes and accruals for these taxes. For example, our effective tax rates could be adversely affected by earnings being lower than anticipated in countries where we have lower statutory tax rates and higher than anticipated in countries where we have higher statutory tax rates, by losses incurred in jurisdictions for which we are not able to realize the related tax benefit, by changes in foreign currency exchange rates, by changes in the valuation of our deferred tax assets and liabilities, or by changes in the relevant tax, accounting and other laws, regulations, principles and interpretations. We are also subject to tax audits in various jurisdictions, and such jurisdictions may assess additional income tax liabilities against us.

We remain liable for potential sales tax and other tax liabilities related to our ownership of the Ticketfly business.

We closed the sale of Ticketfly to Eventbrite pursuant to the Ticketfly Purchase Agreement on September 1, 2017, but we remain obligated to indemnify Eventbrite for any taxes, including indirect taxes, owed with respect to any period ending on or before the closing date of the sale of Ticketfly to Eventbrite (including any period prior to our acquisition of Ticketfly). As described below, such indirect taxes for which we are required to indemnify Eventbrite could be material.

The application of indirect taxes (such as sales, use, excise, admissions, amusement, entertainment or other transaction-based taxes) to internet-based live entertainment ticketing businesses such as Ticketfly is a complex and evolving area. Many of the fundamental statutes and regulations that impose these taxes were established before the adoption and growth of the internet and ecommerce. In many cases, it is not clear how existing statutes apply to the internet or ecommerce. In addition, governments are increasingly looking for ways to increase revenues, which has resulted in discussions about tax reform and other legislative action to increase tax revenues, including through indirect taxes.

Ticketfly is not the seller of tickets sold on the Ticketfly platform. Instead it facilitates the transaction between its venue partners and customers. If a taxing jurisdiction were to treat Ticketfly as the seller and liable for the tax of the venue partners or customers, it could result in a material liability.

During the time Ticketfly was owned by Pandora, Ticketfly did not calculate all applicable indirect taxes on the fees charged when a customer purchased tickets on the Ticketfly platform. Some jurisdictions may interpret their law in a manner that would require Ticketfly to calculate, collect and remit the applicable indirect taxes on the entire charges. Such an interpretation could negatively impact our indemnity obligations pursuant to the sale of Ticketfly.

The issuance of shares of our Series A redeemable convertible preferred stock to Sirius XM dilutes the ownership of holders of our common stock and may adversely affect the market price of our common stock.

The Sirius XM Investment Agreement provided that (i) 172,500 shares of Series A Preferred Stock would be issued and sold to Sirius XM on June 9, 2017 (the "Initial Closing") and (ii) the remaining 307,500 shares would be issued and sold to Sirius XM at a future date (the "Additional Closing"), subject to the satisfaction of certain customary closing conditions.

The Initial Closing occurred on June 9, 2017, whereby Sirius XM paid to the Company \$172.5 million in exchange for 172,500 shares of Series A redeemable convertible preferred stock. The Additional Closing occurred on September 22, 2017 whereby the Company issued and sold to Sirius XM 307,500 shares of Series A Preferred Stock for \$307.5 million. As of December 31, 2017, these shares represented approximately 15% of our outstanding common stock, on an as-converted basis. Holders of Series A redeemable convertible preferred stock are entitled to a cumulative dividend at the rate of 6.0% per annum, payable quarterly in arrears. Beginning on September 22, 2017, the Series A redeemable convertible preferred stock is convertible at the option of the holders at any time into shares of common stock at an initial conversion price of \$10.50 per share of common stock and an initial conversion rate of 95.2381 shares of common stock per share of Series A redeemable convertible preferred stock, subject to certain customary anti-dilution adjustments. Any conversion of Series A redeemable convertible preferred stock may be settled by the Company, at our option, in shares of common stock, cash or any combination thereof. However, subject to explicit stockholder approval, the Series A redeemable convertible preferred stock may not be converted into more than 19.99% of our outstanding common stock.



The conversion of the Series A redeemable convertible preferred stock to common stock would dilute the ownership interest of existing holders of our common stock. Furthermore, any sales in the public market of the common stock issuable upon conversion of the Series A redeemable convertible preferred stock could adversely affect prevailing market prices of our common stock. We granted Sirius XM customary registration rights in respect of their shares of Series A redeemable convertible preferred stock and any shares of common stock issued upon conversion of the Series A redeemable convertible preferred stock and any shares of common stock issued upon conversion of the Series A redeemable convertible preferred stock. These registration rights would facilitate the resale of such securities into the public market, and any such resale would increase the number of shares of our common stock available for public trading. Sales by Sirius XM of a substantial number of shares of our common stock in the public market, or the perception that such sales might occur, could have a material adverse effect on the price of our common stock.

Our Series A redeemable convertible preferred stock has rights, preferences and privileges that are not held by, and are preferential to, the rights of our common stockholders, which could adversely affect our liquidity and financial condition, may result in the interests of Sirius XM differing from those of our common stockholders and could delay or prevent an attempt to take over our Company.

Sirius XM, as the holder of our Series A redeemable convertible preferred stock, has the right to receive a liquidation preference entitling it to be paid out of our assets available for distribution to stockholders before any payment may be made to holders of our common stock in an amount equal to the sum of (a) \$1,000 per share liquidation preference and (b) all accrued dividends as of the date of the liquidation.

In addition, dividends on the Series A redeemable convertible preferred stock accrue and are cumulative at the rate of 6.0% per annum, payable quarterly in arrears. As long as the Series A redeemable convertible preferred stock dividends are in arrears, we may not declare any dividend or make any distributions relating to common stock or redeem any common stock, subject to certain exceptions (including redemptions pursuant to employment contracts and benefit plans).

Sirius XM, as the holder of our Series A Preferred Stock also has certain redemption rights or put rights, including the right to require us to repurchase all or any portion of the Series A Preferred Stock on and immediately following five years after the Additional Closing. As holder of the Series A Preferred Stock, Sirius XM also has the right, subject to certain exceptions, to require us to repurchase all or any portion of the Series A Preferred Stock upon certain change of control events at the greater of (a) 100% of the liquidation preference thereof plus all accrued dividends unpaid through the fifth anniversary of the Initial Closing and (b) the consideration Sirius XM would have received if it had converted its shares of Series A Preferred Stock into common stock immediately prior to the change of control event (disregarding the 19.99% limit).

These dividend and share repurchase obligations could impact our liquidity and reduce the amount of cash flows available for working capital, capital expenditures, growth opportunities, acquisitions, and other general corporate purposes. Our obligations to Sirius XM, as the holder of our Series A Preferred Stock, could also limit our ability to obtain additional financing or increase our borrowing costs, which could have an adverse effect on our financial condition. In certain circumstances, the conversion of Series A Preferred Stock may settle in common stock instead of cash and if such settlement occurs, it would be dilutive to our common stock. The preferential rights could also result in divergent interests between Sirius XM and holders of our common stock. Furthermore, as a sale of our Company will likely require us to repurchase Series A Preferred Stock as a change of control event, this obligation could have the effect of delaying or preventing a sale of our Company that may otherwise be beneficial to our stockholders.

We depend on key personnel to operate our business, and if we are unable to retain, attract and integrate qualified personnel, our ability to develop and successfully grow our business could be harmed.

We believe that our success depends on the contributions of our executive officers as well as our ability to attract and retain qualified sales, technical and other personnel. All of our employees, including our executive officers, are free to terminate their employment relationship with us at any time, and their knowledge of our business and industry may be difficult to replace. Qualified individuals are in high demand, particularly in the digital media industry and in the San Francisco Bay Area, where our headquarters is located, and in New York, and we may incur significant costs to attract them. If we are unable to attract and retain our executive officers and key employees, we may not be able to achieve our strategic objectives, and our business could be harmed. We use stock-based and other performance-based incentive awards such as restricted stock units and cash bonuses to help attract, retain, and motivate qualified individuals. If our stock-based or other compensation programs cease to be viewed as competitive and valuable benefits, our ability to attract, retain, and motivate employees could be weakened, and our business could be harmed.

If we cannot maintain our corporate culture as we grow, we could lose the innovation, teamwork and focus that contribute crucially to our business.

We believe that a critical component of our success is our corporate culture, which we believe fosters innovation, encourages teamwork, cultivates creativity and promotes focus on execution. We have invested substantial time, energy and resources in building a highly collaborative team that works together effectively in a non-hierarchical environment designed to promote openness, honesty, mutual respect and pursuit of common goals. As we continue to develop the infrastructure of a public company and grow, we may find it difficult to maintain these valuable aspects of our corporate culture. Any failure to preserve our culture could negatively impact our future success, including our ability to attract and retain employees, encourage innovation and teamwork and effectively focus on and pursue our corporate objectives.

The impact of worldwide economic conditions, including the effect on advertising budgets and discretionary entertainment spending behavior, may adversely affect our business and operating results.

Our financial condition is affected by worldwide economic conditions and their impact on advertising spending. Expenditures by advertisers generally tend to reflect overall economic conditions, and reductions in spending by advertisers could have a serious adverse impact on our business. In addition, we provide an entertainment service, and payment for our Pandora Plus and Pandora Premium subscription services may be considered discretionary on the part of some of our current and prospective subscribers or listeners who may choose to use a competing free service or to listen to Pandora without subscribing. To the extent that overall economic conditions reduce spending on discretionary activities, our ability to retain current and obtain new subscribers could be hindered, which could reduce our subscription revenue and negatively impact our business.

Our business is subject to the risks of earthquakes, fires, floods and other natural catastrophic events and to interruption by man-made problems such as cybersecurity incidents or terrorism.

Our systems and operations are vulnerable to damage or interruption from earthquakes, fires, floods, power losses, telecommunications failures, terrorist attacks, acts of war, human errors, break-ins or similar events. For example, a significant natural disaster, such as an earthquake, fire or flood, could have a material adverse effect on our business, operating results and financial condition, and our insurance coverage may be insufficient to compensate us for losses that may occur. Our principal executive offices are located in the San Francisco Bay Area, a region known for seismic activity. In addition, acts of terrorism could cause disruptions in our business or the economy as a whole. Our servers may also be vulnerable to computer systems, cybersecurity incidents and similar disruptions caused by unauthorized tampering with our computer systems, which could lead to interruptions, delays, loss of critical data or the unauthorized disclosure of confidential customer data. Our business interruption insurance may be insufficient to compensate us for all such losses. As we rely heavily on our servers and the internet to conduct our business and provide high quality service to our listeners, such disruptions could negatively impact our ability to run our business, resulting in a loss of existing or potential listeners and advertisers and increased maintenance costs, which would adversely affect our operating results and financial condition.

We may not have sufficient cash flow from our business to make payments on our indebtedness.

Our ability to make scheduled payments of the principal of, to pay interest on or to refinance our indebtedness, including our 1.75% convertible senior notes due 2020 (the "Notes") and any borrowings we might make under our credit facility, depends on our performance, which is subject to economic, financial, competitive and other factors beyond our control. Our business may not generate cash flow from operations in the future sufficient to service our debt and make necessary capital expenditures. If we are unable to generate such cash flow, we may be required to adopt one or more alternatives, such as selling assets, restructuring debt or obtaining additional equity capital on terms that may be onerous or highly dilutive. Our ability to refinance our indebtedness will depend on the capital markets and our financial condition at such time. We may not be able to engage in any of these activities or engage in these activities on desirable terms, which could result in a default on our debt obligations.

The conditional conversion feature of the Notes, if triggered, may adversely affect our financial condition and operating results.

In the event the conditional conversion feature of the Notes is triggered, holders of Notes will be entitled to convert the Notes at any time during specified periods at their option. See Note 9, "Debt Instruments". If one or more holders elect to convert their Notes, we may elect to satisfy our conversion obligation in whole or in part through the payment of cash, which could adversely affect our liquidity. In addition, even if holders do not elect to convert their Notes, we could be required under



applicable accounting rules to reclassify all or a portion of the outstanding principal of the Notes as a current rather than long-term liability, which would result in a material reduction of our net working capital.

The accounting method for convertible debt securities that may be settled in cash, such as the Notes, could have a material effect on our reported financial results.

Accounting Standards Codification Subtopic 470-20 (ASC 470-20), Debt with Conversion and Other Options, requires an entity to separately account for the liability and equity components of convertible debt instruments (such as the Notes) that may be settled entirely or partially in cash upon conversion in a manner that reflects the issuer's economic interest cost. The effect of ASC 470-20 on the accounting for the Notes is that the equity component of the Notes is required to be included in the additional paid-in capital section of stockholders' equity on our consolidated balance sheet, and the value of the equity component is treated as original issue discount for purposes of accounting for the debt component of the Notes. As a result, we will be required to recognize a greater amount of non-cash interest expense in current and future periods as a result of the amortization of the discounted carrying value of the Notes to their principal amount over the term of the Notes. We will report lower net income (or greater net losses) in our consolidated financial results because ASC 470-20 will require interest to include both the current period's amortization of the original issue discount and the instrument's coupon interest, which could adversely affect our reported or future consolidated financial results, the trading price of our common stock and the trading price of the Notes.

In addition, under certain circumstances, in calculating earnings per share, convertible debt instruments (such as the Notes) that may be settled entirely or partly in cash are currently accounted for utilizing the treasury stock method, the effect of which is that the shares of common stock issuable upon conversion of the Notes, if any, are not included in the calculation of diluted earnings per share except to the extent that the conversion value of the Notes exceeds their principal amount. Under the treasury stock method, diluted earnings per share is calculated as if the number of shares of common stock that would be necessary to settle such excess, if we were to elect to settle such excess in shares, were issued. We cannot be sure that the accounting standards in the future will continue to permit the use of the treasury stock method. If we are unable to use the treasury stock method in accounting for the shares issuable upon conversion of the Notes (if any) then, to the extent we generate positive net income, our diluted consolidated earnings per share would be adversely affected.

Risks Related to Owning Our Common Stock

Our stock price has been and will likely continue to be volatile, and the value of an investment in our common stock may decline.

The trading price of our common stock has been and is likely to continue to be volatile. In addition to the risk factors described in this section, and elsewhere in this Annual Report on Form 10-K, additional factors that may cause the price of our common stock to fluctuate include, but are not limited to:

- our actual or anticipated operating performance and the operating performance of similar companies in the internet, radio or digital media spaces;
- our ability to grow active users and listener hours;
- competitive conditions and developments;
- our actual or anticipated achievement of financial and non-financial key operating metrics;
- general economic conditions and their impact on advertising spending;
- the overall performance of the equity markets;
- threatened or actual litigation or regulatory proceedings, including the current Phonorecords III rate proceedings in the CRB;
- changes in laws or regulations relating to our service;
- any major change in our board of directors or management;



Table of Contents

- publication of research reports about us or our industry or changes in recommendations or withdrawal of research coverage by securities analysts; and
- sales or expected sales of shares of our common stock by us, and our officers, directors and significant stockholders.

In addition, the stock market has experienced extreme price and volume fluctuations that often have been unrelated or disproportionate to the operating performance of affected companies. Securities class action litigation has often been instituted against companies following periods of volatility in the overall market and in the market price of a company's securities. Such litigation, if instituted against us, could result in substantial costs, divert our management's attention and resources and harm our business, operating results and financial condition.

If securities or industry analysts cease publishing research about our business, publish inaccurate or unfavorable research about our business, or make projections that exceed our actual results, our stock price and trading volume could decline.

The trading market for our common stock depends in part on the research and reports that securities or industry analysts publish about us or our business. If securities or industry analysts who cover us downgrade our stock or publish inaccurate or unfavorable research about our business, our stock price would likely decline. If one or more of these analysts cease coverage of us or fail to publish reports on us regularly, demand for our stock could decrease, which might cause our stock price and trading volume to decline. Furthermore, such analysts publish their own projections regarding our actual results. These projections may vary widely from one another and may not accurately predict the results we actually achieve. Our stock price may decline if we fail to meet securities and industry analysts' projections.

Our charter documents, Delaware law and certain terms of our music licensing arrangements could discourage takeover attempts and lead to management entrenchment.

Our certificate of incorporation and bylaws contain provisions that could delay or prevent a change in control of the Company. These provisions could also make it difficult for stockholders to elect directors who are not nominated by the current members of our board of directors or take other corporate actions, including effecting changes in our management. These provisions include:

- a classified board of directors with three-year staggered terms, which could delay the ability of stockholders to change the membership of a majority of our board of directors;
- no cumulative voting in the election of directors, which limits the ability of minority stockholders to elect director candidates;
- the ability of our board of directors to issue shares of preferred stock and to determine the price and other terms of those shares, including preferences and voting
 rights, without stockholder approval, which could be used to significantly dilute the ownership of a hostile acquiror;
- the exclusive right of our board of directors to elect a director to fill a vacancy created by the expansion of our board of directors or the resignation, death or removal
 of a director, which prevents stockholders from being able to fill vacancies on our board of directors;
- a prohibition on stockholder action by written consent, which forces stockholder action to be taken at an annual or special meeting of our stockholders;
- the requirement that a special meeting of stockholders may be called only by the chairman of our board of directors, our president, our secretary, or a majority vote of our board of directors, which could delay the ability of our stockholders to force consideration of a proposal or to take action, including the removal of directors;
- the requirement for the affirmative vote of holders of at least 66 23% of the voting power of all of the then outstanding shares of the voting stock, voting together as
 a single class, to amend the provisions of our certificate of incorporation relating to the issuance of preferred stock and management of our business or our bylaws,
 which may inhibit the ability of an acquiror to effect such amendments to facilitate an unsolicited takeover attempt;
- the ability of our board of directors, by majority vote, to amend the bylaws, which may allow our board of directors to take additional actions to prevent an unsolicited takeover and inhibit the ability of an acquiror to amend the bylaws to facilitate an unsolicited takeover attempt; and

advance notice procedures with which stockholders must comply to nominate candidates to our board of directors or to propose matters to be acted upon at a
stockholders' meeting, which may discourage or deter a potential acquiror from conducting a solicitation of proxies to elect the acquiror's own slate of directors or
otherwise attempting to obtain control of us.

Section 203 of the Delaware General Corporation Law governs us. These provisions may prohibit large stockholders, in particular those owning 15% or more of our outstanding voting stock, from merging or combining with us for a certain period of time.

In addition, if we are acquired, certain terms of our music licensing arrangements, including favorable rates for content acquisition costs that currently apply to us, may not be available to an acquiror. These terms may discourage a potential acquiror from making an offer to buy us or may reduce the price such a party may be willing to offer.

ITEM 1B. UNRESOLVED STAFF COMMENTS

Not applicable.

ITEM 2. PROPERTIES

Our principal executive offices are located in Oakland, California, under a lease for approximately 250,000 square feet of office space, expiring onSeptember 30, 2020. Our second largest office is located in New York, New York, under a lease for approximately 100,000 square feet of office space, expiring onSeptember 30, 2026. We recently announced a plan to expand our workforce in Atlanta, Georgia, where we have an existing regional office. We also lease regional offices in Chicago, Illinois; Dallas, Texas; Detroit, Michigan; and Santa Monica, California and local sales offices at various locations throughout the United States.

Our data centers are located in colocation facilities operated by Equinix in San Jose, California and Ashburn, Virginia, Digital Realty Trust in Chicago, Illinois and Oakland, California and InfoMart, LLC in San Jose, California. These data centers are designed to be fault-tolerant and operate at maximum uptime. Backup systems in California and Virginia can be brought online in the event of a failure at the other data centers. These redundancies enable fault tolerance and will also support our continued growth.

The data centers host the Pandora.com website and intranet applications that are used to manage the website content. The websites are designed to be fault-tolerant, with a collection of identical web servers connecting to an enterprise database. The design also includes load balancers, firewalls and routers that connect the components and provide connections to the internet. The failure of any individual component is not expected to affect the overall availability of our website.

We believe that our current facilities are adequate to meet our needs for the near future and that suitable additional or alternative space will be available on commercially reasonable terms to accommodate our foreseeable future operations.

ITEM 3. LEGAL PROCEEDINGS

The material set forth in the section entitled "Legal Proceedings" in Note 11 of the Notes to Consolidated Financial Statements in Part II, Item 8 of this Annual Report on Form 10-K is incorporated herein by reference.

ITEM 4. MINE SAFETY DISCLOSURES

Not applicable.

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF EQUITY SECURITIES

Market Information

Our common stock has traded on The New York Stock Exchange ("NYSE") under the symbol "P" since June 15, 2011. Our initial public offering was priced at \$16.00 per share on June 14, 2011. The following table sets forth the range of high and low intra-day sales prices per share of our common stock for the periods indicated, as reported by the NYSE.

	High	Low			
Year ended December 31, 2017					
First quarter (January 1, 2017 - March 31, 2017)	\$ 13.58	\$	11.48		
Second quarter (April 1, 2017 - June 30, 2017)	\$ 11.98	\$	6.91		
Third quarter (July 1, 2017 - September 30, 2017)	\$ 9.83	\$	7.69		
Fourth quarter (October 1, 2017 - December 31, 2017)	\$ 8.19	\$	4.49		
Year Ended December 31, 2016					
First quarter (January 1, 2016 - March 31, 2016)	\$ 12.25	\$	7.88		
Second quarter (April 1, 2016 - June 30, 2016)	\$ 12.52	\$	8.17		
Third quarter (July 1, 2016 - September 30, 2016)	\$ 14.63	\$	12.00		
Fourth quarter (October 1, 2016 - December 31, 2016)	\$ 14.77	\$	10.42		

On December 31, 2017, the closing price per share of our common stock as reported on the NYSE was\$4.82. As of December 31, 2017, there were approximately 65 holders of record of our common stock. The number of beneficial stockholders is substantially greater than the number of holders of record because a large portion of our common stock is held through brokerage firms.

Dividend Policy

We have not declared or paid any cash dividends on our common stock and currently do not anticipate paying any cash dividends in the foreseeable future. Instead, we intend to retain all available funds and any future earnings for use in the operation and expansion of our business. Any future determination relating to dividend policy will be made at the discretion of our board of directors and will depend on our future earnings, capital requirements, financial condition, future prospects, applicable Delaware law, which provides that dividends are only payable out of surplus or current net profits, and other factors that our board of directors deems relevant. In addition, our credit facility restricts our ability to pay dividends. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources—Our Indebtedness—Credit Facility" and Note 9 to our financial statements included elsewhere in this Annual Report on Form 10-K.

Equity Compensation Plan Information

For equity compensation plan information refer to Item 12 in Part III of this Annual Report on Form 10-K.

Stock Price Performance Graph

This performance graph shall not be deemed to be "soliciting material" or "filed" or incorporated by reference in future filings with the SEC, or subject to the liabilities of Section 18 of the Exchange Act except as shall be expressly set forth by specific reference in such filing.

The following graph shows a comparison from June 15, 2011, the date our common stock commenced trading on the NYSE, throughDecember 31, 2017 of the total cumulative return of our common stock with the total cumulative return of the New York Stock Exchange Composite Index (the "NYA Composite"), the Global X Social Media Index (the "SOCL") and the SPDR Morgan Stanley Technology MTK Index (the "MTK"). The figures represented below assume an investment of \$100 in

our common stock at the closing price of \$17.42 on June 15, 2011 and in the NYA Composite and MTK on the same date. The SOCL was modeled from the inception of the index on November 15, 2011. Data for the NYA Composite, MTK and SOCL assume reinvestment of dividends. The comparisons in the graph are historical and are not intended to forecast or be indicative of possible future performance of our common stock.



ITEM 6. SELECTED FINANCIAL DATA

The following selected consolidated financial and other data should be read in conjunction with, and are qualified by reference to, Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," and our audited consolidated financial statements and the accompanying notes included elsewhere in this report. The consolidated statement of operations data for the eleven months ended December 31, 2013 and the year ended December 31, 2014 and the consolidated balance sheet data as of December 31, 2013, 2014 and 2015 were derived from our audited consolidated financial statements not included in this report. The consolidated statements of operations data for the years ended December 31, 2015, 2016 and 2017, and the consolidated balance sheet data as of December 31, 2016 and 2017 were derived from our audited consolidated balance sheet data as of December 31, 2015, 2016 and 2017, and the consolidated statements of operations for the years ended December 31, 2015, 2016 and 2017, and the consolidated statements of operations for the years ended December 31, 2015, 2016 and 2017, include Ticketfly results for the two months ended December 31, 2015, the year ended December 31, 2016 and the period ended September 1, 2017. There is no ticketing service revenue subsequent to the disposition of Ticketfly Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition. The consolidated balance sheets as of December 31, 2015 and 2016 include Ticketfly's financial position as ofDecember 31, 2015 and 2016.

The historical results presented below are not necessarily indicative of financial results to be achieved in future periods.

Income Statement Data:

	E	leven months ended December 31,								
		2013		2014		2015		2016		2017
			(in th	ousands, exc	ept p	oer share dat	a)			
Total revenue	\$	600,233	\$	920,802	\$	1,164,043	\$	1,384,826	\$	1,466,812
Net loss	\$	(27,017)	\$	(30,406)	\$	(169,661)	\$	(342,978)	\$	(518,395)
Net loss available to common stockholders	\$	(27,017)	\$	(30,406)	\$	(169,661)	\$	(342,978)	\$	(558,561)
Net loss per common share, basic and diluted	\$	(0.15)	\$	(0.15)	\$	(0.79)	\$	(1.49)	\$	(2.29)
Weighted-average basic and diluted common shares		180,968		205,273		213,790		230,693		243,637

Key Metrics (unaudited):(1)

	Eleven Months Ended December 31,									
	2013	2014	2015	2016	2017					
		(in bill								
Listener hours	15.31	20.03	21.11	21.96	20.61					
		As of December 31,								
	2013	2014	2015	2016	2017					
Active users	76.2	81.5	81.1	81.0	74.7					

(1) Listener hours and active users are defined in the section entitled "Key Metrics" in Item 7 of this Annual Report on Form 10-K.



Balance Sheet Data:

	As of December 31,									
		2013		2014		2015		2016		2017
					(in	thousands)				
Cash and cash equivalents	\$	245,755	\$	175,957	\$	334,667	\$	199,944	\$	499,597
Working capital	\$	362,777	\$	439,254	\$	451,675	\$	371,704	\$	685,883
Total assets	\$	673,335	\$	749,290	\$	1,240,657	\$	1,184,810	\$	1,166,322
Long-term debt, net	\$	_	\$	—	\$	234,577	\$	342,247	\$	273,014
Total liabilities	\$	165,104	\$	165,933	\$	497,270	\$	630,551	\$	522,795
Redeemable convertible preferred stock	\$	_	\$	_	\$		\$	_	\$	490,849
Common stock and additional paid-in capital	\$	675,123	\$	781,030	\$	1,110,562	\$	1,264,717	\$	1,422,246
Total stockholders' equity	\$	508,231	\$	583,357	\$	743,387	\$	554,259	\$	152,678



ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS (MD&A)

You should read the following discussion of our financial condition and results of operations in conjunction with the financial statements and the notes thereto included elsewhere in this Annual Report on Form 10-K. The following discussion contains forward-looking statements that are subject to risks and uncertainties. Actual results may differ substantially from those referred to herein due to a number of factors, including but not limited to those discussed below and elsewhere in this report, particularly in the sections entitled "Special Note Regarding Forward-Looking Statements and Industry Data" and "Risk Factors."

Overview

Pandora—Streaming Radio and On-Demand Music Services

Pandora is the world's most powerful music discovery platform, offering a personalized experience for each of our listeners wherever and whenever they want to listen to music—whether through mobile devices, car speakers or connected devices in the home. Unlike traditional radio that broadcasts the same content at the same time to all of their listeners, we enable our listeners to create personalized stations and playlists, as well as search and play songs and albums on-demand. The Music Genome Project and our content programming algorithms power our ability to predict listener music preferences, play music content suited to the tastes of each individual listener and introduce listeners to the music we think they will love. The Music Genome Project is a database of over 1.5 million uniquely analyzed songs from over 250 thousand artists, spanning over 660 genres and sub-genres, which our team of trained musicologists has developed one song at a time by evaluating and cataloging each song's particular attributes. The Music Genome Project taste device available to be played. Over time, our service has evolved by using data science to develop playlisting algorithms that further tailor the listener experience based on individual listener and broader audience reactions to the recordings we pick. With billions of data points collected from our listeners, we are able to use listeners' feedback to fuel our ability to choose exactly the right song for our users. Founded by musicians, Pandora also empowers artists with valuable data and tools to help grow their audience and connect with their fans.

Pandora is available as an ad-supported service, a radio subscription service called Pandora Plus and an on-demand subscription service called Pandora Premium. The majority of our listener hours occur on mobile devices, with the majority of our revenue generated from advertising on our ad-supported radio service on these devices. With billions of data points that help us understand our users' preferences, we offer both local and national advertisers the opportunity to deliver targeted messages to our listeners using a combination of audio, display and video advertisements. We also generate increasing revenue from our subscription offerings.

For the year ended December 31, 2017, we streamed 20.61 billion hours of content, and as of December 31, 2017, we had 74.7 million active users during the trailing 30day period and 5.48 million paid subscribers. Since we launched our service in 2005, our listeners have created over12 billion stations.

Ad-Supported Radio Service

Our ad-supported service allows listeners to access our music and podcast catalogs and personalized playlist generating system for free across all of our delivery platforms. We also offer listeners on our ad-supported service access to our on-demand service in exchange for engaging with an ad, which allows advertisers to engage with a user to create a positive brand experience. This service is valued by lean-back listeners, as it uses the Music Genome Project to instantly generate a station that plays music we think that listener will enjoy. Over time, our service has evolved by using data science to further tailor the listener experience based on listener reactions to the content we pick. Listeners also have the ability to add variety to and rename stations, which further allows for the personalization of our service. We also offer listeners on our ad-supported service the option to gain temporary access to on-demand listening experience, which includes some features of our Pandora Premium service, in exchange for viewing a video ad. We call this experience Premium Access.

Subscription Radio Service-Pandora Plus

Pandora Plus is a paid, ad-free subscription version of the Pandora radio service that also includes replays, additional skipping of songs, offline listening, higher quality audio on supported devices and longer timeout-free listening. This service is valued by listeners who want the ability to have limited interactive features such as skips and replays. Similar to the ad-supported service, the more the listener interacts with the platform, the more we tailor the songs we recommend to the listener.

On-Demand Subscription Service—Pandora Premium

Our on-demand subscription service, Pandora Premium, launched in the United States in April 2017. Pandora Premium is an on-demand version of the Pandora service that offers a unique, on-demand experience, providing users with the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button, listen to curated playlists and share playlists on social networks. Unique to Pandora, a listener can create partial playlists and have Pandora complete the playlist based on the user's listening activity using the Music Genome Project. The features of Pandora Plus are also included in Pandora Premium.

A key element of our strategy is to make the Pandora service available in any environment that has internet connectivity. To this end, we make the Pandora service available through a variety of distribution channels. In addition to streaming our service to computers, we have developed Pandora mobile device applications ("apps") for smartphones and mobile operating systems, such as the iPhone and Android and for tablets, including the iPad and Android tablets. We distribute those mobile apps free to listeners via app stores.

We expect to continue to make enhancements to Pandora Plus and Pandora Premium, which will require engineering effort, as well as other resources. In addition, in connection with the launch and continued operation of these services we have entered into direct license agreements with major and independent record labels, some of which include substantial minimum guarantee payments. In order for Pandora Plus and Pandora Premium to be successful, we will need to attract subscribers to these service offerings. The market for subscription-based music services, including on-demand services, is intensely competitive, and our ability to realize a return on our investments on these service offerings will depend on our ability to leverage the existing audience of our ad-supported service, our brand awareness and our ability to deliver differentiated subscription services with features and functionality that listeners find attractive. Refer to our discussion of these matters in Item 1A—"Risk Factors".

Ticketing Service

Prior to September 1, 2017, we operated a ticketing service through our former subsidiary Ticketfly, a leading live events technology company that provides ticketing and marketing software and services for clients, which are venues and event promoters, across North America. We completed the sale of Ticketfly on September 1, 2017. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

Recent Events

Ticketfly Disposition

On September 1, 2017, we completed the sale of Ticketfly, our ticketing service segment, to Eventbrite Inc. ("Eventbrite") for an aggregate unadjusted purchase price of \$200.0 million. The aggregate unadjusted purchase price consists of \$150.0 million in cash and a \$50.0 million Convertible Promissory Note, which were paid and issued at the closing of the transaction. The Convertible Promissory Note was recorded at its fair value at the date of sale, which resulted in a discount of \$13.8 million. The aggregate purchase price of \$174.0 million. Refer to Note 8 "Convertible Promissory Note Receivable" in the Notes to Consolidated Financial Statements for further details on the Convertible Promissory Note.

In the three months ended June 30, 2017, the assets and liabilities of Ticketfly were classified as held for sale, and we recognized a goodwill impairment charge of \$131.7 million. The impairment charge was based on the fair value of the net assets as implied by the estimated purchase price of \$184.5 million as of June 30, 2017. In the year ended December 31, 2017, we recognized a loss on sale of \$9.3 million in the general and administrative line item on our Consolidated Statements of Operations, which was based on an adjusted net purchase price of \$174.0 million as of September 1, 2017 and includes \$1.1 million in transfers of cumulative translation adjustments. Refer to Note 5 "Dispositions" in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

KXMZ Disposition

On August 31, 2017, we completed the sale of KXMZ, an FM radio station based in Rapid City, South Dakota. The sale did not result in a material impact to our consolidated financial statements. Refer to Note 5 "Dispositions" in the Notes to Consolidated Financial Statements for further details on the KXMZ disposition.

Australia and New Zealand

On June 27, 2017, we announced a plan to discontinue business activities in Australia and New Zealand. The related restructuring charges in the year ended December 31, 2017 primarily relate to a reduction of headcount of approximately 50 employees, which resulted in employee severance and benefits costs offset by a credit related to non-cash stock-based compensation expense reversals for unvested equity awards. The dissolution of the Australia and New Zealand business operations was substantially completed in the three months ended September 30, 2017. These restructuring charges did not have a material impact on our financial statements.

Redeemable Convertible Preferred Stock

In June 2017, we entered into an agreement with Sirius XM Radio, Inc. ("Sirius XM") to sell480,000 shares of Series A redeemable convertible preferred stock ("Series A") for \$1,000 per share, with gross proceeds of \$480.0 million. The Series A shares were issued in two rounds: an initial closing of 172,500 shares for \$172.5 million that occurred on June 9, 2017 upon signing the agreement with Sirius XM, and an additional closing of 307,500 shares for \$307.5 million that occurred on September 22, 2017. Refer to <u>Note 10 "Redeemable Convertible Preferred Stock"</u> in the Notes to Consolidated Financial Statements for further details on the redeemable convertible preferred stock sale.

Factors Affecting our Business Model

Content Acquisition Costs

For sound recording rights, we pay content acquisition costs based on the terms of direct license agreements with major and independent music labels and distributors for the significant majority of the sound recordings we stream on our ad-supported service, Pandora Plus and Pandora Premium. Depending on the applicable service, these license agreements generally require us to pay either a per-performance fee based on the number of sound recordings we transmit, a percentage of revenue associated with the service, or a per-subscriber minimum amount. Certain of these license agreements require minimum guarantee payments, some of which are paid in advance.

If we have not entered into a direct license agreement with the copyright owner of a particular sound recording that is streamed on our services, we stream that sound recording pursuant to a statutory license and pay the applicable rates set by the Copyright Royalty Board ("CRB") for the period from January 1, 2016 through December 31, 2020 (the "Section 114 Rates"). The 2016 and 2017 rates for non-subscription services, such as our ad-supported service, were set at \$0.0017 per play and the rates for subscription services, such as Pandora Plus, were set at \$0.0022, adjusted for inflation. Effective January 1, 2018, these rates were adjusted for inflation to \$0.0018 per play for non-subscription services and \$0.0023 per play for subscription services. Sound recordings streamed under the statutory license and paid at the CRB-set rates can only be played in radio mode on our services. These sound recordings cannot be played on-demand or offline and are not eligible for replay or additional skips.

Content acquisition costs for musical works are negotiated with and paid to performing rights organizations ("PROs") such as the American Society of Composers, Authors and Publishers ("ASCAP"), Broadcast Music, Inc. ("BMI"), SESAC, Inc. ("SESAC") and Global Music Rights, and directly to publishing companies. Content acquisition costs for the streaming of musical works on our ad-supported service are calculated such that each copyright holder receives its usage-based and ownership-based share of a royalty pool equal to 20% of the content acquisition costs paid by us for sound recordings on our ad-supported service. Content acquisition costs for the streaming of musical works on our subscription services are equal to the rates determined in accordance with the statutory license set forth in 17 U.S.C. §115 ("Section 115").

The rate structure for the statutory license for reproduction rights under Section 115 expired at the end of 2017. A new rate structure was set in January 2018, covering the period from January 1, 2018 through December 31, 2022, by a three judge CRB panel, and we were one of five commercial music service operators (along with Amazon, Apple, Google and Spotify) that participated in rate-setting proceedings that determined these rates (the "Phonorecords III Proceedings"). The Nashville Songwriters Association International, the National Association of Music Publishers and George Johnson Music Publishing are also participating in the Phonorecords III Proceedings. A trial before the CRB concluded in April 2017, and the CRB rendered a decision in January 2018 ("the Initial Determination"). If the Initial Determination is adopted and affirmed, the "all-in" rate that streaming services, including Pandora, will pay to music publishers and songwriters for the mechanical rights and performance rights needed in connection with interactive streaming will increase annually between 2018 and 2022: from 11.4% of revenues or 22.0% of label payments in 2018, to 15.1% of revenues or 26.2% of label payments in 2022. Certain per-subscriber minimum royalty floors also apply depending on the type of service. The CRB may elect to modify the Initial Determination at



the request of the participants in the Phonorecords III Proceedings, and thereafter the final CRB decision may be appealed for review by the DC Circuit Court of Appeals.

The Phonorecords III Proceedings are important to us because our direct licenses with music publishers use the Section 115 rates. As a result, increases in the Section 115 rates increase our content acquisition costs for our subscription services, which, if such increase were substantial, could materially harm our financial condition and hinder our ability to provide interactive features in our services, or cause one or more of our subscription services to not be economically viable.

Ad-Supported Service

Our ad-supported service is monetized through the sale of display, audio and video advertisements to national, regional and local advertisers.

Our total number of listener hours is a key driver for both advertising revenue generation opportunities and content acquisition costs, which are the largest component of our ad-related expenses.

- Advertising Revenue. Listener hours define the number of opportunities we have to sell advertisements. Our ability to attract advertisers depends in large part on
 our ability to offer sufficient inventory within desired demographics.
- Cost of Revenue—Content Acquisition Costs—Ad-Supported Service. We pay content acquisition costs to the copyright owners and performers, or their agents, of
 each sound recording that we stream, as well as to the publishers and songwriters, or their agents, for the musical works embodied in each of those sound
 recordings, subject to certain exclusions. The majority of the content acquisition costs related to our ad-supported service are driven by direct license agreements
 with major and independent labels and distributors, as discussed above in "Factors Affecting Our Business Model—Content Acquisition Costs". Certain of these
 license agreements include minimum guarantee payments, some of which are paid in advance.

As a result of the structure of our license agreements, our ability to achieve and sustain profitability and operating leverage on our ad-supported service depends on our ability to increase our advertising revenue per thousand listener hours ("ad RPM") of streaming through increased advertising revenue across all of our delivery platforms.

Subscription Services

We monetize our subscription services through subscription payments made by users of the services. We drive subscriber growth by providing the world's most powerful music discovery platform, offering a personalized experience for each of our listeners, and investing in marketing and free-trials to promote our service.

Our total number of paid subscriptions is a key driver for both subscription revenue and content acquisition costs related to our subscription services, which is the largest component of our subscription-related expenses. In order to drive greater subscription revenue, we must increase the number of new subscribers to our subscription services and minimize the number of current subscribers who discontinue their subscriptions.

- Subscription Revenue. Our subscription revenue depends upon the number of paid subscriptions we are able to sell and the price that our subscribers pay for those
 subscriptions. Our ability to attract subscribers depends in large part on our ability to offer features and functionality on our subscription services that are valued
 by consumers within desired demographics, on terms that are attractive to those consumers, and still enable us to maintain adequate gross margins.
- Cost of Revenue—Content Acquisition Costs—Subscription Services. We pay content acquisition costs to the copyright owners, performers, songwriters, or their
 agents, subject to certain exclusions. The majority of our content acquisition costs related to our subscription service are generally driven by direct license
 agreements with major and independent labels and distributors, PROs and publishers, as discussed above in "Factors Affecting Our Business Model—Content
 Acquisition Costs". Certain of these license agreements include minimum guarantee payments, some of which are paid in advance.

Given the structure of our license agreements for our subscription services, the majority of our content acquisition costs increase as subscription revenue increases and are subject to minimum guarantee payments. As such, our ability to achieve and sustain profitability and operating leverage on our subscription services depends on our ability to increase our revenue through



increased paid subscriptions on terms that maintain an adequate gross margin. Refer to our discussion of these matters interms that maintain an adequate gross margin.

Key Metrics

The numbers for our key metrics, which include our listener hours, active users, advertising RPM, advertising licensing costs per thousand listener hours ("ad LPM"), paid subscribers, average monthly revenue per paid subscriber ("ARPU") and average monthly licensing costs per paid subscriber ("LPU") are calculated using internal company data. While these numbers are based on what we believe to be reasonable estimates of our user base for the applicable period of measurement, there are inherent challenges in measuring usage of our products across large online and mobile populations. In addition, we are continually seeking to improve our estimates of our user base, and such estimates may change due to improvements or changes in our methodology.

In the quarter ended December 31, 2016, we began reporting new key metrics on a prospective basis as a result of a change in our service offerings. We discontinued our previous key metrics as of October 1, 2016. Certain of our new key metrics are not comparable to prior periods given the lack of history of our new service offerings. As such, these metrics have not been presented for, nor compared against, these periods. Key metrics that did not change are presented for the year ended December 31, 2017 and are compared to the results for the year ended December 31, 2016.

Total Service

Listener hours

We track listener hours because it is a key indicator of the growth of our business and the engagement of our listeners. We include listener hours related to our non-radio content offerings in the definition of listener hours. These offerings include non-music content such as podcasts, as well as custom music content such as Pandora Premieres and artist mixtapes. We calculate listener hours based on the total bytes served for each track that is requested and served from our servers, as measured by our internal analytics systems, whether or not a listener listens to the entire track. For non-music content such as podcasts, episodes are divided into approximately track-length parts, which are treated as tracks under this definition. To the extent that third-party measurements of listener hours are not calculated using a similar server-based approach, the third-party measurements may differ from our measurements. We have implemented strategic hours control mechanisms, such as time outs, to optimize content acquisition costs for our ad-supported listening, and will continue to use these measures in the future.

The table below sets forth the total listener hours for theyears ended December 31, 2015, 2016 and 2017.

		Year ended December 31,									
	2015	2016	2017								
Service		Listener hours (in billions)									
Advertising	18.47	19.17	16.41								
Subscription	2.64	2.79	4.20								
Total	21.11	21.96	20.61								

Active users

We track the number of active users as an additional indicator of the breadth of audience we are reaching at a given time. We define active users as the number of distinct registered users, including subscribers that have requested audio from our servers within the trailing 30 days to the end of the final calendar month of the period. The number of active users may overstate the number of unique individuals who actively use our service, as one individual may register for, and use, multiple accounts.

The table below sets forth our total active users as ofDecember 31, 2016 and 2017.

		As of December 31,				
	-	2016	2017			
		(in millions)				
users—all services		81.0	74.7			

Advertising-based service

Advertising RPM

We track ad RPM for our non-subscription, ad-supported service because it is a key indicator of our ability to monetize advertising inventory created by our listener hours. We focus on ad RPM across all of our delivery platforms. We believe ad RPM to be the central top-line indicator for evaluating the results of our advertising monetization efforts. Ad RPM is calculated by dividing advertising revenue by the number of thousands of listener hours for our advertising-based service.

Advertising LPM

We track ad LPM for our non-subscription, ad-supported service across all delivery platforms. Prior to September 15, 2016, the content acquisition costs included in our ad LPM calculations were relatively fixed, with scheduled annual rate adjustments as specified by the Section 114 Rates. Subsequent to September 15, 2016, the content acquisition costs included in our ad LPM calculations are based on the rates set by our license agreements with record labels, PROs and music publishers or the Section 114 Rates if we have not entered into a license agreement with the copyright owner of a particular sound recording.

The table below sets forth our RPM and LPM for our ad-supported service for the years endedDecember 31, 2015, 2016 and 2017.

		Year ended December 31,											
		2015 2016						2017					
	1	RPM*		LPM		RPM*		LPM		RPM*		LPM	
Advertising	\$	50.52	\$	26.13	\$	55.94	\$	32.40	\$	65.32	\$	35.70	

*The calculation of RPM does not include revenue generated by Ticketfly or Next Big Sound.

Advertising RPM

For the year ended December 31, 2017 compared to 2016, the increase in ad RPM was primarily due to the increase in the number of ads sold per hour, due in part to an increase in the number of audio ads played.

For the year ended December 31, 2016 compared to 2015, ad RPM increased primarily due to the increase in advertising revenue outpacing the growth in advertising listener hours as a result of an increase in the average price per ad sold.

Advertising LPM

Total ad LPMs in the year ended December 31, 2017 compared to 2016 increased primarily due to rate increases and minimum guarantee accruals related to our direct license agreements with major and independent labels, distributors, PROs and publishers in comparison to the statutory rates used to calculate our content acquisition costs for the majority of the year ended December 31, 2016.

Total ad LPM in the year endedDecember 31, 2016 compared to 2015 increased primarily due to an increase in content acquisition costs subject to the license agreements for recorded music with major and independent labels, distributors, PROs and publishers that were signed in 2015 and 2016. The increase was partially offset by one-time cumulative charges in the year ended December 31, 2015 of \$57.9 million for the pre-1972 sound recordings settlement and \$23.9 million as a result of management's decision to forgo the application of the RMLC publisher royalty rate from June 2013 to September 2015.

Subscription Services-Total

Paid Subscribers



Paid subscribers are defined as the number of distinct users that have current, paid access to our subscription service as of the end of the period. We track paid subscribers because it is a key indicator of the growth of our subscription services.

The below table sets forth our paid subscribers as ofDecember 31, 2016 and 2017.

of December 31,	As of December 31,				
2017	2016				
(in millions)	(in m				
.39 5.48	4.39	5.48			

ARPU and LPU

ARPU is defined as average monthly revenue per paid subscriber on our subscription services. LPU is defined as average monthly licensing costs per paid subscriber on our subscription services. We believe ARPU to be the central top-line indicator for evaluating the results of our monetization efforts on our subscription services. We track LPU because it is a key measure of our ability to manage costs for our subscription services. The below table sets forth our ARPU and LPU for our subscription services for the quarters ended December 31, 2016 and 2017 and the year ended December 31, 2017.

	Quarter ende December 3		Year ended December 31,						
	2016 2017		2016	2017					
Subscription ARPU	4.73	6.08	N/A	5.34					
Subscription LPU	3.12	4.41	N/A	3.62					

Subscription ARPU

For the quarter ended December 31, 2017 compared to 2016, the increase in subscription ARPU is primarily due to the increase in subscription revenue outpacing the increase in subscriptions due to the fact that Pandora Premium launched in 2017 at a higher price point than Pandora Plus.

Subscription LPU

For the quarter ended December 31, 2017 compared to 2016, the increase in subscription LPU was primarily due to an increase in content acquisition costs associated with Pandora Premium and minimum guarantee accruals related to our direct license agreements with major and independent labels, distributors, PROs and publishers.

Basis of Presentation and Results of Operations

The following table presents our results of operations for the periods indicated as a percentage of total revenue. Our results of operations for theyear ended December 31, 2015 include operating results of Ticketfly for the two months ended December 31, 2015. Our results of operations for the year ended December 31, 2017 include operating results of Ticketfly for the period ended September 1, 2017. The period-to-period comparisons of results are not necessarily indicative of results for future periods.



	Year ended December 31,								
	2015	2016	2017						
Revenue									
Advertising	80 %	77 %	73 %						
Subscription and other	19	16	22						
Ticketing service	1	6	5						
Total revenue	100	100	100						
Cost of revenue									
Cost of revenue-Content acquisition costs	52	53	55						
Cost of revenue—Other (1)	7	7	8						
Cost of revenue—Ticketing service (1)	1	4	3						
Total cost of revenue	60	65	66						
Gross profit	40	35	34						
Operating expenses									
Product development (1)	7	10	11						
Sales and marketing (1)	34	35	34						
General and administrative (1)	13	13	13						
Goodwill impairment	_	<u> </u>	9						
Contract termination fees	_	_	2						
Total operating expenses	55	58	68						
Loss from operations	(15)	(23)	(34)						
Interest expense		(2)	(2)						
Other income, net	_	_	_						
Total other expense, net		(2)	(2)						
Loss before (provision for) benefit from income taxes	(15)	(25)	(35)						
(Provision for) benefit from income taxes	_	_	_						
Net loss	(15)%	(25)%	(35)%						
Net loss available to common stockholders	(15)%	(25)%	(38)%						
(1) Includes stock-based compensation as follows:									
Cost of revenue—Other	0.5 %	0.4%	0.2 %						
Cost of revenue—Ticketing service	_	_	_						
Product development	2.0	2.2	2.3						
Sales and marketing	4.5	4.2	3.8						
General and administrative	2.5	3.1	2.4						
Note: Amounts may not recalculate due to rounding									

Note: Amounts may not recalculate due to rounding

Revenue

		Year ended December 31,						Year Decen				
	2015 2016				\$ Change		2016	2017			\$ Change	
				(in thousands)		(in thousands)						
Revenue												
Advertising	\$	933,305	\$	1,072,490	\$	139,185	\$	1,072,490	\$	1,074,927	\$	2,437
Subscription and other		220,571		225,786		5,215		225,786		315,853		90,067
Ticketing service		10,167		86,550		76,383		86,550		76,032		(10,518)
Total revenue	\$	1,164,043	\$	1,384,826	\$	220,783	\$	1,384,826	\$	1,466,812	\$	81,986

Advertising revenue

We generate advertising revenue primarily from audio, display and video advertising, which is typically sold on a cost-per-thousand impressions, or CPM, basis. Advertising campaigns typically range from one to twelve months, and advertisers generally pay us based on the number of delivered impressions or the satisfaction of other criteria, such as click-through rates. For the years ended December 31, 2015, 2016 and 2017, advertising revenue accounted for 80%, 77% and 73%, of our total revenue, respectively. We expect that advertising will comprise a substantial majority of revenue for the foreseeable future.

For the year ended December 31, 2017 compared to 2016, advertising revenue increased \$2.4 million or 0.2% due to the average price per ad and the number of ads sold remaining relatively flat compared to the prior year.

For the year ended December 31, 2016 compared to 2015, advertising revenue increased by \$139.2 million or 15%, primarily due to an increase in the average price per ad sold.

Subscription and other revenue

Subscription and other revenue is generated primarily through the sale of monthly or annual subscriptions to Pandora Plus and Pandora Premium. Pandora Plus is a paid, ad-free version of the Pandora service that includes replays, additional skipping, offline listening, higher quality audio on supported devices and longer timeout-free listening. Pandora Premium is a paid, ad-free version of the Pandora service that also offers a unique, on-demand experience, providing users with the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button and automatically generate playlists based on their listening activity. Subscription revenue is recognized on a straight-line basis over the duration of the subscription period. For the years ended December 31, 2015, 2016 and 2017, subscription and other revenue accounted for 19%, 16% and 22% of our total revenue, respectively.

For the year ended December 31, 2017 compared to 2016, subscription revenue increased \$90.1 million or 40%, primarily due to an increase in the number of subscribers and an increase in the average price per paid subscriber due to the launch and growth of Pandora Premium.

For the year ended December 31, 2016 compared to 2015, subscription and other revenue increased by \$5.2 million or 2%, primarily due to an increase in the number of subscribers.

Ticketing service

Ticketing service revenue was generated primarily from service and merchant processing fees generated on ticket sales through the Ticketfly platform. Ticketfly sold tickets to fans for events on behalf of clients and charged either a fee per ticket or a percentage of the total convenience charge and order processing fee for its services at the time the ticket for an event was sold. Ticketing service revenue was recorded net of the face value of the ticket at the time of the sale, as Ticketfly generally acted as the agent in these transactions. For the years ended December 31, 2015, 2016 and 2017, ticketing service revenue accounted for approximately 1%, 6% and 5% of our total revenue, respectively. On September 1, 2017, we completed the sale of Ticketfly to Eventbrite. Ticketing service revenue is included in our Consolidated Statements of Operations for the period from the acquisition date of October 31, 2015 to the disposition date of September 1, 2017. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

For the year ended December 31, 2017 compared to 2016, ticketing service revenue decreased \$10.5 million or 12%,

Table of Contents

primarily due to a decrease in the number of tickets sold, excluding box office sales, as a result of the sale of Ticketfly on September 1, 2017.

Deferred revenue

Our deferred revenue consists primarily of prepaid but unrecognized subscription revenue and, to a lesser extent, advertising fees received or billed in advance of the delivery or completion of the delivery of services. Deferred revenue is recognized as revenue when the services are provided and all other revenue recognition criteria have been met.

In addition, subscription revenue derived from sales through certain mobile device app stores may be subject to refund or cancellation terms which may affect the timing or amount of the subscription revenue recognition. When refund rights exist, we recognize revenue when services have been provided and the rights lapse or when we have developed sufficient transaction history to estimate a return reserve.

Costs and Expenses

Cost of revenue consists of cost of revenue—content acquisition costs, cost of revenue—other and cost of revenue—ticketing. Our operating expenses consist of product development, sales and marketing, general and administrative costs, goodwill impairment and contract termination fees. Cost of revenue—content acquisition costs are the largest component of our costs and expenses, followed by employee-related costs, which include stock-based compensation expenses.

Cost of revenue—Content acquisition costs

	Year ended December 31,						Year Decer				
	 2015 2016				\$ Change	2016	2017			\$ Change	
	 (in thousands)							(in	thousands)		
Cost of revenue-Content acquisition costs	\$ 610,362	\$	734,353	\$	123,991	\$	734,353	\$	804,032	\$	69,679

Cost of revenue-content acquisition costs primarily consist of licensing fees paid for streaming music or other content to our listeners. In the year ended December 31, 2016, we obtained the rights to stream the majority of sound recordings on our service through statutory licenses, with the costs for such licenses determined according to the per-play rates set by the CRB. We obtained the rights to the majority of the musical works streamed on our service through direct licensing agreements with PROs or publishers, with the costs for such licenses based on a percentage of the content acquisition costs we paid for sound recordings.

During the year ended December 31, 2017, the majority of our content acquisition costs were calculated using negotiated rates in direct license agreements with record labels, music publishers and PROs. Depending on the applicable service, our sound recording license agreements generally require us to pay either a per-performance fee based on the number of sound recordings we transmit, a percentage of revenue associated with the service, or a per-subscriber minimum amount.

For our ad-supported service, the majority of our content acquisition costs for musical works are based on a percentage of content acquisition costs paid for sound recordings.

For our subscription services, content acquisition costs for musical works are determined in accordance with the statutory license set forth in 17 U.S.C. §115. Certain of our direct license agreements are also subject to minimum guarantee payments, some of which are paid in advance and amortized over the minimum guarantee period. For certain content acquisition arrangements, we accrue for estimated content acquisition costs based on the available facts and circumstances and adjust these estimates as more information becomes available. For additional information, see above in "Factors Affecting Our Business Model—Content Acquisition Costs".

For the year ended December 31, 2017 compared to 2016, content acquisition costs increased \$69.7 million or 9%, and total content acquisition costs as a percentage of total revenue increased from 53% to 55%. As we entered into direct license agreements with major and independent labels, distributors, PROs and publishers, we experienced rate increases and minimum guarantee accruals as compared to the statutory rates used to calculate our content acquisition costs for the majority of the year ended December 31, 2016. These rates particularly impacted our subscription services. As we launched Pandora Premium and experienced growth in our subscription services in 2017, this resulted in a corresponding increase in content acquisition costs as dictated by these direct license agreements.



For the year ended December 31, 2016 compared to 2015, content acquisition costs increased by \$124.0 million or 20%, and total content acquisition costs as a percentage of total revenue increased from 52% to 53%, primarily due to rate increases and minimum guarantee accruals related to our direct license agreements with major and independent labels, distributors, PROs and publishers that went into effect beginning September 2016. The increase was partially offset by one-time cumulative charges in the year ended December 31, 2015 of \$57.9 million for the pre-1972 sound recordings settlement and \$23.9 million as a result of management's decision to forgo the application of the RMLC publisher royalty rate from June 2013 to September 2015.

Cost of Revenue-Other

	Year Decen				Year Decer			
	 2015	2016	•	\$ Change	 2016	2017	•	\$ Change
		(in thousands)				(in thousands)		
Cost of revenue—Other	\$ 79,972	\$ 102,717	\$	22,745	\$ 102,717	\$ 112,638	\$	9,921

Cost of revenue—other consists primarily of ad and music serving costs, employee-related costs, facilities and equipment costs, other costs of ad sales and amortization expense related to acquired intangible assets and internal-use software. In the year ended December 31, 2017 we reallocated headcount from cost of revenue—other to other financial statement line items due to a reorganization of the company as a result of a reallocation of resources. Ad and music serving costs consist of content streaming, maintaining our streaming radio and on-demand subscription services and creating and serving advertisements through third-party ad servers. We make payments to third-party ad servers for the period the advertising impressions are delivered or click-through actions occur, and accordingly, we record this as a cost of revenue in the related period. Employee-related costs include salaries and benefits associated with supporting music and ad-serving functions. Other costs of ad sales include costs related to music events that are included as part of certain of our advertising arrangements.

For the year ended December 31, 2017 compared to 2016, cost of revenue—other increased \$9.9 million or 10%, primarily due to an \$11.2 million increase in amortization expense of internal-use software and acquired intangible assets related to the launch of Pandora Premium, a \$5.0 million increase in other cost of ad sales and a \$4.8 million increase in hosting and ad serving costs driven by an increase in the cost of ad sales. The increase is offset by an \$11.6 million decrease in employee-related costs driven by a decrease in average headcount related to a reorganization of the company as a result of a reallocation of resources and a reduction in force in the first quarter of 2017.

For the year ended December 31, 2016 compared to 2015, cost of revenue—other increased by \$22.7 million or 28%, primarily due to a \$6.0 million increase in employee-related expenses and a \$3.3 million increase in facilities and equipment costs driven by an increase in average headcount, a \$7.4 million increase in hosting and adserving costs and a \$3.0 million increase in cost of ad sales driven by an increase in the number of ads sold, a \$1.7 million increase in amortization expense of internal-use software and a \$1.1 million increase in costs related to music events that are sold as part of our advertising arrangements.

Cost of revenue—ticketing service

		ended nber 3				Year Decer			
	2015		2016	-	\$ Change	 2016	2017	-	\$ Change
			(in thousands)				(in thousands)		
Cost of revenue—Ticketing service	\$ 7,121	\$	59,280	\$	52,159	\$ 59,280	\$ 50,397	\$	(8,883)

Cost of revenue—ticketing service consists primarily of ticketing revenue share costs, hosting costs, credit card fees and other cost of revenue and intangible amortization expense. The majority of these costs are related to revenue share costs which consist of fees paid to clients for their share of convenience and order processing fees. Intangible amortization expense is related to amortization of developed technology acquired in connection with the Ticketfly acquisition on October 1, 2015. On September 1, 2017, we completed the sale of Ticketfly to Eventbrite. Cost of revenue—ticketing service is included in our Consolidated Statements of Operations for the period from the acquisition date of October 31, 2015 to the disposition date of

Table of Contents

September 1, 2017. Refer to Note 5 "Dispositions" in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

For the year ended December 31, 2017 compared to 2016, cost of revenue—ticketing service decreased \$8.9 million or 15%, primarily due to the sale of Ticketfly on September 1, 2017.

Gross margin

		ended nber 31	l,		Year Decen		
	2015		2016	\$ Change	2016	2017	\$ Change
		(in thousands)			(in thousands)	
Gross profit							
Total revenue	\$ 1,164,043	\$	1,384,826	\$ 220,783	\$ 1,384,826	\$ 1,466,812	\$ 81,986
Total cost of revenue	697,455		896,350	198,895	896,350	967,067	70,717
Gross profit	\$ 466,588	\$	488,476	\$ 21,888	\$ 488,476	\$ 499,745	\$ 11,269
Gross margin	 40 %		35 %		 35 %	 34 %	

For the year ended December 31, 2017 compared to 2016, gross margin decreased from 35% to 34%, as the growth in cost of revenue—content acquisition costs outpaced the growth in revenue. As we entered into direct license agreements with major and independent labels, distributors, PROs and publishers, we experienced rate increases and minimum guarantee accruals as compared to the statutory rates used to calculate our content acquisition costs for the majority of the year ended December 31, 2016. These rates particularly impacted our subscription services. As we launched Pandora Premium and experienced growth in our subscription services in 2017, this resulted in a corresponding increase in content acquisition costs as dictated by these direct license agreements.

For the year ended December 31, 2016 compared to 2015, gross margin decreased from 40% to 35%, as the growth in content acquisition costs outpaced the growth in revenue, primarily due to rate increases and minimum guarantee accruals related to our direct license agreements with major and independent labels, distributors, PROs and publishers that went into effect in September 2016.

Product development

		ended nber 3				Year Decer			
	2015		2016	-	\$ Change	 2016	2017	-	\$ Change
		((in thousands)				(in thousands)		
Product development	\$ 84,135	\$	140,707	\$	56,572	\$ 140,707	\$ 154,325	\$	13,618

Product development consists primarily of employee-related costs, including salaries and benefits related to employees in software engineering, music analysis and product management departments, facilities and equipment costs, information technology costs and amortization expense related to acquired intangible assets. We incur product development expenses primarily for improvements to our website and the Pandora app, development of new services and enhancement of existing services, development of new advertising products and development and enhancement of our personalized playlisting system. We have generally expensed product development as incurred. These amounts are offset by costs that we capitalize to develop software for internal use. Certain website development and internal use software development costs are capitalized when specific criteria are met. In such cases, the capitalized amounts are amortized over the useful life of the related application once the application is placed in service.

For the year ended December 31, 2017 compared to 2016, product development expenses increased \$13.6 million or 10%, primarily due to a \$8.4 million decrease in capitalized personnel costs driven by an increase in costs for maintenance of and minor enhancements to Pandora Premium that were not capitalized. The increase is also due to an increase in employee related costs, excluding capitalized personnel, of \$7.3 million and a \$2.9 million increase in facilities and equipment costs as a result of an increase in average headcount. This was offset by a \$5.0 million decrease in amortization expense of acquired intangible assets.

For the year ended December 31, 2016 compared to 2015, product development expenses increased by \$56.6 million or 67%, primarily due to a \$59.2 million increase in employee-related costs and a \$7.5 million increase in facilities and equipment costs driven by an increase in average headcount, a \$6.6 million increase in intangible amortization expense and a \$4.0 million increase in professional fees, offset by a \$22.3 million increase in costs that we capitalized to develop software for internal use.

Sales and marketing

		ended nber 3				ended nber 31	,	
	 2015		2016	\$ Change	 2016		2017	\$ Change
			(in thousands)			(in	thousands)	
Sales and marketing	\$ 397,842	\$	490,364	\$ 92,522	\$ 490,364	\$	492,542	\$ 2,178

Sales and marketing consists primarily of employee-related costs, including salaries, commissions and benefits related to employees in sales, sales support, marketing, advertising and industry relations and artist marketing departments and facilities and equipment costs. In addition, sales and marketing expenses include commissions on subscription purchases through mobile app stores ("subscription commissions"), external sales and marketing expenses such as brand marketing, advertising, customer acquisition, direct response and search engine marketing costs, public relations expenses, costs related to music events, agency platform and media measurement expenses and amortization expense related to acquired intangible assets.

For the year ended December 31, 2017 compared to 2016, sales and marketing expenses increased \$2.2 million or 0.4%, primarily due to a \$10.3 million increase in subscription commissions driven by an increase in subscribers as a result of the launches of Pandora Plus and Pandora Premium and a \$4.5 million increase in facilities and equipment costs due to an increase in expensed software. This was offset by a \$5.9 million decrease in employee-related costs driven by a decrease in average headcount, a \$3.8 million decrease in amortization expense of acquired intangible assets and a \$1.3 million decrease in professional fees.

For the year ended December 31, 2016 compared to 2015, sales and marketing expenses increased by \$92.5 million or 23%, primarily due to a \$45.5 million increase in employee-related costs and a \$9.1 million increase in facilities and equipment costs driven by an increase in average headcount, a \$22.1 million increase in brand marketing, advertising, direct response and search engine marketing costs driven by advertising campaigns launched in 2016, a \$5.8 million increase in intangible amortization expense and a \$5.0 million increase in amortization of non-recoupable ticketing contract advances.

General and administrative

	Year Decer			Year Decen		
	 2015	2016	\$ Change	 2016	2017	\$ Change
		(in thousands)			(in thousands)	
General and administrative	\$ 154,602	\$ 176,164	\$ 21,562	\$ 176,164	\$ 190,711	\$ 14,547

General and administrative consists primarily of employee-related costs, including salaries, benefits and severance expense for finance, accounting, legal, internal information technology and other administrative personnel and facilities and equipment costs. In addition, general and administrative expenses include legal expenses, professional fees for outside accounting and other services, credit card fees and sales and other tax expense.

For the year ended December 31, 2017 compared to 2016, general and administrative expenses increased \$14.5 million or 8%, the majority of which is due to a \$9.3 million loss on the sale of Ticketfly, an \$8.7 million increase in provision for bad debt primarily related to our ticketing service, a \$5.0 million increase in professional fees related to consulting and advisory fees and a \$3.9 million increase in legal fees primarily related to the rate-setting proceedings under Section 115, offset by a \$15.4 million decrease in employee-related costs primarily driven by a decrease in average headcount.

For the year ended December 31, 2016 compared to 2015, general and administrative expenses increased by \$21.6 million or 14%, primarily due to a \$32.7 million increase in employee-related costs and a \$2.8 million increase in facilities and

equipment related costs driven by an increase in average headcount and executive severance, a \$2.3 million increase in sales and other tax expense, offset by a \$18.4 million decrease in legal fees related to content acquisition and other matters.

Goodwill impairment

		ended nber 31,			Year Decen	ended 1ber 3			
	 2015	20	16	\$ Change	 2016		2017	5	6 Change
		(in the	ousands)			(in	thousands)		
Goodwill impairment	\$ —	\$	_	\$ _	\$ 	\$	131,997	\$	131,997

We had no goodwill impairment expense for the years ended December 31, 2015 and 2016.

For the year ended December 31, 2017, goodwill impairment was \$132.0 million and consisted primarily of \$131.7 million of impairment expense related to the write down of Ticketfly goodwill, which was based on the fair value of Ticketfly's net assets as implied by the estimated purchase price of \$184.5 million as of June 30, 2017. Refer to <u>Note 5 "Dispositions"</u> and <u>Note 7 "Goodwill and Intangible Assets"</u> in the Notes to Consolidated Financial Statements for further information on the goodwill impairment.

Contract termination fees

		ended nber 31,					ended mber 31			
	2015	20	016	\$ Ch	ange	2016		2017	5	\$ Change
		(in th	ousands)				(in t	thousands)		
Contract termination fees	\$ _	\$	— \$;	— \$	_	\$	23,044	\$	23,044

We had no contract termination fees for the years ended December 31, 2015 and 2016.

For the year ended December 31, 2017, contract termination fees were \$23.0 million and consisted of fees related to the termination of the contractual commitment with KKR Classic Investors L.P. ("KKR"). In May 2017, we entered into an agreement to sell redeemable convertible preferred stock to KKR. In conjunction with the Series A, we terminated the contractual commitment to sell redeemable convertible preferred stock to KKR, which resulted in contract termination fees, including the related legal and professional fees.

Interest expense

Interest expense in the years ended December 31, 2015, 2016 and 2017 consists primarily of interest expense on our 1.75% Convertible Senior Notes due 2020 and interest on our credit facility. Refer to Note 9 "Debt Instruments" in the Notes to Consolidated Financial Statements for further details on our Notes and credit facility.

Benefit from (provision for) income taxes

We have historically been subject to income taxes in the United States and various foreign jurisdictions. If we expand our operations to other foreign locations, we become subject to taxation based on the applicable foreign statutory rates and our effective tax rate could fluctuate accordingly.

Our benefit from (provision for) income taxes is computed using the asset and liability method, under which deferred tax assets and liabilities are determined based on the difference between the financial statement and tax bases of assets and liabilities using enacted statutory income tax rates in effect for the year in which the differences are expected to affect taxable income. Valuation allowances are established when necessary to reduce net deferred tax assets to the amount expected to be realized.

On December 22, 2017, the Tax Cut and Jobs Act was signed into law, which enacted significant changes to U.S. tax and related laws. Some of the provisions of the new tax law affecting corporations include, but are not limited to a reduction of the federal corporate income tax rate from 35% to 21%, limiting the interest expense deduction, expensing of cost of acquired qualified property, elimination of the domestic production activities deduction and allows net operating losses generated in

taxable years ending after December 31, 2017 to be carried forward indefinitely. The reduction in the federal corporate income tax rate reduced our net deferred tax assets by approximately \$129.2 million with an offsetting reduction to our valuation allowance of approximately \$130.0 million, which has been reflected in our financial statements for the tax year ended December 31, 2017. The rate reduction and the valuation allowance offset is included in the change in rate and change in valuation allowance of approximately \$1.1 million, and is included in the change in valuation allowance of approximately \$1.1 million, and is included in the change in valuation allowance of approximately \$1.1 million, and is included in the change in valuation allowance line item in the reconciliation of the effective tax rate. We continue to evaluate the impact the new tax law will have on our financial condition and results of operations.

During the year ended December 31, 2016, we made an adjustment to goodwill and deferred tax liabilities as a result of the impact of final pre-acquisition Ticketfly income tax returns filed. As a result, during the year ended December 31, 2016, we released \$1.9 million of our valuation allowance and recorded an income tax benefit.

During the year ended December 31, 2015, we released \$1.8 million of our valuation allowance as a result of acquisitions. Deferred tax liabilities were established for the book-tax basis difference related to acquired intangible assets. The net deferred tax liabilities provided an additional source of income to support the realizability of preexisting deferred tax assets.

Liquidity and Capital Resources

As of December 31, 2017, we had cash, cash equivalents and investments totaling \$500.8 million, which primarily consisted of cash and money market funds held at major financial institutions and investment-grade corporate debt securities.

Our principal uses of cash during the year ended December 31, 2017 were funding our operations, as described below, repaying our credit facility and capital expenditures.

Sources of Funds

We believe, based on our current operating plan, that our existing cash and cash equivalents and additional sources of funding will be sufficient to meet our anticipated cash needs for at least the next year.

From time to time, we may explore additional financing sources and means to lower our cost of capital, which could include equity, equity-linked and debt financing. In addition, in connection with any future acquisitions, we may require additional funding which may be provided in the form of additional debt, equity or equity-linked financing or a combination thereof. There can be no assurance that any additional financing will be available to us on acceptable terms.

Our Indebtedness

Credit Facility

On December 29, 2017, we entered into a credit facility for an aggregate commitment amount of \$200.0 million, with an option to increase the commitment amount by \$50.0 million and a maturity date of the earliest of December 29, 2022, 120 days prior to the Notes maturity date of December 1, 2020, provided that the Notes have not been converted into common stock prior to such date or 120 days prior to the Series A redeemable convertible preferred stock ("Series A") redemption date of September 22, 2022, provided that the Series A have not been converted into common stock prior to such date. Prior to December 29, 2017, we were party to a credit facility with an aggregate commitment amount of \$120.0 million, with an option to increase the commitment amount by \$20.0 million and a maturity date of September 12, 2018. This credit facility was terminated on December 29, 2017.

As of December 31, 2017, we had no outstanding borrowings, \$1.2 million in letters of credit outstanding and \$194.1 million of available borrowing capacity under the credit facility. We are in compliance with all financial covenants associated with the credit facility as of December 31, 2017. Refer to <u>Note 9 "Debt Instruments"</u> in the Notes to Consolidated Financial Statements for further details regarding our credit facility.

1.75% Convertible Senior Notes Due 2020

On December 9, 2015, we completed an unregistered Rule 144A offering of \$345.0 million aggregate principal amount of our 1.75% Convertible Senior Notes due 2020. The net proceeds from the sale of the Notes were approximately \$336.5 million, after deducting the initial purchaser's fees and other estimated expenses. We used approximately \$43.2 million of the

net proceeds to pay the cost of the capped call transactions. Refer to Note 9 "Debt Instruments" in the Notes to Consolidated Financial Statements for further details on our Notes.

Redeemable Convertible Preferred Stock

In June 2017, we entered into an agreement with Sirius XM Radio, Inc. ("Sirius XM") to sell480,000 shares of Series A redeemable convertible preferred stock ("Series A") for \$1,000 per share, with gross proceeds of \$480.0 million. The Series A shares were issued in two rounds: an initial closing of 172,500 shares for \$172.5 million that occurred on June 9, 2017 upon signing the agreement with Sirius XM, and an additional closing of 307,500 shares for \$307.5 million that occurred on September 22, 2017. Refer to Note 10 "Redeemable Convertible Preferred Stock" in the Notes to Consolidated Financial Statements for further details on the redeemable convertible preferred stock.

Capital Expenditures

Consistent with previous periods, future capital expenditures will primarily focus on acquiring additional hosting and general corporate infrastructure. Our access to capital is adequate to meet our anticipated capital expenditures for our current plans.

Historical Trends

The following table summarizes our cash flow data for the years endedDecember 31, 2015, 2016 and 2017. Our cash flow data includes Ticketfly results for the two months ended December 31, 2015, the year ended December 31, 2016 and the period ended September 1, 2017.

		Y	ear ended December 31,	
	2015		2016	2017
			(in thousands)	
Net cash used in operating activities	\$ (42,082)	\$	(181,691)	\$ (210,709)
Net cash (used in) provided by investing activities	(102,266)		(52,155)	129,946
Net cash provided by financing activities	303,135		99,757	380,193

Operating activities

In the year ended December 31, 2017, net cash used in operating activities was \$210.7 million and primarily consisted of our net loss of \$518.4 million, which was partially offset by non-cash charges of \$363.9 million, primarily related to \$132.0 million in goodwill impairment, \$128.4 million in stock-based compensation expense and \$62.9 million in depreciation and amortization expense. Net cash used in operating activities also included anincrease in accounts receivable of \$36.8 million, an increase in prepaid and other assets of \$16.6 million and a decrease in accrued compensation of \$10.7 million due to a decrease in accrued bonuses and commissions. Cash used in operating activities increased \$29.0 million from the year ended December 31, 2016, primarily due to a \$175.4 million increase in our net loss, offset by a goodwill impairment charge of \$132.0 million and a charge for loss on sale of \$9.4 million, primarily related to the sale of Ticketfly.

In the year ended December 31, 2016 net cash used in operating activities was\$181.7 million and primarily consisted of our net loss of\$343.0 million, which was partially offset by non-cash charges of \$222.3 million, primarily related to\$138.5 million in stock-based compensation charges and \$60.8 million in depreciation and amortization expense. Net cash used in operating activities also included an increase in prepaid content acquisition costs of \$44.2 million due to minimum guarantee payments and an increase in accounts receivable of \$35.7 million. Cash used in operating activities increased \$139.6 million from the year ended December 31, 2015, primarily due to a \$173.3 million increase in our net loss, offset by a \$36.3 million increase in depreciation and amortization expense as a result of an increase in depreciation expense for leasehold improvements and server equipment and an increase in intangible amortization expense.

In the year ended December 31, 2015, net cash used in operating activities was \$42.1 million and primarily consisted of our net loss of \$169.7 million and increases in accounts receivable and prepaid and other assets of \$55.9 million and \$17.5 million, offset by decreases in accrued content acquisition costs and accounts payable, accrued and other current liabilities of \$23.7 million and \$18.1 million, and non-cash charges of \$141.2 million, primarily related to \$111.6 million in stock-based

Table of Contents

compensation charges and \$24.5 million in depreciation and amortization charges. Cash used in operating activities increased \$63.1 million from the year ended December 31, 2014, primarily due to a \$139.3 million increase in our net loss, offset by a \$24.6 million increase in stock-based compensation expense as a result of an increase in headcount.

Investing activities

In the year ended December 31, 2017, net cash provided by investing activities was \$129.9 million and included \$124.3 million in proceeds from sales of subsidiaries, net of cash, related to the sale of Ticketfly and \$42.1 million in proceeds from maturities of investments, offset by \$20.2 million of capital expenditures for internal-use software primarily related to Pandora Premium and \$15.7 million of capital expenditures for leasehold improvements and server equipment. Net cashprovided by investing activities increased \$182.1 million from the year ended December 31, 2016 primarily due to an increase in proceeds from sales of subsidiaries, net of cash, of\$124.3 million related to the sale of Ticketfly, a decrease in capital expenditures for leasehold improvements and server equipment of \$44.1 million and a decrease in purchases of investments of \$12.4 million.

In the year ended December 31, 2016 net cash used in investing activities was \$52.2 million and included \$59.8 million of capital expenditures for leasehold improvements and server equipment, \$30.2 million of capital expenditures for internal-use software and \$12.4 million in purchases of investments, offset by \$51.2 million in proceeds from sales and maturities of investments. Net cash used in investing activities decreased \$50.1 million from the year ended December 31, 2015 primarily due to a decrease of \$268.9 million in payments related to acquisitions, net of cash acquired, due to the acquisitions of Rdio, Ticketfly, NBS and KXMZ, and a decrease in purchases of investments of \$128.6 million, offset by a decrease in proceeds from of sales and maturities of investments of \$289.2 million.

In the year ended December 31, 2015, net cash used in investing activities was \$102.3 million, primarily due to \$269.6 million in payments related to acquisitions, net of cash acquired, due to the acquisitions of Rdio, Ticketfly, NBS and KXMZ, \$141.0 million of purchases of investments, \$23.5 million of capital expenditures for leasehold improvements and server equipment and \$8.6 million of capital expenditures for internal-use software, offset by \$340.4 million in proceeds from sales and maturities of investments. Net cash used in investing activities decreased \$9.9 million from the year ended December 31, 2014 primarily due to a decrease in purchases of investments of \$199.7 million and an increase in proceeds from sales and maturities of investments of \$81.8 million, offset by payments related to acquisitions, net of cash acquired, of \$269.6 million.

Financing activities

In the year ended December 31, 2017, net cash provided by financing activities was \$380.2 million and included \$480.0 million in proceeds from the issuance of redeemable convertible preferred stock to Sirius XM, offset by \$90.0 million in repayment of our line of credit and \$30.5 million in cash paid for issuance costs primarily related to the issuance of redeemable convertible preferred stock to Sirius XM. Net cash provided by financing activities increased \$280.4 million from the year ended December 31, 2016 primarily due to an increase in proceeds from the issuance of redeemable convertible preferred stock to Sirius AM. Net cash provided by financing activities increased \$280.4 million, offset by a decrease proceeds from borrowings from our line of credit of \$90.0 million, an increase in repayment of our line of credit of \$90.0 million and an increase in payments of issuance costs of \$30.5 million, primarily related to the issuance of redeemable convertible preferred stock to Sirius XM.

In the year ended December 31, 2016 net cash provided by financing activities was\$99.8 million and included \$90.0 million in proceeds from borrowings from our line of credit and 9.7 million in proceeds from our employee stock purchase plan. Net cash provided by financing activities decreased\$203.4 million from the year ended December 31, 2015 primarily due to a decrease of \$345.0 million in proceeds from issuance of the Notes, offset by proceeds of \$90.0 million in borrowings from our line of credit.

In the year ended December 31, 2015, net cash provided by financing activities was\$303.1 million, primarily consisting of \$345.0 million in proceeds from issuance of the Notes, offset by \$43.2 million in payments pursuant to the capped call transaction and \$8.9 million in payment of debt issuance costs related to the issuance of the Notes and the amendment to our line of credit facility. Net cash provided by financing activities increased \$281.5 million from the year ended December 31, 2014 primarily due to\$345.0 million in proceeds from issuance of the Notes, offset by \$43.2 million in payments of the Notes and the amendment to the capped call transaction and \$8.9 million from the year ended December 31, 2014 primarily due to\$345.0 million in proceeds from issuance of the Notes, offset by \$43.2 million in payments pursuant to the capped call transaction and \$8.9 million in payment of debt issuance costs related to the issuance of the Notes and the amendment to our line of credit facility.

Contractual Obligations and Commitments

The following summarizes our contractual obligations as of December 31, 2017:

			Paym	ents Due by Period		
		Less Than				More Than
	 Total	1 Year		1 - 3 Years	4 - 5 Years	5 Years
			((in thousands)		
Minimum guarantees-Content						
acquisition costs	\$ 405,199	\$ 394,199	\$	11,000	\$ —	\$ —
Operating lease obligations	 123,454	24,380		43,735	22,695	32,644
Total	\$ 528,653	\$ 418,579	\$	54,735	\$ 22,695	\$ 32,644

Minimum Guarantees—Content Acquisition Costs

As of December 31, 2017, certain of our content acquisition agreements contain minimum guarantees and require that we make upfront minimum guarantee payments. As of December 31, 2017, we have future minimum guarantee commitments of \$405.2 million, of which \$394.2 million will be paid in 2018 and the remainder will be paid thereafter.

Off-Balance Sheet Arrangements

We do not have any off-balance sheet arrangements and did not have any such arrangements in 2016 or 2017.

Business Trends

Our operating results fluctuate from quarter to quarter as a result of a variety of factors. We expect our operating results to continue to fluctuate in future quarters.

Our results reflect the effects of seasonal trends in listener, subscriber and advertising behavior. During the last quarter of each calendar year, and particularly during the holiday season, we expect to experience both higher advertising sales due to greater advertiser demand during the holiday season and increased usage of our service due to the popularity of holiday music. In addition, in the first quarter of each calendar year, we expect to experience lower advertising sales due to reduced advertiser demand, but sustained higher levels of subscriptions and usage by listeners due to increased use of media-streaming devices and subscriptions received as gifts during the holiday season. We believe these seasonal trends have affected, and will continue to affect our operating results, particularly if increases in content acquisition costs from increased usage are not offset by advertising sales in the first calendar quarter.

In addition, expenditures by advertisers tend to be cyclical and discretionary in nature, reflecting overall economic conditions, the economic prospects of specific advertisers or industries, budgeting constraints and buying patterns and a variety of other factors, many of which are outside our control. As a result of these and other factors, the results of any prior quarterly or annual periods should not be relied upon as indications of our future operating performance.

We continue to invest in building a local advertising sales force in major radio markets and as ofDecember 31, 2017, we had 130 local sellers in 37 markets in the United States. As a result, we experienced an increase in local advertising revenue as a percentage of total advertising revenue in the year ended December 31, 2017 compared to the year ended December 31, 2016, and we intend to continue investing in our local market presence for the foreseeable future.

Critical Accounting Policies and Estimates

Our discussion and analysis of our financial condition and results of operations is based upon our consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these consolidated financial statements requires us to make estimates, judgments and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses and the related disclosure of contingent assets and liabilities. We base our estimates on historical experience and on various other assumptions that we believe are reasonable under the circumstances. Our estimates form the basis for our judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates.

An accounting policy is considered to be critical if it requires an accounting estimate to be made based on assumptions about matters that are highly uncertain at the time the estimate is made, and if different estimates that reasonably could have been used, or changes in the accounting estimate that are reasonably likely to occur, could materially impact the consolidated



Table of Contents

financial statements. We believe that our critical accounting policies reflect the critical accounting estimates and assumptions used in the preparation of the consolidated financial statements.

We believe that the assumptions and estimates associated with our content acquisition costs for performance rights of musical works, prepaid content acquisition costs, advertising revenue, subscription and other revenue, business combinations, goodwill and intangible assets, stock based compensation, the valuation of stock option grants, performance stock units and capitalized internal-use software have the greatest potential impact on our financial statements. Therefore, we consider these to be our critical accounting policies and estimates.

Content Acquisition Costs for Performance Rights of Musical Works

We incur content acquisition cost expenses from our streaming of musical works. Content acquisition costs for musical works are negotiated with and paid to PROs such as ASCAP, BMI, SESAC and Global Music Rights and directly to publishing companies. Content acquisition costs for the streaming of musical works on our ad-supported service are calculated such that each copyright holder receives its usage-based and ownership-based share of a royalty pool equal to 20% of the content acquisition costs paid by us for sound recordings on our ad-supported service. Content acquisition costs for the streaming of musical works on our subscription services are equal to the rates determined in accordance with the statutory license set forth in Section 115. We record a liability for content acquisition costs for the streaming of musical works based on our best estimate of the amount owed to each licensor, PRO or individual copyright owner, based on historical rates, third-party evidence and legal developments consistent with our past practices. For each quarterly period, we evaluate our estimates to assess the adequacy of recorded liabilities. If actual content acquisition costs differ from estimates, revisions to the estimated royalty liabilities may be required, which could materially affect our results of operations.

Prepaid Content Acquisition Costs

Prepaid content acquisition costs are primarily comprised of minimum guarantees under content license agreements. In 2015 and 2016, we signed direct license agreements with major and independent labels, distributors, PROs and publishers. Certain of these license agreements include minimum guarantee payments, some of which are paid in advance. These minimum guarantees may take the form of either a contractually obligated minimum over a specified period of time that requires a true-up payment at the end of the specified period, which cash payments are then recoupable against content acquisition costs over the specified period. On a quarterly basis, we record the greater of the cumulative actual content acquisition costs incurred or the cumulative minimum guarantee based on forecasted usage for the minimum guarantee period. The minimum guarantee relates to, as specified in each agreement, which may be annual or a longer period. The cumulative minimum guarantee, based on forecasted usage, considers factors such as listening hours, revenue, subscribers and other terms of each agreement that impact our expected attainment or recoupanent of the minimum guarantees on a non-straight line basis. If we are unable to accurately estimate the forecasted usage for the minimum guarantee period, our content acquisition costs could increase and have a material adverse effect on our business, financial condition and results of operations.

Revenue Recognition

We recognize revenue when four basic criteria are met: (1) persuasive evidence exists of an arrangement with the customer reflecting the terms and conditions under which the products or services will be provided; (2) delivery has occurred or services have been provided; (3) the fee is fixed or determinable; and (4) collection is reasonably assured. We consider a signed agreement, a binding insertion order or other similar documentation to be persuasive evidence of an arrangement. Collectability is assessed based on a number of factors, including transaction history and the creditworthiness of a customer. If it is determined that collection is not reasonably assured, revenue is not recognized until collection becomes reasonably assured, which is generally upon receipt of cash. We record cash received in advance of revenue recognition as deferred revenue.

Advertising revenue

We generate advertising revenue primarily from audio, display and video advertising. We generate the majority of our advertising revenue through the delivery of advertising impressions sold on a cost per thousand, or CPM, basis. In determining whether an arrangement exists, we ensure that a binding arrangement, such as an insertion order or a fully executed customer-specific agreement, is in place. We generally recognize revenue based on delivery information from our campaign trafficking systems. If we are unable to accurately estimate the delivery of information, our advertising revenue could be misstated and have a material adverse effect on our business, financial condition and results of operations.



Subscription and other revenue

Subscription and other revenue is generated primarily through the sale of monthly or annual paid subscriptions to Pandora Plus and Pandora Premium. Pandora Plus is a paid, ad-free version of the Pandora service that includes replays, additional skipping, offline listening, higher quality audio on supported devices and longer timeout-free listening. Pandora Premium is a paid, ad-free version of the Pandora service that also offers a unique, on-demand experience, providing users with the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button and automatically generates playlists based on their listening activity. Subscription revenue derived from direct sales to listeners is recognized on a straight-line basis over the duration of the subscription period. Subscription revenue derived from sales through some mobile device app stores may be subject to refund or cancellation terms which may affect the timing or amount of the subscription revenue recognition. When refund rights exist, we recognize revenue when services have been provided and the rights lapse or when we have developed sufficient transaction history to estimate a reserve. If we are unable to accurately estimate a reserve, our subscription revenue could be misstated and have a material adverse effect on our business, financial condition and results of operations.

Business Combinations, Goodwill and Intangible Assets, net

We allocate the fair value of purchase consideration to the tangible assets acquired, liabilities assumed and intangible assets acquired based on their estimated fair values. The excess of the fair value of purchase consideration over the fair values of these identifiable assets and liabilities is recorded as goodwill. Such valuations require management to make significant estimates and assumptions, especially with respect to intangible assets. Significant estimates in valuing certain intangible assets include, but are not limited to, future expected cash flows from acquired users, acquired technology, and trade names from a market participant perspective, useful lives and discount rates. Management's estimates of fair value are based upon assumptions believed to be reasonable, but which are inherently uncertain and unpredictable and, as a result, actual results may differ from estimates. During the measurement period, which is one year from the acquisition date, we may record adjustments to the assets acquired and liabilities assumed, with the corresponding offset to goodwill. Upon the conclusion of the measurement period, any subsequent adjustments are recorded to earnings.

We do not amortize goodwill and intangible assets with indefinite useful lives, rather such assets are required to be tested for impairment at least annually or sooner whenever events or changes in circumstances indicate that the assets may be impaired. We perform our annual goodwill and intangible asset impairment tests in the fourth quarter of each year. In the year ended December 31, 2017, we recognized \$132.0 million in impairment of goodwill, primarily related to the sale of Ticketfly.

Acquired finite-lived intangible assets are amortized over the estimated useful lives of the assets, which range from three to eleven years. Acquired finite-lived intangible assets consist primarily of patents, customer relationships, developed technology and trade names resulting from business combinations. We evaluate the recoverability of our intangible assets for potential impairment whenever events or circumstances indicate that the carrying amount of such assets may not be recoverable. Recoverability of these assets is measured by a comparison of the carrying amounts to the future undiscounted cash flows the assets are expected to generate. If such review indicates that the carrying amount of intangible assets is not recoverable, the carrying amount of such assets is reduced to the fair value.

In addition to the recoverability assessment, we routinely review the remaining estimated useful lives of finite-lived intangible assets. If we reduce the estimated useful life assumption for any asset, the remaining unamortized balance would be amortized over the revised estimated useful life. We record the amortization of intangible assets to the financial statement line item in our consolidated statement of operations that the asset directly relates to. To the extent that purchased intangibles are used in revenue generating activities, we record the amortization of these intangible assets to cost of revenue.

Stock-Based Compensation—Restricted Stock Units and Stock Options

Stock-based awards granted to employees, including grants of restricted stock units ("RSUs") and stock options, are recognized as expense in our statements of operations based on their grant date fair value. We recognize stock-based compensation expense on a straight-line basis over the service period of the award, which is generally three to four years. We estimate the fair value of RSUs at our stock price on the grant date.

We generally estimate the grant date fair value of stock options using the Black-Scholes option-pricing model. The inputs to the Black-Scholes option-pricing model are our stock price on the date of grant, the expected stock price volatility over the expected term of the award, which is based on projected employee stock option exercise behaviors, the risk-free interest rate for the expected term of the award and expected dividends. If we are unable to accurately estimate the inputs to the

Black-Scholes option-pricing model, our stock compensation expense could be misstated and have a material adverse effect on our business, financial condition and results of operations.

Stock-based compensation expense is recorded in the statement of operations for only those stock-based awards that will vest. In the first quarter of 2017 we adopted new accounting guidance from the Financial Accounting Standards Board ("FASB") on stock compensation, or ASU 2016-09, as described in "Recently Adopted Accounting Standards" in Note 2 of the Notes to Consolidated Financial Statements and have elected to account for forfeitures as they occur, rather than estimating expected forfeitures. In addition, we recognize all income tax effects of awards in the income statement when the awards vest or are settled as required by ASU 2016-09.

Stock-Based Compensation—Performance Stock Units ("PSUs")

In 2016, the compensation committee of the board of directors granted 2016 Performance Awards consisting of stock-settled performance-based RSUs to certain key executives.

We have determined the grant-date fair value of the PSUs granted in 2016 using a Monte Carlo simulation performed by a third-party valuation firm. The Monte Carlo simulation model utilizes multiple input variables to estimate the probability that market conditions will be achieved. The variables used in these models are reviewed on an annual basis and adjusted, as needed. We recognize stock-based compensation for the PSUs over the requisite service period, which is approximately four years, using the accelerated attribution method. If we are unable to accurately estimate the grant-date fair value of the PSUs, our stock-based compensation expense could be misstated and have a material adverse effect on our business, financial condition and results of operations.

Capitalized Internal-Use Software

We capitalize certain costs incurred to develop software for internal use. Costs incurred in the preliminary stages of development are expensed as incurred. Once software has reached the development stage, internal and external costs, if direct and incremental, are capitalized until the software is substantially complete and ready for its intended use. We also capitalize costs related to specific upgrades and enhancements when it is probable the expenditures will result in additional functionality. Capitalized costs are recorded as part of property and equipment. Costs related to minor enhancements, maintenance and training are expensed as incurred.

Capitalized internal-use software costs are amortized on a straight-line basis over their three to five-year estimated useful lives. We evaluate the useful lives of these assets and test for impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets. There were no material impairments to internal-use software during the years ended December 31, 2015, 2016 and 2017. If the assumptions used in estimating in our capitalized internal-use software amounts and the related amortization could be misstated and have a material adverse effect on our business, financial condition and results of operations.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to market risks in the ordinary course of our business, including interest rate and inflation risks.

Interest Rate Fluctuation Risk

On December 9, 2015, we completed an unregistered Rule 144A offering for the issuance of \$345.0 million aggregate principal amount of our 1.75% Convertible Senior Notes due 2020 (the "Notes"). The Notes were offered only to qualified institutional buyers pursuant to Rule 144A under the Securities Act. In connection with the issuance of the Notes, we entered into capped call transactions with the initial purchaser of the Notes and an additional financial institution. The Notes are unsecured, senior obligations of Pandora, and interest is payable semi-annually at a rate of 1.75% per annum, with no interest rate fluctuation risk.

Our exposure to interest rates relates to the increase or decrease in the amount of interest we must pay on our outstanding debt instruments. We are party to \$200.0 million credit facility with a syndicate of financial institutions, which expires on the earliest of December 29, 2022, 120 days prior to the Notes maturity date of December 1, 2020, provided that the Notes have not been converted in accordance with their terms prior to such date or 120 days prior to the Series A redeemable convertible preferred stock ("Series A") redemption date of September 22, 2022, provided that the Series A have not been converted into common stock prior to such date. As of December 31, 2017, we had \$1.2 million in letters of credit outstanding and \$194.1 million of available borrowing capacity under the credit facility.

Refer to Note 9 "Debt Instruments" in the Notes to Consolidated Financial Statements for further details regarding our credit facility and convertible notes.

The primary objective of our investment activities is to preserve principal while maximizing income without significantly increasing risk. Approximately100% of our portfolio consists of cash and cash equivalents that have a relatively short maturity, and a fair value relatively insensitive to interest rate changes. Our available-for-sale investments consist of corporate debt securities which may be subject to market risk due to changes in prevailing interest rates that may cause the fair values of our investments to fluctuate. Based on a sensitivity analysis, we have determined that a hypothetical 100 basis points increase in interest rates would have no impact to the fair values of our investments as of December 31, 2017. In future periods, we will continue to evaluate our investment policy in order to ensure that we continue to meet our overall objectives.

Inflation Risk

We do not believe that inflation has had a material effect on our business, financial condition or results of operations. However, if our costs were to become subject to significant inflationary pressures, we may not be able to fully offset such higher costs through price increases. Our inability or failure to do so could harm our business, financial condition and results of operations.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

PANDORA MEDIA, INC. INDEX TO CONSOLIDATED FINANCIAL STATEMENTS

	Page No.
Reports of Ernst & Young LLP, Independent Registered Public Accounting Firm	<u>63</u>
Consolidated Financial Statements:	<u>65</u>
Consolidated Balance Sheets	<u>65</u>
Consolidated Statements of Operations	<u>66</u>
Consolidated Statements of Comprehensive Loss	<u>67</u>
Consolidated Statements of Stockholders' Equity	<u>68</u>
Consolidated Statements of Cash Flows	<u>69</u>
Notes to Consolidated Financial Statements	<u>71</u>

Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of Pandora Media, Inc.

Opinion on the Financial Statements

We have audited the accompanying consolidated balance sheets of Pandora Media, Inc. ("the Company") as of December 31, 2017 and 2016, the related consolidated statements of operations, comprehensive loss, stockholders' equity and cash flows for each of the three years in the period ended December 31, 2017, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the consolidated financial position of the Company at December 31, 2017 and 2016, and the consolidated results of its operations and its cash flows for each of the three years in the period ended December 31, 2017, in conformity with U.S. generally accepted accounting principles.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of December 31, 2017, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework) and our report dated February 26, 2018 expressed an unqualified opinion thereon.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ Ernst & Young LLP

We have served as the Company's auditor since 2010. San Francisco, California February 26, 2018

Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of Pandora Media, Inc.

Opinion on Internal Control over Financial Reporting

We have audited Pandora Media, Inc.'s internal control over financial reporting as of December 31, 2017, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework) (the COSO criteria). In our opinion, Pandora Media, Inc. (the Company) maintained, in all material respects, effective internal control over financial reporting as of December 31, 2017, based on the COSO criteria.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the 2017 consolidated financial statements of Pandora Media, Inc. and our report dated February 26, 2018 expressed an unqualified opinion thereon.

Basis for Opinion

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting included in the accompanying Management's Report on Internal Control over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

Definition and Limitations of Internal Control Over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ Ernst & Young LLP

San Francisco, California February 26, 2018

Pandora Media, Inc. Consolidated Balance Sheets (in thousands, except share and per share amounts)

	As of De	cember	31,
	 2016		2017
Assets			
Current assets			
Cash and cash equivalents	\$ 199,944	\$	499,597
Short-term investments	37,109		1,250
Accounts receivable, net of allowance of \$3,633 at December 31, 2016 and \$5,352 at December 31, 2017	309,267		336,429
Prepaid content acquisition costs	46,310		55,668
Prepaid expenses and other current assets	33,191		19,220
Total current assets	625,821		912,164
Convertible promissory note receivable	_		35,471
Long-term investments	6,252		_
Property and equipment, net	124,088		116,742
Goodwill	306,691		71,243
Intangible assets, net	90,425		19,409
Other long-term assets	31,533		11,293
Total assets	\$ 1,184,810	\$	1,166,322
Liabilities, redeemable convertible preferred stock and stockholders' equity			
Current liabilities			
Accounts payable	\$ 15,224	\$	14,896
Accrued liabilities	35,465		34,535
Accrued content acquisition costs	93,723		97,751
Accrued compensation	60,353		47,635
Deferred revenue	28,359		31,464
Other current liabilities	20,993		—
Total current liabilities	 254,117		226,281
Long-term debt	342,247		273,014
Other long-term liabilities	34,187		23,500
Total liabilities	630,551	-	522,795
Commitments and contingencies (Note 11)			
Redeemable convertible preferred stock, \$0.0001 par value, 10,000,000 shares authorized: 480,000 shares issued and outstanding at December 31, 2017 (liquidation preference \$480,000,000)	_		490,849
Stockholders' equity			
Common stock, \$0.0001 par value, 1,000,000,000 shares authorized: 235,162,757 shares issued and outstanding at December 31, 2016 and 250,867,462 at December 31, 2017	24		25
Additional paid-in capital	1,264,693		1,422,221
Accumulated deficit	(709,636)		(1,269,351)
Accumulated other comprehensive loss	(822)		(217)
Total stockholders' equity	554,259		152,678
Total liabilities, redeemable convertible preferred stock and stockholders' equity	\$ 1,184,810	\$	1,166,322

The accompanying notes are an integral part of the consolidated financial statements.

Pandora Media, Inc. Consolidated Statements of Operations (in thousands, except per share amounts)

	Year ended December 31,				
	 2015		2016		2017
Revenue					
Advertising	\$ 933,305	\$	1,072,490	\$	1,074,927
Subscription and other	220,571		225,786		315,853
Ticketing service	10,167		86,550		76,032
Total revenue	 1,164,043		1,384,826	<u> </u>	1,466,812
Cost of revenue					
Cost of revenue—Content acquisition costs	610,362		734,353		804,032
Cost of revenue—Other	79,972		102,717		112,638
Cost of revenue—Ticketing service	7,121		59,280		50,397
Total cost of revenue	697,455		896,350		967,067
Gross profit	 466,588		488,476		499,745
Operating expenses					
Product development	84,135		140,707		154,325
Sales and marketing	397,842		490,364		492,542
General and administrative	154,602		176,164		190,711
Goodwill impairment	_		—		131,997
Contract termination fees	 		—		23,044
Total operating expenses	636,579		807,235		992,619
Loss from operations	(169,991)		(318,759)		(492,874)
Interest expense	(1,976)		(26,144)		(29,335)
Other income, net	756		1,697		3,024
Total other expense, net	(1,220)		(24,447)		(26,311)
Loss before benefit from income taxes	(171,211)		(343,206)		(519,185)
Benefit from income taxes	1,550		228		790
Net loss	\$ (169,661)	\$	(342,978)	\$	(518,395)
Net loss available to common stockholders	\$ (169,661)	\$	(342,978)	\$	(558,561)
Basic and diluted net loss per common share	\$ (0.79)	\$	(1.49)	\$	(2.29)
Weighted-average basic and diluted common shares	 213,790		230,693		243,637

The accompanying notes are an integral part of the consolidated financial statements.

Pandora Media, Inc. Consolidated Statements of Comprehensive Loss (in thousands)

	Year ended December 31,				
	 2015		2016		2017
Net loss	\$ (169,661)	\$	(342,978)	\$	(518,395)
Change in foreign currency translation adjustment	53		(594)		553
Change in net unrealized loss on marketable securities	106		289		52
Other comprehensive income (loss)	 159		(305)		605
Total comprehensive loss	\$ (169,502)	\$	(343,283)	\$	(517,790)

The accompanying notes are an integral part of the consolidated financial statements.

Other comprehensive income

Balances as of December 31, 2017

Pandora Media, Inc. Consolidated Statements of Stockholders' Equity (in thousands, except share amounts) **Common Stock** Additional Paid-Accumulated Other **Total Stockholders'** Accumulated Shares Par Amount **Comprehensive Loss** Deficit in Capital Equity 209,071,488 21 781,009 \$ (676) (196,997) 583,357 Balances as of December 31, 2014 \$ \$ \$ \$ Issuance of common stock upon exercise of stock options 1,077,797 5,156 5,156 Issuance of common stock related to 10,246,616 2 148,488 148,490 acquisitions Stock-based compensation 111,645 111,645 Vesting of restricted stock units 4,184,415 Share cancellations to satisfy tax withholding on vesting of restricted stock units (148,302) (2,540)(2,540) Stock issued under employee stock purchase 538,398 6,973 6,973 plan Equity component of convertible note issuance, net of issuance costs 102,968 102,968 Purchase of capped call (43, 160)(43, 160)Components of comprehensive loss: Net loss (169,661) (169,661) Other comprehensive loss 159 159 224,970,412 23 1,110,539 (517) (366,658) 743,387 Balances as of December 31, 2015 \$ \$ \$ \$ \$ Issuance of common stock upon exercise of stock options 1,588,781 3,464 3,464 Stock-based compensation 145,968 145,968 Vesting of restricted stock units 7,666,647 1 (1)Vesting of market stock units 56,903 Share cancellations to satisfy tax withholding on vesting of restricted stock units (354,638) (3,368) (3,368) Stock issued under employee stock purchase 1,254,910 8,484 8,484 plan (393) (20,258) (393) Fair value of escrow settlement Components of comprehensive loss: Net loss (342,978) (342,978) Other comprehensive income (305) (305) Balances as of December 31, 2016 235,162,757 \$ 24 \$ 1,264,693 \$ (822) \$ (709,636) \$ 554,259 Issuance of common stock upon exercise of 4,968,577 9,789 9,789 stock options 135,788 135,788 Stock-based compensation ____ Vesting of restricted stock units 9,478,804 1 (1)Stock issued under employee stock purchase 1,287,687 11,387 11,387 plan Fair value of escrow settlement (30,363) (589) (589) Cumulative effect adjustment from adoption of ASU 2016-09 1.154 (1.154)Series A preferred stock dividends (40,166) (40,166) Components of comprehensive loss: Net loss (518,395) (518,395)

The accompanying notes are an integral part of the consolidated financial statements.

1,422,221

\$

25 \$

250,867,462

\$

605

(217)

\$

(1,269,351)

\$

605

152,678

Pandora Media, Inc. Consolidated Statements of Cash Flows (in thousands)

		Year ended December 31,		
	2015	2016	2017	
Operating activities				
Net loss	\$ (169,661)	\$ (342,978)	\$ (518,395)	
Adjustments to reconcile net loss to net cash used in operating activities				
Goodwill impairment	—	—	131,997	
Loss on sales of subsidiaries	_	—	9,378	
Depreciation and amortization	24,458	60,757	62,948	
Stock-based compensation	111,645	138,458	128,431	
Amortization of premium on investments, net	1,911	405	81	
Accretion of discount on convertible promissory note receivable	_	_	(687)	
Other operating activities	196	881	447	
Amortization of debt discount	1,084	18,315	20,153	
Interest income	_	—	(1,081)	
Provision for bad debt	1,938	3,522	12,211	
Changes in restricted cash	_	_	(1,257)	
Changes in assets and liabilities				
Accounts receivable	(55,904)	(35,710)	(36,760)	
Prepaid content acquisition costs	(1,399)	(44,211)	(9,358)	
Prepaid expenses and other assets	(17,519)	(12,321)	(16,566)	
Accounts payable, accrued and other current liabilities	18,080	5,294	9,053	
Accrued content acquisition costs	23,736	(3,668)	4,016	
Accrued compensation	7,378	15,364	(10,679)	
Other long-term liabilities	6,005	1,384	(3,007)	
Deferred revenue	4,946			
		8,420	3,105	
Reimbursement of cost of leasehold improvements	1,024	4,397	5,261	
Net cash used in operating activities	(42,082)	(181,691)	(210,709)	
Investing activities		(20.8.0.)		
Purchases of property and equipment	(23,512)	(59,769)	(15,656)	
Internal-use software costs	(8,562)	(30,210)	(20,157)	
Changes in restricted cash		(250)	(642)	
Purchases of investments	(140,980)	(12,413)	—	
Proceeds from maturities of investments	228,998	47,656	42,082	
Proceeds from sales of investments	111,356	3,507	-	
Proceeds from sales of subsidiaries, net of cash	_	-	124,319	
Payments related to acquisitions, net of cash acquired	(269,566)	(676)		
Net cash (used in) provided by investing activities	(102,266)	(52,155)	129,946	
Financing activities				
Proceeds from issuance of redeemable convertible preferred stock	_	_	480.000	
Payments of issuance costs			480,000	
-		(32)	(30,511)	
Proceeds from issuance of convertible notes	345,000			
Payments for purchase of capped call	(43,160)	-	-	
Repayment of debt arrangements		-	(90,000)	
Borrowings under debt arrangements		90,000	-	
Proceeds from employee stock purchase plan	7,552	9,701	10,926	
Proceeds from exercise of stock options	5,192	3,457	9,778	
Tax payments from net share settlements of restricted stock units	(2,540)	(3,369)		
Net cash provided by financing activities	303,135	99,757	380,193	
Effect of exchange rate changes on cash and cash equivalents	(77)	(634)	223	
Net increase (decrease) in cash and cash equivalents	158,710	(134,723)	299,653	
Cash and cash equivalents at beginning of period	175,957	334,667	199,944	
Cash and cash equivalents at end of period	\$ 334,667	\$ 199,944	\$ 499,597	

Table of Contents

Supplemental disclosures of cash flow information			
Cash paid during the period for income taxes	\$ 389	\$ 297	\$ 209
Cash paid during the period for interest	\$ 351	\$ 7,222	\$ 8,981
Purchases of property and equipment recorded in accounts payable and accrued liabilities	\$ 5,890	\$ 1,129	\$ 2,106
Fair value of shares issued related to the acquisition of a business	\$ 146,855	\$ —	\$ —
Accretion of preferred stock issuance costs	\$ —	\$ —	\$ 29,317
Stock dividend payable to preferred stockholders	\$ _	\$ _	\$ 10,849
Fair value of convertible promissory note receivable received as partial consideration for sale of subsidiary	\$ _	\$ _	\$ 36,203

The accompanying notes are an integral part of the consolidated financial statements.

1. Description of Business and Basis of Presentation

Pandora-Streaming Radio Service and On Demand Music Services

Pandora is the world's most powerful music discovery platform, offering a personalized experience for each of our listeners wherever and whenever they want to listen to music—whether through mobile devices, car speakers or connected devices in the home.

Pandora is available as an ad-supported radio service, a radio subscription service called Pandora Plus and an on-demand subscription service called Pandora Premium. The majority of our listener hours occur on mobile devices, with the majority of our revenue generated from advertising on our ad-supported service on these devices. With billions of data points that help us understand our users' preferences, we offer both local and national advertisers the opportunity to deliver targeted messages to our listeners using a combination of audio, display and video advertisements. We also generate increasing revenue from subscriptions to Pandora Plus and Pandora Premium. We were incorporated as a California corporation in January 2000 and reincorporated as a Delaware corporation in December 2010. Our principal operations are located in the United States.

Ticketing Service

Prior to September 1, 2017, we operated a ticketing service through our former subsidiary Ticketfly, a leading live events technology company that provides ticketing and marketing software and services for clients, which are venues and event promoters, across North America. We completed the sale of Ticketfly on September 1, 2017. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

As used herein, "Pandora," "we," "our," "the Company" and similar terms include Pandora Media, Inc. and its subsidiaries, unless the context indicates otherwise.

Basis of Presentation

The consolidated financial statements and accompanying notes have been prepared in accordance with United States generally accepted accounting principles ("U.S. GAAP") and include the accounts of Pandora and our wholly owned subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

Certain changes in presentation have been made to conform the prior period presentation to current period reporting. We have reclassified a portion of amortization of internal-use software costs from the product development and sales and marketing line items to the cost of revenue—other and general and administrative line items of our consolidated statements of operations. We have also reclassified bad debt and goodwill impairment from the other operating activities line item to the bad debt and goodwill impairment line items of the consolidated statements of cash flows.

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make certain estimates, judgments and assumptions that affect the reported amounts of assets and liabilities and the related disclosures at the date of the financial statements, as well as the reported amounts of revenue and expenses during the periods presented. Estimates are used in several areas including, but not limited to determining accrued content acquisition costs, amortization of minimum guarantees under content acquisition agreements, selling prices for elements sold in multiple-element arrangements, the allowance for doubtful accounts, the fair value of stock options, stock-settle performance-based restricted stock units ("PSUs"), the Employee Stock Purchase Plan ("ESPP"), the provision for (benefit from) income taxes, the fair value of the convertible subordinated promissory note ("Convertible Promissory Note"), fair value of acquired intangible assets and goodwill and the useful lives of acquired intangible assets. To the extent there are material differences between these estimates, judgments or assumptions and actual results, our financial statements could be affected. In many cases, the accounting treatment of a particular transaction is specifically dictated by U.S. GAAP and does not require management's judgment in its application. There are also areas in which management's judgment in selecting among available alternatives would not produce a materially different result.

2. Summary of Significant Accounting Policies

Revenue Recognition

We recognize revenue when four basic criteria are met: (1) persuasive evidence exists of an arrangement with the customer reflecting the terms and conditions under which the products or services will be provided; (2) delivery has occurred or services have been provided; (3) the fee is fixed or determinable; and (4) collection is reasonably assured. We consider a signed agreement, a binding insertion order or other similar documentation to be persuasive evidence of an arrangement. Collectability is assessed based on a number of factors, including transaction history and the creditworthiness of a customer. If it is determined that collection is not reasonably assured, revenue is not recognized until collection becomes reasonably assured, which is generally upon receipt of cash. We record cash received in advance of revenue recognition as deferred revenue.

Gross versus net revenue recognition. We report revenue on a gross or net basis based on management's assessment of whether we act as a principal or agent in the transaction. To the extent we act as the principal, revenue is reported on a gross basis. The determination of whether we act as a principal or an agent in a transaction is based on an evaluation of whether we have the substantial risks and rewards of ownership under the terms of an arrangement.

Advertising revenue. We generate advertising revenue primarily from audio, display and video advertising. We generate the majority of our advertising revenue through the delivery of advertising impressions sold on a cost per thousand, or CPM, basis. In determining whether an arrangement exists, we ensure that a binding arrangement, such as an insertion order or a fully executed customer-specific agreement, is in place. We generally recognize revenue based on delivery information from our campaign trafficking systems.

We also generate advertising revenue pursuant to arrangements with advertising agencies and brokers. Under these arrangements, we provide the agencies and brokers the ability to sell advertising inventory on our service directly to advertisers. We report this revenue net of amounts due to agencies and brokers because we are not the primary obligor under these arrangements, we do not set the pricing nor do we establish or maintain the relationships with the advertisers.

Subscription and other revenue. Subscription and other revenue is generated primarily through the sale of monthly or annual paid subscriptions to Pandora Plus and Pandora Premium. Pandora Plus is a paid, ad-free version of the Pandora service that includes replays, additional skipping, offline listening, higher quality audio on supported devices and longer timeout-free listening. Pandora Premium is a paid, ad-free version of the Pandora service that also offers a unique, on-demand experience, providing users with the ability to search, play and collect songs and albums, build playlists on their own or with the tap of a button and automatically generates playlists based on their listening activity. Subscription revenue derived from direct sales to listeners is recognized on a straight-line basis over the duration of the subscription period and is recognized net of sales tax amounts collected from subscription revenue derived from sales through some mobile device app stores may be subject to refund or cancellation terms which may affect the timing or amount of the subscription revenue recognition. When refund rights exist, we recognize revenue when services have been provided and the rights lapse or when we have developed sufficient transaction history to estimate a reserve.

Multiple-element arrangements. We enter into arrangements with customers to sell advertising packages that include different media placements or ad services that are delivered at the same time, or within close proximity of one another. We recognize the relative fair value of the media placements or ad services as they are delivered assuming all other revenue recognition criteria are met.

We allocate arrangement consideration in multiple-deliverable revenue arrangements at the inception of an arrangement to all deliverables or those packages in which all components of the package are delivered at the same time, based on the relative selling price method in accordance with the selling price hierarchy, which includes: (1) vendor-specific objective evidence ("VSOE") if available; (2) third-party evidence ("TPE") if VSOE is not available; and (3) best estimate of selling price ("BESP") if neither VSOE nor TPE is available.

We determine VSOE based on our historical pricing and discounting practices for the specific product or service when sold separately. In determining VSOE, we require that a substantial majority of the selling prices for these services fall within a reasonably narrow pricing range. We have not historically priced our advertising products within a narrow range. As a result, we have not been able to establish VSOE for any of our advertising products.

When VSOE cannot be established for deliverables in multiple element arrangements, we apply judgment with respect to whether it can establish a selling price based on TPE. TPE is determined based on competitor prices for similar deliverables when sold separately. Generally, our go-to-market strategy differs from that of our peers and our offerings contain a significant level of differentiation such that the comparable pricing of services cannot be obtained. Furthermore, we are unable to reliably

determine what similar competitor services' selling prices are on a stand-alone basis. As a result, we have not been able to establish selling price based on TPE.

When we are unable to establish selling price using VSOE or TPE, we use BESP in our allocation of arrangement consideration. The objective of BESP is to determine the price at which we would transact a sale if the service were sold on a stand-alone basis. BESP is generally used to allocate the selling price to deliverables in our multiple element arrangements. We determine BESP for deliverables by considering multiple factors including, but not limited to, prices we charge for similar offerings, market conditions, competitive landscape and pricing practices. We limit the amount of allocable arrangement consideration to amounts that are fixed or determinable and that are not contingent on future performance or future deliverables. We regularly review BESP. Changes in assumptions or judgments or changes to the elements in the arrangement may cause an increase or decrease in the amount of revenue that we report in a particular period.

Ticketing service revenue. Prior to the sale of Ticketfly, ticketing service revenue was generated primarily from service and merchant processing fees generated on ticket sales through the Ticketfly platform. Ticketfly sells tickets to fans for events on behalf of clients and charges either a fee per ticket or a percentage of the total convenience charge and order processing fee for its services at the time the ticket for an event is sold. Ticketfly service revenue is recorded net of the face value of the ticket at the time of the sale, as Ticketfly generally acts as an agent with respect to the ticket price in these transactions.

Concentration of Credit Risk

Financial instruments that potentially subject us to concentrations of credit risk consist principally of cash and cash equivalents, investments and trade accounts receivable. We maintain cash and cash equivalents with domestic financial institutions of high credit quality. We perform periodic evaluations of the relative credit standing of such institutions.

We perform ongoing credit evaluations of customers to assess the probability of accounts receivable collection based on a number of factors, including past transaction experience with the customer, evaluation of their credit history, and review of the invoicing terms of the contract. We generally do not require collateral. We maintain reserves for potential credit losses on customer accounts when deemed necessary. Actual credit losses during the years ended December 31, 2015, 2016 and 2017 were \$1.1 million, \$2.0 million and \$3.6 million, respectively.

For the years ended December 31, 2015, 2016 and 2017, we had no individual customers that accounted for 10% or more of total revenue. As of December 31, 2016 and 2017, there were no individual customers that accounted for 10% or more of our total accounts receivable.

Cash, Cash Equivalents and Investments

We classify our highly liquid investments with maturities of three months or less at the date of purchase as cash equivalentsOur investments consist of corporate debt securities. These investments are classified as available-for-sale securities and are carried at fair value with the unrealized gains and losses reported as a component of stockholders' equity. Management determines the appropriate classification of our investments at the time of purchase and reevaluates the available-for-sale designations as of each balance sheet date. We classify our investments as either short-term or long-term based on each instrument's underlying contractual maturity date. Investments with maturities of twelve months or less are classified as short-term and those with maturities greater than twelve months are classified as long-term. The cost basis for investments sold is based upon the specific identification method.

Accounts Receivable and Allowance for Doubtful Accounts

Accounts receivable are recorded net of an allowance for doubtful accounts. Our allowance for doubtful accounts is based upon historical loss patterns, the number of days that billings are past due and an evaluation of the potential risk of loss associated with delinquent accounts. We also consider any changes to the financial condition of our customers and any other external market factors that could impact the collectability of our receivables in the determination of our allowance for doubtful accounts. Accounts receivable amounts that are deemed uncollectable are charged against the allowance for doubtful accounts when identified.

Prepaid Content Acquisition Costs

Prepaid content acquisition costs are primarily comprised of minimum guarantees under content license agreements. In 2015 and 2016, we signed direct license agreements with major and independent labels, distributors and publishers. Certain of these license agreements include minimum guarantee payments, some of which are paid in advance. These minimum



guarantees may take the form of either a contractually obligated minimum over a specified period of time that requires a true-up payment at the end of the specified period if the cumulative payments have not met or exceeded the specified minimum, or cash advance payments made at the beginning of, or at intervals during, the specified period, which cash payments are then recoupable against content acquisition costs over the specified period. On a quarterly basis, we record the greater of the cumulative actual content acquisition costs incurred or the cumulative minimum guarantee based on forecasted usage for the minimum guarantee period. The minimum guarantee period is the period of time that the minimum guarantee relates to, as specified in each agreement, which may be annual or a longer period. The cumulative minimum guarantee, based on forecasted usage, considers factors such as listening hours, revenue, subscribers and other terms of each agreement that impact our expected attainment or recoupment of the minimum guarantees on a non-straight line basis. As of December 31, 2016 and 2017, we had prepaid content acquisition costs of \$46.3 million and \$55.7 million.

Property and Equipment, net

Property and equipment is recorded at cost, less accumulated depreciation and amortization. Depreciation is computed using the straight-line method based on the estimated useful lives of the assets, which typically range from three to five years. Leasehold improvements are amortized over the shorter of the lease term or expected useful lives of the improvements.

Property and equipment is reviewed for impairment whenever events or changes in circumstances indicate the carrying amount of an asset may not be recoverable. Recoverability of these assets is measured by a comparison of the carrying amounts to the future undiscounted cash flows the assets are expected to generate. If property and equipment are considered to be impaired, the impairment to be recognized equals the amount by which the carrying value of the asset exceeds its fair market value. If we reduce the estimated useful life assumption for any asset, the remaining unamortized balance would be amortized or depreciated over the revised estimated useful life.

Capitalized Internal-Use Software

We capitalize certain costs incurred to develop software for internal use. Costs incurred in the preliminary stages of development are expensed as incurred. Once software has reached the development stage, internal and external costs, if direct and incremental, are capitalized until the software is substantially complete and ready for its intended use. We also capitalize costs related to specific upgrades and enhancements when it is probable the expenditures will result in additional functionality. Capitalized costs are recorded as part of property and equipment. Costs related to minor enhancements, maintenance and training are expensed as incurred.

Capitalized internal-use software costs are amortized on a straight-line basis over theirthree to five-year estimated useful lives. As of December 31, 2016 and 2017, we had approximately \$25.7 million and \$50.9 million of capitalized internal use software and website development costs, net of accumulated amortization. We evaluate the useful lives of these assets and test for impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets. There were no material impairments to internal-use software during the years ended December 31, 2015, 2016 and 2017.

Business Combinations, Goodwill and Intangible Assets, net

We allocate the fair value of purchase consideration to the tangible assets acquired, liabilities assumed and intangible assets acquired based on their estimated fair values. The excess of the fair value of purchase consideration over the fair values of these identifiable assets and liabilities is recorded as goodwill. Such valuations require management to make significant estimates and assumptions, especially with respect to intangible assets. Significant estimates in valuing certain intangible assets include, but are not limited to, future expected cash flows from acquired users, acquired technology, and trade names from a market participant perspective, useful lives and discount rates. Management's estimates of fair value are based upon assumptions believed to be reasonable, but which are inherently uncertain and unpredictable and, as a result, actual results may differ from estimates. During the measurement period, which is one year from the acquisition date, we may record adjustments to the assets acquired and liabilities assumed, with the corresponding offset to goodwill. Upon the conclusion of the measurement period, any subsequent adjustments are recorded to earnings.

We do not amortize goodwill and intangible assets with indefinite useful lives, rather such assets are required to be tested for impairment at least annually or sooner whenever events or changes in circumstances indicate that the assets may be impaired. We perform our annual goodwill and intangible asset impairment tests in the fourth quarter of each year. In the year ended December 31, 2017, we recognized \$132.0 million impairment of goodwill, primarily related to the sale of Ticketfly.

Acquired finite-lived intangible assets are amortized over the estimated useful lives of the assets, which range from three to eleven years. Acquired finite-lived intangible assets consist primarily of patents, customer relationships,

developed technology and trade names resulting from business combinations. We evaluate the recoverability of our intangible assets for potential impairment whenever events or circumstances indicate that the carrying amount of such assets may not be recoverable. Recoverability of these assets is measured by a comparison of the carrying amounts to the future undiscounted cash flows the assets are expected to generate. If such review indicates that the carrying amount of intangible assets is not recoverable, the carrying amount of such assets is reduced to the fair value.

In addition to the recoverability assessment, we routinely review the remaining estimated useful lives of finite-lived intangible assets. If we reduce the estimated useful life assumption for any asset, the remaining unamortized balance would be amortized over the revised estimated useful life. We record the amortization of intangible assets to the financial statement line item in our consolidated statement of operations that the asset directly relates to. To the extent that purchased intangibles are used in revenue generating activities, we record the amortization of these intangible assets to cost of revenue.

Stock-Based Compensation—Restricted Stock Units and Stock Options

Stock-based awards granted to employees, including grants of restricted stock units ("RSUs") and stock options, are recognized as expense in our statements of operations based on their grant date fair value. We recognize stock-based compensation expense on a straight-line basis over the service period of the award, which is generally three to four years. We estimate the fair value of RSUs at our stock price on the grant date. We generally estimate the grant date fair value of stock options using the Black-Scholes option-pricing model. The inputs to the Black-Scholes option-pricing model are our stock price on the date of grant, the expected stock price volatility over the expected term of the award, which is based on projected employee stock option exercise behaviors, the risk-free interest rate for the expected term of the award and expected dividends.

Stock-based compensation expense is recorded in the statement of operations for only those stock-based awards that will vest. In the first quarter of 2017 we adopted new accounting guidance from the Financial Accounting Standards Board ("FASB") on stock compensation, or ASU 2016-09, as described in "Recently Adopted Accounting Standards" below and have elected to account for forfeitures as they occur, rather than estimating expected forfeitures. In addition, we recognize all income tax effects of awards in the income statement when the awards vest or are settled as required by ASU 2016-09.

Stock-Based Compensation—Employee Stock Purchase Plan

We estimate the fair value of shares to be issued under the Employee Stock Purchase Plan ("ESPP") on the first day of the offering period using the Black-Scholes valuation model. The inputs to the Black-Scholes option-pricing model are our stock price on the date of grant, the expected stock price volatility over the expected term of the award, which is based on projected employee stock option exercise behaviors, the risk-free interest rate for the expected term of the award and expected dividends. Stock-based compensation expense related to the ESPP is recognized on a straight-line basis over the offering period, net of estimated forfeitures.

Stock-Based Compensation—PSUs

We implemented a performance stock unit program in April 2016 for stock-settled performance-based RSUs to certain key executives.

We have determined the grant-date fair value of the PSUs granted in 2016 using a Monte Carlo simulation performed by a third-party valuation firm. The Monte Carlo simulation model utilizes multiple input variables to estimate the probability that market conditions will be achieved. The variables used in these models are reviewed on an annual basis and adjusted, as needed. We recognize stock-based compensation for the PSUs over the requisite service period, which is approximately four years, using the accelerated attribution method.

Cost of Revenue—Content Acquisition Costs

Cost of revenue—content acquisition costs principally consist of licensing fees paid for streaming music or other content to our listeners. Content acquisition costs are currently calculated using negotiated rates documented in direct license agreements with major and independent record labels, music publishers and performing rights organizations ("PROs") for our ad-supported service, Pandora Plus and Pandora Premium. Depending on the applicable service, these license agreements generally require us to pay either a per-performance fee based on the number of sound recordings we transmit, a percentage of revenue associated with the service, or a per-subscriber minimum amount. Certain of these license agreements require minimum guarantee payments, some of which are paid in advance.

If we have not entered into a direct license agreement with the copyright owner of a particular sound recording that is streamed on our services, we stream that sound recording pursuant to a statutory license and pay the applicable rates set by the Copyright Royalty Board ("CRB") for the period from January 1, 2016 through December 31, 2020. Content acquisition costs for the streaming of musical works on our ad-supported service are calculated such that each copyright holder receives its usage-based and ownership-based share of a royalty pool equal to 20% of the content acquisition costs paid by us for sound recordings on our ad-supported service. Content acquisition costs for the streaming of musical works on our subscription services are equal to the rates determined in accordance with the statutory license set forth in 17 U.S.C. §115 ("Section 115").

Certain of our direct license agreements are also subject to minimum guarantee payments, some of which are paid in advance and amortized over the minimum guarantee period. For certain content acquisition arrangements, we accrue for estimated content acquisition costs based on the available facts and circumstances and adjust these estimates as more information becomes available. Several of our direct license agreements also include "most favored nations" provisions, which, if triggered, could cause our payments under those agreements to escalate. We recognize an accrual when it is probable that we will make additional payments under these provisions. The expense related to these accruals is recognized in cost of revenue—content acquisition costs.

Cost of Revenue—Ticketing Service

Prior to the sale of Ticketfly, cost of revenue—ticketing service consisted primarily of ticketing revenue share costs, credit card fees and intangible amortization expense. The majority of the cost was related to revenue share costs, which consist of royalties paid to clients for their share of convenience and order processing fees. Payments to clients were recorded as an expense to the extent that the fair value of the identifiable benefit received in the exchange exceeded the amount of the payment to the client. Intangible amortization expense was related to amortization of developed technology.

Cost of Revenue-Other

Cost of revenue—other consists primarily of ad and music serving costs, employee-related costs, facilities and equipment costs, other costs of ad sales and amortization expense related to acquired intangible assets and internal-use software. Ad and music serving costs consist of content streaming, maintaining our streaming radio and ondemand music services and creating and serving advertisements through third-party ad servers. We make payments to third-party ad servers for the period the advertising impressions are delivered or click-through actions occur, and accordingly, we record this as a cost of revenue in the related period. Employee-related costs include salaries and benefits associated with supporting music and ad-serving functions. Other costs of ad sales include costs related to music events that are sold as part of certain of our advertising arrangements.

Product Development

Product development consists primarily of employee-related costs, including salaries and benefits for employees in software engineering, music analysis and product management departments, information technology and amortization expense related to acquired intangible assets. We incur product development expenses primarily for improvements to our website and the Pandora app, development of new advertising products and development and enhancement of our personalized station generating system. We have generally expensed product development as incurred.

Certain website development and internal use software development costs are capitalized when specific criteria are met. In such cases, the capitalized amounts are amortized over the useful life of the related application once the application is placed in service.

Sales and Marketing

Sales and marketing consists primarily of employee-related costs, including salaries, commissions and benefits for employees in sales, sales support, marketing, advertising and industry relations and artist marketing departments and facilities and equipment costs. In addition, sales and marketing expenses include commissions on subscription purchases through mobile app stores ("subscription commissions"), external sales and marketing expenses such as brand marketing, advertising, customer acquisition, direct response and search engine marketing costs, public relations expenses, costs related to music events, agency platform and media measurement expenses, infrastructure costs and amortization expense related to acquired intangible assets.

We expense the costs of producing advertisements as they are incurred and expense the cost of communicating advertisements at the time the advertisement airs or the event occurs, in each case as sales and marketing expense within the



accompanying consolidated statements of operations. During the years ended December 31, 2015, 2016 and 2017, we recorded advertising expenses of \$35.1 million, \$45.7 million and \$48.5 million, respectively.

General and Administrative

General and administrative consists primarily of employee-related costs, including salaries, benefits and severance expense for finance, accounting, legal, internal information technology and other administrative personnel and facilities and equipment costs. In addition, general and administrative expenses include legal expenses, professional services costs for accounting and other services, credit card fees and sales and other tax expense.

Provision for (Benefit from) Income Taxes

Our provision for (benefit from) income taxes is computed using the asset and liability method, under which deferred tax assets and liabilities are determined based on the difference between the financial statement and tax bases of assets and liabilities using enacted statutory income tax rates in effect for the year in which the differences are expected to affect taxable income. Valuation allowances are established when necessary to reduce net deferred tax assets to the amount expected to be realized.

We recognize a tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. The tax benefits recognized in the financial statements from such positions are then measured based on the largest benefit that has a greater than 50% likelihood of being realized upon settlement. We will recognize interest and penalties related to unrecognized tax benefits in the provision for (benefit from) income taxes in the accompanying statement of operations.

We calculate the current and deferred income tax provision based on estimates and assumptions that could differ from the actual results reflected in income tax returns filed in subsequent years. Adjustments based on filed income tax returns are recorded when identified. The amount of income taxes paid is subject to examination by U.S. federal, state and international tax authorities. The estimate of the potential outcome of any uncertain tax issue is subject to management's assessment of relevant risks, facts and circumstances existing at that time. To the extent that the assessment of such tax positions change, the change in estimate is recorded in the period in which the determination is made.

Net Loss per Common Share

Basic net loss per common share is computed by dividing the net loss available to common stockholders by the weighted-average number of shares of common stock outstanding during the period. Diluted net loss per common share is computed by giving effect to all potential shares of common stock, including stock options, restricted stock units, market stock units, performance-based RSUs, potential ESPP shares and instruments convertible into common stock, to the extent dilutive. Basic and diluted net loss per common share were the same for each period presented as the inclusion of all potential common shares outstanding would have been anti-dilutive.

Recently Issued Accounting Standards

In May 2014, the FASB issued Accounting Standards Update No. 2014-09, Revenue from Contracts with Customers (Topic 606) ("ASU 2014-09"), which amends the existing accounting standards for revenue recognition. ASU 2014-09 outlines a single comprehensive model for entities to use in accounting for revenue. Under the guidance, revenue is recognized when a company transfers promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. The standard is effective for public entities with annual and interim reporting periods beginning after December 15, 2017. Entities have the option of using either a full retrospective or a modified retrospective approach to adopt the guidance. We expect to adopt ASU 2014-09 as of January 1, 2018 using the modified retrospective method. This method requires to retrospectively adjust open contracts as of December 31, 2017 to apply the guidance under Topic 606 and to make a cumulative adjustment for such impact to opening retained earnings in the year of adoption. It also requires additional disclosures in the first year of adoption on the impact of adoption primarily affects certain aspects of our advertising revenue. We have determined that the cumulative adjustment to retained earnings will not result in a material impact to our consolidated financial statements.

In February 2016, the FASB issued Accounting Standards Update No. 2016-02, Leases (Topic 842) ("ASU 2016-02"). ASU 2016-02 requires lessees to put most leases on their balance sheets and recognize expenses on their income statements and also eliminates the real estate-specific provisions for all entities. The guidance is effective for fiscal years beginning after



December 15, 2018, including interim periods within those fiscal years. We have completed our initial assessment and expect to adopt ASU 2016-02 as of January 1, 2019 using the modified retrospective method. We are in the process of evaluating the impact of ASU 2016-02 on our consolidated financial statements and expect there to be a material impact related to the recognition of new right-of-use assets and lease liabilities on our balance sheet for operating leases.

In June 2016, the FASB issued Accounting Standards Update No. 2016-13, Credit Losses—Measurement of Credit Losses on Financial Instruments ("ASU 2016-13"). ASU 2016-13 will replace today's incurred loss approach with an expected loss model for instruments measured at amortized cost and require entities to record allowances for available-for-sale debt securities rather than reduce the carrying amount. The guidance is effective for fiscal years beginning after December 15, 2019, and interim periods within that fiscal year, although early adoption is permitted. We are currently evaluating the impact that this standard update will have on our consolidated financial statements.

In November 2016, the FASB issued Accounting Standards Update No. 2016-18, Restricted Cash ("ASU 2016-18"). ASU 2016-18 requires amounts generally described as restricted cash and restricted cash and restricted cash equivalents be included with cash and cash equivalents when reconciling the total beginning and ending amounts for the periods shown on the statement of cash flows. The guidance is effective retrospectively for fiscal years beginning after December 15, 2017, and interim periods within that fiscal year. We will adopt ASU 2016-18 in the first quarter of 2018. We do not expect the adoption of ASU 2016-09 will have a material impact on our financial statements.

In May 2017, the FASB issued Accounting Standards Update No. 2017-09, Compensation—Stock Compensation (Topic 718), Scope of Modification Accounting ("ASU 2017-09"). ASU 2017-09 clarifies when changes to the terms or conditions of a share-based payment award must be accounted for as modifications. The guidance is effective prospectively for fiscal years beginning after December 15, 2017, including interim periods within those fiscal years, and early adoption is permitted. We do not expect the adoption of ASU 2017-09 will have a material impact on our financial statements.

Recently Adopted Accounting Standards

In March 2016, the FASB issued Accounting Standards Update No. 2016-09, Compensation—Stock Compensation (Topic 718) ("ASU 2016-09"). ASU 2016-09 requires all income tax effects of awards to be recognized in the income statement when the awards vest or are settled. Additionally, it allows an employer to repurchase more of an employee's shares for tax withholding purposes without triggering liability accounting and to make a policy election to account for forfeitures as they occur. We adopted this guidance in the first quarter of 2017 using the modified retrospective transition method. Upon adoption, we recognized the previously unrecognized excess tax benefits as of January 1, 2017 through retained earnings. The previously unrecognized excess tax benefits were recorded as a deferred tax asset, which was fully offsets by a valuation allowance. As a result, the net impact resulted in no effect on net deferred tax assets or our accumulated deficit as of January 1, 2017. Without the valuation allowance, the Company's net deferred tax assets would have increased by approximately \$142.0 million. Additionally, we elected to account for forfeitures as they occur, rather than estimating expected forfeitures. The net cumulative effect of this change was an increase to additional paid in capital as of January 1, 2017 of \$1.2 million.

In January 2017, the FASB issued Accounting Standards Update No. 2017-04, Intangibles—Goodwill and Other (Topic 350), Simplifying the Test for Goodwill Impairment ("ASU 2017-04"). ASU 2017-04 eliminated the requirement to calculate the implied fair value of goodwill, which is step two of the previous goodwill impairment test, to measure a goodwill impairment charge. By eliminating step two of the goodwill impairment test, entities will record an impairment charge based on the excess of a reporting unit's carrying amount over its fair value. The guidance is effective for calendar-year public business entities that meet the definition of an SEC filer for fiscal years beginning after December 15, 2019, although early adoption is permitted for annual and interim goodwill impairment testing dates following January 1, 2017. We have elected to early adopt this guidance beginning in the second quarter of 2017 using the prospective method, as we believe the elimination of step two of the goodwill impairment test will make testing for goodwill impairment less costly.

3. Composition of Certain Financial Statement Captions

Cash, Cash Equivalents and Investments

Cash, cash equivalents and investments consisted of the following:

	As of Dec	ember 3	1,
	2016		2017
	 (in tho	usands)	
Cash and cash equivalents			
Cash	\$ 144,192	\$	146,294
Money market funds	55,752		353,303
Total cash and cash equivalents	\$ 199,944	\$	499,597
Short-term investments			
Corporate debt securities	37,109		1,250
Total short-term investments	\$ 37,109	\$	1,250
Long-term investments			
Corporate debt securities	\$ 6,252	\$	—
Total long-term investments	\$ 6,252	\$	_
Cash, cash equivalents and investments	\$ 243,305	\$	500,847

Our short-term investments have maturities of twelve months or less and are classified as available-for-sale. Our long-term investments have maturities of greater than twelve months and are classified as available-for-sale.

The following table summarizes our available-for-sale securities' adjusted cost, gross unrealized gains, gross unrealized losses and fair value by significant investment category as of December 31, 2016. As of December 31, 2017, the adjusted cost and fair value by significant investment category are the same.

	As of December 31, 2016								
	 Adjusted Cost		Unrealized Gains		Unrealized Losses		Fair Value		
			(in tho	usands)					
Cash equivalents and marketable securities									
Money market funds	\$ 55,752	\$	_	\$	_	\$	55,752		
Corporate debt securities	43,413		3		(55)		43,361		
Total cash equivalents and marketable securities	\$ 99,165	\$	3	\$	(55)	\$	99,113		

The following tables present available-for-sale securities by contractual maturity date as of December 31, 2016 and 2017:

		6,251 6 \$ 99,165 \$ 99				
	Α	Čost		air Value		
		(in the	usands)			
Due in one year or less	\$	92,914	\$	92,861		
Due after one year through three years		6,251		6,252		
Total	\$	99,165	\$	99,113		
		As of Decer	nber 31, 201	7		
	A	Adjusted Cost	F	air Value		
		(in the	ousands)			
Due in one year or less	\$	354,553	\$	354,553		
Total	\$	354,553	\$	354,553		

The following table summarizes our available-for-sale securities' fair value and gross unrealized losses aggregated by investment category and length of time that the individual securities have been in a continuous unrealized loss position as of December 31, 2016. We had no securities that had been in a continuous unrealized loss position for greater than 12 months as of December 31, 2017.

					As of Dece	ember 31,	2016			
	 Twelve Months or Less More than Twelve Months					Total				
	 Fair Value	Gr	oss Unrealized Losses		Fair Value		Unrealized Losses	 Fair Value	Gro	ss Unrealized Losses
					(in th	ousands)				
Corporate debt securities	\$ 34,257	\$	(52)	\$	4,099	\$	(3)	\$ 38,356	\$	(55)
Total	\$ 34,257	\$	(52)	\$	4,099	\$	(3)	\$ 38,356	\$	(55)

Our investment policy requires investments to be investment grade, primarily rated "A1" by Standard & Poor's or "P1" by Moody's or better for short-term investments and rated "A" by Standard & Poor's or "A2" by Moody's or better for long-term investments, with the objective of minimizing the potential risk of principal loss. In addition, the investment policy limits the amount of credit exposure to any one issuer.

As of December 31, 2017, we owned no securities that were in an unrealized loss position. When evaluating investments for other-than-temporary impairment, we review factors such as the length of time and extent to which fair value has been below cost basis, the financial condition of the issuer and any changes thereto, and our intent to sell, or whether it is more likely than not we will be required to sell, the investment before recovery of the investment's amortized cost basis. During the year ended December 31, 2017, we did not recognize any impairment charges. There were no sales of available-for-sale securities during theyear ended December 31, 2017.

Accounts Receivable, net

Accounts receivable, net consisted of the following as of December 31, 2016 and 2017:

	 As of December 31,				
	 2016	2017			
	 (in thousands)				
Accounts receivable, net					
Accounts receivable	\$ 312,900 \$	341,781			
Allowance for doubtful accounts	(3,633)	(5,352)			
Total accounts receivable, net	\$ 309,267 \$	336,429			

The following table summarizes our beginning allowance for doubtful accounts balance for each period, additions, write-offs net of recoveries and the balance at the end of each period for the years December 31, 2015, 2016 and 2017:

Allowance for Doubtful Accounts	t Beginning of eriod	Additions	Write-offs, Net of Recoveries	Balance a	t End of Period
		(in thousands)		
For the year ended December 31, 2015	\$ 1,218	2,085	(1,138)	\$	2,165
For the year ended December 31, 2016	\$ 2,165	3,508	(2,040)	\$	3,633
For the year ended December 31, 2017	\$ 3,633	5,274	(3,555)	\$	5,352

Prepaid and Other Current Assets

Prepaid and other current assets consisted of the following as of December 31, 2016 and 2017:

			As of De	ember 31,		
		2016	2017			
		(in thousands)				
Prepaid expenses and other current assets						
Other current assets	\$		13,858	\$		6,636
Prepaid expenses			13,533			12,584
Ticketing contract advances-short term, net			5,800			_
Total prepaid expenses and other current assets	\$		33,191	\$		19,220
Total prepaid expenses and other current assets	3		55,191	¢		19,220

As of December 31, 2016 and 2017, other current assets consisted primarily of \$9.1 million and \$3.8 million in receivables for the reimbursement of costs of leasehold improvements in connection with our operating leases.

Other Long-Term Assets

Other long-term assets consisted of the following as of December 31, 2016 and 2017:

 As of December 31,				
2016		2017		
 (in thousands)				
\$ 9,090	\$	8,810		
7,048		2,483		
 15,395		_		
\$ 31,533	\$	11,293		
\$ \$	2016 (in tho \$ 9,090 7,048 15,395	(in thousands) \$ 9,090 \$ 7,048 15,395		

Property and Equipment, net

Property and equipment, net consisted of the following as of December 31, 2016 and 2017:

	 As of Deco	ember 31,	
	2016		2017
	(in thou	isands)	
Property and equipment, net			
Servers, computers and other related equipment	\$ 85,541	\$	91,803
Software developed for internal use	34,983		74,260
Leasehold improvements	63,519		64,968
Construction in progress	20,393		8,460
Office furniture and equipment	9,037		8,040
Total property and equipment	\$ 213,473	\$	247,531
Less accumulated depreciation and amortization	(89,385)		(130,789)
Total property and equipment, net	\$ 124,088	\$	116,742

Depreciation expenses totaled \$20.4 million, \$34.2 million and \$46.2 million for the years ended December 31, 2015, 2016 and 2017, respectively. There were no material write-offs during the years ended December 31, 2015, 2016 and 2017.

Software developed for internal use generally has an expected useful life offhree to five years from the date placed in service. As ofDecember 31, 2016 and 2017 the net carrying amount was \$25.7 million and \$50.9 million, including accumulated amortization of \$9.3 million and \$23.4 million. Amortization expense for the years ended December 31, 2015, 2016 and 2017 was \$2.2 million, \$5.3 million and \$14.1 million, respectively.

Other Current Liabilities

Other current liabilities consisted of the following as of December 31, 2016 and 2017:

	As of December 31,					
	 2016		2017			
	 (in tho	usands)				
Other current liabilities						
Ticketing amounts due to clients	\$ 20,666	\$		_		
Other	327			—		
Total other current liabilities	\$ 20,993	\$		_		

Ticketing amounts due to clients consists of the face value of tickets sold and the revenue share costs related to tickets sold on the Ticketfly ticketing platform that are owed to clients. The face value of tickets sold on the Ticketfly ticketing platform is collected by Ticketfly and remitted to clients. Revenue share costs owed to clients related to tickets sold on the Ticketfly ticketing platform consist of fees paid to clients for their share of convenience and order processing fees. We completed the sale of Ticketfly on September 1, 2017, which resulted in the disposition of these liabilities. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

Other Long-Term Liabilities

Other long-term liabilities consisted of the following as of December 31, 2016 and 2017:

	 As of Dec	ember 31,	
	2016		2017
	 (in tho	usands)	
Other long-term liabilities			
Long-term deferred rent	\$ 24,245	\$	19,822
Other	9,942		3,678
Total other long-term liabilities	\$ 34,187	\$	23,500

For operating leases that include escalation clauses over the term of the lease, tenant improvement reimbursements and rent abatement periods, we recognize rent expense on a straight-line basis over the lease term including expected renewal periods. The difference between rent expense and rent payments is recorded as deferred rent.

4. Fair Value

We record cash equivalents and investments at fair value. Fair value is an exit price, representing the amount that would be received from the sale of an asset or paid to transfer a liability in an orderly transaction between market participants. As such, fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or liability. Fair value measurements are required to be disclosed by level within the following fair value hierarchy:

Level 1 — Inputs are unadjusted, quoted prices in active markets for identical assets or liabilities at the measurement date.

Level 2 — Inputs (other than quoted prices included in Level 1) are either directly or indirectly observable for the asset or liability through correlation with market data at the measurement date and for the duration of the instrument's anticipated life.

Level 3 — Inputs lack observable market data to corroborate management's estimate of what market participants would use in pricing the asset or liability at the measurement date. Consideration is given to the risk inherent in the valuation technique and the risk inherent in the inputs to the model.

When determining fair value, whenever possible we use observable market data and rely on unobservable inputs only when observable market data is not available.

The following fair value hierarchy tables categorize information regarding our financial assets and liabilities measured at fair value on a recurring basis aDecember 31, 2016 and 2017:

			As of December 31, 2016					
]	Fair Value Measurement Using					
	A	Quoted Prices in Active Markets for Identical truments (Level 1)	Significant Other Observable Inputs (Level 2)	Total				
			(in thousands)					
Assets								
Money market funds	\$	55,752 \$		\$ 55,752				
Corporate debt securities		—	43,361	43,361				
Total assets measured at fair value	\$	55,752 \$	43,361	\$ 99,113				
		As of December 31, 2017						
		I	Fair Value Measurement Using					
	Ā	uoted Prices in Active Markets for Identical ruments (Level 1)	Significant Other Observable Inputs (Level 2)	Total				
			(in thousands)					
Assets								
Money market funds	\$	353,303 \$	_	\$ 353,303				
Corporate debt securities		—	1,250	1,250				

Our cash equivalents and short-term investments are classified as Level 2 within the fair value hierarchy because they are valued using professional pricing sources for identical or comparable instruments, rather than direct observations of quoted prices in active markets.

As of December 31, 2016 and 2017, we held no Level 3 assets or liabilities measured on a recurring basis. The fair value of our Convertible Promissory Note was calculated on a nonrecurring basis as of September 1, 2017 and is classified as a Level 3 measurement within the fair value hierarchy. Refer to <u>Note 8 "Convertible Promissory</u> <u>Note Receivable"</u> in the Notes to Consolidated Financial Statements for further details on the Convertible Promissory Note.

Refer to Note 9, "Debt Instruments", in the Notes to Consolidated Financial Statements for the carrying amount and estimated fair value of our convertible senior notes, which are not recorded at fair value as of December 31, 2017.

5. Dispositions

Ticketfly

On September 1, 2017, we completed the sale of Ticketfly, our ticketing service segment, to Eventbrite Inc. ("Eventbrite") for an aggregate unadjusted purchase price of \$200.0 million. The aggregate unadjusted purchase price consists of \$150.0 million in cash and a \$50.0 million Convertible Promissory Note, which were paid and issued at the closing of the

transaction. The Convertible Promissory Note was recorded at its fair value at the date of sale, which resulted in a discount o**\$**13.8 million. The aggregate purchase price was further reduced by \$4.8 million in costs to sell and \$7.5 million in working capital adjustments and certain indemnification provisions, for a net purchase price of\$174.0 million. Refer to <u>Note 8 "Convertible Promissory Note Receivable"</u> in the Notes to Consolidated Financial Statements for further details on the Convertible Promissory Note.

In the three months ended June 30, 2017, the assets and liabilities of Ticketfly were classified as held for sale, and we recognized a goodwill impairment charge of \$131.7 million. The impairment charge was based on the fair value of the net assets as implied by the estimated purchase price of \$184.5 million as of June 30, 2017. We consider the fair value of these net assets to be classified as Level 2 within the fair value hierarchy because Ticketfly is not a publicly traded company. Instead, the fair value was based on other observable inputs, such as the selling price, which represents an exit price. In the year ended December 31, 2017, we recognized a loss on sale of \$9.3 million in the general and administrative line item on our Consolidated Statements of Operations, which was based on an adjusted net purchase price of \$174.0 million as of September 1, 2017 and includes \$1.1 million in transfers of cumulative translation adjustments.

Net cash proceeds from the sale of Ticketfly were \$124.1 million and consisted of the cash purchase price of \$150.0 million, less cash held for sale of \$22.2 million and cash purchase price adjustments of \$3.7 million.

Prior to the sale of Ticketfly, we operated in two reportable segments. Subsequent to the sale of Ticketfly, we operate inone reportable segment.

The revenues and expenses of Ticketfly are included in our Consolidated Statements of Operations for the years endedDecember 31, 2015, 2016 and 2017, from the date of acquisition of October 31, 2015 through the disposition date of September 1, 2017. The following table provides Ticketfly's loss before benefit from (provision for) income taxes for the years ended December 31, 2015, 2016 and 2017:

	 Year ended December 31,					
	 2015 2016 20				2017	
		(in t	housands)			
Loss before benefit from (provision for) income taxes	\$ 7,223	\$	37,904	\$	149,803	

The sale of Ticketfly did not represent a strategic shift in our business, and therefore we have not classified the operations of Ticketfly as discontinued operations in our Consolidated Statements of Operations.

KXMZ

On August 31, 2017, we completed the sale of KXMZ, an FM radio station based in Rapid City, South Dakota. Net cash proceeds from the sale of KXMZ wer\$0.2 million. The sale did not result in a material impact to our consolidated financial statements.

Disposal of Assets and Liabilities

As of the disposal date, the total current assets of Ticketfly and KXMZ were \$37.8 million, long-term assets, including goodwill and intangibles, were \$187.9 million and total liabilities were \$43.4 million.

6. Business Combinations

Year Ended December 31, 2017

We had no business combinations during the year ended December 31, 2017.

Year Ended December 31, 2016

During the year ended December 31, 2016, we completed a business combination that was not material to our consolidated financial statements. We had no material business combinations during the period.

We included the financial results of the acquired business in our consolidated financial statements from the respective date of acquisition through the disposition on September 1, 2017. Pro forma results of operations related to this acquisition



during the year ended December 31, 2016 were not presented because they were not material to our consolidated statements of operations.

The fair value of assets acquired and liabilities assumed from our acquisition were based on a preliminary valuation.

Year Ended December 31, 2015

Ticketfly

On October 31, 2015, we completed the acquisition of Ticketfly, a leading live events technology company that provides ticketing and marketing software and services for venues and event promoters across North America, for an aggregate purchase price of \$335.3 million of common stock and cash, including11,193,847 shares of the Company's common stock and approximately \$191.5 million in cash paid by the Company. In addition to the purchase price, unvested options and unvested RSUs of Ticketfly held by Ticketfly employees were converted into unvested options to acquire our common stock and unvested RSUs in our common stock. On September 1, 2017, we completed the sale of Ticketfly. Refer to Note 5 "Dispositions" in the Notes to Consolidated Financial Statements for further details on the disposition.

Upon acquisition, Ticketfly became a wholly owned subsidiary of Pandora. The acquisition was accounted for as a business combination, and the financial results of Ticketfly were included in our consolidated financial statements from the date of acquisition through the date of disposition.

Rdio, Inc. ("Rdio")

On December 23, 2015, we completed the acquisition of technology and intellectual property from Rdio for\$77.5 million, which includes \$2.5 million in additional purchase consideration transferred prior to the closing of the acquisition. Goodwill generated from the assets acquired was primarily attributable to synergies that allowed us to broaden our subscription business and roll out our new subscription services, Pandora Plus and Pandora Premium. Pandora Plus launched in September 2016 and Pandora Premium launched in April 2017. We have accounted for this acquisition as a business combination, and the financial results of Rdio are included in our consolidated financial statements from the date of acquisition. As a result of the sale of assets, Rdio discontinued its service as of December 22, 2015.

Other acquisitions

During the year ended December 31, 2015, we completed the acquisitions of Next Big Sound ("NBS") and KXMZ-FM ("KXMZ"). These acquisitions were not material to our consolidated financial statements, either individually or in the aggregate. We completed the sale of KXMZ on August 31, 2017. Refer to Note 5 "Dispositions" in the Notes to Consolidated Financial Statements for further details on the disposition.

We have included the financial results of Ticketfly, Rdio, NBS and KXMZ in our consolidated financial statements from their respective dates of acquisition and through the respective disposition dates of Ticketfly and KXMZ. Pro forma results of operations related to our acquisitions, other than Ticketfly, during the year ended December 31, 2015 have not been presented because they are not material to our consolidated statements of operations, either individually or in the aggregate.

The following table summarizes the allocation of estimated fair values of the net assets acquired during the year endedDecember 31, 2015, including the related estimated useful lives, where applicable. Intangible assets acquired from Ticketfly and KXMZ were disposed of during the year ended December 31, 2017.



		Ticketfly			Rdio			Other			
	Estimated fair value				imated fair value	Estimated useful life in years	Estimated fair value		Estimated useful life in years(1)		
			(ir	thous:	ands, except fo	or estimated useful	life)				
Intangible assets:											
Customer relationships-clients	\$	37,300	8	\$	_		\$	_			
Developed technology		28,100	5		26,400	2-5		1,550	4		
Tradename		10,400	8		1,000	3		320	2		
Customer relationships—users		1,000	2		_			940	2		
FCC license-broadcast radio		—			_			193			
Tangible assets acquired, net		27,640			1,969			(490)			
Deferred tax liabilities		(1,738)			_			(49)			
Net assets acquired	\$	102,702		\$	29,369		\$	2,464			
Goodwill		232,641			48,131			23,103			
Total fair value consideration	\$	335,343		\$	77,500		\$	25,567			

(1) Estimated useful lives reflect the useful lives as of the date of acquisition, and does not take into consideration the disposition of intangible assets

Goodwill generated from the Ticketfly acquisition was primarily attributable to expected synergies from growth and strategic advantages in the ticketing industry. Goodwill generated from Rdio is primarily attributable to expected synergies from future growth and strategic advantages in the online streaming music industry. Goodwill generated from all other business acquisitions during the year ended December 31, 2015 is primarily attributed to expected synergies from future growth and, also for NBS, the potential to expand our Artist Marketing Platform. Goodwill generated during the period related to Ticketfly and NBS is not deductible for tax purposes and goodwill generated during the period related to Rdio and KXMZ is deductible for tax purposes.

7. Goodwill and Intangible Assets

During the year ended December 31,2017, we completed the sales of both Ticketfly and KXMZ. In the three months ended June 30, 2017, we recognized a goodwill impairment of \$131.7 million related to the Ticketfly sale. The impairment charge was based on the fair value of Ticketfly's net assets as implied by the estimated purchase price of \$184.5 million as of June 30, 2017. As a result of the KXMZ agreement, we recognized a goodwill impairment of \$0.3 million in the three months ended June 30, 2017, which was based on the fair value of these net assets as implied by the estimated purchase price.

The changes in the carrying amount of goodwill for the years endedDecember 31, 2016 and 2017, are as follows:

	(Goodwill
	(in	thousands)
Balance as of December 31, 2015	\$	303,875
Goodwill resulting from business combination and purchase price adjustments		2,816
Balance as of December 31, 2016	\$	306,691
Goodwill impairment		(131,997)
Goodwill related to disposed assets		(103,474)
Effect of currency translation adjustment		
		23
Balance as of December 31, 2017	\$	71,243

The following summarizes information regarding the gross carrying amounts and accumulated amortization of intangibles:

	As of December 31, 2016						As of December 31, 2017							
	Gross Carrying Amount			Accumulated Amortization	N	et Carrying Value	Gr	oss Carrying Amount		Accumulated Amortization	Int	Disposal of angible Assets	Ν	et Carrying Value
							(iı	n thousands)						
Finite-lived intangible assets														
Patents	\$	8,030	\$	(2,556)	\$	5,474	\$	8,030	\$	(3,289)	\$	—	\$	4,741
Developed technology		56,162		(13,599)		42,563		56,162		(22,586)		(19,235)		14,341
Customer relationships-clients		37,399		(5,487)		31,912		37,399		(7,449)		(29,950)		—
Customer relationships—users		1,940		(1,288)		652		1,940		(1,732)		(208)		—
Trade names		11,735		(2,104)		9,631		11,735		(3,062)		(8,346)		327
Total finite-lived intangible assets	\$	115,266	\$	(25,034)	\$	90,232	\$	115,266	\$	(38,118)	\$	(57,739)	\$	19,409
Indefinite-lived intangible assets														
FCC license-broadcast radio	\$	193	\$	—	\$	193	\$	193	\$	—	\$	(193)	\$	_
Total intangible assets	\$	115,459	\$	(25,034)	\$	90,425	\$	115,459	\$	(38,118)	\$	(57,932)	\$	19,409

Amortization expense of intangible assets was \$3.4 million, \$20.5 million and \$13.1 million for the years ended December 31, 2015, 2016 and 2017, respectively.

The following is a schedule of future amortization expense related to finite-lived intangible assets as ofDecember 31, 2017.

	As of December 31, 2017
	(in thousands)
2018	\$ 6,067
2019	5,546
2020	5,251
2021	727
2022	727
Thereafter	1,091
Total future amortization expense	\$ 19,409

8. Convertible Promissory Note Receivable

On September 1, 2017, we completed the sale of Ticketfly, our ticketing service segment, to Eventbrite for an aggregate unadjusted purchase price of 200.0 million. The aggregate unadjusted purchase price consists of \$150.0 million in cash and a \$50.0 million Convertible Promissory Note, which were paid and issued at the closing of the transaction. The Convertible Promissory Note will be due five years from its issuance date (the "Convertible Promissory Note Maturity Date") and will accrue interest at a rate of 6.5% per annum, payable quarterly in cash or in-kind for the first year at the discretion of Eventbrite, and in cash thereafter. Prior to the Convertible Promissory Note Maturity Date, the Convertible Promissory Note is convertible at our option into shares of Eventbrite's common stock. The Convertible Promissory Note may be prepaid at any time.

The Convertible Promissory Note was recorded at its fair value of \$36.2 million as of the issuance date of September 1, 2017, which resulted in a discount of \$13.8 million. The note was further reduced by \$2.5 million in purchase price adjustments. As of December 31, 2017, the balance of the Convertible Promissory Note also included \$1.1 million in interest

receivable and \$0.7 million in accretion of the discount, for a total balance of \$35.5 million. We assess the collectability of the Convertible Promissory Note on a quarterly basis through a review of Eventbrite's financial results.

The fair value of the Convertible Promissory Note was based on a methodology that combines inputs based on comparable debt instruments and market-corroborated inputs with quantitative pricing models. At issuance, our Convertible Promissory Note was classified as Level 3 within the fair value hierarchy because the fair value was based on unobservable inputs in an inactive market. However, our Convertible Promissory Note will not be remeasured at each reporting date.

The discount on the Convertible Promissory Note is being amortized to interest income using the effective interest method over the period from the date of issuance through the Convertible Promissory Note Maturity Date. The following table outlines the effective interest rate, contractually stated interest income and amortization of the discount for the Convertible Promissory Note:

	Ye	ar ended
	Decem	ber 31, 2017
	(in thousands exc	ept for effective interest rate)
Effective interest rate		14.73 %
Contractually stated interest income	\$	1,082
Amortization of discount	\$	687

9. Debt Instruments

Long-term debt, net consisted of the following:

	 As of December 31,					
	2016	2017				
	(in thousands)					
1.75% convertible senior notes due 2020	\$ 345,000 \$	345,000				
Credit facility	90,000	_				
Unamortized discount and deferred issuance costs	(92,753)	(71,986)				
Long-term debt, net	\$ 342,247 \$	273,014				

Convertible Debt Offering

On December 9, 2015, we completed an unregistered Rule 144A offering for the issuance of \$345.0 million aggregate principal amount of our 1.75% Convertible Senior Notes due 2020 (the "Notes"). In connection with the issuance of the Notes, we entered into capped call transactions with the initial purchaser of the Notes and an additional financial institution (the "capped call transactions").

The net proceeds from the sale of the Notes were approximately\$336.5 million, after deducting the initial purchasers' fees and other estimated expenses. We used approximately \$43.2 million of the net proceeds to pay the cost of the capped call transactions.

The Notes are unsecured, senior obligations of Pandora, and interest is payable semi-annually at a rate of 1.75% per annum. The Notes will mature on December 1, 2020, unless earlier repurchased or redeemed by Pandora or converted in accordance with their terms prior to such date. Prior to July 1, 2020, the Notes are convertible at the option of holders only upon the occurrence of specified events or during certain periods as further described below; thereafter, until the second scheduled trading day prior to maturity, the Notes will be convertible at the option of holders at any time.

The conversion rate for the Notes is initially60.9050 shares of common stock per\$1,000 principal amount of the Notes, which is equivalent to an initial conversion price of approximately \$16.42 per share of our common stock, and is subject to adjustment in certain circumstances.

We will not have the right to redeem the Notes prior to December 5, 2018. We may redeem all or any portion of the Notes for cash at our option on or after December 5, 2018 if the last reported sale price of our common stock is at least 130% of



the conversion price then in effect for at least20 trading days, whether or not consecutive, during any 30 consecutive trading day period, including the last trading day of such period, ending on, and including, any of the five trading days immediately preceding the date on which we provide notice of redemption. Any optional redemption of the Notes will be at a redemption price equal to 100% of the principal amount of the Notes to be redeemed, plus accrued and unpaid interest to, but excluding, the redemption date. The maximum number of shares of common stock the Notes are convertible into is approximately 27.3 million, and is subject to adjustment under certain circumstances.

The Notes will be convertible at the option of holders only under the following circumstances:

- Prior to the close of business on the business day immediately preceding July 1, 2020, during any calendar quarter commencing after the calendar quarter ended on March 31, 2016 (and only during such calendar quarter), if the last reported sale price of our common stock for at least 20 trading days (whether or not consecutive), during a period of 30 consecutive trading days ending on the last trading day of the immediately preceding calendar quarter is greater than or equal to 130% of the conversion price on each applicable trading day;
- Prior to the close of business on the business day immediately preceding July 1, 2020, during thefive business day period after any ten consecutive trading day period (the "measurement period") in which the trading price per \$1,000 principal amount of Notes for each trading day of the measurement period was less than 98% of the product of the last reported sale price of our common stock and the conversion rate on each such trading day;
- Prior to the business day immediately preceding July 1, 2020, upon the occurrence of specified corporate events; or
- At any time on or after July 1, 2020 until the close of business on the second scheduled trading day immediately preceding the December 1, 2020 maturity date.

Upon the occurrence of a make-whole fundamental change or if we call all or any portion of the Notes for redemption prior to July 1, 2020, we will, in certain circumstances, increase the conversion rate by a number of additional shares for a holder that elects to convert its Notes in connection with such make-whole fundamental change or during the related redemption period.

The Notes were separated into debt and equity components and assigned a fair value. The value assigned to the debt component is the estimated fair value as of the issuance date of similar debt without the conversion feature. The difference between the cash proceeds and this estimated fair value represents the value which has been assigned to the equity component and recorded as a debt discount. The debt discount is being amortized using the effective interest method over the period from the date of issuance through the December 1, 2020 maturity date.

The initial debt component of the Notes was valued at \$233.5 million, based on the contractual cash flows discounted at an appropriate market rate for non-convertible debt at the date of issuance. The carrying value of the permanent equity component reported in additional paid-in-capital was initially valued at \$103.0 million, which is net of \$2.6 million of fees and expenses allocated to the equity component.

The following table outlines the effective interest rate, contractually stated interest expense and costs related to the amortization of the discount for the Notes:

	Year ended December 31,			
	 2016 2017			
	 (in thousands except for effective interest rate)			
Effective interest rate	10.18 %)	10.18 %	
Contractually stated interest expense	\$ 6,046	\$	6,029	
Amortization of discount	\$ 18,315	\$	20,153	

The capped call transactions are expected to reduce the potential dilution to our common stock and/or offset the cash payments we would be required to make in excess of the principal amount of the converted Notes in the event that the market price of our common stock, as measured under the terms of the capped call transaction, is greater than the strike price of the capped call transaction, with such reduction and/or offset subject to a cap based on the cap price of the capped call transactions. The strike price of the capped call transactions corresponds to the initial conversion price of the Notes and is subject to certain adjustments under the terms of the capped call transactions. The capped call transactions have an initial cap price of \$25.26 per share and are subject to certain adjustments under the terms of the capped call transactions have been included as a net reduction to additional paid-in capital within stockholders' equity.

The total estimated fair value of the Notes of December 31, 2017 was \$301.5 million. The fair value was determined using a methodology that combines direct market observations with quantitative pricing models to generate evaluated prices. We consider the fair value of the Notes to be a Level 2 measurement due to the limited trading activity of the Notes.

The closing price of our common stock was\$4.82 on December 31, 2017, which was less than the initial conversion price for the Notes of approximately\$16.42 per share. As such, the if-converted value of the Notes was less than the principal amount of \$345.0 million.

Credit Facility

On December 29, 2017, we entered into a credit facility for an aggregate commitment amount of \$200.0 million, with an option to increase the commitment amount by \$50.0 million and a maturity date of the earliest of December 29, 2022, 120 days prior to the Notes maturity date of December 1, 2020, provided that the Notes have not been converted into common stock prior to such date or 120 days prior to the Series A redeemable convertible preferred stock ("Series A") redemption date of September 22, 2022, provided that the Series A has not been converted into common stock prior to such date. Prior to December 29, 2017, we were party to a credit facility with an aggregate commitment amount of \$120.0 million, with an option to increase the commitment amount by \$20.0 million and a maturity date of September 12, 2018. This credit facility was terminated on December 29, 2017.

The credit facility interest rate on borrowings is either based on a base rate plus 0.50% - 1.00% or a LIBOR rate plus 1.50% - 2.00%, both of which are per annum rates based on outstanding borrowings. The unused line fee is 0.25% - 0.375% per annum based on outstanding borrowings. The available letters of credit under the credit facility is \$15.0 million, and the annual charge for outstanding letters of credit is 1.50% - 2.00% per annum in addition to a fronting fee of 0.125% per annum based on outstanding borrowings.

The amount of borrowings available under the credit facility at any time is limited by our monthly accounts receivable balance at such time and the amounts borrowed are collateralized by substantially all of our personal property, including such accounts receivable but excluding intellectual property. The credit facility contains customary events of default, conditions to borrowing and covenants, including restrictions on our ability to dispose of assets, make acquisitions, incur debt, incur liens and make distributions to stockholders. During the continuance of an event of default, the lenders may accelerate amounts outstanding, terminate the credit facility and foreclose on all collateral.

In September 2016, we borrowed \$90.0 million from the previous credit facility to enhance our working capital position. This amount was repaid in full in September 2017.

As of December 31, 2016 we had \$90.0 million outstanding borrowings, \$1.2 million in letters of credit outstanding and \$28.8 million of available borrowing capacity under the credit facility. As of December 31, 2017, we had no outstanding borrowings, \$1.2 million in letters of credit outstanding and \$194.1 million of available borrowing capacity under the credit facility.

Total debt issuance costs associated with the 2015 credit facility amendment and 2017 credit facility were \$0.4 million and \$1.1 million, which are recorded as deferred issuance costs that will be amortized as interest expense over the term of the credit facility. We did not incur any debt issuance costs in the year ended December 31, 2016. For the years ended December 31, 2015, 2016 and 2017, \$0.2 million, \$0.4 million and \$0.4 million of debt issuance costs, respectively, were amortized and included in interest expense.

10. Redeemable Convertible Preferred Stock

In June 2017, we entered into an agreement with Sirius XM Radio, Inc. ("Sirius XM") to sel480,000 shares of Series A for\$1,000 per share, with gross proceeds of \$480.0 million. The Series A shares were issued in two rounds: an initial closing of 172,500 shares for \$172.5 million that occurred on June 9, 2017 upon signing the agreement with Sirius XM, and an additional closing of 307,500 shares for \$307.5 million that occurred on September 22, 2017. In the year ended December 31, 2017, total proceeds from the initial and additional closing, net of preferred stock issuance costs of \$29.3 million, was \$450.7 million.

Liquidation Preference

In the event of any voluntary or involuntary liquidation, dissolution or winding up of the Company, holders of the Series A have the right to receive a liquidation preference entitling them to be paid out of our assets available for distribution to stockholders before any payment may be made to holders of our common stock in an amount equal to the sum of the liquidation preference, or \$1,000 per share plus all accrued dividends as of the date of the liquidation.

Voting Rights

Holders of the Series A shares are entitled to vote as a single class with the holders of our common stock and the holders of any other class or series of capital stock of the Company then entitled to vote with the common stock on all matters submitted to a vote of the holders of common stock. Each holder of the Series A is entitled to the number of votes equal to the largest number of whole shares of common stock into which all shares of Series A held of record by such holder could then be converted, subject to certain other restrictions.

Conversion Feature

Holders of the Series A shares have the option to convert their shares plus any accrued dividends into common stock. We have the right to settle the conversion in cash, common stock or a combination thereof. The conversion rate for the Series A is initially 95.2381 shares of common stock per each share of Series A, which is equivalent to an initial conversion price of approximately \$10.50 per share of our common stock, and is subject to adjustment in certain circumstances. Dividends on the Series A will accrue on a daily basis, whether or not declared, and will be payable on a quarterly basis at a rate of 6% per year. We have the option to pay dividends in cash when authorized by the Board and declared by the Company or accumulate dividends in lieu of paying cash. Dividends accumulated in lieu of paying cash will continue to accrue and accumulate at rate of 6% per year.

Redemption Feature

As of December 31, 2017, there is no ability to redeem or require redemption of the Series A. Under certain circumstances, we will have the right to redeem the Series A on or after September 22, 2020. The Series A holders will have the right to require us to redeem the Series A on or after September 22, 2022. Any optional redemption of the Series A will be at a redemption price equal to 100% of the liquidation preference, or \$1,000 per share, plus accrued and unpaid dividends to, but excluding, the redemption date. In the event of a future redemption, whether initiated by us or by the holders, we will have the option to redeem the Series A in cash, common stock or a combination thereof.

Fundamental Changes

If certain fundamental changes involving the Company occur, including change in control or liquidation, the Series A will be redeemed subject to certain adjustments, as determined by the date of the fundamental change. The change in control amount is the greater of the redemption value of 100% of the liquidation preference, plus all accrued dividends unpaid through September 22, 2022, assuming the shares would have remained outstanding through that date, or the price that common stockholders would receive if the Series A shares had been redeemed immediately prior to the announcement of the change in control.

Recognition

Since the redemption of the Series A is contingently or optionally redeemable and therefore not certain to occur, the Series A is not required to be classified as a liability under ASC 480, *Distinguishing Liabilities from Equity*. As the Series A is redeemable at the option of the holders and is redeemable in certain circumstances upon the occurrence of an event that is not solely within the Company's control, we have classified the Series A in the redeemable convertible preferred stock line item in our consolidated balance sheets. We did not identify any embedded features that would require bifurcation from the equity-like



host instrument. We have elected to recognize the Series A at the redemption value at each period end, and have recorded the issuance costs through retained earnings as a deemed preferred stock dividend. In addition, we have elected to account for the 6% dividend at the stated rate.

As of December 31, 2017, redeemable convertible preferred stock consisted of the following. We did not have any redeemable convertible preferred stock as of December 31, 2016.

	As of December 31, 2017
	 (in thousands)
Series A redeemable convertible preferred stock	\$ 480,000
Issuance costs	(29,318)
Accretion of issuance costs	29,318
Stock dividend payable to preferred stockholders	10,849
Redeemable convertible preferred stock	\$ 490,849

Contract Termination Fees

In May 2017, we entered into an agreement to sell redeemable convertible preferred stock to KKR Classic Investors L.P. ("KKR"). In June 2017, in conjunction with the Series A, we terminated the previous contractual commitment to sell redeemable convertible preferred stock to KKR, which resulted in a contract termination fee and related legal and professional fees, totaling \$23.0 million. This is included in the contract termination fees line item of our consolidated statements of operations for the year ended December 31, 2017.

11. Commitments and Contingencies

Leases

The following is a schedule of future minimum lease payments and future minimum sublease income under noncancelable operating leases as of December 31, 2017:

	As of December 31, 2017			
	 Future Minimum Lease Payments	Future Minimum Sublease Income		
	 (in thousands)			
018	\$ 24,380	\$ 2,053		
2019	23,313	2,065		
2020	20,422	2,103		
2021	13,073	1,783		
2022	9,622	_		
Thereafter	32,644	_		
Total	\$ 123,454	\$ 8,004		

We conduct our operations using leased office facilities in various locations. We lease office space under arrangements expiring through 2027. Rent expense for the years ended December 31, 2015, 2016 and 2017 was \$12.2 million, \$20.5 million and \$20.6 million, respectively.

For operating leases that include escalation clauses over the term of the lease, tenant improvement reimbursements and rent abatement periods, we recognize rent expense on a straight-line basis over the lease term including expected renewal periods. The difference between rent expense and rent payments is recorded as deferred rent in current and long-term liabilities. As of December 31, 2016 and 2017 deferred rent was \$27.6 million and \$23.8 million.

Minimum Guarantees and Other Provisions-Content Acquisition Costs

Certain of our content acquisition agreements contain minimum guarantees and require that we make upfront minimum guarantee payments. Refer to our discussion of these matters in <u>Item 1A—"Risk Factors"</u>. During the year ended December 31, 2017, we prepaid\$324.1 million in content acquisition costs related to minimum guarantees, which were offset by amortization of prepaid content acquisition costs of \$268.4 million. As of December 31, 2017, we have future minimum guarantee commitments of\$405.2 million, of which \$394.2 million will be paid in 2018 and the remainder will be paid thereafter. On a quarterly basis, we record the greater of the cumulative actual content acquisition costs incurred or the cumulative minimum guarantee based on forecasted usage for the minimum guarantee period. The minimum guarantee, based on forecasted usage, considers factors such as listening hours, revenue, subscribers and other terms of each agreement that impact our expected attainment or recoupment of the minimum guarantees based on the relative attribution method.

Several of our content acquisition agreements also include provisions related to the royalty payments and structures of those agreements relative to other content licensing arrangements, which, if triggered, could cause our payments under those agreements to escalate. In addition, record labels, publishers and PROs with whom we have entered into direct license agreements have the right to audit our content acquisition payments, and any such audit could result in disputes over whether we have paid the proper content acquisition costs. However, as of December 31, 2017, we do not believe it is probable that these provisions of our agreements discussed above will, individually or in the aggregate, have a material adverse effect on our business, financial position, results of operations or cash flows.

Legal Proceedings

We have been in the past, and continue to be, a party to various legal proceedings, which have consumed, and may continue to consume, financial and managerial resources. We record a liability when we believe that it is both probable that a loss has been incurred and the amount can be reasonably estimated. Our management periodically evaluates developments that could affect the amount, if any, of liability that we have previously accrued and make adjustments as appropriate. Determining both the likelihood and the estimated amount of a loss requires significant judgment, and management's judgment may be incorrect. We do not believe the ultimate resolution of any pending legal matters is likely to have a material adverse effect on our business, financial position, results of operations or cash flows.

Pre-1972 copyright litigation

On October 2, 2014, Flo & Eddie Inc. filed a class action suit against Pandora Media Inc. in the federal district court for the Central District of California. The complaint alleges misappropriation and conversion in connection with the public performance of sound recordings recorded prior to February 15, 1972. On December 19, 2014, Pandora filed a motion to strike the complaint pursuant to California's Anti-Strategic Lawsuit Against Public Participation ("Anti-SLAPP") statute, which was appealed to the Ninth Circuit Court of Appeals. The district court litigation is currently stayed pending the Ninth Circuit's decision. On December 8, 2016, the Ninth Circuit heard oral arguments on the Anti-SLAPP motion. On March 15, 2017, the Ninth Circuit requested certification to the California Supreme Court on the substantive legal questions. The California Supreme Court has accepted certification and the Company filed its opening brief on August 4, 2017.

Between September 14, 2015 and October 19, 2015, Arthur and Barbara Sheridan filed separate class action suits against the Company in the federal district courts for the Northern District of California and the District of New Jersey. The complaints allege a variety of violations of common law and state copyright statutes, common law misappropriation, unfair competition, conversion, unjust enrichment and violation of rights of publicity arising from allegations that we owe royalties for the public performance of sound recordings recorded prior to February 15, 1972. The actions in California and New Jersey are currently stayed pending the Ninth Circuit's decision in *Flo & Eddie, Inc. v. Pandora Media, Inc.*

On September 7, 2016, Ponderosa Twins Plus One et al. filed a class action suit against the Company alleging claims similar to that of *Flo & Eddie, Inc. v. Pandora Media Inc.* The action is currently stayed in the Northern District of California pending the Ninth Circuit's decision in *Flo & Eddie, Inc. v. Pandora Media, Inc.*

The outcome of any litigation is inherently uncertain. Except as noted above, we do not believe it is probable that the final outcome of the matters discussed above will, individually or in the aggregate, have a material adverse effect on our business, financial position, results of operations or cash flows; however, in light of the uncertainties involved in such matters,

there can be no assurance that the outcome of each case or the costs of litigation, regardless of outcome, will not have a material adverse effect on our business.

Indemnification Agreements and Contingencies

In the ordinary course of business and in connection with the sale of Ticketfly, we are party to certain contractual agreements under which we may provide indemnifications of varying scope, terms and duration to customers, vendors, lessors, business partners and other parties with respect to certain matters, including, but not limited to, losses arising out of breach of such agreements, services to be provided by us or from intellectual property infringement claims made by third parties. In addition, we have entered into indemnification agreements with directors and certain officers and employees that will require us, among other things, to indemnify them against certain liabilities that may arise by reason of their status or service as directors, officers or employees. In connection with the sale of Ticketfly, we have accrued approximately \$3.9 million related to these indemnifications, which is the probable indemnification liability as estimated in accordance with the accounting guidance for loss contingencies. Other than this amount, to date, we have not incurred, do not anticipate incurring and therefore have not accrued for, any costs related to such indemnification provisions.

While the outcome of these matters cannot be predicted with certainty, we do not believe that the outcome of any claims under indemnification arrangements will have a material adverse effect on our consolidated financial position, results of operations or cash flows.

12. Provision for Income Taxes

Loss before provision for income taxes by jurisdiction consists of the following:

		Yea	r ended December 31,	
	 2015		2016	2017
			(in thousands)	
Jurisdiction				
Domestic	\$ (163,460)	\$	(328,414)	\$ (499,803)
Foreign	(7,751)		(14,792)	(19,382)
Loss before provision for income taxes	\$ (171,211)	\$	(343,206)	\$ (519,185)

The provision for income taxes consists of the following:

	Year ended December 31,					
		2015		2016		2017
				(in thousands)		
Current						
Federal	\$	—	\$	—	\$	—
State and local		9		26		43
International		214		443		(2)
Total current income tax expense	\$	223	\$	469	\$	41
Deferred						
Federal		(17,943)		(96,852)		58,116
State and local		(2,174)		(10,750)		(20,267)
International		—		(1,032)		2,178
Valuation allowance		18,344		107,937		(40,858)
Total deferred income tax expense (benefit)	\$	(1,773)	\$	(697)	\$	(831)
Total provision for (benefit from) income taxes	\$	(1,550)	\$	(228)	\$	(790)

The income tax provision decreased by \$0.6 million during the year ended December 31, 2017 as a result of valuation allowance releases resulting from provisions of the Tax Cut and Jobs Act.

The following table presents a reconciliation of the statutory federal rate and our effective tax rate:

		Year ended December 31,				
	2015	2016	2017			
U.S. federal taxes at statutory rate	34 %	34 %	34 %			
State taxes, net of federal benefit	_		_			
Permanent differences	3	2	(14)			
Foreign rate differential	(1)	(2)	(1)			
Federal and state credits, net of reserve	2	2	1			
Impact of acquired deferred tax assets and liabilities	1	1				
Change in valuation allowance	(33)	(32)	8			
Change in rate	(1)		(25)			
Deferred adjustments	(4)	(5)	(3)			
Effective tax rate	1 %	- %	— %			

The major components of deferred tax assets and liabilities consist of the following:

	As of December 31,		
2016		2017	
(in tho	usands)		
\$ 167,961	\$	245,442	
21,111		31,575	
27,729		15,293	
32,986		20,162	
3,704		5,746	
\$ 253,491	\$	318,218	
(200,797)		(302,298)	
\$ 52,694	\$	15,920	
(31,592)		(16,348)	
(22,360)		_	
\$ (53,952)	\$	(16,348)	
\$ (1,258)	\$	(428)	
\$ \$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$(in thousands)$ $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	

During the year ended December 31, 2016, we released \$1.9 million of our valuation allowance as a result of acquisitions. Deferred tax liabilities were established for the book-tax basis difference related to acquired intangible assets. The net deferred tax liabilities provided an additional source of income to support the realizability of preexisting deferred tax assets.

At December 31, 2017, we had federal net operating loss carryforwards ("NOLs") of approximately \$972.7 million and tax credit carryforwards of approximately \$20.5 million. The federal net operating losses and tax credits expire in years beginning in2021. At December 31, 2017, we had state net operating loss carryforwards of approximately \$640.2 million, which expire in years beginning in 2018. In addition, we had state tax credit carryforwards of approximately \$22.0 million that do not expire and \$6.2 million of credits that will expire beginning in 2024.

Included in the net operating loss carryforward amounts above are approximately \$62.4 million of federal and \$41.3 million of state net operating loss carryforwards related to acquisitions.

Under Section 382 and 383 of the Internal Revenue Code of 1986, as amended (the "Code"), if a corporation undergoes an "ownership change," the corporation's ability to use its pre-change net operating loss carryforwards and other pre-change attributes, such as research tax credits, to offset its post-change income may be limited. In general, an "ownership change" will occur if there is a cumulative change in our ownership by "5-percent shareholders" that exceeds 50 percentage points over a rolling three-year period. Similar rules may apply under state tax laws. Utilization of our net operating loss and tax credit carryforwards may be subject to annual limitations due to ownership changes. Such annual limitations could result in the expiration of our net operating loss and tax credit carryforwards before utilized.

On December 22, 2017, the Tax Cut and Jobs Act was signed into law, which enacted significant changes to U.S. tax and related laws. Some of the provisions of the new tax law affecting corporations include, but are not limited to a reduction of the federal corporate income tax rate from 35% to 21%, limiting the interest expense deduction, expensing of cost of acquired qualified property, elimination of the domestic production activities deduction and allows net operating losses generated in taxable years ending after December 31, 2017 to be carried forward indefinitely. The reduction in the federal corporate income tax rate reduced our net deferred tax assets by approximately \$129.2 million with an offsetting reduction to our valuation allowance of approximately\$130.0 million, which has been reflected in our financial statements for the tax year ended December 31, 2017. The rate reduction and the valuation allowance offset is included in the change in rate and change in valuation allowance line items in the reconciliation of the offset is a release of federal valuation allowance of approximately \$1.1 million, and is included in the change in valuation allowance line item in the reconciliation of the effective tax rate. We continue to evaluate the impact the new tax law will have on our financial condition and results of operations.

Additionally, on December 22, 2017, the SEC staff issued Staff Accounting Bulletin No. 118 ("SAB 118") to address the

application of U.S. GAAP in situations when a registrant does not have the necessary information available, prepared or analyzed (including computations) in reasonable detail to complete the accounting for certain income tax effects of the Tax Cut and Jobs Act. The Company has recognized the provisional tax impacts related to the revaluation of net deferred tax assets of approximately \$129.2 million, offset by a reduction in the valuation allowance of approximately \$130.0 million, and the indefinite carryforward of NOLs generated in taxable years ending after December 31, 2017 of approximately \$1.1 million and included these amounts in its consolidated financial statements for the year ended December 31, 2017. The Company has also recognized the impact to the changes to the limitation on the deductibility of executive compensation, offset by the valuation allowance, which did not have a material impact to financial statement for the year ended December 31, 2017. The ultimate impact is not expected to be material, due to, among other things, additional analysis, changes in interpretations and assumptions as applicable and additional although the net impact is not expected to be material, due to, among other things, additional analysis, changes in interpretations and assumptions as applicable and additional additional Jobs Act, any subsequent adjustments will be made in the quarter of 2018 when the analysis is complete.

During the year ended December 31, 2017, our valuation allowance increased by \$101.5 million. At December 31, 2016 and 2017, we maintained a full valuation allowance on our net deferred tax assets. The valuation allowance was determined in accordance with the provisions of Accounting Standards Codification 740—Accounting for Income Taxes, which requires an assessment of both positive and negative evidence when determining whether it is more likely than not that deferred tax assets are recoverable. Such assessment is required on a jurisdiction by jurisdiction basis. Our history of cumulative losses, along with expected future U.S. losses required that a full valuation allowance be recorded against all net deferred tax assets. We intend to maintain a full valuation allowance on net deferred tax assets until sufficient positive evidence exists to support reversal of the valuation allowance.

At December 31, 2016 and 2017 we have unrecognized tax benefits of approximately \$9.4 million and \$12.4 million. The increase in our unrecognized tax benefits was primarily attributable to current year activities. A reconciliation of the beginning and ending amounts of unrecognized tax benefits (excluding interest and penalties) is as follows:

	Year ended December 31,			
	2016 2017			
	 (in thousands)			
Beginning balance	\$ 6,864	\$		9,412
Increases related to tax positions taken during a prior year	—			—
Decreases related to tax positions taken during a prior year	(13)			(37)
Increases related to tax positions taken during the current year	2,561			2,995
Ending balance	\$ 9,412	\$		12,370

The total unrecognized tax benefits, if recognized, would not affect the Company's effective tax rate as the tax benefit would increase a deferred tax asset, which is currently offset with a full valuation allowance. We do not anticipate that the amount of existing unrecognized tax benefits will significantly increase or decrease within the next twelve months. Accrued interest and penalties related to unrecognized tax benefits are recorded in the provision for income taxes. We did not have such interest, penalties or tax benefits during the years ended December 31, 2015, 2016 and 2017.

We file income tax returns in the United States, California, other states and international jurisdictions. Tax years 2000 to 2017 remain subject to examination for U.S. federal, state and international purposes. All net operating loss and tax credits generated to date are subject to adjustment for U.S. federal and state purposes. We are not currently under examination in federal, state or international jurisdictions.

13. Stock-based Compensation Plans and Awards

Stock Compensation Plans

In 2011, our board of directors adopted the Pandora Media, Inc. 2011 Equity Incentive Plan (the "2011 Plan") which encompasses the 2000 Stock Incentive Plan, as amended (the "2000 Plan") and the 2004 Stock Option Plan (the "2004 Plan", together the "Plans")). The Plans are administered by the compensation committee of our board of directors (the "Plan Administrator").

The 2011 Plan provides for the issuance of stock options, restricted stock units and other stock-based awards. Shares of common stock reserved for issuance under the 2011 Plan include 12,000,000 shares of common stock reserved for issuance under the 2011 Plan and1,506,424 shares of common stock previously reserved but unissued under the 2004 Plan as of June 14, 2011. Each year, the number of shares in the reserve under the Plan may be increased by the lesser of 10,000,000 shares, 4.0% of the outstanding shares of common stock on the last day of the prior fiscal year or another amount determined by our board of directors. The 2011 Plan is scheduled to terminate in 2021, unless our board of directors determines otherwise.

Under the 2011 Plan, the Plan Administrator determines various terms and conditions of awards including option expiration dates (no more tharten years from the date of grant), vesting terms (generally over a three to four-year period) and payment terms. For stock option grants the exercise price is determined by the Plan Administrator, but generally may not be less than the fair market value of the common stock on the date of grant.

We have an approved ESPP that allows eligible employees to purchase shares of our common stock through payroll deductions of up td 5% of their eligible compensation, subject to a maximum of \$25,000 per calendar year. Shares reserved for issuance under the ESPP include 10,000,000 shares of common stock. The ESPP provides forsix-month offering periods, commencing in February and August of each year. At the end of each offering period employees are able to purchase shares at 85% of the lower of the fair market value of our common stock on the first trading day of the offering period or on the last day of the offering period.

Shares available for grant as of December 31, 2017 and the activity during the year ended December 31, 2017 are as follows:

	Sha	Shares Available for Grant			
	Equity Awards	ESPP	Total		
Balance as of December 31, 2016	8,344,668	2,064,427	10,409,095		
Additional shares authorized	9,406,510	6,000,000	15,406,510		
Restricted stock units granted	(13,627,053)	_	(13,627,053)		
Options granted	(2,226,443)	_	(2,226,443)		
ESPP shares issued	_	(1,287,687)	(1,287,687)		
Options forfeited	382,769	_	382,769		
Options expired	1,469,714	_	1,469,714		
Restricted stock units forfeited	7,089,976	_	7,089,976		
Performance units forfeited*	883,169	_	883,169		
Balance as of December 31, 2017	11,723,310	6,776,740	18,500,050		
*Includes shares forfeited for MSUs and PSUs					

ESPP

We estimate the fair value of shares to be issued under the ESPP on the first day of the offering period using the Black-Scholes valuation model. The inputs to the Black-Scholes option-pricing model are our stock price on the date of grant, the expected stock price volatility over the expected term of the award, which is based on projected employee stock option exercise behaviors, the risk-free interest rate for the expected term of the award and expected dividends. Stock-based compensation expense related to the ESPP is recognized on a straight-line basis over the offering period. Forfeitures are recognized as they occur.

The following assumptions for the Black-Scholes option pricing model were used to determine the per-share fair value of shares to be granted under the ESPP:

	Year Ended December 31,			
	2015	2016	2017	
Expected term (in years)	0.5	0.5	0.5	
Risk-free interest rate	0.12 %	0.36 %	0.78 %	
Expected volatility	52 %	44 %	45 %	
Expected dividend yield	0%	0 %	0 %	

During the years ended December 31, 2015, 2016 and 2017, we withheld \$7.6 million, \$9.7 million and \$10.9 million in contributions from employees and recognized \$3.3 million, \$3.4 million and \$4.0 million of stock-based compensation expense related to the ESPP, respectively. In the years endedDecember 31, 2015, 2016 and 2017, 538,398, 1,254,910 and 1,287,687 shares of common stock were issued under the ESPP at a weighted average purchase price of \$17.80, \$6.76 and \$8.84, respectively.

Stock Options

Stock option activity during the year ended December 31, 2017 was as follows:

		Options Outstanding					
	Outstanding Weighted-Average Re Stock Options Exercise Price			Aggregate Intrinsic Value (1)			
		(in thousands, except share and per share data)					
Balance as of December 31, 2016	9,436,380	\$ 5.74	0.47	\$ 77,752			
Granted	2,226,938	9.80					
Exercised	(4,968,577)	1.97					
Forfeited	(903,112)	7.69					
Expired	(1,469,714)	16.15					
Balance as of December 31, 2017	4,321,915	8.90	5.84	2,922			
Vested and exercisable as of December 31, 2017	2,616,163	8.55	3.43	2,918			
Expected to vest as of December 31, 2017	1,705,752	\$ 9.45	9.53	\$ 4			

(1)Amounts represent the difference between the exercise price and the fair value of common stock at each period end for all in the money options outstanding based on the fair value per share of common stock of \$13.04 and \$4.82 as of December 31, 2016 and 2017.

The per-share fair value of each stock option was determined on the grant date using the Black-Scholes option pricing model using the following assumptions. There were no option grants during the year ended December 31, 2016.

	Ye	ar Ended December 31,	
	2015	2016	2017
Expected term (in years)	6.08	N/A	5.93 - 6.25
Risk-free interest rate	1.75% - 1.92%	N/A	1.92% - 2.18%
Expected volatility	49% - 50%	N/A	61 %
Expected dividend yield	0 %	N/A	0 %

The expected term of stock options granted represents the weighted average period that the stock options are expected to

remain outstanding. We determined the expected term assumption based on our historical exercise behavior combined with estimates of the post-vesting holding period. Expected volatility is based on the historical volatility of our common stock over the expected term. The risk-free interest rate is based on the implied yield currently available on U.S. Treasury issues with terms approximately equal to the expected life of the option. We currently have no history or expectation of paying cash dividends on our common stock.

During the years ended December 31, 2015, 2016 and 2017, we recorded stock-based compensation expense related to stock options of approximately \$10.7 million, \$13.8 million and \$7.6 million, respectively.

As of December 31, 2017, there was \$8.1 million of unrecognized compensation cost related to outstanding employee stock options. This amount is expected to be recognized over a weighted-average period of 3.52 years.

The weighted-average fair value of stock option grants made during the years endedDecember 31, 2015 and 2017 was \$9.08 and \$5.55.

The aggregate intrinsic value of stock options exercised during the years endedDecember 31, 2015, 2016 and 2017 was \$9.5 million, \$17.3 million and \$15.1 million, respectively. The total fair value of options vested during the years ended December 31, 2015, 2016 and 2017 was \$17.6 million, \$10.2 million and \$4.6 million, respectively.

RSUs

The fair value of the restricted stock units is expensed ratably over the vesting period. RSUs vest annually on a cliff basis over the service period, which is generally three to four years. During the years ended December 31, 2015, 2016 and 2017, we recorded stock-based compensation expense related to RSUs of approximately\$96.1 million, \$124.3 million and \$114.4 million, respectively. As of December 31, 2017, total compensation cost not yet recognized of approximately\$175.2 million related to non-vested restricted stock units, is expected to be recognized over a weighted average period of 2.03 years.

The following table summarizes the activities for our RSUs for theyear ended December 31, 2017:

	Number of RSUs	Weighted-Average Grant Date Fair Value
Unvested as of December 31, 2016	22,404,417 \$	3 13.26
Granted	13,627,053	10.13
Vested	(9,478,804)	13.80
Forfeited	(7,277,071)	12.16
Unvested as of December 31, 2017	19,275,595	11.19
Expected to vest as of December 31, 2017	19,275,595 \$	5 11.19

MSUs

In March 2015, the compensation committee of the board of directors granted performance awards consisting of market stock units to certain key executives under our 2011 Plan.

MSUs granted in March 2015 are earned as a function of Pandora's TSR performance measured against that of the Russell 2000 Index acrossthree performance periods:

- One-third of the target MSUs are eligible to be earned for a performance period that is the first calendar year of the MSU grant (the "One-Year Performance Period");
- One-third of the target MSUs are eligible to be earned for a performance period that is the first two calendar years of the MSU grant (the "Two-Year Performance Period"); and
- Any remaining portion of the total potential MSUs are eligible to be earned for a performance period that is the entire three calendar years of the MSU grant (the "Three-Year Performance Period").

For each performance period, a "performance multiplier" is calculated by comparing Pandora's TSR for the period to the



Russell 2000 Index TSR for the same period, using the average adjusted closing stock price of Pandora stock, and the Russell 2000 Index, fominety calendar days prior to the beginning of the performance period and the last ninety calendar days of the performance period. In each period, the target number of shares will vest if the Pandora TSR is equal to the Russell 2000 Index TSR. For each percentage point that the Pandora TSR falls below the Russell 2000 Index TSR for the period, the performance multiplier is decreased by three percentage points. The performance multiplier is capped at 100% for the One-Year and Two-Year Performance Periods. However, the full award is eligible for a payout up to 200% of target, less any shares earned in prior periods, in the Three-Year Performance Period. Specifically, for each percentage point that the Pandora TSR exceeds the Russell 2000 Index TSR for the Three-Year Performance Period. As such, the ability to exceed the target number of shares is determined exclusively with respect to Pandora's three-year TSR during the term of the award.

We have determined the grant-date fair value of the MSUs using a Monte Carlo simulation performed by a third-party valuation firm. We recognize stock-based compensation for the MSUs over the requisite service period, which is approximately three years, using the accelerated attribution method.

During the year ended December 31, 2015, we granted 776,000 MSUs at a total grant-date fair value of \$4.3 million. There were no MSUs granted in the years ended December 31, 2016 and 2017, we recorded stock-based compensation expense from MSUs of approximately \$1.5 million, \$0.7 million and \$0.4 million, respectively.

In February 2016, January 2017 and January 2018, the compensation committee of the board of directors certified the results of the One-Year Performance Period, Two-Year Performance Period and Three-Year Performance Period of the 2015 MSU grant, which concluded December 31, 2015, 2016 and 2017. During the One-Year Performance Period, our relative TSR declined 26 percentage points relative to the Russell 2000 Index TSR for the period, which resulted in the vesting of the One-Year Performance Period at 22% of the one-third vesting opportunity for the period. During the Two-Year Performance Period, our relative TSR declined48 percentage points relative to the Russell 2000 Index TSR for the period, which resulted in vesting of the Three-Year Performance Period at 0% of the one-third vesting opportunity for the period. During the Two-Year Performance Period at 0% of the one-third vesting opportunity for the period. During the Two-Year Performance Period at 0% of the one-third vesting opportunity for the period. During the Two-Year Performance Period at 0% of the one-third vesting opportunity for the period. During the Two-Year Performance Period at 0% of the one-third vesting opportunity for the period. During the Three-Year Performance Period, our relative TSR declined 101 percentage points relative to the Russell 2000 Index TSR for the period, which resulted in vesting of the Three-Year Performance Period at 0% of the full vesting opportunity.

The following table summarizes the activities for our MSUs for theyear ended December 31, 2017:

	Number of MSUs	Weighted-Average Gran Fair Value	nt Date
Unvested as of December 31, 2016	533,383	\$	5.49
Forfeited	(217,387)		5.64
Unvested as of December 31, 2017	315,996		5.38
Expected to vest as of December 31, 2017	113,672	\$	5.38

PSUs

In 2016, the compensation committee of the board of directors granted 2016 Performance Awards consisting of stock-settled performance-based RSUs to certain key executives under our 2011 Plan.

PSUs have a vesting period that includes a four-year service period, during which one fourth of the awards will vest after one year and the remainder will vest quarterly thereafter. The PSUs are earned when our trailing average ninety-day stock price is equal to or greater than \$20.00. If the trailing average ninety-day stock price does not equal or exceed \$20.00 on the applicable vesting date, then the portion of the award that was scheduled to vest on such vesting date shall not vest but shall vest on the next vesting date on which the trailing average ninety-day stock price equals or exceeds \$20.00. Any portion of the award that remains unvested as of the final vesting date shall be canceled and forfeited.

We determined the grant-date fair value of the PSUs using a Monte Carlo simulation performed by a third-party valuation firm. We recognize stock-based compensation for the PSUs over the requisite service period, which is approximately four years, using the accelerated attribution method. During the year ended December 31, 2016, we granted 1,835,250 PSUs at

a total grant-date fair value of \$9.8 million. No PSUs were granted in the years endedDecember 31, 2015 and 2017. During the years ended December 31, 2016 and 2017, we recorded stock-based compensation expense from PSUs of approximately \$3.8 million and \$2.0 million. As of December 31, 2017, total compensation cost not yet recognized of approximately \$0.9 million related to non-vested PSUs is expected to be recognized over a weighted average period of 2.14 years. There was no stock-based compensation expense related to PSUs or shares of common stock issued as a result of vesting of PSUs in the year ended December 31, 2015.

The following table summarizes the activities for our PSUs for theyear ended December 31, 2017:

	Number of PSUs	Weighted-Average Grant Date Fair Value	è
Unvested as of December 31, 2016	1,835,250	\$ 5.3	3
Forfeited	(665,782)	5.5	;5
Unvested as of December 31, 2017	1,169,468	5.2	21
Expected to vest as of December 31, 2017	544,392	\$ 5.0	19

Stock-based Compensation Expense

Stock-based compensation expense related to all employee and non-employee stock-based awards was as follows:

		Year	ended December 31,	
	 2015		2016	2017
			(in thousands)	
Stock-based compensation expense				
Cost of revenue—Other	\$ 5,531	\$	6,108	\$ 3,249
Cost of revenue—Ticketing service	40		188	69
Product development	23,671		30,975	33,243
Sales and marketing	52,747		58,118	56,116
General and administrative	29,656		43,069	35,754
Total stock-based compensation expense	\$ 111,645	\$	138,458	\$ 128,431

During the years ended December 31, 2016 and 2017, we capitalized \$7.5 million and \$7.4 million of stock-based compensation as internal use software and website development costs.

During the years ended December 31, 2016 and 2017, we recorded \$8.7 million and \$5.4 million related to accelerated awards in connection with executive severance. The majority of this amount is included in the general and administrative line item of our consolidated statement of operations.

14. Common Stock and Net Loss per Common Share

Each share of common stock has the right toone vote per share. The holders of common stock are also entitled to receive dividends as and when declared by our board of directors, whenever funds are legally available.

Net Loss per Common Share

Basic net loss per common share is computed by dividing the net loss available to common stockholders by the weighted-average number of shares of common stock outstanding during the period.

Diluted net loss per common share is computed by giving effect to all potential shares of common stock, including stock options, restricted stock units, market stock units, performance-based RSUs, potential ESPP shares and instruments convertible



into common stock, to the extent dilutive. Basic and diluted net loss per common share were the same for the years endedDecember 31, 2015, 2016 and 2017, as the inclusion of all potential common shares outstanding would have been anti-dilutive.

The following table sets forth the computation of historical basic and diluted net loss per common share:

	Year Ended December 31,										
		2015		2016		2017					
Numerator											
Net loss	\$	(169,661)	\$	(342,978)	\$	(518,395)					
Less: Stock dividend payable and transaction costs		—		—		40,166					
Net loss available to common stockholders		(169,661)		(342,978)		(558,561)					
Denominator											
Weighted-average common shares outstanding used in computing basic and diluted net loss per common share		213,790		230,693		243,637					
Net loss per common share, basic and diluted	\$	(0.79)	\$	(1.49)	\$	(2.29)					

The following potential common shares outstanding were excluded from the computation of diluted net loss per common share because including them would have been anti-dilutive:

		As of December 31,	
	2015	2016	2017
		(in thousands)	
Shares issuable upon conversion of preferred stock	—	—	46,747
Restricted stock units	17,272	22,404	19,276
Options to purchase common stock	12,816	9,436	4,322
Performance awards*	776	2,369	1,486
Shares issuable pursuant to the ESPP			
	296	577	820
Total common stock equivalents	31,160	34,786	72,651
*Includes potential common shares outstanding for MSUs and PSUs			

On December 9, 2015, we completed an offering of our1.75% convertible senior notes due 2020. As we have the intention to settle the Notes either partially or wholly in cash, under the treasury stock method, the Notes will generally have a dilutive impact on earnings per share if we are in a net income position for the period and if our average stock price for the period exceeds approximately \$16.42 per share of our common stock, the conversion price of the Notes. For the period from the issuance of the offering of the Notes through December 31, 2017, the conversion feature of the Notes was anti-dilutive, as we were in a net loss position for all periods and our average stock price was less than the conversion price.

In connection with the pricing of the Notes, we entered into capped call transactions which increase the effective conversion price of the Notes, and are designed to reduce potential dilution upon conversion of the Notes. Since the beneficial impact of the capped call is anti-dilutive, it is excluded from the calculation of earnings per share. Refer to <u>Note 9 "Debt Instruments"</u> for further details regarding our Notes.

15. Segment Data and Revenue by Geographic Area

Segment Data

We report segment information based on the "management" approach. The management approach designates the internal reporting used by management for making decisions and assessing performance as the source of our reportable operating segments.

Our chief operating decision maker (the "CODM"), our Chief Executive Officer, manages our operations on a consolidated basis for purposes of allocating resources. When evaluating our financial performance, the CODM reviews separate revenue information for our advertising, subscription and other offerings, while all other financial information is reviewed on a consolidated basis. There are no segment managers who are held accountable by the CODM, or anyone else, for operations, operating results and planning for levels or components below the consolidated unit level. Accordingly, we have determined that we have one reportable segment and operating unit structure. Prior to the sale of Ticketfly, we operated in two reportable segments. Refer to <u>Note 5 "Dispositions"</u> in the Notes to Consolidated Financial Statements for further details on the Ticketfly disposition.

Revenue by Geographic Area

The following table sets forth revenue by geographic area:

	Year ended December 31,								
		2015		2016		2017			
			(in thousands)						
Revenue by geographic area									
United States	\$	1,155,210	\$	1,366,330	\$	1,456,673			
International		8,833		18,496		10,139			
Total revenue	\$	1,164,043	\$	1,384,826	\$	1,466,812			

As of December 31, 2015 and 2016, long-lived assets were predominantly located in the United States, and as ofDecember 31, 2017, all long-lived assets were located in the United States. On June 27, 2017, we announced a plan to discontinue business activities in Australia and New Zealand. The discontinuance of the Australia and New Zealand business operations was substantially completed in the year ended December 31, 2017. No individual foreign country represented a material portion of our consolidated revenue during the years ended December 31, 2015, 2016 and 2017.

16. Restructuring Charges

Reduction in Force

On January 12, 2017, we announced a reduction in force plan affecting approximately7% of our U.S. employee base, excluding Ticketfly. In the year ended December 31, 2017, we incurred approximately \$6.0 million of cash expenditures, substantially all of which were related to employee severance and benefits costs. In the year ended December 31, 2017, total reduction in force expenses were \$5.6 million, which was lower than cash reduction in force costs due to a credit related to non-cash stock-based compensation expense reversals for unvested equity awards. Reduction in force expenses are recognized in the cost of revenue—other, product development, sales and marketing and general and administrative line items on our Consolidated Statements of Operations. The reduction in force plan was completed and all amounts were paid in the year ended December 31, 2017.

Australia and New Zealand Exit Costs

On June 27, 2017, we announced a plan to discontinue business activities in Australia and New Zealand. The related restructuring charges in the year ended December 31, 2017 primarily relate to a reduction of headcount of approximately 50 employees, which resulted in employee severance and benefits costs offset by a credit related to noncash stock-based compensation expense reversals for unvested equity awards. Restructuring charges are recognized in the cost of revenue—other, product development, sales and marketing and general and administrative line items on our Consolidated Statements of



Operations. The discontinuance of the Australia and New Zealand business operations was substantially completed in the year ended December 31, 2017. These restructuring charges did not have a material impact on our financial statements.

17. Selected Quarterly Financial Data (unaudited)

	Quarter ended															
	March 31, June		June 30,	September 30,			December 31,		March 31,		June 30,		eptember 30,	December 31,		
		2016		2016		2016		2016		2017		2017		2017		2017
						(in th	ousands, exc	ept p	er share data	ı)					
Revenue																
Advertising	\$	220,308	\$	265,126	\$	273,716	\$	313,340	\$	223,308	\$	278,204	\$	275,741	\$	297,674
Subscription and other		54,732		55,125		56,100		59,829		64,878		68,900		84,414		97,661
Ticketing service (1)		22,265		22,771		22,085		19,429		27,818		29,730		18,484		_
Total revenue		297,305		343,022		351,901		392,598		316,004		376,834		378,639		395,335
Cost of revenue																
Cost of revenue—Content acquisition costs		171,264		176,633		174,334		212,122		187,420		195,875		204,222		216,515
Cost of revenue—Other		21,195		25,106		25,896		30,520		25,532		27,440		27,287		32,379
Cost of revenue—Ticketing service (1)		14,646		15,259		15,318		14,057		18,618		20,510		11,269		_
Total cost of revenue		207,105		216,998		215,548		256,699		231,570	_	243,825		242,778		248,894
Gross profit		90,200		126,024		136,353		135,899		84,434		133,009		135,861		146,441
Operating expenses																
Product development (1)		35,611		33,560		33,560		37,976		39,588		41,233		39,469		34,035
Sales and marketing (1)		117,433		123,589		116,091		133,251		125,102		145,891		107,588		113,961
General and administrative (1)		46,524		40,760		41,909		46,971		44,525		57,954		48,171		40,061
Goodwill impairment		_		_		_		_		_		131,997				_
Contract termination fees (benefit)		_		_		_		_		_		23,467		(423)		—
Total operating expenses		199,568		197,909		191,560		218,198		209,215		400,542		194,805		188,057
Loss from operations		(109,368)		(71,885)		(55,207)		(82,299)		(124,781)		(267,533)		(58,944)		(41,616)
Net loss		(115,102)		(76,333)		(61,534)		(90,009)		(132,267)		(275,136)		(66,243)		(44,749)
Net loss available to common stockholders		(115,102)		(76,333)		(61,534)		(90,009)		(132,267)		(289,664)		(84,562)		(52,068)
Net loss per common share, basic and diluted	\$	(0.51)	\$	(0.33)	\$	(0.27)	\$	(0.38)	\$	(0.56)	\$	(1.20)	\$	(0.34)	\$	(0.21)
(1) Includes approximately two more	ths o	f revenue and	expe	nse for Ticketf	ly fro	om July 1, 201	7 to 1	the disposition	n date	e of Septembe	r 1, 2	017.				

Pandora Media, Inc. Notes to Consolidated Financial Statements - Continued

18. Subsequent Event

On January 31, 2018, we announced efforts to prioritize our strategic growth initiatives and optimize overall business performance, including a reduction in force plan affecting approximately 5% of our employee base. Our Board of Directors approved the plan on January 11, 2018 and affected employees were informed of the plan on January 31, 2018. We expect the reduction in force plan to be substantially completed by the end of the first quarter of 2018.

Total costs and cash expenditures for the reduction in force plan are estimated at \$6.5 million to \$8.5 million, substantially all of which are related to employee severance and benefits costs. We expect to recognize most of these pre-tax reduction in force charges in the first quarter of 2018.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURES

None.

ITEM 9A. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

We maintain "disclosure controls and procedures," as such term is defined in Rule 13a-15(e) under the Exchange Act, that are designed to ensure that information required to be disclosed by us in reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms, and that such information is accumulated and communicated to our management, including our chief executive officer and chief financial officer, as appropriate to allow timely decisions regarding required disclosure. In designing and evaluating our disclosure controls and procedures, management recognizes that disclosure controls and procedures, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the disclosure controls and procedures are met. Additionally, in designing disclosure controls and procedures, uranagement necessarily was required to apply judgment in evaluating the cost-benefit relationship of possible disclosure controls and procedures. Based on their evaluation at the end of the period covered by this Annual Report on Form 10-K, our chief executive officer and chief financial officer have concluded that our disclosure controls and procedures were effective at the reasonable assurance level as of December 31, 2017.

Management's Report on Internal Control Over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act. Our internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of consolidated financial statements for external purposes in accordance with generally accepted accounting principles. Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of internal control effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

Our management has assessed the effectiveness of the internal control over financial reporting as ofDecember 31, 2017. In making this assessment, our management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission ("COSO") in Internal Control-Integrated Framework (2013 framework).

Based on the results of this evaluation, our management has concluded that our internal control over financial reporting was effective as oDecember 31, 2017.

The effectiveness of our internal control over financial reporting as of December 31, 2017 has been audited by Ernst & Young LLP, an independent registered public accounting firm, as stated in their report which is included in this Annual Report on Form 10-K.

Changes in Internal Control over Financial Reporting

There have been no changes in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the year ended December 31, 2017, that have materially affected, or are reasonably likely to materially affect our internal control over financial reporting.

106

Pandora Media, Inc. Notes to Consolidated Financial Statements - Continued

ITEM 9B. OTHER INFORMATION

None.

107

PART III

ITEM 10. DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

Information required by this Item regarding our directors and executive officers is incorporated by reference to the sections of our proxy statement to be filed with the SEC in connection with our 2018 Annual Meeting of Stockholders (the "Proxy Statement") entitled "Election of Class III Directors" and "Management."

Information required by this Item regarding our corporate governance, including our audit committee and code of business conduct and ethics, is incorporated by reference to the sections of the Proxy Statement entitled "Corporate Governance" and "Board of Directors."

Information required by this Item regarding compliance with Section 16(a) of the Exchange Act required by this Item is incorporated by reference to the section of the Proxy Statement entitled "Section 16(a) Beneficial Ownership Reporting Compliance."

ITEM 11. EXECUTIVE COMPENSATION

Information required by this Item is incorporated by reference to the sections of the Proxy Statement entitled "Executive Compensation," "Board of Directors—Director Compensation," "Corporate Governance—Compensation Committee Interlocks and Insider Participation."

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

Information regarding security ownership of certain beneficial owners and management is incorporated by reference to the section of the Proxy Statement entitled "Security Ownership of Certain Beneficial Owners and Management."

Information regarding our stockholder approved and non-approved equity compensation plans is incorporated by reference to the section of the Proxy Statement entitled "Equity Compensation Plan Information."

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

Information required by this Item is incorporated by reference to the sections of the Proxy Statement entitled "Certain Relationships and Related-Party Transactions" and "Corporate Governance—Director Independence."

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

Information required by this Item is incorporated by reference to the section of the Proxy Statement entitled "Ratification of Appointment of Independent Registered Public Accounting Firm."



PART IV

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES (a) The following documents are included as part of this Annual Report on Form 10-K.

1. Index to Financial Statements

- Reports of Ernst & Young LLP, Independent Registered Public Accounting Firm
- Consolidated Balance Sheets
- Consolidated Statements of Operations
- Consolidated Statements of Comprehensive Loss
- Consolidated Statements of Stockholders' Equity
- Consolidated Statements of Cash Flows
- Notes to Consolidated Financial Statements

2. Financial Statement Schedules

All other schedules are omitted as the information required is inapplicable or the information is presented in the consolidated financial statements or the related notes.

3. Exhibits

See the Exhibit Index immediately following the signature page of this Annual Report on Form 10-K.

SIGNATURES

Pursuant to the requirements Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized on February 26, 2018.

PANDORA MEDIA, INC.

By: /s/ ROGER LYNCH

 Name:
 Roger Lynch

 Chief Executive Officer (Principal Executive Officer) and

 Title:
 Director

POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints Roger Lynch, Naveen Chopra and Stephen Bené and each of them, his or her true and lawful attorneys-in-fact and agents, with full power to act separately and full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign any and all amendments to this Annual Report on Form 10-K, and to file the same, with all exhibits thereto, and all other documents in connection therewith, with the Securities and Exchange Commission, granting unto each said attorney-in-fact and agent full power and authority to do and perform each and every act in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or either of them or his or her or their substitute or substitutes may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1934, as amended, this report has been signed by the following persons in the capacities and on the dates indicated.

110

Signature	Title	Date
/s/ ROGER LYNCH	_	February 26, 2018
Roger Lynch	President, Chief Executive Officer (Principal Executive Officer) and Director	
/s/ NAVEEN CHOPRA	_	February 26, 2018
Naveen Chopra	Chief Financial Officer (Principal Financial Officer)	
/s/ KAREN WALKER	_	February 26, 2018
Karen Walker	Chief Accounting Officer (Principal Accounting Officer)	
/s/ GREGORY B. MAFFEI	— —	February 26, 2018
Gregory B. Maffei	Director	
/s/ ROGER FAXON		February 26, 2018
Roger Faxon	Director	
		February 26, 2018
David J. Frear	Director	
		February 26, 2018
Jason Hirschhorn	Director	
/s/ TIMOTHY LEIWEKE		February 26, 2018
Timothy Leiweke	Director	
/s/ MICHAEL M. LYNTON		February 26, 2018
Michael M. Lynton	Director	
/s/ JAMES E. MEYER		February 26, 2018
James E. Meyer	Director	
/s/ MICKIE ROSEN		February 26, 2018
Mickie Rosen	Director	

EXHIBIT INDEX

	Incorporated by Reference						
Exhibit No.	Exhibit Description	Form	File No.	Exhibit	Filing Date	Filed By	Filed Herewith
	Agreement and Plan of Merger, dated as of October 7, 2015, among						
	the Company, Ticketfly, Inc., Tennessee Acquisition Sub I, Inc.,						
2.01	<u>Tennessee Acquisition Sub II, LLC and Shareholder Representative</u> Services LLC	8-K/A	001-35198	2.1	10/8/2015		
2.01	Asset Purchase Agreement, dated as of November 16, 2015, by and	0 1211	001 00190	2.1	10/0/2010		
<u>2.02</u>	between Pandora Media, Inc. and Rdio, Inc.	10-K	001-35198	2.02	2/18/2016		
<u>3.01</u>	Amended and Restated Certificate of Incorporation	S-1/A	333-172215	3.1	5/4/2011		
	Certificate of Amendment to the Amended and Restated Certificate of						
<u>3.02</u>	Incorporation	10-Q	001-35198	3.02	7/26/2016		
<u>3.03</u>	Amended and Restated Bylaws						Х
3.04	<u>Certificate of Designation of Preferences, Rights and Limitations of</u> Series A Convertible Preferred Stock	8-K	001-35198	3.1	6/14/2017		
<u>3.04</u>	Fifth Amended and Restated Investor Rights Agreement, by and	0-K	001-55158	5.1	0/14/2017		
	among Pandora Media, Inc. and the investors listed on Exhibit A						
<u>4.01</u>	thereto, dated May 20, 2010, as amended	S-1/A	333-172215	4.2	2/22/2011		
4.02	Indenture, dated as of December 9, 2015, between Pandora Media,	0.17	001 25100	4.1	12/0/2015		
<u>4.02</u>	Inc. and Citibank, N.A., as Trustee	8-K	001-35198	4.1	12/9/2015		
4.03	Form of 1.75% Convertible Senior Note due 2020 (included in Exhibit 4.02)						
	2011 Equity Incentive Plan and Form of Stock Option Agreement						
<u>10.01†</u>	under 2011 Equity Incentive Plan	S-1/A	333-172215	10.1	5/26/2011		
<u>10.02†</u>	Amendment No. 1 to the 2011 Equity Incentive Plan	S-8	333-216087	99.2	2/16/2017		
	2004 Stock Plan, as amended, and Forms of Stock Option Agreement						
<u>10.03†</u>	and Restricted Stock Purchase Agreement under 2004 Stock Plan	S-1/A	333-172215	10.3	2/22/2011		
	2000 Stock Incentive Plan, as amended, and Forms of NSO Stock Option Agreement and ISO Stock Option Agreement under 2000						
<u>10.04†</u>	Stock Plan	S-1/A	333-172215	10.4	2/22/2011		
<u>10.05†</u>	Amended and Restated 2014 Employee Stock Purchase Plan	8-K	001-35198	10.1	8/9/2017		
	Form of Indemnification Agreement by and between Pandora						
10.064	Media, Inc. and each of its executive officers and its directors not	C 1/A	222 172215	10.5	2/22/2011		
<u>10.06†</u>	affiliated with an investment fund Form of Indemnification Agreement by and between Pandora	S-1/A	333-172215	10.5	2/22/2011		
10.07†	Media, Inc. and each of its directors affiliated with an investment fund	S-1/A	333-172215	10.5A	2/22/2011		
10.08†	Offer Letter with John Trimble, dated February 18, 2009	S-1/A	333-172215	10.10	2/22/2011		
	Office Lease between CIM/Oakland Center 21, LP and Pandora						
<u>10.09</u>	Media, Inc., dated July 23, 2009	S-1/A	333-172215	10.12	2/22/2011		
	First Amendment to Lease between CIM/Oakland Center 21, LP and	~					
<u>10.10A</u>	Pandora Media, Inc., dated April 13, 2010	S-1/A	333-172215	10.12A	2/22/2011		
10.10B	Second Amendment to Lease between CIM/Oakland Center 21, LP and Pandora Media, Inc., dated June 16, 2010	S-1/A	333-172215	10.12B	2/22/2011		
<u>10.10D</u>	Third Amendment to Lease between CIM/Oakland Center 21. LP and	D -1/2 X	555-172215	10.12D	2/22/2011		
<u>10.10C</u>	Pandora Media, Inc., dated December 15, 2010	10-Q	001-35198	10.12C	9/4/2012		
	Fourth Amendment to Lease between CIM/Oakland Center 21, LP						
<u>10.10D</u>	and Pandora Media, Inc., dated March 10, 2011	10-Q	001-35198	10.12D	9/4/2012		
		112					
		112					

Table of Contents

	Fifth Amendment to Lease between CIM/Oakland Center 21, LP and				
<u>10.10E</u>	Pandora Media, Inc., dated July 1, 2011	10-Q	001-35198	10.12E	9/4/2012
10.10F	Sixth Amendment to Lease between CIM/Oakland Center 21, LP and Pandora Media, Inc., dated September 27, 2011	10-Q	001-35198	10.12F	9/4/2012
10.101	Seventh Amendment to Lease between CIM/Oakland Center 21, LP	10-Q	001-55198	10.121	9/4/2012
<u>10.10G</u>	and Pandora Media, Inc., dated July 12, 2012	10-Q	001-35198	10.12G	9/4/2012
	Eighth Amendment to Lease between CIM/Oakland Center 21, LP				
<u>10.10H</u>	and Pandora Media, Inc., dated February 1, 2013	10-Q	001-35198	10.12H	5/29/2013
10.11I	Ninth Amendment to Lease between CIM/Oakland Center 21, LP and Pandora Media, Inc., dated August 15, 2013	10-Q	001-35198	10.12I	10/28/2014
<u>10.111</u>	Tenth Amendment to Lease between CIM/Oakland Center 21, LP and	10-Q	001-55170	10.121	10/20/2014
<u>10.10J</u>	Pandora Media, Inc., dated October 1, 2014	10-Q	001-35198	10.12J	10/28/2014
	Sublease between Cerexa, Inc. and Pandora Media, Inc. dated January	40.77		10 1077	
<u>10.11K</u>	<u>1,2015</u>	10-K	001-35198	10.10K	2/11/2015
10.10L	Eleventh Amendment to Lease between CIM/Oakland Center 21, LP and Pandora Media, Inc., dated July 28, 2015*	10-K	001-35198	10.10M	2/18/2016
101101	Twelfth Amendment to Lease between CIM/Oakland Center 21, LP	10 11	001 00190	10110111	2,10,2010
<u>10.10M</u>	and Pandora Media, Inc., dated January 11, 2017	10-Q	001-35198	10.01	5/9/2017
	Form of Restricted Stock Unit Agreement under the 2011 Equity				
<u>10.11†</u>	Incentive Plan	10-Q	001-35198	10.01	9/2/2011
<u>10.12†</u>	Form of MSU Grant Notice and Award Agreement	10-Q	001-35198	10.28	4/27/2015
10.124	Form of Performance Award Agreement under the 2011 Equity	10.0	001 25108	10.01	7/20/2010
<u>10.13†</u>	Incentive Plan Form of Retention Restricted Stock Unit Agreement under the 2011	10-Q	001-35198	10.01	7/26/2016
10.14†	Equity Incentive Plan				
	Amended and Restated Executive Severance and Change of Control				
<u>10.15†</u>	Policy	10-Q	001-35198	10.04	11/2/2017
<u>10.16†</u>	Amended and Restated 2016 Corporate Incentive Plan	10-K	001-35198	10.19A	2/16/2017
<u>10.17†</u>	2017 Corporate Incentive Plan	10-Q	001-35198	10.02	5/9/2017
<u>10.18†</u>	Offer Letter with Stephen Bené, dated October 14, 2014	10-Q	001-35198	10.26	4/27/2015
<u>10.19†</u>	Offer Letter with Christopher Phillips, dated October 20, 2014	10-Q	001-35198	10.27	4/27/2015
<u>10.20†</u>	Offer Letter with David Gerbitz, dated June 3, 2014	10-Q	001-35198	10.04	5/9/2017
	Severance and Release Agreement between Sara Clemens and				
<u>10.21†</u>	Pandora Media, Inc., dated January 12, 2017	10-Q	001-35198	10.03	5/9/2017
<u>10.22†</u>	Offer Letter with Naveen Chopra, dated February 27, 2017	10-Q	001-35198	10.05	5/9/2017
10.23†	Separation Agreement and General Release between Tim Westergren and Pandora Media, Inc., dated July 15, 2017	10-Q	001-35198	10.01	11/2/2017
10.23	First Amendment to Offer Letter with Naveen Chopra, dated August	10-Q	001-55198	10.01	11/2/2017
10.24†	7, 2017	10-Q	001-35198	10.02	11/2/2017
	Restricted Stock Unit Agreement under the 2011 Equity Incentive				
	Plan between Naveen Chopra and Pandora Media, Inc., dated August				
<u>10.25</u> †	<u>7, 2017</u>	10-Q	001-35198	10.03	11/2/2017
<u>10.26†</u>	Offer Letter with Roger Lynch, dated August 9, 2017	10-Q	001-35198	10.05	11/2/2017
10.27*	Separation Agreement and General Release between Michael Herring	10.0	001 25109	10.00	11/2/2017
<u>10.27†</u>	and Pandora Media, Inc., dated August 15, 2017	10-Q	001-35198	10.06	11/2/2017
		113			

	Restricted Stock Unit Agreement under the 2011 Equity Incentive					
<u>10.28†</u>	Plan between Roger Lynch and Pandora Media, Inc., dated September 18, 2017	10-Q	001-35198	10.08	11/2/2017	
	Restricted Stock Unit Agreement under the 2011 Equity Incentive Plan between Roger Lynch and Pandora Media, Inc., dated September					
<u>10.29†</u>	18, 2017	10-Q	001-35198	10.09	11/2/2017	
<u>10.30†</u>	Option Agreement under the 2011 Equity Incentive Plan between Roger Lynch and Pandora Media, Inc., dated September 18, 2017	10-Q	001-35198	10.10	11/2/2017	
<u>10.31</u>	Capped call transaction confirmation, dated as of December 3, 2015, by and between Morgan Stanley & Co. LLC and Pandora Media, Inc.	8-K	001-35198	10.1	12/9/2015	
	Additional capped call transaction confirmation, dated as of December 4, 2015, by and between Morgan Stanley & Co, LLC and					
<u>10.32</u>	Pandora Media, Inc.	8-K	001-35198	10.2	12/9/2015	
10.22	Capped call transaction confirmation, dated as of December 3, 2015, by and between JPMorgan Chase Bank, National Association, London	0.12	001 25100	10.2	12/0/2015	
<u>10.33</u>	Branch and Pandora Media, Inc. Additional capped call transaction confirmation, dated as of	8-K	001-35198	10.3	12/9/2015	
	December 4, 2015, by and between JPMorgan Chase Bank, National					
<u>10.34</u>	Association, London Branch and Pandora Media, Inc. Investment Agreement dated as of May 8, 2017, by and among	8-K	001-35198	10.4	12/9/2015	
	Pandora Media, Inc., KKR Classic Investors LLC and the other					
<u>10.35</u>	purchasers listed on the signature pages thereto	8-K	001-35198	10.1	5/12/2017	
10.36	<u>First Amendment to Investment Agreement, dated as of June 8, 2017,</u> by and between Pandora Media, Inc. and KKR Classic Investors L.P.	8-K	001-35198	10.1	6/14/2017	
10.37	Notice of Termination, dated as of June 9, 2017	8-K	001-35198	10.2	6/14/2017	
	Investment Agreement, dated as of June 9, 2017, by and between					
<u>10.38</u>	Pandora Media, Inc. and Sirius XM Radio Inc.	8-K	001-35198	10.3	6/14/2017	
<u>10.39</u>	Registration Rights Agreement, dated as of June 9, 2017, by and between Pandora Media, Inc. and Sirius XM Radio Inc.	8-K	001-35198	10.4	6/14/2017	
	Membership Interest Purchase Agreement, dated as of June 9, 2017					
<u>10.40</u>	by and among Eventbrite, Inc., Pandora Media, Inc., and Ticketfly, LLC	8-K	001-35198	10.5	6/14/2017	
	Amendment No. 1 to Membership Interest Purchase Agreement, dated					
<u>10.41</u>	as of June 9, 2017 by and among Eventbrite, Inc., Pandora Media, Inc., and Ticketfly, LLC, dated September 1, 2017	10-Q	001-35198	10.07	11/2/2017	
	<u>Credit Agreement, by and among the Company and Pandora Media</u> California, LLC, the lenders party thereto, Wells Fargo Bank,					
	National Association, as Administrative Agent, and JPMorgan Chase					
	Bank, N.A., Morgan Stanley Senior Funding, Inc. and Wells Fargo Bank, National Association as joint lead arrangers and joint book					
<u>10.42*</u>	runners, dated as of December 29, 2017					
22.01	Consent of Ernst & Young LLP, Independent Registered Public					
<u>23.01</u>	Accounting Firm Power of Attorney (included on signature page of this Annual Report					
<u>24.01</u>	on Form 10-K)					
31.01	<u>Certification of the Principal Executive Officer Pursuant to</u> Section 302 of the Sarbanes-Oxley Act					
	Certification of the Principal Financial Officer Pursuant to					
<u>31.02</u>	Section 302 of the Sarbanes-Oxley Act					
<u>32.01</u>	Certification of the Principal Executive Officer and Principal Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act					
		114				

114

x x x x x x XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within

101.INS the Inline XBRL document
101.SCH XBRL Taxonomy Schema Linkbase Document
101.CAL XBRL Taxonomy Calculation Linkbase Document
101.DEF XBRL Taxonomy Definition Linkbase Document
101.LAB XBRL Taxonomy Labels Linkbase Document
101.PRE XBRL Taxonomy Presentation Linkbase Document
† Indicates management contract or compensatory plan. Schedules and exhibits have been omitted pursuant to Item 601(b)(2)

Х

Х

Х

Х

Х

Х

- Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. A copy of any omitted schedule or exhibit will be furnished on a supplemental basis to the Securities and Exchange Commission upon request; provided, however that we may request confidential treatment pursuant to Rule 24b-2 of the Securities Exchange Act of 1934, as amended, for any schedules or exhibits so
- furnished.
 Confidential treatment requested as to certain portions of this exhibit,
- which portions have been omitted and submitted separately to the
- ** Securities and Exchange Commission.

115

AMENDED AND RESTATED BYLAWS OF PANDORA MEDIA, INC.

* * * * *

ARTICLE 1 OFFICES

Section 1.01. Registered Office. The registered office of Pandora Media, Inc. (the 'Corporation') shall be in the City of Wilmington, County of New Castle, State of Delaware.

Section 1.02. Other Offices. The Corporation may also have offices at such other places both within and without the State of Delaware as the Board of Directors may from time to time determine or the business of the Corporation may require.

Section 1.03. Books. The books of the Corporation may be kept within or without the State of Delaware as the Board of Directors may from time to time determine or the business of the Corporation may require.

ARTICLE 2 MEETINGS OF STOCKHOLDERS

Section 2.01. *Time and Place of Meetings*. All meetings of stockholders shall be held at such place, either within or without the State of Delaware, on such date and at such time as may be determined from time to time by the Board of Directors (or the Chairman in the absence of a designation by the Board of Directors).

Section 2.02. Annual Meetings. An annual meeting of stockholders shall be held once every year for the election of directors and to transact such other business as may properly be brought before the meeting. The annual stockholder meeting shall be held as closely as practicable to the same month of each year so as to ensure that the terms of office of directors shall approximate a complete year in length.

Section 2.03. Special Meetings. Special meetings of the stockholders may be called by a majority of the Board of Directors, the Chairman of the Board of Directors, the President or the Secretary of the Corporation and may not be called by any other person.

Section 2.04. Notice of Meetings and Adjourned Meetings; Waivers of Notice.

(a) Whenever stockholders are required or permitted to take any action at a meeting, a written notice of the meeting shall be given which shall state the place, if any, date and hour of the meeting, the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, and, in the case of a special meeting, the purpose or purposes for which the meeting is called. Unless otherwise provided by the General Corporation Law of the State of Delaware ("Delaware Law"), such notice shall be given not less than 10 nor more than 60 days before the date of the meeting to each stockholder of record entitled to vote at such meeting. Unless these bylaws otherwise require, when a meeting is adjourned to another time or place (whether or not a quorum is present), notice need not be given of the adjourned meeting if the time, place, if any, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting. If the adjournment is taken. At the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If the adjournment is for more than 30 days, or after the adjournment a new record date is fixed for the adjourned meeting.

(b) A written waiver of any such notice signed by the person entitled thereto, or a waiver by electronic transmission by the person entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Business transacted at any special meeting of stockholders shall be limited to the purposes stated in the notice.

Section 2.05. *Quorum.* Unless otherwise provided under the certificate of incorporation or these bylaws and subject to Delaware Law, the presence, in person or by proxy, of the holders of a majority of the outstanding capital stock of the Corporation entitled to vote at a meeting of stockholders shall constitute a quorum for the transaction of business. If, however, such quorum shall not be present or represented at any meeting of the stockholders, a majority in voting interest of the stockholders present in person or represented by proxy may adjourn the meeting, without notice other than announcement at the meeting, until a quorum shall be present or represented any business may be transacted which might have been transacted at the meeting as originally notified.

Section 2.06. Voting.

(a) Unless otherwise provided in the certificate of incorporation and subject to Delaware Law, each stockholder shall be entitled to one vote for each outstanding share of capital stock of the Corporation held by such stockholder. Any share of capital stock of the Corporation held by the Corporation shall have no voting rights. Except as otherwise provided by law, the certificate of incorporation or these bylaws, in all matters other than the election of directors, the affirmative vote of the majority of the shares of capital stock of the Corporation present in person or represented by proxy at the meeting and entitled to vote on the subject matter shall be the act of the stockholders. Subject to the rights of the holders of any series of preferred stock to elect additional directors under specific circumstances, directors shall be elected by a plurality of the votes of the shares of capital stock of the Corporation present in person or represented by proxy at the meeting and entitled to vote on the election of directors.

(b) Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to a corporate action in writing without a meeting may authorize another person or persons to act for such stockholder by proxy, appointed by an instrument in writing, subscribed by such stockholder or by his attorney thereunto authorized, or by proxy sent by cable, telegram or by any means of electronic communication permitted by law, which results in a writing from such stockholder or by his attorney, and delivered to the secretary of the meeting. No proxy shall be voted after three (3) years from its date, unless said proxy provides for a longer period.

(c) In determining the number of votes cast for or against a proposal or nominee, shares abstaining from voting on a matter will not be treated as a vote cast. A non-vote by a broker will be counted for purposes of determining a quorum but not for purposes of determining the number of votes cast.

Section 2.07. No Actions by Written Consent. Subject to the rights of the holders of any series of preferred stock then outstanding, as may be set forth in the certificate of designation relating to such preferred stock, any action required or permitted to be taken at any annual or special meeting of stockholders may be taken only upon the vote of stockholders at an annual or special meeting duly noticed and called in accordance with Delaware Law and may not be taken by written consent of stockholders without a meeting.

Section 2.08 *Organization*. At each meeting of stockholders, the Chairman of the Board, if one shall have been elected, or in the Chairman's absence or if one shall not have been elected, the director designated by the vote of the majority of the directors present at such meeting, shall act as chairman of the meeting. The Secretary (or in the Secretary's absence or inability to act, the person whom the chairman of the meeting shall appoint secretary of the meeting) shall act as secretary of the meeting and keep the minutes thereof.

Section 2.09. Order of Business. The order of business at all meetings of stockholders shall be as determined by the chairman of the meeting.

Section 2.10. Notice of Nominations and Stockholder Business.

(a) Annual Meetings of Stockholders.

(i) Nominations of persons for election to the Board of Directors of the Corporation or the proposal of other business to be transacted by the stockholders at an annual meeting of stockholders may be made only (A) pursuant to the Corporation's notice of meeting (or any supplement thereto), (B) by or at the direction of the Board of Directors or any committee thereof or (C) by any stockholder of the Corporation who is a stockholder of record at the time of giving of notice provided for in this Section 2.10(a) and at the time of the annual meeting, who shall be entitled to vote at the meeting and who complies with the procedures set forth in this Section 2.10(a).

(ii) For nominations or other business to be properly brought before an annual meeting of stockholders by a stockholder pursuant to clause (i)(C) of paragraph (i) of this Section 2.10(a), the stockholder must have given

timely notice thereof in writing to the Secretary of the Corporation and any such proposed business (other than the nominations of persons for election to the Board of Directors) must constitute a proper matter for stockholder action. To be timely, a stockholder's notice shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not less than 90 days nor more than 120 days prior to the first anniversary of the preceding year's annual meeting of stockholders; *provided, however*, that in the event that the date of the annual meeting is advanced more than 30 days prior to such anniversary date or delayed more than 70 days after such anniversary date then to be timely such notice must be received by the Corporation no earlier than 120 days prior to such annual meeting and no later than the later of 70 days prior to the date of the meeting or the 10th day following the day on which public announcement of the date of the meeting was first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of an annual meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice relating to nominations of persons for election to the Board of Directors of the Corporation such annual meeting shall be timely if it is delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation no later than the later of (A) the latest date as determined in accordance with the second preceding sentence and (B) May 8, 2017.

(iii) A stockholder's notice to the Secretary shall set forth (A) as to each person whom the stockholder proposes to nominate for election or reelection as a director all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected, (B) as to any other business that the stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend these bylaws, the text of the proposed amendment), the reasons for conducting such business and any material interest in such business of such stockholder Associated Person (as defined below), if any, on whose behalf the proposal is made and (C) as to the stockholder giving the notice and the Stockholder forces in farming the proposal is made:

(1) the name and address of such stockholder (as they appear on the Corporation's books) and any such Stockholder Associated Person;

(2) the class or series and number of shares of capital stock of the Corporation which are held of record or are beneficially owned by such stockholder and by any such Stockholder Associated Person;

(3) a description of any agreement, arrangement or understanding between or among such stockholder and any such Stockholder Associated Person, any of their respective affiliates or associates, and any other person or persons (including their names) in connection with the proposal of such nomination or other business;

(4) a description of any agreement, arrangement or understanding (including, regardless of the form of settlement, any derivative, long or short positions, profit interests, forwards, futures, swaps, options, warrants, convertible securities, stock appreciation or similar rights, hedging transactions and borrowed or loaned shares) that has been entered into by or on behalf of, or any other agreement, arrangement or understanding that has been made, the effect or intent of which is to create or mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of, such stockholder or any such Stockholder Associated Person or any such nominee with respect to the Corporation's securities;

(5) a representation that the stockholder is a holder of record of stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to bring such nomination or other business before the meeting; and

(6) a representation as to whether such stockholder or any such Stockholder Associated Person intends or is part of a group that intends to (i) deliver a proxy statement and/or form of proxy to holders of at least the percentage of the voting power of the Corporation's outstanding capital stock required to approve or adopt the proposal or to elect each such nominee and/or (ii) otherwise to solicit proxies from stockholders in support of such proposal or nomination.

If requested by the Corporation, the information required under clauses 2.10(a)(iii)(C)(2), (3) and (4) of the preceding sentence of this Section 2.10 shall be supplemented by such stockholder and any such Stockholder Associated Person not later than 10 days after the record date for the meeting to disclose such information as of the record date.

(iv) For purposes of this Section 2.10, the term **'Stockholder Associated Person**' of any stockholder shall mean (A) any person controlling, directly or indirectly, or acting in concert with, such stockholder, (B) any beneficial owner of shares of stock of the Corporation owned of record or beneficially by such stockholder and (C) any person controlling, controlled by or under common control with such Stockholder Associated Person.

(b) Special Meetings of Stockholders. If the election of directors is included as business to be brought before a special meeting in the Corporation's notice of meeting, then nominations of persons for election to the Board of Directors of the Corporation at a special meeting of stockholders may be made by any stockholder who is a stockholder of record at the time of giving of notice provided for in this Section 2.10(b) and at the time of the special meeting, who shall be entitled to vote at the meeting and who complies with the procedures set forth in this Section 2.10(b). For nominations to be properly brought by a stockholder before a special meeting of stockholder's notice shall be delivered to or mailed and received at the principal executive offices of the Corporation (A) not earlier than 120 days prior to the date of the special meeting or the 10th day following the day on which public announcement of the date of the special meeting was first made. A stockholder's notice to the Secretary shall comply with the notice requirements of Section 2.10(a)(iii).

(c) General.

(i) At the request of the Board of Directors, any person nominated by the Board of Directors for election as a director shall furnish to the Secretary of the Corporation that is required to be set forth in a stockholder's notice of nomination that pertains to the nominee. No person shall be eligible to be nominated by a stockholder to serve as a director of the Corporation unless nominated in accordance with the procedures set forth in this Section 2.10. No business proposed by a stockholder shall be conducted at a stockholder meeting except in accordance with this Section 2.10. The chairman of the meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the procedures prescribed by these bylaws or that business was not properly brought before the meeting, and if he should so determine, he shall so declare to the meeting and the defective nomination shall be disregarded or such business shall not be transacted, as the case may be. Notwithstanding the foregoing provisions of this Section 2.10, unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the Corporation to present a nomination or other proposed business, such nomination shall be disregarded or such proposed business shall not be transacted, as the case may be, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 2.10, to be considered a qualified representative of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders.

(ii) Without limiting the foregoing provisions of this Section 2.10, a stockholder shall also comply with all applicable requirements of the Exchange Act, and the rules and regulations thereunder with respect to the matters set forth in this Section 2.10; *provided, however*, that any references in these bylaws to the Exchange Act or such rules and regulations are not intended to and shall not limit any requirements applicable to nominations or proposals as to any other business to be considered pursuant to this Section 2.10, and compliance with paragraphs (a)(i)(C) and (b) of this Section 2.10 shall be the exclusive means for a stockholder to make nominations or submit other business (other than as provided in paragraph 2.10(c)(iii)).

(d) *Proxy Access Rules.* Notwithstanding anything to the contrary, the notice requirements set forth herein with respect to director nominations or other business pursuant to this Section 2.10 shall be deemed satisfied by a stockholder if such stockholder has submitted a proposal to the Corporation, a nomination in compliance with Rule 14a-11 or a proposal in compliance with Rule 14a-8 under the Exchange Act, and such stockholder's nomination or proposal, as the case may be, has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for the meeting of stockholders.

ARTICLE 3 DIRECTORS

Section 3.01. *General Powers*. Except as otherwise provided in Delaware Law or the certificate of incorporation, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. All directors of the Corporation will be in compliance with all applicable requirements of the Communications Act of 1934, as amended, and the rules and policies of the Federal Communications Commission.

Section 3.02. Number, Election, Classes and Term of Office. The number of directors which shall constitute the Board of Directors shall initially be eight (8) and, thereafter, shall be fixed exclusively by one or more resolutions adopted from time to time by the affirmative vote of a majority of the Board of Directors. The directors shall be divided into three classes, designated Class I, Class II and Class III. Each class shall consist, as nearly as may be possible, of one-third of the total number of directors constituting the entire Board of Directors. Except as otherwise provided in the certificate of incorporation, each director shall serve for a term ending on the date of the third annual meeting of stockholders next following the annual meeting at which such director's earlier death, resignation or removal. Directors need not be stockholders.

Section 3.03. *Quorum and Manner of Acting.* Unless the certificate of incorporation or these bylaws require a greater number, a majority of the total number of directors shall constitute a quorum for the transaction of business, and, except as otherwise expressly required by law or by the certificate of incorporation, the act of a majority of the directors present at a meeting at which a quorum is present shall be the act of the Board of Directors. When a meeting is adjourned to another time or place (whether or not a quorum is present), notice need not be given of the adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the Board of Directors may transact any business which might have been transacted at the original meeting. If a quorum shall not be present at any meeting of the Board of Directors the directors present thereat shall adjourn the meeting, from time to time, without notice other than announcement at the meeting, until a quorum shall be present.

Section 3.04. *Time and Place of Meetings*. The Board of Directors shall hold its meetings at such place, either within or without the State of Delaware, and at such time as may be determined from time to time by the Board of Directors (or the Chairman of the Board in the absence of a determination by the Board of Directors).

Section 3.05. Annual Meeting. The Board of Directors shall meet for the purpose of organization, the election of officers and the transaction of other business, as soon as practicable after each annual meeting of stockholders, on the same day and at the same place where such annual meeting shall be held. Notice of such meeting need not be given. In the event such annual meeting is not so held, the annual meeting of the Board of Directors may be held at such place either within or without the State of Delaware, on such date and at such time as shall be specified in a notice thereof given as hereinafter provided in Section 3.07 herein or in a waiver of notice thereof signed by any director who chooses to waive the requirement of notice.

Section 3.06. *Regular Meetings*. After the place and time of regular meetings of the Board of Directors shall have been determined and notice thereof shall have been once given to each member of the Board of Directors, regular meetings may be held without further notice being given.

Section 3.07. *Special Meetings*. Special meetings of the Board of Directors may be called by the Chairman of the Board, the Lead Director designated in accordance with the Corporation's Corporate Governance Guidelines ("Lead Director") or the President and shall be called by the Chairman of the Board, Lead Director, President or Secretary on the written request of three directors. Notice of special meetings of the Board of Directors shall be given to each director at least 48 hours before the date of the meeting in such manner as is determined by the Board of Directors.

Section 3.08. *Committees.* The Board of Directors may designate one or more committees, each committee to consist of one or more of the directors of the Corporation. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to the following matters: (a) approving or adopting, or recommending to the stockholders, any action or matter expressly required by Delaware Law to be submitted to the stockholders for approval or (b) adopting, amending or repealing any bylaw of the Corporation. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required.

Section 3.09. Action by Consent. Unless otherwise restricted by the certificate of incorporation or these bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing or writings or electronic transmissions, are filed with the minutes of proceedings

of the Board or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

Section 3.10. *Telephonic Meetings*. Unless otherwise restricted by the certificate of incorporation or these bylaws, members of the Board of Directors, or any committee designated by the Board of Directors, may participate in a meeting of the Board of Directors, or such committee, as the case may be, by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such participation in a meeting shall constitute presence in person at the meeting.

Section 3.11. *Resignation*. Any director may resign at any time by giving notice in writing or by electronic transmission to the Board of Directors, the Chairman of the Board of Directors, the President or the Secretary of the Corporation. The resignation of any director shall take effect upon receipt of notice thereof or at such later time as shall be specified in such notice; and unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 3.12. *Vacancies*. Unless otherwise provided in the certificate of incorporation, vacancies on the Board of Directors resulting from death, resignation, removal or otherwise and newly created directorships resulting from any increase in the number of directors may be filled solely by a majority of the directors then in office (although less than a quorum) or by the sole remaining director, and each director so elected shall hold office for a term that shall coincide with the term of the class to which such director shall have been elected. If there are no directors in office, then an election of directors may be held in accordance with Delaware Law. Unless otherwise provided in the certificate of incorporation, when one or more directors shall resign from the Board, effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have the power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office as provided in the filling of the other vacancies.

Section 3.13. *Removal.* No director may be removed from office by the stockholders except for cause with the affirmative vote of the holders of not less than a majority of the total voting power of all outstanding securities of the Corporation then entitled to vote generally in the election of directors, voting together as a single class.

Section 3.14. Compensation. Unless otherwise restricted by the certificate of incorporation or these bylaws, the Board of Directors shall have authority to fix the compensation of directors, including fees and reimbursement of expenses.

Section 3.15. *Preferred Stock Directors*. Notwithstanding anything else contained herein, whenever the holders of one or more classes or series of preferred stock shall have the right, voting separately as a class or series, to elect directors, the election, term of office, filling of vacancies, removal and other features of such directorships shall be governed by the terms of the resolutions applicable thereto adopted by the Board of Directors pursuant to the certificate of incorporation, and such directors so elected shall not be subject to the provisions of Section 3.02, Section 3.12 and Section 3.13 of this Article 3 unless otherwise provided therein.

ARTICLE 4 OFFICERS

Section 4.01. *Principal Officers*. The principal officers of the Corporation shall be a President, a Chief Executive Officer, a Chief Financial Officer, one or more Vice Presidents, a Treasurer and a Secretary who shall have the duty, among other things, to record the proceedings of the meetings of stockholders and directors in a book kept for that purpose. The Board of Directors may, by resolution, designate the Chairman of the Board of Directors of the Corporation as a principal officer. The Corporation may also have such other principal officers, including one or more Controllers, as the Board of Directors may in its discretion appoint. One person may hold the offices and perform the duties of any two or more of said offices, except that no one person shall hold the offices and perform the duties of President and Secretary. All officers of the Corporation will be in compliance with all applicable requirements of the Communications Act of 1934, as amended, and the rules and policies of the Federal Communications Commission.

Section 4.02. Appointment, Term of Office and Remuneration. The principal officers of the Corporation shall be appointed by the Board of Directors in the manner determined by the Board of Directors. Each such officer shall hold office until his or her successor is appointed, or until his or her earlier death, resignation or removal. The remuneration of all officers of the Corporation shall be fixed by the Board of Directors. Any vacancy in any office shall be filled in such manner as the Board of Directors shall determine.

Section 4.03. Subordinate Officers. In addition to the principal officers enumerated in Section 4.01 herein, the Corporation may have one or more Assistant Treasurers, Assistant Secretaries and Assistant Controllers and such other subordinate officers, agents and employees as the Board of Directors may deem necessary, each of whom shall hold office for such period as the Board of Directors may from time to time determine. The Board of Directors may delegate to any principal officer the power to appoint and to remove any such subordinate officers, agents or employees.

Section 4.04. Removal. Except as otherwise permitted with respect to subordinate officers, any officer may be removed, with or without cause, at any time, by resolution adopted by the Board of Directors.

Section 4.05. *Resignations*. Any officer may resign at any time by giving written notice to the Board of Directors (or to a principal officer if the Board of Directors has delegated to such principal officer the power to appoint and to remove such officer). The resignation of any officer shall take effect upon receipt of notice thereof or at such later time as shall be specified in such notice; and unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 4.06. *Powers and Duties*. The officers of the Corporation shall have such powers and perform such duties incident to each of their respective offices and such other duties as may from time to time be conferred upon or assigned to them by the Board of Directors.

ARTICLE 5 CAPITAL STOCK

Section 5.01. *Certificates For Stock; Uncertificated Shares.* The shares of the Corporation shall be represented by certificates, provided that the Board of Directors of the Corporation may provide by resolution or resolutions that some or all of any or all classes or series of its stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the Corporation. Except as otherwise provided by law, the rights and obligations of the holders of uncertificated shares and the rights and obligations of the holders of shares represented by certificates shall be identical. Every holder of stock represented by certificates shall be entitled to have a certificate signed by, or in the name of the Corporation by the Chairman or Vice Chairman of the Board of Directors, or the President, and by the Chief Financial Officer, Treasurer or an assistant Treasurer, or the Secretary or an assistant Secretary of such Corporation representing the number of shares registered in certificate form. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar at the date of issue. A Corporation shall not have power to issue a certificate in bearer form.

Section 5.02. *Transfer Of Shares*. Shares of the stock of the Corporation may be transferred on the record of stockholders of the Corporation by the holder thereof or by such holder's duly authorized attorney upon surrender of a certificate therefor properly endorsed or upon receipt of proper transfer instructions from the registered holder of uncertificated shares or by such holder's duly authorized attorney and upon compliance with appropriate procedures for transferring shares in uncertificated form, unless waived by the Corporation.

Section 5.03. Authority for Additional Rules Regarding Transfer. The Board of Directors shall have the power and authority to make all such rules and regulations as they may deem expedient concerning the issue, transfer and registration of certificated or uncertificated shares of the stock of the Corporation, as well as for the issuance of new certificates in lieu of those which may be lost or destroyed, and may require of any stockholder requesting replacement of lost or destroyed certificates, bond in such amount and in such form as they may deem expedient to indemnify the Corporation, and/or the transfer agents, and/or the registrars of its stock against any claims arising in connection therewith.

ARTICLE 6 GENERAL PROVISIONS

Section 6.01. Fixing the Record Date.

(a) In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than 60 nor less than 10 days before the date of such meeting. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at

the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; *provided* that the Board of Directors may fix a new record date for the adjourned meeting.

(b) In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which date shall not be more than 10 days after the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which date shall not be more than 10 days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. If no record date has been fixed by the Board of Directors is required by Delaware Law, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by the Board of Directors and prior action by the Board of Directors is required by Delaware Law, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting the upon which the resolution's registered office shall be by hand or by certified or registered mail, return receipt requested. If no record date has been fixed by the Board of Directors and prior action by the Board of Directors is required by Delaware Law, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.

(c) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

Section 6.02. *Dividends*. Subject to limitations contained in Delaware Law and the certificate of incorporation, the Board of Directors may declare and pay dividends upon the shares of capital stock of the Corporation, which dividends may be paid either in cash, in property or in shares of the capital stock of the Corporation.

Section 6.03. Year. Except as otherwise determined by the Board of Directors, the fiscal year of the Corporation shall commence on February 1 and end on January 31 of each year.

Section 6.04. Corporate Seal. The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization and the words "Corporate Seal, Delaware". The seal may be used by causing it or a facsimile thereof to be impressed, affixed or otherwise reproduced.

Section 6.05. Voting of Stock Owned by the Corporation. The Board of Directors may authorize any person, on behalf of the Corporation, to attend, vote at and grant proxies to be used at any meeting of stockholders of any corporation (except this Corporation) in which the Corporation may hold stock.

Section 6.06. Amendments. These bylaws or any of them may be altered, amended or repealed, or new bylaws may be made, by the stockholders entitled to vote thereon at any annual or special meeting thereof or by the Board of Directors. Unless a higher percentage is required by the certificate of incorporation as to any matter that is the subject of these bylaws, all such amendments must be approved by the affirmative vote of the holders of not less than sixty-six and two-thirds percent (66-2/3%) of the total voting power of all outstanding securities of the Corporation then entitled to vote generally in the election of directors, voting together as a single class, or by a majority of the Board of Directors.

The undersigned hereby certifies that he is the duly elected, qualified, and acting Secretary of the Corporation, and that the foregoing Amended and Restated Bylaws are solely a restatement of the Amended and Restated Bylaws, as amended, that were adopted by the Corporation's Board of Directors effective as of September 21, 2017.

PANDORA MEDIA, INC.

By: /s/ Stephen Bené

Name:Stephen BenéTitle:General Counsel and Corporate Secretary

PANDORA MEDIA, INC.

2011 Equity Incentive Plan

NOTICE OF RESTRICTED STOCK UNIT GRANT

[Participant Name]

You have been granted the number of restricted stock units (the "<u>RSUs</u>"), each representing one share of Common Stock of Pandora Media, Inc. (the "<u>Company</u>") (the "<u>Shares</u>"), as follows:

Date of Grant: [Grant Date]	
Total Number of RSUs Granted:	[Number of Shares Granted]
Grant Number: [Client Grant ID]	
Vesting/Exercise Schedule:	So long as your Continuous Service Status continues, the RSUs shall vest in accordance with the following schedule:
Please refer to	Appendix attached to the Restricted Stock Unit Agreement attached hereto

By accepting these RSUs, you agree that these RSUs are granted under and governed by the terms and conditions of the Pandora Media, Inc. 2011 Equity Incentive Plan and the Restricted Stock Unit Agreement attached hereto and incorporated by reference herein.

In addition, you agree and acknowledge that your rights to any Shares underlying the RSUs will be earned only as you provide services to the Company over time and that nothing in this Notice or the attached documents confers upon you any right to continue your employment or consulting relationship with the Company for any period of time, nor does it interfere in any way with your right or the Company's right to terminate that relationship at any time, for any reason, with or without cause.

Title:

PANDORA MEDIA, INC.

[Signed Electronically]	By:
[Participant Name]	Name:

PANDORA MEDIA, INC.

2011 EQUITY INCENTIVE PLAN

RESTRICTED STOCK UNIT AGREEMENT

1. <u>Grant of RSUs</u>. Pandora Media, Inc., a Delaware corporation (the "<u>Company</u>"), hereby grants to you ("<u>Participant</u>") the number of RSUs (each representing a share of Common Stock of the Company) set forth in the Notice of Restricted Stock Unit Grant (the "<u>Notice</u>"), subject to the terms, definitions and provisions of the Pandora Media, Inc. 2011 Equity Incentive Plan (the "<u>Plan</u>") adopted by the Company, which is incorporated in this Agreement by reference. Unless otherwise defined in this Agreement, the terms used in this Agreement shall have the meanings defined in the Notice or the Plan, as applicable. The terms and conditions of this Restricted Stock Unit Award Agreement (this "<u>Agreement</u>"), to the extent not controlled by the terms and conditions contained in the Plan, are as follows:

1. *Vesting*. The RSUs shall become vested on the vesting schedule set forth in the Appendix attached hereto, subject to Participant remaining in Continuous Service Status on the applicable vesting date.

2. *Forfeiture of Unvested RSUs*. Except as set forth in the Appendix attached hereto, immediately upon termination of Participant's employment (including due to death or disability), any unvested RSUs shall be forfeited without consideration.

3. Conversion into Common Stock. Shares will be issued on the applicable vesting date (or, to the extent not administratively feasible, within 30 days thereafter). As a condition to such issuance, Participant shall have satisfied his or her tax withholding obligations as specified in this Agreement and shall have completed, signed and returned any documents and taken any additional action that the Company deems appropriate to enable it to accomplish the delivery of the Shares. In no event will the Company be obligated to issue a fractional share. Notwithstanding the foregoing, (i) the Company shall not be obligated to deliver any Shares during any period when the Company determines that the conversion of a RSU or the delivery of shares hereunder would violate any federal, state or other applicable laws and/or may issue shares subject to any restrictive legends that, as determined by the Company's counsel, is necessary to comply with securities or other regulatory requirements, and (ii) the date on which shares are issued may include a delay in order to provide the Company such time as it determines appropriate to address tax withholding and other administrative matters.

4. *Tax Treatment*. Any withholding tax liabilities (whether as a result of federal, state or other law and whether for the payment and satisfaction of any income tax, social security tax, payroll tax, or payment on account of other tax related to withholding obligations that arise by reason of the RSUs) incurred in connection with the RSUs becoming vested and Shares issued, or otherwise incurred in connection with the RSUs, may be satisfied in any of the following manners determined by the Participant (and the Committee may with prior written notice to Participant require any of the following methods): (i) by the sale by Participant of a number of

2

Shares that are issued under the RSUs, which the Company determines is sufficient to generate an amount that meets the tax withholding obligations plus additional Shares to account for rounding and market fluctuations, and payment of such tax withholding to the Company, and such Shares may be sold as part of a block trade with other participants of the Plan; (ii) by the Company withholding a number of Shares that would otherwise be issued under the RSUs that the Company determines have a fair market value equal to the minimum amount of taxes that the Company concludes it is required to withhold under applicable law; or (iii) by payment by Participant to the Company in cash or by check an amount equal to the minimum amount of taxes that the Company concludes it is required to withhold such tax withholding amount from any amounts owing to Participant to the Company and to take any action necessary in accordance with this paragraph.

Notwithstanding the foregoing, Participant acknowledges and agrees that he is responsible for all taxes that arise in connection with the RSUs becoming vested and Shares being issued or otherwise incurred in connection with the RSUs, regardless of any action the Company takes pursuant to this Section.

5. *Restrictions on Transfer*. Participant understands and agrees that the RSUs may not be sold, given, transferred, assigned, pledged or otherwise hypothecated by the holder.

6. *Certificates*. Certificates, transfer agent book entries or other evidence of ownership as determined by the Company issued in respect of the Shares shall, unless the Committee otherwise determines, be registered in the name of Participant. The stock certificate, if any, shall carry such appropriate legends, and such written instructions shall be given to the Company transfer agent, as may be deemed necessary or advisable by counsel to the Company in order to comply with the requirements of the Securities Act of 1933, any state securities laws or any other applicable laws.

7. *No Stockholder Rights*. Participant will have no voting or other rights as the Company's other stockholders with respect to the Shares until issuance of the Shares.

8. *No Employment/Service Rights*. Neither this Agreement nor the grant of the RSUs hereby confers on Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes in any way with the right of the Company or any Subsidiary to determine the terms of Participant's employment or service.

9. *Entire Agreement; Terms of Plan, Interpretations.* Participant acknowledges that he has received and reviewed a copy of the Plan. This Agreement (including the Notice and the Appendix attached hereto) contains the entire understanding of the parties hereto in respect of the subject matter contained herein. This Agreement together with the Plan supersedes all prior agreements and understandings between the parties hereto with respect to the subject matter hereof. This Agreement and the terms and conditions herein set forth are subject in all respects to the terms and conditions of the Plan, which shall be controlling. All interpretations or determinations of the Committee and/or the Board shall be binding and conclusive upon Participant and his legal representatives on any question arising hereunder.

3

Appendix: Vesting Schedule

- [•]% of the RSUs set forth in the Notice shall vest on [•] and [•]% of the RSUs set forth in the Notice shall vest on [•], subject to the Participant remaining in Continuous Service Status on the applicable vesting date.
- In the case of an Involuntary Termination (as defined in the Company's Executive Severance Change of Control Policy, as amended, (the "*Policy*")) of Participant, the RSUs set forth in the Notice shall fully vest on the effective date of the Release (as defined in the Policy), and such acceleration shall supersede and replace any other acceleration benefit with respect to the RSUs set forth in the Notice that would have been provided under the Policy in the case of an Involuntary Termination. The RSUs that vest pursuant to this provision shall be settled within 30 days following Participant's Involuntary Termination.
- In the case of a voluntary termination for Good Reason (as defined below) (and regardless of whether or not a Change of Control (as defined in the Company's 2011 Equity Incentive Plan) precedes said voluntary termination) of Participant, the RSUs set forth in the Notice shall fully vest on the effective date of the Release (as defined in the Policy), and such acceleration shall supersede and replace any other acceleration benefit with respect to the RSUs set forth in the Notice that would have been provided under the Policy in the case of a voluntary termination for Good Reason. The RSUs that vest pursuant to this provision shall be settled within 30 days following Participant's termination of employment for Good Reason.
 - O "Good Reason" means Participant's resignation from employment after the occurrence of one of the following events without Participant's consent: (A) a material reduction of Participant's base salary or target annual incentive bonus; (B) any requirement by the Company (or its successor) that Participant engage in any illegal or unethical conduct, after Participant has given the Company thirty (30) days' notice and opportunity to cure; (C) the Company's failure to fully cure within thirty (30) days any material breach by the Company of the Policy or of any other material agreement between Participant and the Company, in each case which Participant has notified the Committee in writing; or (D) a relocation of Participant's principal place of employment by more than fifty (50) miles; *provided* that in any event, Participant notifies the Company of the event constituting Good Reason and gives the Company thirty (30) days to cure (to the extent capable of cure), and then Participant resigns within thirty (30) days thereafter.
- The RSUs are intended to be exempt from or comply with Section 409A of the Code and shall be interpreted and construed accordingly, and each vesting hereunder shall be considered a separate payment. Notwithstanding any other provision in the RSU Agreement, to the extent the RSUs constitute nonqualified deferred compensation, within the meaning of Section 409A, (i) if Participant is a specified employee (within the meaning of Section 409A of the Code) as of the date of Participant's "separation from service" (within the meaning of Section 409A of the Code), then the RSUs that vest upon Participant's separation from service shall be settled on the first business day after the date that is six months following Participant's separation from service or Participant's death, if earlier, and (ii) if the RSUs are conditioned upon the effectiveness of the Release and are to be settled during a designated period that begins in one taxable year and ends in a second taxable year, such RSUs shall be settled in the later of the two taxable years.



CREDIT AGREEMENT

by and among

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent,

WELLS FARGO BANK, NATIONAL ASSOCIATION, JPMORGAN CHASE BANK, N.A. AND MORGAN STANLEY SENIOR FUNDING, INC.,

as Joint Lead Arrangers,

WELLS FARGO BANK, NATIONAL ASSOCIATION, JPMORGAN CHASE BANK, N.A. AND MORGAN STANLEY SENIOR FUNDING, INC.,

as Joint Book Runners,

THE LENDERS THAT ARE PARTIES HERETO

as the Lenders, and

PANDORA MEDIA, INC. and

PANDORA MEDIA CALIFORNIA, LLC,

as Borrowers

Dated as of December 29, 2017

CONSTRUCTION 1	1.DEFINITIONS AND
	1.1Definitions 1
	<u>1.2</u> Accounting Terms 46
	<u>1.3</u> Code 46
	<u>1.4</u> Construction 46
	<u>1.5</u> Time References 47
	<u>1.6</u> Schedules and Exhibits 47
	2.LOANS AND TERMS OF
PAYMENT 47	
	2.1 Revolving Loans 47
	2.2[Intentionally Omitted] 48
Settlements 48	2.3Borrowing Procedures and
Commitments; Prepayments 55	2.4 Payments; Reductions of
	2.5 Promise to Pay; Promissory
Notes 59	
Fee: Rates, Payments, and Calculations 60	2.6 Interest Rates and Letter of Credit
ree. Rates, rayments, and Calculations 00	2.7 Crediting Payments 62
	2.8 Designated Account 62
	<u>2.9</u> Maintenance of Loan Account;
Statements of Obligations 62	
	<u>2.10</u> Fees 62
	2.11 Letters of Credit 63
	2.12LIBOR Option 71
	2.13 Capital Requirements 73
	2.14 Incremental Facilities 74
	2.15 Joint and Several Liability of
Borrowers 76	2 CONDITIONS, TEDM OF
AGREEMENT 79	<u>3.</u> CONDITIONS; TERM OF
Extension of Credit 79	3.1 Conditions Precedent to the Initial
Extensions of Credit 79	<u>3.2</u> Conditions Precedent to all
	<u>3.3</u> Maturity 79
	<u>3.4</u> Effect of Maturity 79
	<u>3.5</u> Early Termination by
Borrowers 80	
	<u>3.6</u> Conditions Subsequent 80
WARRANTIES 80	4.REPRESENTATIONS AND
	4.1 Due Organization and Qualification;
Subsidiaries 80	
	<u>4.2</u> Due Authorization; No
Conflict 81	
	<u>4.3</u> Governmental Consents 81

Encumbrances 82

4.5 Title to Assets; No

4.6Litigation 82

(continued)

Page

4.7 Compliance with Laws 82 4.8No Material Adverse Effect 82 <u>4.9</u>Solvency 82 4.10 Employee Benefits 83 4.11Environmental Condition 83 4.12Complete Disclosure 83 4.13Patriot Act 83 4.14Indebtedness 84 4.15 Payment of Taxes 84 4.16 Margin Stock 84 4.17 Governmental Regulation 84 4.18OFAC; Sanctions; Anti-Corruption 4.19 Employee and Labor Matters 84 4.20 Material Contracts 85 <u>4.21</u>Leases 85 4.22Eligible Accounts 85 4.23 Intellectual Property 85 4.24Music Performance Rights 85 4.25Location of Books and 4.26 Inventory Records 85 4.27 FCC Licenses 86 4.28 Material Subsidiaries 86 4.29 Hedge Agreements 86 5.AFFIRMATIVE COVENANTS 86 5.1 Financial Statements, Reports, 5.2 Reporting 86 5.3 Existence 86 5.4 Maintenance of Properties 86

5.5 Taxes 86 5.6Insurance 86 5.7 Inspection 87 5.8 Compliance with Laws 87 5.9 Environmental 88 5.10Disclosure Updates 88 5.11 Formation of Subsidiaries 88 5.12Further Assurances 89

5.13Material Contracts 89

Laws; Anti-Money Laundering Laws 84

Records 85

Certificates 86

(continued)

Page

Chief Executive Office 89	5.14 Location of Books and Records;
Laws; Anti-Money Laundering Laws 89	5.15OFAC; Sanctions; Anti-Corruption
	6.NEGATIVE COVENANTS 89
	<u>6.1</u> Indebtedness 90
	<u>6.2</u> Liens 90
Changes 90	6.3 Restrictions on Fundamental
	6.4 Disposal of Assets 90
	<u>6.5</u> Nature of Business 90
	6.6 Prepayments and Amendments 90
	6.7 Restricted Payments 91
	<u>6.8</u> Accounting Methods 92
	<u>6.9</u> Investments 92
	<u>6.10</u> Transactions with Affiliates 92
	<u>6.11</u> Use of Proceeds 93
Interests 93	6.12Limitation on Issuance of Equity
	6.13 Inventory with Bailees 94
	<u>6.14</u> Intellectual Property 94
	6.15 Agency Relationships 94
	7.FINANCIAL COVENANTS 94
	8.EVENTS OF DEFAULT 94
	<u>8.1</u> Payments 94
	8.2Covenants 94
	8.3 Judgments 95
	8.4 Voluntary Bankruptcy, etc 95
	<u>8.5</u> Involuntary Bankruptcy, etc 95
	8.6Default Under Other
Agreements 95	
	<u>8.7</u> Representations, etc 95
	<u>8.8</u> Guaranty 95
	<u>8.9</u> Security Documents 95
	8.10 Loan Documents 96
	<u>8.11</u> Change of Control 96
Programming. 96	8.12Material Cessation of
Licenses 96	8.13 Termination of Material
	9. RIGHTS AND REMEDIES 96
	9.1 Rights and Remedies 96
	9.2 Remedies Cumulative 97
	<u>10.</u> WAIVERS;
INDEMNIFICATION 07	

(continued)

Page

	10.1 Demand; Protest; etc 97
	10.2 The Lender Group's Liability for
Collateral 97	
	<u>10.3</u> Indemnification 97
CHOICE OF LAW AND VENUE, HUNY TRIAL WARREN, HUNGLAD	<u>11.</u> NOTICES 98
CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE	
	PROVISION 99
	13.ASSIGNMENTS AND
PARTICIPATIONS; SUCCESSORS 102	
Participations 102	<u>13.1</u> Assignments and
	<u>13.2</u> Successors 106
	<u>14.</u> AMENDMENTS; WAIVERS 106
	14.1 Amendments and Waivers 106
	<u>14.2</u> Replacement of Certain
Lenders 108	
Remedies 108	<u>14.3</u> No Waivers; Cumulative
	15.AGENT; THE LENDER
GROUP 108	
A	<u>15.1</u> Appointment and Authorization of
Agent 108	15.2Delegation of Duties 109
	<u>15.3</u> Liability of Agent 109
	<u>15.4</u> Reliance by Agent 110
	<u>15.5</u> Notice of Default or Event of
Default 110	
	15.6Credit Decision 110
Indemnification 111	<u>15.7</u> Costs and Expenses;
	15.8 Agent in Individual Capacity 111
	<u>15.9</u> Successor Agent 112
	<u>15.10</u> Lender in Individual
Capacity 112	
	15.11 Collateral Matters 112
Lenders; Sharing of Payments 114	<u>15.12</u> Restrictions on Actions by
Lenders, Sharing of Layments 114	<u>15.13</u> Agency for Perfection 115
	<u>15.14</u> Payments by Agent to the
Lenders 115	<u></u>
Balatad Laan Daaumanta 115	15.15 Concerning the Collateral and
Related Loan Documents 115	15.16 Field Examination Reports;
Confidentiality; Disclaimers by Lenders; Other Reports and	<u>13.10</u> 1 for Examination Reports,
	Information 115
	15.17 Several Obligations; No
Liability 116	

<u>16.</u>WITHHOLDING TAXES 116

<u>16.1</u>Payments 116

<u>16.2</u>Exemptions 117

<u>16.3</u>Reductions 119

(continued)

Page

	<u>16.4</u> Refunds 120
	17.GENERAL PROVISIONS 120
	<u>17.1</u> Effectiveness 120
	<u>17.2</u> Section Headings 120
	<u>17.3</u> Interpretation 120
	<u>17.4</u> Severability of Provisions 120
	<u>17.5</u> Bank Product Providers 120
Relationship 121	<u>17.6</u> Debtor-Creditor
Execution 121	17.7 Counterparts; Electronic
Obligations; Certain Waivers 121	17.8 Revival and Reinstatement of
	<u>17.9</u> Confidentiality 122
	<u>17.10</u> Survival 124
	<u>17.11</u> Patriot Act; Due Diligence 124
	17.12Integration 124
	17.13 Pandora as Agent for
Borrowers 124	
to Bail-In of EEA Financial Institutions 125	<u>17.14</u> Acknowledgement and Consent

-v-

EXHIBITS AND SCHEDULES

- Exhibit A-1 Form of Assignment and Acceptance Exhibit B-1 Form of Borrowing Base Certificate Exhibit B-2 Form of Bank Product Provider Agreement Exhibit C-1 Form of Compliance Certificate Exhibit J-1 Form of Joinder Exhibit L-1 Form of LIBOR Notice Exhibit P-1 Form of Perfection Certificate Schedule A-1 Agent's Account Schedule A-2 Authorized Persons Schedule C-1 Commitments Schedule D-1 Designated Account Schedule P-1 Permitted Investments Schedule P-2 Permitted Liens Schedule 3.1 Conditions Precedent Schedule 3.6 Conditions Subsequent Schedule 4.1(b) Capitalization of Borrowers Schedule 4.1(c) Capitalization of Borrowers' Subsidiaries Schedule 4.1(d) Subscriptions, Options, Warrants, Calls Schedule 4.6 Litigation Schedule 4.11 Environmental Matters Schedule 4.14 Permitted Indebtedness Schedule 4.20 Material Contracts Schedule 4.25 Location of Books and Records Schedule 5.1 Financial Statements, Reports, Certificates Collateral Reporting Schedule 5.2
- Schedule 6.5 Nature of Business

-vi-

CREDIT AGREEMENT

THIS CREDIT AGREEMENT, is entered into as of December 29, 2017, by and among the lenders identified on the signature pages hereof (each of such lenders, together with its successors and permitted assigns, is referred to hereinafter as a "Lender", as that term is hereinafter further defined), WELLS FARGO BANK, NATIONAL ASSOCIATION, a national banking association ("WFB"), as administrative agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and assigns in such capacity, "Agent"), JPMORGAN CHASE BANK, N.A. ("JPM"), MORGAN STANLEY SENIOR FUNDING, INC. ("MSSE"), and WFB, as joint lead arrangers (in such capacities, together with their respective successors and assigns in such capacity, individually, each, a "Joint Lead Arranger" and collectively, the "Joint Lead Arrangers"), JPM, MSSF, and WFB, as joint book runners (in such capacities, together with their respective successors and assigns in such capacity, individually, each, a "Joint Lead Arranger" and collectively, the "Joint Book Runners"), PANDORA MEDIA, INC., a Delaware corporation ("Pandora"), the Subsidiaries of Pandora identified on the signature pages hereof as "Borrowers", and those additional entities that hereafter become parties hereto as Borrowers in accordance with the terms hereof by executing the form of Joinder attached hereto as Exhibit J-1 (together with Pandora, each, a "Borrower" and individually and collectively, jointly and severally, the "Borrowers").

The parties agree as follows:

1.DEFINITIONS AND CONSTRUCTION.

1.1 **Definitions**. As used in this Agreement, the following terms shall have the following definitions:

"Account" means an account (as that term is defined in the Code).

"Account Debtor" means any Person who is obligated on an Account, chattel paper, or a general intangible.

"Account Party" has the meaning specified therefor in Section 2.11(h) of this Agreement.

"Accounting Changes" means changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants (or successor thereto or any agency with similar functions).

"Acquired Indebtedness" means Indebtedness of a Person whose assets or Equity Interests are acquired by a Loan Party or any of its Subsidiaries in a Permitted Acquisition; provided, that such Indebtedness (a) is either purchase money Indebtedness or a Capital Lease with respect to Equipment or mortgage financing with respect to Real Property, (b) was in existence prior to the date of such Permitted Acquisition, and (c) was not incurred in connection with, or in contemplation of, such Permitted Acquisition.

"Acquisition" means (a) the purchase or other acquisition by a Person or its Subsidiaries of all or substantially all of the assets of (or any division or business line of) any other Person, or (b) the purchase or other acquisition (whether by means of a merger, consolidation, or otherwise) by a Person or its Subsidiaries of all of the Equity Interests of any other Person.

"Additional Documents" has the meaning specified therefor in Section 5.12 of this Agreement.

"Administrative Borrower" has the meaning specified therefor in Section 17.13 of this Agreement.

"Administrative Questionnaire" has the meaning specified therefor in Section 13.1(a) of this Agreement.

"Affected Lender" has the meaning specified therefor in Section 2.13(b) of this Agreement.

"Affiliate" means, as applied to any Person, any other Person who controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" means the possession, directly or indirectly through one or more intermediaries, of the power to direct the management and policies of a Person, whether through the ownership of Equity Interests, by contract, or otherwise; provided, that for purposes of the definition of Eligible Accounts and <u>Section 6.10</u> of this Agreement: (a) if any Person owns directly or indirectly 10% or more of the Equity Interests having ordinary voting power for the election of directors or other members of the governing body of a Person or 10% or more of the partnership interests of a Person (other than as a limited partner of such Person), then both such Persons shall be Affiliates of each other, (b) each director (or comparable manager) of a Person shall be deemed to be an Affiliate of such Person, and (c) each partnership in which a Person is a general partner shall be deemed an Affiliate of such Person.

"Agent" has the meaning specified therefor in the preamble to this Agreement.

"Agent-Related Persons" means Agent, together with its Affiliates, officers, directors, employees, attorneys, and agents.

"Agent's Account" means the Deposit Account of Agent identified on <u>Schedule A-1</u> to this Agreement (or such other Deposit Account of Agent that has been designated as such, in writing, by Agent to Borrowers and the Lenders).

"Agent's Liens" means the Liens granted by each Loan Party or its Subsidiaries to Agent under the Loan Documents and securing the Obligations.

"Agreement" means this Credit Agreement, as amended, restated, amended and restated, supplemented or otherwise modified from time to time.

"Anti-Corruption Laws" means the FCPA, the U.K. Bribery Act of 2010, as amended, and all other applicable laws and regulations or ordinances concerning or relating to bribery, money laundering or corruption in any jurisdiction in which any Loan Party or any of its Subsidiaries or Affiliates is located or is doing business.

"Anti-Money Laundering Laws" means the applicable laws or regulations in any jurisdiction in which any Loan Party or any of its Subsidiaries or Affiliates is located or is doing business that relates to money laundering, any predicate crime to money laundering, or any financial record keeping and reporting requirements related thereto.

"<u>Applicable Margin</u>" means, as of any date of determination and with respect to Base Rate Loans or LIBOR Rate Loans, as applicable, the applicable margin set forth in the following table that corresponds to the Average Excess Availability of Borrowers for the most recently completed quarter; <u>provided</u>, that for the period from the Closing Date through and including December 31, 2017, the Applicable Margin shall be set at the margin in the row styled "Level II":

-2-

<u>Level</u>	Average Excess Availability	Applicable Margin for Base Rate Loans (the "Base Rate Margin")	Applicable Margin for LIBOR Rate Loans (the "LIBOR Rate Margin")
Ι	> 66% of the Maximum Revolver Amount	0.50 percentage points	1.50 percentage points
Π	\leq 66% of the Maximum Revolver Amount and > 33% of the Maximum Revolver Amount		1.75 percentage points
Ш	\leq 33% of the Maximum Revolver Amount	1.00 percentage points	2.00 percentage points

The Applicable Margin shall be re-determined as of the first day of each quarter.

"Applicable Unused Line Fee Percentage" means, as of any date of determination, the applicable percentage set forth in the following table that corresponds to the Average Revolver Usage of Borrowers for the most recently completed month as determined by Agent in its Permitted Discretion; provided, that for the period from the Closing Date through and including December 31, 2017, the Applicable Unused Line Fee Percentage shall be set at the rate in the row styled "Level II":

Level	Average Revolver Usage	Applicable Unused Line Fee Percentage
Ι	> 50% of the Maximum Revolver Amount	0.25 percentage points
П	\leq 50% of the Maximum Revolver Amount	0.375 percentage points

The Applicable Unused Line Fee Percentage shall be re-determined on the first date of each month by Agent.

"Application Event" means the occurrence of (a) a failure by Borrowers to repay all of the Obligations in full on the Maturity Date, or (b) an Event of Default and the election by Agent or the Required Lenders to require that payments and proceeds of Collateral be applied pursuant to Section 2.4(b)(iii) of this Agreement.

"Assignee" has the meaning specified therefor in Section 13.1(a) of this Agreement.

"Assignment and Acceptance" means an Assignment and Acceptance Agreement substantially in the form of Exhibit A-1 to this Agreement.

"Authorized Person" means any one of the individuals identified as an officer of a Borrower on <u>Schedule A-2</u> to this Agreement, or any other individual identified by Administrative Borrower as an authorized person and authenticated through Agent's electronic platform or portal in accordance with its procedures for such authentication.

"<u>Availability</u>" means, as of any date of determination, the amount that Borrowers are entitled to borrow as Revolving Loans under <u>Section 2.1</u> of this Agreement (after giving effect to the then outstanding Revolver Usage).

-3-

"Available Increase Amount" means, as of any date of determination, an amount equal to the result of (a) \$50,000,000, <u>minus</u> (b) the aggregate principal amount of Increases to the Revolver Commitments previously made pursuant to <u>Section 2.14</u> of this Agreement.

"Average Excess Availability " means, with respect to any period, the sum of the aggregate amount of Excess Availability for each day in such period (as calculated by Agent as of the end of each respective day) divided by the number of days in such period.

"Average Revolver Usage" means, with respect to any period, the sum of the aggregate amount of Revolver Usage for each day in such period (calculated as of the end of each respective day) divided by the number of days in such period.

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"<u>Bail-In Legislation</u>" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Bank Product" means any one or more of the following financial products or accommodations extended to any Loan Party by a Bank Product Provider: (a) credit cards (including commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards")), (b) payment card processing services, (c) debit cards, (d) stored value cards, (e) Cash Management Services, or (f) transactions under Hedge Agreements.

"Bank Product Agreements" means those agreements entered into from time to time by any Loan Party with a Bank Product Provider in connection with the obtaining of any of the Bank Products.

"Bank Product Collateralization" means providing cash collateral (pursuant to documentation reasonably satisfactory to Agent) to be held by Agent for the benefit of the Bank Product Providers (other than the Hedge Providers) in an amount determined by Agent as sufficient to satisfy the reasonably estimated credit exposure, operational risk or processing risk with respect to the then existing Bank Product Obligations (other than Hedge Obligations).

"Bank Product Obligations" means (a) all obligations, liabilities, reimbursement obligations, fees, or expenses owing by each Loan Party and its Subsidiaries to any Bank Product Provider pursuant to or evidenced by a Bank Product Agreement and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, (b) all Hedge Obligations, and (c) all amounts that Agent or any Lender is obligated to pay to a Bank Product Provider as a result of Agent or such Lender purchasing participations from, or executing guarantees or indemnities or reimbursement obligations to, a Bank Product Provider with respect to the Bank Products provided by such Bank Product Provider to a Loan Party or its Subsidiaries.

"Bank Product Provider" means any Lender or any of its Affiliates, including each of the foregoing in its capacity, if applicable, as a Hedge Provider, provided, that no such Person (other than Wells Fargo or its Affiliates) shall constitute a Bank Product Provider with respect to a Bank Product unless and until Agent receives a Bank Product Provider Agreement from such Person (a) on or prior to the Closing Date (or such later date as Agent shall agree to in writing in its sole discretion) with respect to Bank Products provided on or prior to the Closing Date, or (b) on or prior to the date that is 10 days after the provision of such Bank Product to a Loan Party or its Subsidiaries (or such later date as Agent shall agree to in writing in its sole discretion) with respect to Bank Products provided after the Closing Date; provided further, that if, at any time, a Lender ceases to be a Lender under this Agreement (prior to the payment in full of the Obligations), then, from and after the



date on which it so ceases to be a Lender thereunder, neither it nor any of its Affiliates shall constitute Bank Product Providers and the obligations with respect to Bank Products provided by such former Lender or any of its Affiliates shall no longer constitute Bank Product Obligations.

"Bank Product Provider Agreement," means an agreement in substantially the form attached hereto as Exhibit B-2 to this Agreement, in form and substance satisfactory to Agent, duly executed by the applicable Bank Product Provider, the applicable Loan Parties, and Agent.

"Bank Product Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate to establish (based upon the Bank Product Providers' determination of the liabilities and obligations of each Loan Party and its Subsidiaries in respect of Bank Product Obligations) in respect of Bank Products then provided or outstanding.

"Bankruptcy Code" means title 11 of the United States Code, as in effect from time to time.

"Base Rate" means the greatest of (a) the Federal Funds Rate <u>plus</u> $\frac{1}{2}$ %, (b) the LIBOR Rate (which rate shall be calculated based upon an Interest Period of one month and shall be determined on a daily basis), <u>plus</u> one percentage point, and (c) the rate of interest announced, from time to time, within Wells Fargo at its principal office in San Francisco as its "prime rate", with the understanding that the "prime rate" is one of Wells Fargo's base rates (not necessarily the lowest of such rates) and serves as the basis upon which effective rates of interest are calculated for those loans making reference thereto and is evidenced by the recording thereof after its announcement in such internal publications as Wells Fargo may designate (and, if any such announced rate is below zero, then the rate determined pursuant to this clause (d) shall be deemed to be zero).

"Base Rate Loan" means each portion of the Revolving Loans that bears interest at a rate determined by reference to the Base Rate.

"Base Rate Margin" has the meaning set forth in the definition of "Applicable Margin".

"Benefit Plan" means a "defined benefit plan" (as defined in Section 3(35) of ERISA) for which any Loan Party or any of its Subsidiaries or ERISA Affiliates has been an "employer" (as defined in Section 3(5) of ERISA) within the past six years.

"Board of Directors" means, as to any Person, the board of directors (or comparable managers) of such Person, or any committee thereof duly authorized to act on behalf of the board of directors (or comparable managers).

"Board of Governors" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" and "Borrowers" have the respective meanings specified therefor in the preamble to this Agreement.

"Borrower Materials" has the meaning specified therefor in Section 17.9(c) of this Agreement.

"Borrowing" means a borrowing consisting of Revolving Loans made on the same day by the Lenders (or Agent on behalf thereof), or by Swing Lender in the case of a Swing Loan, or by Agent in the case of an Extraordinary Advance.

"Borrowing Base" means, as of any date of determination, the result of:

(a) 85% of the amount of Eligible Advertising Accounts, *less* the amount, if any, of the Dilution Reserve, <u>plus</u>

(b) The lesser of (i) \$15,000,000, (ii) 50% of the amount of Eligible Subscription Accounts, and (iii) 10% of the Line Cap, <u>less</u> the amount, if any, of the Dilution Reserve, less

(c) the aggregate amount of Reserves, if any, established by Agent from time to time under Section 2.1(c) of this Agreement.

"Borrowing Base Certificate" means a certificate in the form of Exhibit B-1 to this Agreement.

"Business Day" means any day that is not a Saturday, Sunday, or other day on which banks are authorized or required to close in the state of California, except that, if a determination of a Business Day shall relate to a LIBOR Rate Loan, the term "Business Day" also shall exclude any day on which banks are closed for dealings in Dollar deposits in the London interbank market.

"<u>Capital Expenditures</u>" means, with respect to any Person for any period, the amount of all expenditures by such Person and its Subsidiaries during such period that are capital expenditures as determined in accordance with GAAP, whether such expenditures are paid in cash or financed, but excluding, without duplication (a) with respect to the purchase price of assets that are purchased substantially contemporaneously with the trade-in of existing assets during such period, the amount that the gross amount of such purchase price is reduced by the credit granted by the seller of such assets for the assets being traded in at such time, and (b) expenditures made during such period to consummate one or more Permitted Acquisitions.

GAAP.

"Capitalized Lease Obligation" means that portion of the obligations under a Capital Lease that is required to be capitalized in accordance with

GAAr.

"Capital Lease" means a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

"<u>Cash Equivalents</u>" means (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within one year from the date of acquisition thereof, (b) marketable direct obligations issued or fully guaranteed by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within one year from the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from either Standard & Poor's Rating Group ("<u>S&P</u>") or Moody's Investors Service, Inc. ("<u>Moody's</u>"), (c) commercial paper maturing no more than 270 days from the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's, (d) certificates of deposit, time deposits, overnight bank deposits or bankers' acceptances maturing within one year from the date of acquisition thereof or any state thereof or the District of Columbia or any United States branch of a foreign bank having at the date of acquisition thereof capital and surplus of not less than \$1,000,000,000, (e) Deposit Accounts maintained with (i) any bank that satisfies the criteria described in clause (d) above, or (ii) any other bank organized under the laws of the United States or any state thereof so long as the full amount maintained with any such other bank is insured by the Federal Deposit Insurance Corporation, (f) repurchase obligations of any commercial bank satisfying the criteria described in clauses (a) or (d) above, (g) debt securities with maturities of six months or less from the date of acquisition backed by standby letters of credit issued by any commercial bank satisfying the criteria described in clauses (a) through (g) above.

-6-

"<u>Cash Management Services</u>" means any cash management or related services including treasury, depository, return items, overdraft, controlled disbursement, merchant store value cards, e-payables services, electronic funds transfer, interstate depository network, automatic clearing house transfer (including the Automated Clearing House processing of electronic funds transfers through the direct Federal Reserve Fedline system) and other cash management arrangements.

"<u>CFC</u>" means a controlled foreign corporation (as that term is defined in the IRC) in which any Loan Party is a "United States shareholder" within the meaning of Section 951(b) of the IRC.

"<u>Change of Control</u>" means that:

(a) any Person or two or more Persons acting in concert (other than Permitted Holders), shall have acquired beneficial ownership, directly or indirectly, of Equity Interests of Administrative Borrower (or other securities convertible into such Equity Interests) representing 35% or more of the combined voting power of all Equity Interests of Administrative Borrower entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Administrative Borrower,

(b) any Person or two or more Persons acting in concert (other than Permitted Holders), shall have acquired by contract or otherwise, or shall have entered into a contract or arrangement that, upon consummation thereof, will result in its or their acquisition of the power to exercise, directly or indirectly, a controlling influence over the management or policies of Administrative Borrower or control over the Equity Interests of such Person entitled to vote for members of the Board of Directors of Administrative Borrower on a fully-diluted basis (and taking into account all such Equity Interests that such Person or group has the right to acquire pursuant to any option right) representing 35% or more of the combined voting power of such Equity Interests,

(c) during any period of 24 consecutive months commencing on or after the Closing Date, the occurrence of a change in the composition of the Board of Directors of Administrative Borrower such that a majority of the members of such Board of Directors are not Continuing Directors,

(d) Borrowers fail to own and control, directly or indirectly, 100% of the Equity Interests of each other Loan Party, or

(e) there is a "Change of Control" under the 2020 Unsecured Convertible Notes, the Preferred Equity Documents, any Permitted Convertible Notes, or any Permitted Subordinated Notes.

"<u>Change in Law</u>" means the occurrence after the date of this Agreement of: (a) the adoption of any law, rule, regulation, judicial ruling, judgment or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application by any Governmental Authority of any law, rule, regulation, guideline or treaty, or (c) the making or issuance by any Governmental Authority of any request, rule, guideline or directive, whether or not having the force of law; <u>provided</u>, that notwithstanding anything in this Agreement to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith, and (ii) all requests, rules, guidelines or directives concerning capital adequacy promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities shall, in each case, be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Closing Date" means December 29, 2017.

-7-

"Code" means the California Uniform Commercial Code, as in effect from time to time.

"<u>Collateral</u>" means all assets and interests in assets and proceeds thereof now owned or hereafter acquired by any Loan Party or its Subsidiaries in or upon which a Lien is granted by such Person in favor of Agent or the Lenders under any of the Loan Documents.

"<u>Collateral Access Agreement</u>" means a landlord waiver or acknowledgement agreement of any lessor or other Person in possession of, having a Lien upon, or having rights or interests in any Loan Party's or its Subsidiaries' books and records, in each case, in form and substance reasonably satisfactory to Agent.

"Collections" means, all cash, checks, notes, instruments, and other items of payment (including insurance proceeds, cash proceeds of asset sales, rental proceeds and tax refunds).

"Commitment" means, with respect to each Lender, its Revolver Commitment, and, with respect to all Lenders, their Revolver Commitments, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on <u>Schedule C-1</u> to this Agreement or in the Assignment and Acceptance pursuant to which such Lender became a Lender under this Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of <u>Section 13.1</u> of this Agreement.

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.), as amended from time to time, and any successor statute.

"Communications Laws" means the Communications Act of 1934, and any similar or successor federal statute, together with all published rules, regulations, policies, orders and decisions of the FCC promulgated thereunder.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C-1 to this Agreement delivered by the chief financial officer or treasurer of Administrative Borrower to Agent.

"Confidential Information" has the meaning specified therefor in Section 17.9(a) of this Agreement.

"<u>Continuing Director</u>" means (a) any member of the Board of Directors who was a director (or comparable manager) of Administrative Borrower on the Closing Date, and (b) any individual who becomes a member of the Board of Directors after the Closing Date if such individual was approved, appointed or nominated for election to the Board of Directors by either the Permitted Holders or a majority of the Continuing Directors.

"Control Agreement" means a control agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by a Loan Party or one of its Subsidiaries, Agent, and the applicable securities intermediary (with respect to a Securities Account) or bank (with respect to a Deposit Account).

"Copyright Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Covenant Testing Period" means a period (a) commencing on the last day of the fiscal month of Administrative Borrower most recently ended prior to a Covenant Trigger Event for which Borrowers are required to deliver to Agent monthly, quarterly or annual financial statements pursuant to <u>Schedule 5.1</u> to this Agreement, and (b) continuing through and including the first day after such Covenant Trigger Event that Liquidity has equaled or exceeded the greater of (i) 10% of the Line Cap, and (ii) \$15,000,000 for 90 consecutive days.

-8-

"Covenant Trigger Event" means if at any time Liquidity is less than the greater of (i) 10% of the Line Cap, and (ii) \$15,000,000.

"Default" means an event, condition, or default that, with the giving of notice, the passage of time, or both, would be an Event of Default.

"Defaulting Lender" means any Lender that (a) has failed to (i) fund all or any portion of its Loans within two Business Days of the date such Loans were required to be funded hereunder unless such Lender notifies Agent and Administrative Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable Default or Event of Default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to Agent, Issuing Bank, or any other Lender any other amount required to be paid by it hereunder (including in respect of its participation in Letters of Credit) within two Business Days of the date when due, (b) has notified any Borrower, Agent or Issuing Bank in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable Default or Event of Default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by Agent or Administrative Borrower, to confirm in writing to Agent and Administrative Borrower that it will comply with its prospective funding obligations hereunder (provided, that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by Agent and Administrative Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of any Insolvency Proceeding, (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity, or (iii) become the subject of a Bail-in Action; provided, that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender upon delivery of written notice of such determination to Administrative Borrower, Issuing Bank, and each Lender.

"Defaulting Lender Rate" means (a) for the first three days from and after the date the relevant payment is due, the Base Rate, and (b) thereafter, the interest rate then applicable to Revolving Loans that are Base Rate Loans (inclusive of the Base Rate Margin applicable thereto).

"Deposit Account" means any deposit account (as that term is defined in the Code).

"Designated Account" means the Deposit Account of Administrative Borrower identified on Schedule D-1 to this Agreement (or such other Deposit Account of Administrative Borrower located at Designated Account Bank that has been designated as such, in writing, by Borrowers to Agent).

"Designated Account Bank" has the meaning specified therefor in <u>Schedule D-1</u> to this Agreement (or such other bank that is located within the United States that has been designated as such, in writing, by Borrowers to Agent).

"Dilution" means, as of any date of determination, a percentage, based upon the experience of the immediately prior 12 months, that is the result of dividing the Dollar amount of (a) bad debt write-downs,

-9-

discounts, advertising allowances, credits, or other dilutive items with respect to Borrowers' Accounts during such period, by (b) Borrowers' billings with respect to Accounts during such period.

"Dilution Reserve" means, as of any date of determination, an amount sufficient to reduce the advance rate against Eligible Accounts by the extent to which Dilution is in excess of 5%.

"Disqualified Equity Interests" means any Equity Interests that, by their terms (or by the terms of any security or other Equity Interests into which they are convertible or for which they are exchangeable), or upon the happening of any event or condition (a) matures or are mandatorily redeemable (other than solely for Qualified Equity Interests and solely in connection with such redemption into Qualified Equity Interests, conversion into cash of any *de minimis* Equity Interest that is less than one (1) share or interest), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event shall be subject to the prior repayment in full of the Loans and all other Obligations that are accrued and payable and the termination of the Commitments), (b) are redeemable at the option of the holder thereof (other than solely for Qualified Equity Interests and solely in connection with such redemption into Qualified Equity Interests, conversion into cash of any *de minimis* Equity Interest that is less than one (1) share or interest), in whole or in part, (c) provide for the scheduled payments of dividends in cash, or (d) are or become convertible into or exchangeable for Indebtedness (and solely in connection with such conversion into Indebtedness, conversion into cash of any *de minimis* Equity Interest that is less than one (1) share or interest) or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is 180 days after the Maturity Date.

"Disqualified Institution" means those Persons who are direct competitors of the Borrowers identified in writing by Administrative Borrower to Agent from time to time, subject to the written consent of Agent (it being understood and agreed that any Person shall be a Disqualified Lender from and after the date of such written identification by Administrative Borrower to Agent and such written consent of Agent and not on a retroactive basis); provided, that "Disqualified Institutions" shall exclude any Person that Administrative Borrower has designated as no longer being a "Disqualified Institution" by written notice delivered to Agent from time to time; provided further, that in connection with any assignment or participation, the Assignee or Participant with respect to such proposed assignment or participation that is an investment bank, a commercial bank, a finance company, a fund, or other Person which merely has an economic interest in any such direct competitor, and is not itself such a direct competitor of any Borrower or its Subsidiaries, shall not be deemed to be a Disqualified Institution for the purposes of this definition.

"Disregarded Domestic Person" means any direct or indirect Domestic Subsidiary that holds no material assets other than the equity of (or equity of and debt obligations owed or treated as owed by) one or more direct or indirect Foreign Subsidiaries that are CFCs or other Disregarded Domestic Persons.

"Dollars" or "§" means United States dollars.

"Domestic Subsidiary" means any Subsidiary of any Loan Party that is not a Foreign Subsidiary.

"Drawing Document" means any Letter of Credit or other document presented for purposes of drawing under any Letter of Credit, including by electronic transmission such as SWIFT, electronic mail, facsimile or computer generated communication.

"<u>Earn-Outs</u>" means unsecured liabilities of a Loan Party arising under an agreement to make any deferred payment as a part of the Purchase Price for a Permitted Acquisition, including performance bonuses or consulting payments in any related services, employment or similar agreement, in an amount that is subject to or contingent upon the revenues, income, cash flow or profits (or the like) of the target of such Permitted Acquisition.

-10-

"EBITDA" means, with respect to any fiscal period and with respect to Borrowers determined, in each case, on a consolidated basis in accordance with GAAP:

(a) the consolidated net income (or loss),

<u>minus</u>

(b) without duplication, the sum of the following amounts for such period to the extent included in determining consolidated net income (or loss) for such period:

- (i) any unusual or non-recurring gains,
- (ii) interest income,

(iii) the amount of any non-controlling interest income consisting of losses attributed to non-controlling interests of third parties in any non-wholly owned Subsidiaries of Borrowers to the extent included in consolidated net income (or loss) and not received in cash by Borrowers,

(iv) unrealized or realized gains in respect of obligations under hedging transactions, and gains resulting from currency translation or transaction gains related to currency re-measurements of Indebtedness, and all other foreign currency translation or transaction gains,

(v) expenses or charges arising from the sale or issuance of Equity Interests, the granting of stock options, and the granting of stock appreciation rights and similar arrangements (including any repricing, amendment, modification, substitution, or change of any such Equity Interests, stock option, stock appreciation rights, or similar arrangements) to the extent that such expenses or charges are funded with cash proceeds contributed to the capital of Borrowers or an issuance of capital stock of Administrative Borrower,

- (vi) federal, state and local income tax credits,
- (vii) gains on sales of fixed assets or discontinued or disposed of operations, and
- (viii) income arising by reason of the application of FAS 141R,
- (ix) non-operating income, and
- (x) income tax benefits and decreases in any change in LIFO reserves,

<u>plus</u>

(c) without duplication, the sum of the following amounts for such period to the extent deducted in determining consolidated net income (or loss) for such period:

(i) any unusual or non-recurring non-cash losses; provided, that if any such non-cash losses referred to in this clause represent an accrual or reserve for potential cash items in any future period, the cash payment in respect thereof in such future period shall be deducted from EBITDA to such extent; provided further that any add-back pursuant to this subclause (c)(i) shall be subject to the cap in the proviso at the end of this definition,

(ii) Interest Expense,

-11-

(iii) tax expense based on income, profits or capital, including federal, foreign, state, franchise and similar taxes (and for the avoidance of doubt, specifically excluding any sales taxes or any other taxes held in trust for a Governmental Authority),

(iv) depreciation and amortization,

(v) with respect to any Permitted Acquisition, costs, fees, charges, or expenses consisting of out-of-pocket expenses owed by the Loan Parties or any of their Subsidiaries to any Person for services performed by such Person in connection with such Permitted Acquisition incurred within 180 days of the consummation of such Permitted Acquisition, (i) up to an aggregate amount (for all such items in this clause (B)) for such Permitted Acquisition not to exceed the greater of (1) \$2,000,000, and (2) 10% of the Purchase Price of such Permitted Acquisition, and (ii) in any amount to the extent such costs, fees, charges, or expenses in this clause (B) are paid with proceeds of new equity investments in exchange for Qualified Equity Interests of Administrative Borrower contemporaneously made by Permitted Holders,

(vi) with respect to any Permitted Acquisitions: (1) purchase accounting adjustments, including a dollar for dollar adjustment for that portion of revenue that would have been recorded in the relevant period had the balance of deferred revenue (unearned income) recorded on the closing balance sheet and before application of purchase accounting not been adjusted downward to fair value to be recorded on the opening balance sheet in accordance with GAAP purchase accounting rules; and (2) non-cash adjustments in accordance with GAAP purchase accounting rules under FASB Statement No. 141 and EITF Issue No. 01-3, in the event that such an adjustment is required by Borrowers' independent auditors, in each case, as determined in accordance with GAAP,

(vii) fees, costs, charges and expenses, in respect of Earn-Outs incurred in connection with any Permitted Acquisition to the extent permitted to be incurred under this Agreement that are required by the application of FAS 141R to be and are expensed by the Loan Parties and their Subsidiaries,

(viii) [reserved],

(ix) non-cash compensation expense (including deferred non-cash compensation expense), or other non-cash expenses or charges, arising from the sale or issuance of Equity Interests, the granting of stock options, and the granting of stock appreciation rights and similar arrangements (including any repricing, amendment, modification, substitution, or charge of any such Equity Interests, stock option, stock appreciation rights, or similar arrangements) <u>minus</u> the amount of any such expenses or charges when paid in cash to the extent not deducted in the computation of net income (or loss),

(x) non-recurring expenses for severance, recruitment and hiring of senior management (including signing bonuses in connection therewith), in an aggregate amount not to exceed 1,500,000 during any 12 month period or 5,000,000 during the term of this Agreement; provided that any add-back pursuant to this subclause (c)(x) shall be subject to the cap in the proviso at the end of this definition,

(xi) one-time non-cash restructuring charges; <u>provided</u> that any add-back pursuant to this subclause (c)(xi) shall be subject to the cap in the proviso at the end of this definition,

(xii) the amount of any non-controlling interest expense consisting of losses attributed to non-controlling interests of third parties in any non-wholly owned Subsidiaries of Administrative Borrower,

-12-

(xiii) expenses, charges and fees (including expenses, charges and fees paid to Agent and Lenders) incurred during such period and after the Closing Date in connection with the administration (including in connection with any waiver, amendment, supplementation or other modification thereto of the Loan Documents) of the Loan Documents,

(xiv) documented expenses, charges and fees related to an investment or incurrence of Indebtedness permitted by the Agreement (whether or not consummated or incurred, and including any non-consummated sale of Equity Interests to the extent the proceeds thereof were intended to be contributed to the equity of Borrowers); provided that all such expenses, charges, and fees, in the aggregate pursuant to this subclause (c)(xiv) shall not exceed \$2,000,000 in the aggregate in any fiscal year; provided further that any add-back pursuant to this subclause (c)(xiv) shall not include any contract termination benefit owing to KKR Classic Investors L.P. subject to that certain Notice of Termination of Investment Agreement, dated as of June 9, 2017, from Pandora to KKR Classic Investors L.P.,

(xv) costs and expenses incurred (A) to the extent covered by indemnification or reimbursement provisions in any agreement with a Person in connection with any Permitted Acquisition, or (B) to the extent indemnified or reimbursed by a Person that is not an Affiliate of the Loan Parties, and in each case, solely to the extent such indemnification or reimbursement did not increase consolidated net income for such period,

(xvi) non-cash losses in respect of obligations under hedging transactions, and losses resulting from currency translation or non-cash transaction losses related to currency re-measurements of Indebtedness, and all other non-cash foreign currency translation or transaction losses,

(xvii) documented expenses, charges and fees related to the contract termination benefit owing by Pandora to KKR Classic Investors L.P. in connection with that certain Notice of Termination of Investment Agreement, dated as of June 9, 2017, from Pandora to KKR Classic Investors L.P., in an amount not to exceed \$23,000,000 in the aggregate,

(xviii) [reserved],

(xix) non-cash losses on sales of fixed assets or discontinued or disposed of operations or write-downs of fixed or intangible assets (excluding write-downs of Accounts or Inventory) (provided, that to the extent any non-cash item added back to EBITDA in any period results in a cash payment in such period or a subsequent period such cash payment shall result in a reduction of EBITDA in the period when such payment is made), and

(xx) increases in any change in LIFO reserves;

<u>provided</u> that for purposes of calculating EBITDA, the aggregate amount of charges and expenses which are added back pursuant to clauses (c)(i), (c)(x) and (c)(xi) above shall not exceed 20% of EBITDA (without giving effect to clauses (c)(i), (c)(x) and (c)(xi) above).

For the purposes of calculating EBITDA for any period of twelve consecutive months (each, a "<u>Reference Period</u>"), if at any time during such Reference Period (and on or after the Closing Date), any Loan Party or any of its Subsidiaries shall have made a Permitted Acquisition, EBITDA for such Reference Period shall be calculated after giving *pro forma* effect thereto (including *pro forma* adjustments arising out of events which are directly attributable to such Permitted Acquisition, are factually supportable, and are expected to have a continuing impact, in each case to be mutually and reasonably agreed upon by Borrowers

-13-

and Agent) or in such other manner acceptable to Agent as if any such Permitted Acquisition or adjustment occurred on the first day of such Reference Period.

In addition, notwithstanding the above, (a) EBITDA for the quarter ended March 31, 2017, shall be deemed to be negative \$77,483,000, (b) EBITDA for the quarter ended June 30, 2017, shall be deemed to be negative \$56,026,000, and (c) EBITDA for the quarter ended September 30, 2017, shall be deemed to be negative \$5,801,000.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"<u>Eligible Accounts</u>" means those Accounts created by a Borrower in the ordinary course of its business, that arise out of such Borrower's sale of goods or rendition of services, that comply with each of the representations and warranties respecting Eligible Accounts made in the Loan Documents, and that are not excluded as ineligible by virtue of one or more of the excluding criteria set forth below; <u>provided</u>, that such criteria may be revised from time to time by Agent in Agent's Permitted Discretion to address the results of any information with respect to the Borrowers' business or assets of which Agent becomes aware after the Closing Date, including any field examination performed by (or on behalf of) Agent from time to time after the Closing Date. In determining the amount to be included, Eligible Accounts shall be calculated net of customer deposits, unapplied cash, taxes, finance charges, service charges, discounts, credits, allowances, and rebates. Eligible Accounts shall not include the following:

(a) Accounts that the Account Debtor has failed to pay within 120 days of original invoice date or 90 days of due date,

(b) Accounts owed by an Account Debtor (or its Affiliates) where 50% or more of all Accounts owed by that Account Debtor (or its Affiliates) are deemed ineligible under clause (a) above,

(c) [reserved],

(d) Accounts with respect to which the Account Debtor is an Affiliate of any Borrower or an employee or agent of any Borrower or any Affiliate of any Borrower,

(e) Accounts (i) arising in a transaction wherein goods are placed on consignment or are sold pursuant to a guaranteed sale, a sale or return, a sale on approval, a bill and hold, or any other terms by reason of which the payment by the Account Debtor may be conditional, or (ii) with respect to which the payment terms are "C.O.D.", cash on delivery or other similar terms,

(f) Accounts that are not payable in Dollars,

-14-

(g) Accounts with respect to which the Account Debtor either (i) does not maintain its chief executive office in the United States or Canada, or (ii) is not organized under the laws of the United States or Canada or any state or province thereof, or (iii) is the government of any foreign country or sovereign state, or of any state, province, municipality, or other political subdivision thereof, or of any department, agency, public corporation, or other instrumentality thereof, unless (A) the Account is supported by an irrevocable letter of credit reasonably satisfactory to Agent (as to form, substance, and issuer or domestic confirming bank) that has been delivered to Agent and, if requested by Agent, is directly drawable by Agent, or (B) the Account is covered by credit insurance in form, substance, and amount, and by an insurer, reasonably satisfactory to Agent,

(h) Accounts with respect to which the Account Debtor is either (i) the United States or any department, agency, or instrumentality of the United States (exclusive, however, of Accounts with respect to which Borrowers have complied, to the reasonable satisfaction of Agent, with the Assignment of Claims Act, 31 USC §3727), or (ii) any state of the United States or any other Governmental Authority,

(i) Accounts with respect to which the Account Debtor is a creditor of a Borrower, has or has asserted a right of recoupment or setoff, or has disputed its obligation to pay all or any portion of the Account, to the extent of such claim, right of recoupment or setoff, or dispute,

(j) Accounts with respect to an Account Debtor whose Eligible Accounts owing to Borrowers exceed 15% (such percentage, as applied to a particular Account Debtor, being subject to reduction by Agent in its Permitted Discretion if the creditworthiness of such Account Debtor deteriorates) of all Eligible Accounts, to the extent of the obligations owing by such Account Debtor in excess of such percentage; provided, that in each case, the amount of Eligible Accounts that are excluded because they exceed the foregoing percentage shall be determined by Agent based on all of the otherwise Eligible Accounts prior to giving effect to any eliminations based upon the foregoing concentration limit,

(k) Accounts with respect to which the Account Debtor is subject to an Insolvency Proceeding, is not Solvent, has gone out of business, or as to which any Borrower has received notice of an imminent Insolvency Proceeding or a material impairment of the financial condition of such Account Debtor,

(1) Accounts, the collection of which, Agent, in its Permitted Discretion, believes to be doubtful, including by reason of the Account Debtor's financial condition,

(m) Accounts that are not subject to a valid and perfected first priority Agent's Lien,

(n) Accounts with respect to which (i) the goods giving rise to such Account have not been shipped and billed to the Account Debtor, or (ii) the services giving rise to such Account have not been performed and billed to the Account Debtor,

(o) Accounts with respect to which the Account Debtor is a Sanctioned Person or Sanctioned Entity,

(p) Accounts (i) that represent the right to receive progress payments or other advance billings that are due prior to the completion of performance by the applicable Borrower of the subject contract for goods or services, or (ii) that represent credit card sales (other than credit card receipts from customers to Apple Inc. and Google Inc. relative to the Eligible Subscription Accounts), or

(q) Accounts owned by a target acquired in connection with a Permitted Acquisition or Permitted Investment, or Accounts owned by a Person that is joined to this Agreement as a Borrower

-15-

pursuant to the provisions of this Agreement, until the completion of a field examination with respect to such Accounts, in each case, satisfactory to Agent in its Permitted Discretion.

"Eligible Advertising Accounts" means those Eligible Accounts that arise from advertising services rendered or to be rendered by any Borrower to advertisers or to advertising agencies.

"Eligible Subscription Accounts" means those Eligible Accounts that arise from the transmission, broadcast, delivery or other performance of music or other programming by any Borrower to Account Debtors distributed through (i) Google Play (or any successor mobile device app distribution platform operated by or on behalf of Google LLC or any successor or assignee thereof) pursuant to that certain Google Play Developer Distribution Agreement as in effect on the Closing Date (including, without limitation, all schedules, addendums, exhibits, annexes, appendices, and other additions or amendments thereto from time to time), by and between Pandora and Google LLC; (ii) the App Store (or any successor mobile device app distribution platform operated by or on behalf of Apple Inc. or any successor or assignee thereof) pursuant to (A) that certain Apple Developer Agreement, by and between Pandora and Apple Inc., as in effect on the Closing Date (including, without limitation, all schedules, addendums, exhibits, annexes, appendices, and other additions thereto from time to time) or (B) that certain Apple Developer Program License Agreement, by and between Pandora and Apple Inc., as in effect on the Closing Date (including, without limitation, all schedules, addendums, exhibits, annexes, appendices, and other additions thereto from time to time) or (B) that certain Apple Developer Program License Agreement, by and between Pandora and Apple Inc., as in effect on the Closing Date (including, without limitation, all schedules, addendums, exhibits, annexes, appendices, and other additions thereto from time to time).

"Eligible Transferee" means (a) any Lender (other than a Defaulting Lender), any Affiliate of any Lender and any Related Fund of any Lender; (b) (i) a commercial bank organized under the laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (ii) a savings and loan association or savings bank organized under the laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (iii) a commercial bank organized under the laws of any other country or a political subdivision thereof; <u>provided</u>, that (A) (x) such bank is acting through a branch or agency located in the United States, or (y) such bank is organized under the laws of a country that is a member of the Organization for Economic Cooperation and Development or a political subdivision of such country, and (B) such bank has total assets in excess of \$1,000,000,000; (c) any other entity (other than a natural person) that is an "accredited investor" (as defined in Regulation D under the Securities Act) that extends credit or buys loans as one of its businesses including insurance companies, investment or mutual funds and lease financing companies, and having total assets in excess of \$1,000,000,000; and (d) during the continuation of an Event of Default, any other Person approved by Agent.

"Environmental Action" means any written complaint, summons, citation, notice, directive, order, claim, litigation, investigation, judicial or administrative proceeding, judgment, letter, or other written communication from any Governmental Authority, or any third party involving violations of Environmental Laws or releases of Hazardous Materials (a) from any assets, properties, or businesses of any Borrower, any Subsidiary of any Borrower, or any of their predecessors in interest, (b) from adjoining properties or businesses, or (c) from or onto any facilities which received Hazardous Materials generated by any Borrower, any Subsidiary of any Borrower, or any of their predecessors in interest.

"Environmental Law" means any applicable federal, state, provincial, foreign or local statute, law, rule, regulation, ordinance, code, binding and enforceable guideline, binding and enforceable written policy, or rule of common law now or hereafter in effect and in each case as amended, or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, in each case, to the extent binding on any Loan Party or its Subsidiaries, relating to the environment, the effect of the environment on employee health, or Hazardous Materials, in each case as amended from time to time.

"Environmental Liabilities" means all liabilities, monetary obligations, losses, damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts, or consultants, and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any



claim or demand, or Remedial Action required, by any Governmental Authority or any third party, and which relate to any Environmental Action.

"Environmental Lien" means any Lien in favor of any Governmental Authority for Environmental Liabilities.

"Equipment" means equipment (as that term is defined in the Code).

"Equity Interests" means, with respect to a Person, all of the shares, options, warrants, interests, participations, or other equivalents (regardless of how designated) of or in such Person, whether voting or nonvoting, including capital stock (or other ownership or profit interests or units), preferred stock, or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statute thereto.

"ERISA Affiliate" means (a) any Person subject to ERISA whose employees are treated as employed by the same employer as the employees of any Loan Party or its Subsidiaries under IRC Section 414(b), (b) any trade or business subject to ERISA whose employees are treated as employed by the same employer as the employees of any Loan Party or its Subsidiaries under IRC Section 414(c), (c) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any organization subject to ERISA that is a member of an affiliated service group of which any Loan Party or any of its Subsidiaries is a member under IRC Section 414(m), or (d) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any Person subject to ERISA that is a party to an arrangement with any Loan Party or any of its Subsidiaries and whose employees are aggregated with the employees of such Loan Party or its Subsidiaries under IRC Section 414(o).

"EU Bail-In Legislation Schedule" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"Event of Default" has the meaning specified therefor in Section 8 of this Agreement.

"Eventbrite Note" means that certain Convertible Subordinated Promissory Note, dated as of September 1, 2017, in the face amount of \$50,000,000, issued by Eventbrite, Inc., a Delaware corporation, in favor of Pandora.

"Excess Availability" means, as of any date of determination, the amount equal to the Line Cap <u>minus</u> the sum of (a) outstanding Revolving Loans, and (b) the aggregate amount, if any, of all trade payables of the Loan Parties and their Subsidiaries aged in excess of historical levels with respect thereto and all book overdrafts of the Loan Parties and their Subsidiaries in excess of historical practices with respect thereto, in each case as determined by Agent in its Permitted Discretion.

"Exchange Act" means the Securities Exchange Act of 1934, as in effect from time to time.

"Excluded Subsidiary" means (a) Immaterial Subsidiaries, (b) any Subsidiary of a Loan Party to the extent that the burden or cost (including any potential tax liability) of obtaining a guarantee outweighs the benefit afforded thereby as reasonably determined by Borrowers and Agent, (c) any Disregarded Domestic Persons, (d) any Subsidiary of a Loan Party that is a CFC, (e) any Subsidiary of a Loan Party that is a direct or indirect subsidiary of a Foreign Subsidiary that is a CFC, or (f) any not-for-profit subsidiary or captive insurance subsidiary.

"Excluded Swap Obligation" means, with respect to any Loan Party, any Swap Obligation if, and to the extent that, all or a portion of the guaranty of such Loan Party of (including by virtue of the joint and

-17-

several liability provisions of <u>Section 2.15</u>), or the grant by such Loan Party of a security interest to secure, such Swap Obligation (or any guaranty thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Loan Party's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act and the regulations thereunder at the time the guaranty of such Loan Party or the grant of such security interest becomes effective with respect to such Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such guaranty or security interest is or becomes illegal.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (i) any Tax imposed on or measured by the net income or net profits (however denominated) of any Recipient (including any branch profits and franchise Taxes), in each case (A) imposed by the jurisdiction (or by any political subdivision or taxing authority thereof) in which such Recipient is organized or the jurisdiction (or by any political subdivision or taxing authority thereof) in which such Recipient is organized or the jurisdiction in which its applicable lending office is located or (B) that are Other Connection Taxes, (ii) United States federal withholding Taxes that would not have been imposed but for a Recipient's failure to comply with the requirements of Section 16.2 of this Agreement, (iii) any United States federal withholding Taxes that would be imposed on amounts payable to or for the account of a Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which such Lender acquires such interest in the Loan or Commitment (other than pursuant to an assignment request by the Borrower under Section 14.2) (or designates a new lending office, other than a designation made at the request of a Loan Party), <u>except</u> any amount that such Lender (or its assignre, if any) was previously entitled to receive pursuant to <u>Section 16.1</u> of this Agreement, if any, with respect to such withholding Tax at the time such Lender becomes a party to this Agreement (or designates a new lending office), and (iv) any Taxes imposed under FATCA.

"Existing Credit Facility" means that certain Credit Agreement, dated as of May 13, 2011, as amended and restated as of December 31, 2015, by and among Administrative Borrower, the lenders party thereto, JPMorgan Chase Bank, N.A., as administrative agent, J.P. Morgan Securities LLC, as sole lead arranger, J.P. Morgan Securities LLC and Morgan Stanley Senior Funding, Inc., as joint bookrunners, and Morgan Stanley Senior Funding, Inc. and Silicon Valley Bank, as co-documentation agents (as amended, restated, supplemented, or modified prior to the date hereof).

"Extraordinary Advances" has the meaning specified therefor in Section 2.3(d)(iii) of this Agreement.

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), and (a) any current or future regulations or official interpretations thereof, (b) any agreements entered into pursuant to Section 1471(b)(1) of the IRC, and (c) any intergovernmental agreement entered into by the United States (or any fiscal or regulatory legislation, rules, or practices adopted pursuant to any such intergovernmental agreement entered into in connection therewith).

"FCC" means the Federal Communications Commission or any Governmental Authority substituted therefor.

"FCC Licenses" means a License issued or granted by the FCC.

"FCPA" means the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder.

-18-

"Fee Letter" means that certain fee letter, dated as of even date with this Agreement, among Borrowers and Agent, in form and substance reasonably satisfactory to Agent.

"Federal Funds Rate" means, for any period, a fluctuating interest rate per annum equal to, for each day during such period, the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from three Federal funds brokers of recognized standing selected by it (and, if any such rate is below zero, then the rate determined pursuant to this definition shall be deemed to be zero).

"<u>Fixed Charges</u>" means, with respect to any fiscal period and with respect to Borrowers determined on a consolidated basis in accordance with GAAP, the sum, without duplication, of (a) Interest Expense required to be paid (other than interest paid-in-kind, amortization of financing fees, and other non-cash Interest Expense) during such period, (b) scheduled principal payments in respect of Indebtedness that are required to be paid during such period), (c) all federal, state, and local income taxes required to be paid during such period , and (d) all Restricted Payments paid (whether in cash or other property, other than common Equity Interests) during such period.

"Fixed Charge Coverage Ratio" means, with respect to any fiscal period and with respect to Borrowers determined on a consolidated basis in accordance with GAAP, the ratio of (a) EBITDA for such period <u>minus</u> Unfinanced Capital Expenditures made (to the extent not already incurred in a prior period) or incurred during such period, to (b) Fixed Charges for such period.

For the purposes of calculating Fixed Charge Coverage Ratio for any Reference Period, if at any time during such Reference Period (and after the Closing Date), any Loan Party or any of its Subsidiaries shall have made a Permitted Acquisition, Fixed Charges and Unfinanced Capital Expenditures for such Reference Period shall be calculated after giving *pro forma* effect thereto or in such other manner acceptable to Agent as if any such Permitted Acquisition occurred on the first day of such Reference Period.

"Flow of Funds Agreement," means a flow of funds agreement, dated as of even date with this Agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by Borrowers and Agent.

"Foreign Lender" means any Lender or Participant that is not a United States person within the meaning of IRC section 7701(a)(30).

"Foreign Subsidiary" means any direct or indirect subsidiary of any Loan Party that is organized under the laws of any jurisdiction other than the United States, any state thereof or the District of Columbia.

"<u>Funded Indebtedness</u>" means, as of any date of determination, all Indebtedness for borrowed money or letters of credit of Borrowers, determined on a consolidated basis in accordance with GAAP, including, in any event, but without duplication, with respect to the Loan Parties and their Subsidiaries, the Revolver Usage, and the amount of their Capitalized Lease Obligations.

"Funding Date" means the date on which a Borrowing occurs.

"Funding Losses" has the meaning specified therefor in Section 2.12(b)(ii) of this Agreement.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

-19-

"Governing Documents" means, with respect to any Person, the certificate or articles of incorporation, by-laws, or other organizational documents of such Person.

"Governmental Authority" means the government of any nation or any political subdivision thereof, whether at the national, state, territorial, provincial, county, municipal or any other level, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of, or pertaining to, government (including any supra-national bodies such as the European Union or the European Central Bank).

"<u>Guarantor</u>" means (a) each Person that guaranties all or a portion of the Obligations, including any Person that is a "Guarantor" under the Guaranty and Security Agreement, and (b) each other Person that becomes a guarantor after the Closing Date pursuant to <u>Section 5.11</u> of this Agreement.

"Guaranty and Security Agreement," means a guaranty and security agreement, dated as of even date with this Agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by each of the Loan Parties to Agent.

"Hazardous Materials" means (a) substances that are defined or listed in, or otherwise classified pursuant to, any applicable laws or regulations as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, or "EP toxicity", (b) oil, petroleum, or petroleum derived substances, natural gas, natural gas liquids, synthetic gas, drilling fluids, produced waters, and other wastes associated with the exploration, development, or production of crude oil, natural gas, or geothermal resources, (c) any flammable substances or explosives or any radioactive materials, and (d) asbestos in any form or electrical equipment that contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of 50 parts per million.

"Hedge Agreement" means a "swap agreement" as that term is defined in Section 101(53B)(A) of the Bankruptcy Code.

"Hedge Obligations" means any and all obligations or liabilities, whether absolute or contingent, due or to become due, now existing or hereafter arising, of each Loan Party and its Subsidiaries arising under, owing pursuant to, or existing in respect of Hedge Agreements entered into with one or more of the Hedge Providers.

"Hedge Provider" means any Bank Product Provider that is a party to a Hedge Agreement with a Loan Party or its Subsidiaries or otherwise provides Bank Products under clause (f) of the definition thereof; <u>provided</u>, that if, at any time, a Lender ceases to be a Lender under this Agreement (prior to the payment in full of the Obligations), then, from and after the date on which it ceases to be a Lender thereunder, neither it nor any of its Affiliates shall constitute Hedge Providers and the obligations with respect to Hedge Agreements entered into with such former Lender or any of its Affiliates shall no longer constitute Hedge Obligations.

"Immaterial Subsidiary" means each Subsidiary of a Borrower that is not a Material Subsidiary.

"Increase" has the meaning specified therefor in Section 2.14.

"Increase Date" has the meaning specified therefor in Section 2.14.

"Increase Joinder" has the meaning specified therefor in Section 2.14.

"Increased Reporting Event" means if at any time Liquidity is less than the greater of (a) 10% of the Line Cap, and (b) \$15,000,000.

-20-

"Increased Reporting Period" means the period commencing after the continuance of an Increased Reporting Event for 3 consecutive Business Days and continuing until the date when no Increased Reporting Event has occurred for 30 consecutive days.

"Indebtedness" as to any Person means (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments and all reimbursement or other obligations in respect of letters of credit, bankers acceptances, or other financial products, (c) all obligations of such Person as a lessee under Capital Leases, (d) all obligations or liabilities of others secured by a Lien on any asset of such Person, irrespective of whether such obligation or liability is assumed, (e) all obligations of such Person to pay the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business and repayable in accordance with customary trade practices and, for the avoidance of doubt, other than royalty payments payable in the ordinary course of business in respect of non-exclusive licenses) and any earn-out or similar obligations, (f) all monetary obligations of such Person owing under Hedge Agreements (which amount shall be calculated based on the amount that would be payable by such Person if the Hedge Agreement were terminated on the date of determination), (g) any Disqualified Equity Interests of such Person, (h) the Preferred Equity Investment, and (i) any obligation of such Person guaranteeing or intended to guarantee (whether directly or indirectly guaranteed, endorsed, co-made, discounted, or sold with recourse) any obligation of any other Person that constitutes Indebtedness under any of clauses (a) through (i) above. For purposes of this definition, (i) the amount of any Indebtedness represented by a guaranty or other similar instrument shall be the lesser of the principal amount of the obligations guaranteed and still outstanding and the maximum amount for which the guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Indebtedness, and (ii) the amount of any Indebtedness which is limited or is non-recourse to a Person or for which recourse is limited to an identified a

"Indemnified Liabilities" has the meaning specified therefor in Section 10.3 of this Agreement.

"Indemnified Person" has the meaning specified therefor in Section 10.3 of this Agreement.

"Indemnified Taxes" means, (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by, or on account of any obligation of, any Loan Party under any Loan Document, and (b) to the extent not otherwise described in the foregoing clause (a), Other Taxes.

"Insolvency Proceeding" means any proceeding commenced by or against any Person under any provision of the Bankruptcy Code or under any other state or federal bankruptcy or insolvency law, assignments for the benefit of creditors, formal or informal moratoria, compositions, extensions generally with creditors, or proceedings seeking reorganization, arrangement, or other similar relief.

"Intellectual Property" has the meaning specified therefor in the Guaranty and Security Agreement.

"Intercompany Subordination Agreement," means an intercompany subordination agreement, dated as of even date with this Agreement, executed and delivered by each Loan Party and each of its Subsidiaries, and Agent, the form and substance of which is reasonably satisfactory to Agent.

"Interest Expense" means, for any period, the aggregate of the interest expense of Borrowers for such period, determined on a consolidated basis in accordance with GAAP.

"Interest Period" means, with respect to each LIBOR Rate Loan, a period commencing on the date of the making of such LIBOR Rate Loan (or the continuation of a LIBOR Rate Loan or the conversion of a Base Rate Loan to a LIBOR Rate Loan) and ending 1, 2, 3, 6, or with the approval of all Lenders, 12 months thereafter; provided, that (a) interest shall accrue at the applicable rate based upon the LIBOR Rate from and

-21-

including the first day of each Interest Period to, but excluding, the day on which any Interest Period expires, (b) any Interest Period that would end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day, (c) with respect to an Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period), the Interest Period shall end on the last Business Day of the calendar month that is 1, 2, 3, 6 or, with the approval of all Lenders, 12 months after the date on which the Interest Period began, as applicable, and (d) Borrowers may not elect an Interest Period which will end after the Maturity Date.

"Inventory" means inventory (as that term is defined in the Code).

"Investment" means, with respect to any Person, any investment by such Person in any other Person (including Affiliates) in the form of loans, guarantees, advances, capital contributions (excluding (a) commission, travel, and similar advances to officers and employees of such Person made in the ordinary course of business, and (b) *bona fide* accounts receivable arising in the ordinary course of business), or acquisitions of Indebtedness, Equity Interests, or all or substantially all of the assets of such other Person (or of any division or business line of such other Person), and any other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP. The amount of any Investment shall be the original cost of such Investment <u>plus</u> the cost of all additions thereto, without any adjustment for increases or decreases in value, or write-ups, write-downs, or write-offs with respect to such Investment.

"IRC" means the Internal Revenue Code of 1986, as in effect from time to time.

"ISP" means, with respect to any Letter of Credit, the International Standby Practices 1998 (International Chamber of Commerce Publication No. 590) and any version or revision thereof accepted by the Issuing Bank for use.

"Issuer Document" means, with respect to any Letter of Credit, a letter of credit application, a letter of credit agreement, or any other document, agreement or instrument entered into (or to be entered into) by a Borrower in favor of Issuing Bank and relating to such Letter of Credit.

"Issuing Bank" means Wells Fargo or any other Lender that, at the request of Borrowers and with the consent of Agent, agrees, in such Lender's sole discretion, to become an Issuing Bank for the purpose of issuing Letters of Credit pursuant to Section 2.11 of this Agreement, and Issuing Bank shall be a Lender.

"Joinder" means a joinder agreement substantially in the form of Exhibit J-1 to this Agreement.

"Joint Book Runner" and "Joint Book Runners" have the respective meanings set forth in the preamble to this Agreement.

"Joint Lead Arranger" and "Joint Lead Arrangers" have the respective meanings set forth in the preamble to this Agreement.

"Landlord Reserve" means, as to the chief executive office and each other location at which a Borrower maintains its primary books and records and as to which a Collateral Access Agreement has not been received by Agent, a reserve in an amount equal to two (2) months' rent, storage charges, fees or other amounts under the lease or other applicable agreement relative to such location or, if greater and Agent so elects, the number of months' rent, storage charges, fees or other amounts for which the landlord or other property owner will have, under applicable law, a Lien in the Inventory of such Borrower to secure the payment of such amounts under the lease or other applicable agreement relative to such location; <u>provided</u>, that no Landlord Reserve shall be imposed for the first thirty (30) days following the Closing Date with respect to the Loan Parties' premises located at 2101 Webster Street, Oakland, CA 94612, so long as the Loan Parties use reasonable best efforts to enter into a Collateral Access Agreement with respect to such location.

"Lender" has the meaning set forth in the preamble to this Agreement, shall include Issuing Bank and the Swing Lender, and shall also include any other Person made a party to this Agreement pursuant to the provisions of <u>Section 13.1</u> of this Agreement and "Lenders" means each of the Lenders or any one or more of them.

"Lender Group" means each of the Lenders (including Issuing Bank and the Swing Lender) and Agent, or any one or more of them.

"Lender Group Expenses" means all (a) costs or expenses (including taxes and insurance premiums) required to be paid by any Loan Party or its Subsidiaries under any of the Loan Documents that are paid, advanced, or incurred by the Lender Group, (b) documented out-of-pocket fees or charges paid or incurred by Agent in connection with the Lender Group's transactions with each Loan Party and its Subsidiaries under any of the Loan Documents, including, photocopying, notarization, couriers and messengers, telecommunication, public record searches, filing fees, recording fees, publication, real estate surveys, real estate title policies and endorsements, and environmental audits, (c) Agent's customary fees and charges imposed or incurred in connection with any background checks or OFAC/PEP searches related to any Loan Party or its Subsidiaries, (d) Agent's customary fees and charges (as adjusted from time to time) with respect to the disbursement of funds (or the receipt of funds) to or for the account of any Borrower (whether by wire transfer or otherwise), together with any out-of-pocket costs and expenses incurred in connection therewith, (e) customary charges imposed or incurred by Agent resulting from the dishonor of checks payable by or to any Loan Party, (f) reasonable, documented out-of-pocket costs and expenses paid or incurred by the Lender Group to correct any default or enforce any provision of the Loan Documents, or during the continuance of an Event of Default, in gaining possession of, maintaining, handling, preserving, storing, shipping, selling, preparing for sale, or advertising to sell the Collateral, or any portion thereof, irrespective of whether a sale is consummated, (g) field examination, appraisal, and valuation fees and expenses of Agent related to any field examinations, appraisals, or valuation to the extent of the fees and charges (and up to the amount of any limitation) provided in Section 2.10 of this Agreement, (h) Agent's and Lenders' reasonable, documented costs and expenses (including reasonable and documented attorneys' fees and expenses) relative to third party claims or any other lawsuit or adverse proceeding paid or incurred, whether in enforcing or defending the Loan Documents or otherwise in connection with the transactions contemplated by the Loan Documents, Agent's Liens in and to the Collateral, or the Lender Group's relationship with any Loan Party or any of its Subsidiaries, (i) Agent's reasonable and documented costs and expenses (including reasonable and documented attorneys' fees and due diligence expenses) incurred in advising, structuring, drafting, reviewing, administering (including travel, meals, and lodging), syndicating (including reasonable costs and expenses relative to CUSIP, DXSyndicateTM, SyndTrak or other communication costs incurred in connection with a syndication of the loan facilities), or amending, waiving, or modifying the Loan Documents, and (j) Agent's and each Lender's reasonable and documented costs and expenses (including reasonable and documented attorneys, accountants, consultants, and other advisors fees and expenses) incurred in terminating, enforcing (including attorneys, accountants, consultants, and other advisors fees and expenses incurred in connection with a "workout," a "restructuring," or an Insolvency Proceeding concerning any Loan Party or any of its Subsidiaries or in exercising rights or remedies under the Loan Documents), or defending the Loan Documents, irrespective of whether a lawsuit or other adverse proceeding is brought, or in taking any enforcement action or any Remedial Action with respect to the Collateral; provided, that the fees and expenses of counsel that shall constitute Lender Group Expenses shall in any event be limited to one primary counsel to Agent and one primary counsel to the Lenders, one local counsel to Agent in each reasonably necessary jurisdiction, one specialty counsel to Agent in each reasonably necessary specialty area (including insolvency law), and one or more additional counsel to Lenders if one or more conflicts of interest arise.

-23-

"Lender Group Representatives," has the meaning specified therefor in Section 17.9 of this Agreement.

"Lender-Related Person" means, with respect to any Lender, such Lender, together with such Lender's Affiliates, officers, directors, employees, attorneys, and agents.

"Letter of Credit" means a letter of credit (as that term is defined in the Code) issued by Issuing Bank.

"Letter of Credit Collateralization" means either (a) providing cash collateral (pursuant to documentation reasonably satisfactory to Agent (including that Agent has a first priority perfected Lien in such cash collateral), including provisions that specify that the Letter of Credit Fees and all commissions, fees, charges and expenses provided for in Section 2.11(k) of this Agreement (including any fronting fees) will continue to accrue while the Letters of Credit are outstanding) to be held by Agent for the benefit of the Revolving Lenders in an amount equal to 105% of the then existing Letter of Credit Usage, (b) delivering to Agent documentation executed by all beneficiaries under the Letters of Credit, in form and substance reasonably satisfactory to Agent and Issuing Bank, terminating all of such beneficiaries' rights under the Letters of Credit, or (c) providing Agent with a standby letter of credit, in form and substance reasonably satisfactory to Agent (in its sole discretion) in an amount equal to 105% of the then existing Letter of Credit Usage (it being understood that the Letter of Credit Fee and all fronting fees set forth in this Agreement will continue to accrue while the Letters of Credit are outstanding and that any such fees that accrue must be an amount that can be drawn under any such standby letter of credit).

"Letter of Credit Disbursement" means a payment made by Issuing Bank pursuant to a Letter of Credit.

"Letter of Credit Exposure" means, as of any date of determination with respect to any Lender, such Lender's participation in the Letter of Credit Usage pursuant to Section 2.11(e) on such date.

"Letter of Credit Fee" has the meaning specified therefor in Section 2.6(b) of this Agreement.

"Letter of Credit Indemnified Costs" has the meaning specified therefor in Section 2.11(f) of this Agreement.

"Letter of Credit Related Person" has the meaning specified therefor in Section 2.11(f) of this Agreement.

"Letter of Credit Sublimit" means \$15,000,000.

"Letter of Credit Usage" means, as of any date of determination, the sum of (a) the aggregate undrawn amount of all outstanding Letters of Credit, plus (b) the aggregate amount of outstanding reimbursement obligations with respect to Letters of Credit which remain unreimbursed or which have not been paid through a Revolving Loan.

"LIBOR Deadline" has the meaning specified therefor in Section 2.12(b)(i) of this Agreement.

"LIBOR Notice" means a written notice in the form of Exhibit L-1 to this Agreement.

"LIBOR Option" has the meaning specified therefor in Section 2.12(a) of this Agreement.

"LIBOR Rate" means the rate per annum as published by ICE Benchmark Administration Limited (or any successor page or other commercially available source as the Agent may designate from time to

-24-

time) as of 11:00 a.m., London time, two Business Days prior to the commencement of the requested Interest Period, for a term, and in an amount, comparable to the Interest Period and the amount of the LIBOR Rate Loan requested (whether as an initial LIBOR Rate Loan or as a continuation of a LIBOR Rate Loan) by Borrowers in accordance with this Agreement (and, if any such published rate is below zero, then the LIBOR Rate shall be deemed to be zero). Each determination of the LIBOR Rate shall be made by the Agent and shall be conclusive in the absence of manifest error.

"LIBOR Rate Loan" means each portion of a Revolving Loan that bears interest at a rate determined by reference to the LIBOR Rate.

"LIBOR Rate Margin" has the meaning specified therefor in the definition of "Applicable Margin".

"License" means any authorization, permit, consent, special temporary authorization, franchise, ordinance, registration, certificate, license, agreement or other right filed with, granted by or entered into with a Governmental Authority which permits or authorizes the acquisition, construction, ownership or operation of (i) a radio broadcast station, including any part thereof or (ii) the operation of the business of the Loan Parties.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, assignment, charge, deposit arrangement, encumbrance, easement, lien (statutory or other), security interest, or other security arrangement and any other preference, priority, or preferential arrangement of any kind or nature whatsoever, including any conditional sale contract or other title retention agreement, the interest of a lessor under a Capital Lease and any synthetic or other financing lease having substantially the same economic effect as any of the foregoing.

"Line Cap" means, as of any date of determination, the lesser of (a) the Maximum Revolver Amount, and (b) the Borrowing Base as of such date of determination.

"Liquidity" means, as of any date of determination, the sum of (a) Excess Availability at such time <u>plus</u> (b) Qualified Cash at such time, <u>provided</u> that, for purposes of satisfying any Liquidity test under any Loan Document, at least 50% of Liquidity shall be attributable to Excess Availability.

"Loan" means any Revolving Loan, Swing Loan, or Extraordinary Advance made (or to be made) hereunder.

"Loan Account" has the meaning specified therefor in Section 2.9 of this Agreement.

"Loan Documents" means this Agreement, the Control Agreements, the Copyright Security Agreement, any Borrowing Base Certificate, the Fee Letter, the Guaranty and Security Agreement, the Intercompany Subordination Agreement, any Issuer Documents, the Letters of Credit, the Patent Security Agreement, the Trademark Security Agreement, any note or notes executed by Borrowers in connection with this Agreement and payable to any member of the Lender Group, and any other instrument or agreement entered into, now or in the future, by any Loan Party or any of its Subsidiaries and any member of the Lender Group in connection with this Agreement (but specifically excluding Bank Product Agreements).

"Loan Party" means any Borrower or any Guarantor.

"Margin Stock" as defined in Regulation U of the Board of Governors as in effect from time to time.

"<u>Material Adverse Effect</u>" means (a) a material adverse effect in the business, operations, results of operations, assets, liabilities or financial condition of the Loan Parties and their Subsidiaries, taken as a whole, (b) a material impairment of the Loan Parties' and their Subsidiaries' ability to perform their payment obligations

-25-

under the Loan Documents to which they are parties or satisfy the covenant set forth in <u>Section 7</u> of this Agreement, or of the Lender Group's ability to enforce the Obligations or realize upon the Collateral (other than as a result of an action taken or not taken that is solely in the control of Agent), or (c) a material impairment of the enforceability or priority of Agent's Liens with respect to all or a material portion of the Collateral.

"<u>Material Contract</u>" means (a) all contracts or agreements, the loss of which could reasonably be expected to result in a Material Adverse Effect and (b) any contract or agreement in respect of Music Performance Rights material to the business of a Loan Party.

"<u>Material Subsidiary</u>" means (a) each Borrower, (b) each Subsidiary of a Loan Party that (i) owns at least 5.0% of the consolidated total assets of the Loan Parties and their Subsidiaries, (ii) generates at least 5.0% of the consolidated revenues of the Loan Parties and their Subsidiaries, (iii) is the owner of Equity Interests of any Subsidiary of a Loan Party that otherwise constitutes a Material Subsidiary, and (iv) any group comprising Subsidiaries of a Loan Party that each would not have been a Material Subsidiary under clauses (i), (ii), or (iii) but that, taken together, had revenues or total assets in excess of 10% of the consolidated revenues or total assets, as applicable, of the Loan Parties and their Subsidiaries.

"<u>Maturity Date</u>" means the earliest of (a) the 5th anniversary of the Closing Date; or (b) the date that is 120 days prior to the 2020 Unsecured Convertible Notes Initial Maturity Date; <u>provided</u> that if on or before the date that is 120 days prior to the 2020 Unsecured Convertible Notes Maturity Date, the aggregate principal amount of the 2020 Unsecured Convertible Notes and all other Indebtedness arising therefrom (whether or not such Indebtedness is due and payable) is converted in full into Qualified Equity Interests of Administrative Borrower, the Maturity Date shall be the 5th anniversary of the Closing Date; and (c) the date that is 120 days prior to the Preferred Equity Investment Mandatory Redemption Date; <u>provided</u> that if on or before the date that is 120 days prior to the Preferred Equity Investment Mandatory Redemption Date, all of the Preferred Equity Investments (and all additional preferred equity investments that require conversion to cash prior to the 5th anniversary of the Closing Date) are converted in full into Qualified Equity Interests of Administrative Borrower, then the Maturity Date shall be the 5th anniversary of the Closing Date.

"<u>Maximum Revolver Amount</u>" means \$200,000,000, decreased by the amount of reductions in the Revolver Commitments made in accordance with <u>Section 2.4(c)</u> of this Agreement and increased by the amount of any Increase made in accordance with <u>Section 2.14</u> of this Agreement.

"Moody's" has the meaning specified therefor in the definition of Cash Equivalents.

"<u>Music Performance Rights</u>" means all rights granted by record labels, publishers, performance rights societies, and similar rights, including without limitation by ASCAP, BMI, SESAC, and SoundExchange, that are necessary for the transmission, broadcast, delivery or other performance of music or other programming, whether by terrestrial radio, satellite radio, internet transmission or other means of distribution or public performance.

"Net Cash Proceeds" means:

(a) with respect to any sale or disposition by any Loan Party or any of its Subsidiaries of assets, the amount of cash proceeds received (directly or indirectly) from time to time (whether as initial consideration or through the payment of deferred consideration) by or on behalf of such Loan Party or such Subsidiary, in connection therewith after deducting therefrom only (i) the amount of any Indebtedness secured by any Permitted Lien on any asset (other than (A) Indebtedness owing to Agent or any Lender under this Agreement or the other Loan Documents and (B) Indebtedness assumed by the purchaser of such asset) which is required to be, and is, repaid in connection with such sale or disposition, (ii) reasonable fees, commissions, and expenses related thereto and required to be paid by such Loan Party or such Subsidiary

in connection with such sale or disposition, (iii) taxes paid or payable to any taxing authorities by such Loan Party or such Subsidiary in connection with such sale or disposition, in each case to the extent, but only to the extent, that the amounts so deducted are, at the time of receipt of such cash, actually paid or payable to a Person that is not an Affiliate of any Loan Party or any of its Subsidiaries, and are properly attributable to such transaction, and (iv) all amounts that are set aside as a reserve (A) for adjustments in respect of the purchase price of such assets, (B) for any liabilities associated with such sale or casualty, to the extent such reserve is required by GAAP, and (C) for the payment of unassumed liabilities relating to the assets sold or otherwise disposed of at the time of, or within 30 days after, the date of such sale or other disposition, to the extent that in each case the funds described above in this clause (iv) are (x) deposited into escrow with a third party escrow agent or set aside in a separate Deposit Account that is subject to a Control Agreement in favor of Agent, and (y) paid to Agent as a prepayment of the applicable Obligations in accordance with <u>Section 2.4(e)</u> of this Agreement at such time when such amounts are no longer required to be set aside as such a reserve; and

(b) with respect to the issuance or incurrence of any Indebtedness by any Loan Party or any of its Subsidiaries, or the issuance by any Loan Party or any of its Subsidiaries of any Equity Interests, the aggregate amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment or disposition of deferred consideration) by or on behalf of such Loan Party or such Subsidiary in connection with such issuance or incurrence, after deducting therefrom only (i) reasonable fees, commissions, and expenses related thereto and required to be paid by such Loan Party or such Subsidiary in connection with such issuance or incurrence, and (ii) taxes paid or payable to any taxing authorities by such Loan Party or such Subsidiary in connection with such issuance or incurrence, in each case to the extent, but only to the extent, that the amounts so deducted are, at the time of receipt of such cash, actually paid or payable to a Person that is not an Affiliate of any Loan Party or any of its Subsidiaries, and are properly attributable to such transaction.

"Non-Consenting Lender" has the meaning specified therefor in Section 14.2(a) of this Agreement.

"Non-Defaulting Lender" means each Lender other than a Defaulting Lender.

"Obligations" means (a) all loans (including the Revolving Loans (inclusive of Extraordinary Advances and Swing Loans)), debts, principal, interest (including any interest that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), reimbursement or indemnification obligations with respect to Letters of Credit (irrespective of whether contingent), premiums, liabilities (including all amounts charged to the Loan Account pursuant to this Agreement), obligations (including indemnification obligations), fees (including the fees provided for in the Fee Letter), Lender Group Expenses (including any fees or expenses that accrue after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), guaranties, and all covenants and duties of any other kind and description owing by any Loan Party arising out of, under, pursuant to, in connection with, or evidenced by this Agreement or any of the other Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all other expenses or other amounts that any Loan Party is required to pay or reimburse by the Loan Documents or by law or otherwise in connection with the Loan Documents, and (b) all Bank Product Obligations; <u>provided</u> that, anything to the contrary contained in the foregoing notwithstanding, the Obligations shall exclude any Excluded Swap Obligation. Without limiting the generality of the foregoing, the Obligations of Borrowers under the Loan Documents include the obligation to pay (i) the principal of the Revolving Loans, (ii) interest accrued on the Revolving Loans, (iii) the amount necessary to reimburse Issuing Bank for amounts paid or payable pursuant to Letters of Credit, (iv) Letter of Cre

-27-

Lender Group Expenses, (vi) fees payable under this Agreement or any of the other Loan Documents, and (vii) indemnities and other amounts payable by any Loan Party under any Loan Document. Any reference in this Agreement or in the Loan Documents to the Obligations shall include all or any portion thereof and any extensions, modifications, renewals, or alterations thereof, both prior and subsequent to any Insolvency Proceeding.

"OFAC" means The Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Originating Lender" has the meaning specified therefor in Section 13.1(e) of this Agreement.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction or taxing authority imposing the tax (other than any such connection arising solely from such Recipient having executed, delivered, become a party to, or performed its obligations or received payment under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced its rights or remedies under this Agreement or any other Loan Document).

"Other Taxes" means all present or future stamp, court, documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to <u>Section 14.2</u>).

"Overadvance" means, as of any date of determination, that the Revolver Usage is greater than any of the limitations set forth in Section 2.11 or Section 2.11 of this Agreement.

"Pandora" has the meaning specified therefor in the preamble to this Agreement.

"Participant" has the meaning specified therefor in Section 13.1(e) of this Agreement.

"Participant Register" has the meaning set forth in Section 13.1(i) of this Agreement.

"Patent Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Patriot Act" has the meaning specified therefor in Section 4.13 of this Agreement.

"Payment Conditions" means, at the time of determination with respect to a proposed payment to fund a Specified Transaction, that:

(a) no Default or Event of Default then exists or would arise as a result of the consummation of such Specified Transaction,

(b) either

(i) Liquidity (x) at all times during the 30 consecutive days immediately preceding the date of such proposed payment and the consummation of such Specified Transaction, calculated on a *pro forma basis* as if such proposed payment was made, and the Specified Transaction was consummated, on the first day of such period, and (y) after giving effect to such proposed payment and Specified Transaction, in each case, is not less than the greater of (1) 30% of the Line Cap, and (2) 33,000,000, or

(ii) both (A) the Fixed Charge Coverage Ratio of the Loan Parties and their Subsidiaries is equal to or greater than 1.10:1.00 for the trailing 12 month period most recently ended for which

-28-

financial statements are required to have been delivered to Agent pursuant to <u>Schedule 5.1</u> to this Agreement (calculated on a pro forma basis as if such proposed payment is a Fixed Charge made on the last day of such 12 month period (it being understood that such proposed payment shall also be a Fixed Charge made on the last day of such 12 month period for purposes of calculating the Fixed Charge Coverage Ratio under this clause (ii) for any subsequent proposed payment to fund a Specific Transaction)), and (B) Liquidity (x) at all times during the 30 consecutive days immediately preceding the date of such proposed payment and the consummation of such Specified Transaction, calculated on a *pro forma basis* as if such proposed payment was made, and the Specified Transaction was consummated, on the first day of such period, and (y) after giving effect to such proposed payment and Specified Transaction, in each case, is not less than the greater of (1) 20% of the Line Cap, and (2) \$20,000,000, and

satisfied.

(c) Administrative Borrower has delivered a certificate to Agent certifying that all conditions described in clauses (a) and (b) above have been

"Perfection Certificate" means a certificate in the form of Exhibit P-1 to this Agreement.

"Permitted Acquisition" means any Acquisition so long as:

(a) no Default or Event of Default shall have occurred and be continuing or would result from the consummation of the proposed Acquisition and the proposed Acquisition is consensual,

(b) no Indebtedness will be incurred, assumed, or would exist with respect to any Loan Party or its Subsidiaries as a result of such Acquisition, other than Indebtedness permitted under clause (g) of the definition of Permitted Indebtedness and no Liens will be incurred, assumed, or would exist with respect to the assets of any Loan Party or its Subsidiaries as a result of such Acquisition other than Permitted Liens,

(c) [reserved],

(d) Upon the request of Agent in its Permitted Discretion, Borrowers have provided Agent with its due diligence package relative to the proposed Acquisition, including forecasted balance sheets, profit and loss statements, and cash flow statements of the Person or assets to be acquired, all prepared on a basis consistent with such Person's (or assets') historical financial statements, together with appropriate supporting details and a statement of underlying assumptions for the one year period following the date of the proposed Acquisition, on a quarter by quarter basis), in form and substance (including as to scope and underlying assumptions) reasonably satisfactory to Agent,

(e) [reserved],

(f) the assets being acquired or the Person whose Equity Interests are being acquired did not have negative EBITDA during the 12 consecutive month period most recently concluded prior to the date of the proposed Acquisition, unless otherwise agreed to by Agent; provided, <u>however</u>, that this clause (f) does not need to be satisfied with respect to the Acquisition of the entity identified to Agent as "Almaden" prior to the date hereof ("<u>Almaden</u>"), so long as the EBITDA of Almaden during the 12 consecutive month period most recently concluded prior to the date of the proposed Acquisition is between negative \$5,000,000 and \$0.

(g) Borrowers have provided Agent with written notice of the proposed Acquisition at least 15 Business Days prior to the anticipated closing date of the proposed Acquisition and, not later than five Business Days prior to the anticipated closing date of the proposed Acquisition, copies of the acquisition

-29-

agreement and other material documents relative to the proposed Acquisition, which agreement and documents must be reasonably acceptable to Agent,

(h) the assets being acquired (other than a *de minimis* amount of assets in relation to Borrowers' and their Subsidiaries' total assets), or the Person whose Equity Interests are being acquired, are useful in or engaged in, as applicable, the business of the Loan Parties and their Subsidiaries or a business reasonably related thereto,

(i) the assets being acquired (other than a *de minimis* amount of assets in relation to the assets being acquired) are located within the United States, Canada or the United Kingdom, or the Person whose Equity Interests are being acquired is organized in a jurisdiction located within the United States, Canada or the United Kingdom,

(j) the subject assets or Equity Interests, as applicable, are being acquired directly by a Borrower or one of its Subsidiaries that is a Loan Party, and, in connection therewith, the applicable Loan Party shall have complied with <u>Section 5.11</u> or <u>5.12</u> of this Agreement, as applicable, of this Agreement and, in the case of an acquisition of Equity Interests, the Person whose Equity Interests are acquired shall become a Loan Party and the applicable Loan Party shall have demonstrated to Agent that the new Loan Parties have received consideration sufficient to make the joinder documents binding and enforceable against such new Loan Parties, and

(k) the Payment Conditions are satisfied.

"Permitted Convertible Notes" means any notes issued by Administrative Borrower that are convertible into common stock of Administrative Borrower or cash in lieu of all or any portion of such shares of common stock <u>provided</u> that (a) the stated final maturity thereof shall be no earlier than 91 days after the Maturity Date, and shall not be subject to any conditions that could result in such stated final maturity occurring on a date that precedes the 91st day after the Maturity Date (it being understood that any purchased of such notes on account of the occurrence of a "fundamental change" or any redemption of such notes at the option of Administrative Borrower shall not be deemed to constitute a change in the stated final maturity thereof), (b) such notes shall not be required to be repaid, prepaid, redeemed, repurchased or defeased, whether on one or more fixed dates, upon the occurrence of one or more events or at the option of any holder thereof (except, in each case, upon the occurrence of an event of default, a "fundamental change" or following the Administrative Borrower's election to redeem such notes) prior to the 91st day after the Maturity Date, (c) the terms, conditions and covenants of such notes shall be such as are typical and customary for notes of such type (as determined by the Board of Directors of Administrative Borrower in good faith), (d) no Subsidiary that is not a Loan Party shall guarantee obligations of Administrative Borrower thereunder, and each guarantee shall provide for the release and termination thereof, without action by any Person, upon release and termination of the guarantee by such Subsidiary of the Obligations, and (e) the obligations in respect thereof (and any guarantee thereof) shall not be secured by any Lien on any asset of the Administrative Borrower or any Subsidiary.

judgment.

"Permitted Discretion" means a determination made in the exercise of reasonable (from the perspective of a secured asset-based lender) business

"Permitted Dispositions" means:

(a) sales, abandonment, or other dispositions of Equipment that is substantially worn, damaged, or obsolete or no longer used or useful in the ordinary course of business and leases or subleases of Real Property not useful in the conduct of the business of the Loan Parties and their Subsidiaries,

(b) sales of Inventory to buyers in the ordinary course of business,

-30-

(c) the use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents,

(d) the licensing, on a non-exclusive basis, of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,

(e) the granting of Permitted Liens,

(f) the sale or discount, in each case without recourse, of accounts receivable (other than Eligible Accounts) arising in the ordinary course of business, but only in connection with the compromise or collection thereof,

(g) any involuntary loss, damage or destruction of property,

(h) any involuntary condemnation, seizure or taking, by exercise of the power of eminent domain or otherwise, or confiscation or requisition of use of property,

(i) the leasing or subleasing of assets of any Loan Party or its Subsidiaries in the ordinary course of business,

(j) the sale or issuance of Equity Interests (other than Disqualified Equity Interests) of Administrative Borrower,

(k) (i) the lapse of registered patents, trademarks, copyrights and other intellectual property of any Loan Party or any of its Subsidiaries to the extent not economically desirable in the conduct of its business, or (ii) the abandonment of patents, trademarks, copyrights, or other intellectual property rights in the ordinary course of business so long as (in each case under clauses (i) and (ii)), (A) with respect to copyrights, such copyrights are not material revenue generating copyrights, and (B) such lapse is not materially adverse to the interests of the Lender Group,

(1) the making of Restricted Payments that are expressly permitted to be made pursuant to this Agreement,

(m) the making of Permitted Investments,

(n) so long as no Event of Default has occurred and is continuing or would immediately result therefrom, transfers of assets (i) from any Loan Party or any of its Subsidiaries to a Loan Party, and (ii) from any Subsidiary of any Loan Party that is not a Loan Party to any other Subsidiary of any Loan Party,

(o) dispositions of Equipment or Real Property to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property, or (ii) the proceeds of such disposition are promptly applied to the purchase price of such replacement property; provided, that to the extent the property being transferred constitutes Collateral, such replacement property shall constitute Collateral,

(p) the sale, for all cash consideration, of the Eventbrite Note, so long as no Event of Default has occurred and is continuing or would result therefrom, and

(q) sales or other dispositions of assets not otherwise permitted in clauses (a) through (p) above (other than sales or other dispositions of Accounts in connection with securitization or factoring arrangements), so long as (i) no Default or Event of Default has occurred and is continuing or would

-31-

immediately result therefrom, (ii) each such sale or disposition is in an arm's-length transaction and the applicable Loan Party or its Subsidiary receives at least the fair market value of the assets so disposed, (iii) the consideration received by the applicable Loan Party or its Subsidiary consists of at least 75% cash and Cash Equivalents and is paid at the time of the closing of such sale or disposition, and (iv) the aggregate amount of the cash and non-cash proceeds received from all assets sold or disposed of pursuant to this clause (q) shall not exceed \$5,000,000 in any fiscal year of Administrative Borrower (for this purpose, using the fair market value of property other than cash);

provided, that if, as of any date of determination, sales or dispositions by the Loan Parties during the period of time from the first day of the month in which such date of determination occurs until such date of determination, either individually or in the aggregate, involve \$1,000,000 or more of assets included in the Borrowing Base (based on the fair market value of the assets so disposed) (the "<u>Threshold Amount</u>"), then Borrowers shall have, prior to consummation of the sale or disposition that causes the assets included in the Borrowing that are disposed of during such period to exceed the Threshold Amount, delivered to Agent an updated Borrowing Base Certificate that reflects the removal of the applicable assets from the Borrowing Base.

Notwithstanding the foregoing, no Loan Party nor any Subsidiary shall grant any exclusive license of, or sell, transfer or otherwise dispose of, in one transaction or a series of transactions, Intellectual Property material to the conduct of business of the Loan Parties and the Subsidiaries, taken as a whole, other than (i) licenses to a Borrower or any Domestic Subsidiary that is a Loan Party, (ii) licenses to any Foreign Subsidiary of any Intellectual Property that are exclusive as to any jurisdiction or jurisdictions, in each case other than the United States of America and Canada, or any sale, transfer, or other disposition to any Foreign Subsidiary of any Intellectual Property that is not material to the conduct of business of the Loan Parties and the Subsidiaries, taken as a whole, in the United States of America or Canada; provided that each such Foreign Subsidiary (A) shall be a wholly owned Subsidiary all the Equity Interests in which are directly owned by one or more of the Loan Parties and shall have been pledged to the Agent, for the benefit of the Lender Group and Bank Product Providers, in accordance with (and to the extent required by) this Agreement and the Guaranty and Security Agreement and (B) shall not create, incur, assume or permit to exist any Indebtedness (other than any Indebtedness owed to a Loan Party)) (each Foreign Subsidiary described in this clause (ii) being referred to as the "Foreign IP Holdco"), (iii) licenses by any Foreign IP Holdco to any Foreign Subsidiary or joint venture of any Jurisdiction or jurisdiction or jurisdictions, in each case other than the United States of America and Canada, in which such Foreign Subsidiary or joint venture conducts business, (iv) licenses to any jurisdiction or jurisdiction or jurisdiction or joint venture that conducts business in Canada of any Intellectual Property that are exclusive as to any jurisdiction or jurisdiction or jurisdiction at the United States of America and Canada, in which such Foreign Subsidiary or joint venture conducts

"Permitted Holder" means Sirius XM Radio Inc. and the Sirius Affiliates.

"Permitted Indebtedness" means:

(a) Indebtedness in respect of the Obligations other than Bank Product Obligations,

(b) Indebtedness as of the Closing Date set forth on <u>Schedule 4.14</u> to this Agreement and any Refinancing Indebtedness in respect of such Indebtedness,

- (c) Permitted Purchase Money Indebtedness and any Refinancing Indebtedness in respect of such Indebtedness,
- (d) Indebtedness arising in connection with the endorsement of instruments or other payment items for deposit,

-32-

(e) Indebtedness consisting of (i) unsecured guarantees incurred in the ordinary course of business with respect to surety and appeal bonds, performance bonds, bid bonds, appeal bonds, completion guarantee and similar obligations; (ii) unsecured guarantees arising with respect to customary indemnification obligations to purchasers in connection with Permitted Dispositions; and (iii) unsecured guarantees with respect to Indebtedness of any Loan Party or one of its Subsidiaries, to the extent that the Person that is obligated under such guaranty could have incurred such underlying Indebtedness,

(f) [reserved],

(g) Acquired Indebtedness in an amount not to exceed \$10,000,000 outstanding at any one time,

(h) [reserved],

(i) Indebtedness owed to any Person providing property, casualty, liability, or other insurance to any Loan Party or any of its Subsidiaries, so long as the amount of such Indebtedness is not in excess of the amount of the unpaid cost of, and shall be incurred only to defer the cost of, such insurance for the year in which such Indebtedness is incurred and such Indebtedness is outstanding only during such year,

(j) the incurrence by any Loan Party or its Subsidiaries of Indebtedness under Hedge Agreements that is incurred for the bona fide purpose of hedging the interest rate, commodity, or foreign currency risks associated with such Loan Party's or such Subsidiary's operations and not for speculative purposes,

(k) Indebtedness incurred in the ordinary course of business in respect of credit cards, credit card processing services, debit cards, stored value cards, commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards"), or Cash Management Services,

(1) unsecured Indebtedness of any Loan Party owing to employees, former employees, former officers, directors, or former directors (or any spouses, ex-spouses, or estates of any of the foregoing) incurred in connection with the repurchase or redemption by such Loan Party of the Equity Interests of Administrative Borrower that has been issued to such Persons, so long as (i) no Default or Event of Default has occurred and is continuing or would result from the incurrence of such Indebtedness, (ii) the aggregate amount of all such Indebtedness outstanding at any one time does not exceed \$500,000, and (iii) such Indebtedness is subordinated in right of payment to the Obligations on terms and conditions reasonably acceptable to Agent,

(m) contingent liabilities in respect of any indemnification obligation, adjustment of purchase price, non-compete, or similar obligation of any Loan Party incurred in connection with the consummation of one or more Permitted Acquisitions,

(n) Indebtedness composing Permitted Investments,

(o) unsecured Indebtedness incurred in respect of netting services, overdraft protection, and other like services, in each case, incurred in the ordinary course of business,

(p) unsecured Indebtedness of any Loan Party or its Subsidiaries in respect of Earn-Outs owing to sellers of assets or Equity Interests to such Loan Party or its Subsidiaries that is incurred in connection with the consummation of one or more Permitted Acquisitions so long as such unsecured Indebtedness is on terms and conditions reasonably acceptable to Agent,

-33-

(q) Indebtedness in an aggregate outstanding principal amount not to exceed \$5,000,000 at any time outstanding for all Subsidiaries of each Loan Party that are CFCs; provided, that such Indebtedness is not directly or indirectly recourse to any of the Loan Parties or of their respective assets,

(r) accrual of interest, accretion or amortization of original issue discount, or the payment of interest in kind, in each case, on Indebtedness that otherwise constitutes Permitted Indebtedness,

(s) Permitted Subordinated Notes if (i) the Loan Parties will take all such actions as shall be necessary to cause the Obligations to constitute "senior indebtedness" and "designated senior indebtedness" (however denominated) in respect of such Permitted Subordinated Notes and to enable the Lenders, or any agent on their behalf, to have and exercise any payment blockage or other remedies available or potentially available to holders of senior indebtedness under the terms of such Permitted Subordinated Notes, and (ii) at the time of incurrence thereof and after giving effect thereto, no Default or Event of Default shall have occurred and be continuing, and any Refinancing Indebtedness in respect thereof; provided that the aggregate principal amount of Indebtedness permitted by this clause (s) may not exceed \$100,000,000 at any time outstanding,

(t) any other unsecured Indebtedness incurred by any Loan Party or any of its Subsidiaries in an aggregate outstanding amount not to exceed \$10,000,000 at any one time,

(u) Permitted Convertible Notes; <u>provided</u> that (a) at the time of incurrence thereof and after giving effect thereto, (i) no Default or Event of Default shall have occurred and be continuing and (ii) the Liquidity shall not be less than the greater of (x) 16.67% of the Maximum Revolver Amount and (y) 10,000,000 and (b) Administrative Borrower shall have delivered to Agent a certificate of a financial officer of Administrative Borrower certifying that (i) all the requirements set forth in this clause (u) have been satisfied with respect to such incurrence of Permitted Convertible Notes and (ii) based on the information then available to Administrative Borrower, Administrative Borrower in good faith expects that Liquidity will not be less than the greater of (x) 16.67% of the Maximum Revolver Amount and (y) 10,000,000 at any time during the six month period immediately following such incurrence, together with a calculation in support of the satisfaction of the requirement referred to in clause (a)(ii) above, and any Refinancing Indebtedness in respect thereof; <u>provided</u> that the aggregate principal amount of Indebtedness permitted by this clause (u) may not exceed \$500,000,000 less the aggregate principal amount of any outstanding 2020 Unsecured Convertible Notes at any time outstanding,

- (v) the 2020 Unsecured Convertible Notes, and
- (w) the Preferred Equity Investment.

"<u>Permitted Intercompany Advances</u>" means loans made by (a) a Loan Party to another Loan Party, (b) a Subsidiary of a Loan Party that is not a Loan Party to another Subsidiary of a Loan Party that is not a Loan Party, (c) a Subsidiary of a Loan Party to a Loan Party to a Loan Party, so long as the parties thereto are party to the Intercompany Subordination Agreement, and (d) a Loan Party to a Subsidiary of a Loan Party that is not a Loan Party so long as (i) the aggregate amount of all such loans (by type, not by the borrower) does not exceed \$5,000,000 outstanding at any one time, (ii) at the time of the making of such loan, no Event of Default has occurred and is continuing or would result therefrom, and (iii) Borrowers have Liquidity of \$25,000,000 or greater immediately after giving effect to each such loan.

"Permitted Investments" means:

(a) Investments in cash and Cash Equivalents,

-34-

- (b) Investments in negotiable instruments deposited or to be deposited for collection in the ordinary course of business,
- (c) advances made in connection with purchases of goods or services in the ordinary course of business,

(d) Investments received in settlement of amounts due to any Loan Party or any of its Subsidiaries effected in the ordinary course of business or owing to any Loan Party or any of its Subsidiaries as a result of Insolvency Proceedings involving an account debtor or upon the foreclosure or enforcement of any Lien in favor of a Loan Party or its Subsidiaries,

(e) Investments owned by any Loan Party or any of its Subsidiaries on the Closing Date and set forth on <u>Schedule P-1</u> to this Agreement,

(f) guarantees permitted under the definition of Permitted Indebtedness,

(g) Permitted Intercompany Advances,

(h) Equity Interests or other securities acquired in connection with the satisfaction or enforcement of Indebtedness or claims due or owing to a Loan Party or its Subsidiaries (in bankruptcy of customers or suppliers or otherwise outside the ordinary course of business) or as security for any such Indebtedness or claims,

(i) deposits of cash made in the ordinary course of business to secure performance of operating leases,

(j) (i) non-cash loans and advances to employees, officers, and directors of a Loan Party or any of its Subsidiaries for the purpose of purchasing Equity Interests in Administrative Borrower so long as the proceeds of such loans are used in their entirety to purchase such Equity Interests in Administrative Borrower, and (ii) loans and advances to employees and officers of a Loan Party or any of its Subsidiaries in the ordinary course of business for any other business purpose and in an aggregate amount not to exceed \$500,000 at any one time,

(k) Permitted Acquisitions,

(l) Investments in the form of capital contributions and the acquisition of Equity Interests made by any Loan Party in any other Loan Party (other than capital contributions to or the acquisition of Equity Interests of Administrative Borrower),

(m) Investments resulting from entering into (i) Bank Product Agreements, or (ii) agreements relative to obligations permitted under clause (j) of the definition of Permitted Indebtedness,

(n) equity Investments by any Loan Party in any Subsidiary of such Loan Party which is required by law to maintain a minimum net capital requirement or as may be otherwise required by applicable law,

(o) Investments held by a Person acquired in a Permitted Acquisition to the extent that such Investments were not made in contemplation of or in connection with such Permitted Acquisition and were in existence on the date of such Permitted Acquisition,

-35-

(p) The investment of Pandora's corporate cash assets in accordance with the investment policy approved by the Administrative Borrower's Board of Directors in January 2016 and provided to Agent prior to the Closing Date, <u>provided that</u> any issuance of a corporate note or bond pursuant to clause (e) must be Permitted Indebtedness pursuant to this Agreement.

(q) Investments consisting of non-cash consideration received in connection with Permitted Dispositions, so long as the non-cash consideration received in connection with any Permitted Disposition does not exceed 25% of the total consideration received in connection with such Permitted Disposition,

(r) any Investment in the form of a contribution of any Specified Intercompany Indebtedness to the Subsidiary that is the obligor thereunder,

(s) any Investment in the form of a contribution by the Borrower or any Subsidiary of Equity Interests in any Foreign Subsidiary to any other Foreign Subsidiary,

(t) Investments by any Loan Party or any Subsidiary that result solely from the receipt by the loan Party or such Subsidiary from any of its subsidiaries of a dividend or other Restricted Payment in the form of Equity Interests, evidences of Indebtedness or other securities (but not any additions thereto made after the date of the receipt thereof),

(u) payroll, travel and similar advances to directors and employees of the Loan Parties or any Subsidiary to cover matters that are expected at the time of such advances to be treated as expenses of the Loan Party or such Subsidiary for accounting purposes and that are made in the ordinary course of business,

(v) [reserved], and

(w) other Investments so long as the Payment Conditions are satisfied, in an amount not to exceed \$10,000,000 in the aggregate at any

one time.

"Permitted Liens" means:

(a) Liens granted to, or for the benefit of, Agent to secure the Obligations,

(b) Liens for unpaid taxes, assessments, or other governmental charges or levies that either (i) are not yet delinquent, or (ii) do not have priority over Agent's Liens and the underlying taxes, assessments, or charges or levies are the subject of Permitted Protests,

(c) judgment Liens arising solely as a result of the existence of judgments, orders, or awards that do not constitute an Event of Default under Section 8.3 of this Agreement,

(d) Liens set forth on <u>Schedule P-2</u> to this Agreement; <u>provided</u>, that to qualify as a Permitted Lien, any such Lien described on <u>Schedule P-2</u> to this Agreement shall only secure the Indebtedness that it secures on the Closing Date and any Refinancing Indebtedness in respect thereof,

(e) the interests of lessors under operating leases and non-exclusive licensors under license agreements,

(f) purchase money Liens on fixed assets or the interests of lessors under Capital Leases to the extent that such Liens or interests secure Permitted Purchase Money Indebtedness and so long as (i)

-36-

such Lien attaches only to the fixed asset purchased or acquired and the proceeds thereof, and (ii) such Lien only secures the Indebtedness that was incurred to acquire the fixed asset purchased or acquired or any Refinancing Indebtedness in respect thereof,

(g) Liens arising by operation of law in favor of warehousemen, landlords, carriers, mechanics, materialmen, laborers, or suppliers, incurred in the ordinary course of business and not in connection with the borrowing of money, and which Liens either (i) are for sums not yet delinquent, or (ii) are the subject of Permitted Protests,

(h) Liens on amounts deposited to secure any Borrower's and its Subsidiaries' obligations in connection with worker's compensation or other unemployment insurance,

(i) Liens on amounts deposited to secure any Borrower's and its Subsidiaries' obligations in connection with the making or entering into of bids, tenders, or leases in the ordinary course of business and not in connection with the borrowing of money,

(j) Liens on amounts deposited to secure any Borrower's and its Subsidiaries' reimbursement obligations with respect to surety or appeal bonds obtained in the ordinary course of business,

(k) with respect to any Real Property, easements, rights of way, and zoning restrictions that do not materially interfere with or impair the use or operation thereof,

(1) non-exclusive licenses of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,

(m) Liens that are replacements of Permitted Liens to the extent that the original Indebtedness is the subject of permitted Refinancing Indebtedness and so long as the replacement Liens only encumber those assets that secured the original Indebtedness,

(n) rights of setoff or bankers' liens upon deposits of funds in favor of banks or other depository institutions, solely to the extent incurred in connection with the maintenance of such Deposit Accounts in the ordinary course of business,

(o) Liens granted in the ordinary course of business on the unearned portion of insurance premiums securing the financing of insurance premiums to the extent the financing is permitted under the definition of Permitted Indebtedness,

(p) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods,

(q) Liens solely on any cash earnest money deposits made by a Loan Party or any of its Subsidiaries in connection with any letter of intent or purchase agreement with respect to a Permitted Acquisition,

(r) Liens assumed by any Loan Party or its Subsidiaries in connection with a Permitted Acquisition that secure Acquired Indebtedness that is Permitted Indebtedness,

(s) other Liens which do not secure Indebtedness for borrowed money or letters of credit and as to which the aggregate amount of the obligations secured thereby does not exceed \$10,000,000,

-37-

(t) in connection with the sale or transfer of any Equity Interests or other assets in a transaction permitted hereunder, customary rights and restrictions contained in agreements relating to such sale or transfer pending the completion thereof,

(u) any Lien on assets of any Foreign Subsidiary; <u>provided</u> that (i) such Lien shall not apply to any Collateral (including any Equity Interests in any Subsidiary that constitute Collateral) or any other asset of any Loan Party and (ii) such lien shall secure only Indebtedness or other obligations of such Foreign Subsidiary permitted hereunder, and

(v) in the case of (i) any Subsidiary that is not a wholly-owned Subsidiary or (ii) the Equity Interests in any Person that is not a Subsidiary, any encumbrance or restriction including any put and call arrangements, related to Equity Interests in such Subsidiary or such other Person set forth in the organizational documents of such Subsidiary or such other Person or any related joint venture shareholders or similar agreement.

Notwithstanding the foregoing, no Lien, other than Liens permitted under (a), (b), (c), (e), (f), (g), (r), and (t) above, may attach to any Account or Intellectual Property of any Loan Party.

"<u>Permitted Protest</u>" means the right of any Loan Party or any of its Subsidiaries to protest any Lien (other than any Lien that secures the Obligations), taxes (other than payroll taxes or taxes that are the subject of a United States federal tax lien), or rental payment; <u>provided</u>, that (a) a reserve with respect to such obligation is established on such Loan Party's or its Subsidiaries' books and records in such amount as is required under GAAP, (b) any such protest is instituted promptly and prosecuted diligently by such Loan Party or its Subsidiary, as applicable, in good faith, and (c) Agent is satisfied that, while any such protest is pending, there will be no impairment of the enforceability, validity, or priority of any of Agent's Liens.

"<u>Permitted Purchase Money Indebtedness</u>" means, as of any date of determination, Indebtedness (other than the Obligations, but including Capitalized Lease Obligations), incurred after the Closing Date and at the time of, or within 30 days after, the acquisition of any fixed assets for the purpose of financing all or any part of the acquisition cost thereof, in an aggregate principal amount outstanding at any one time not in excess of \$10,000,000.

"<u>Permitted Subordinated Notes</u>" means any notes issued by Administrative Borrower; <u>provided</u> that (a) such notes, and any guarantees thereof, are subordinated in right of payment to the Obligations on terms customary at the time of the issuance of such notes for high yield subordinated debt securities issued in a public offering, (b) the stated final maturity thereof shall be no earlier than 91 days after the Maturity Date, and shall not be subject to any conditions that could result in such stated final maturity occurring on a date that precedes the 91st day after the Maturity Date, (c) such notes shall not be required to be repaid, prepaid, redeemed, repurchased or defeased, whether on one or more fixed dates, upon the occurrence of one or more events or at the option of any holder thereof (except, in each case, upon the occurrence of an event of default, a change in control, an asset disposition or an event of loss) prior to the 91st day after the Maturity Date, (d) the terms, conditions and covenants of such notes shall be such as are typical and customary for notes of such type (as determined by the Board of Directors of Administrative Borrower in good faith) (e) no Subsidiary that is not a Loan Party shall guarantee obligations of the Administrative Borrower thereunder, and each such guarantee (i) shall provide for the release and termination thereof, without action by any Person, upon any release and termination of the guarantee by such Subsidiary of the Obligations and (ii) shall be subordinated to the guarantee by such Subsidiary of the Obligations of such notes, and (f) the obligations in respect thereof (and any guarantee thereof) shall not be secured by any Lien on any asset of Administrative Borrower or any Subsidiary.

-38-

"<u>Person</u>" means natural persons, corporations, limited liability companies, limited partnerships, general partnerships, limited liability partnerships, joint ventures, trusts, land trusts, business trusts, or other organizations, irrespective of whether they are legal entities, and governments and agencies and political subdivisions thereof.

"Platform" has the meaning specified therefor in Section 17.9(c) of this Agreement.

"Post-Increase Revolver Lenders" has the meaning specified therefor in Section 2.14 of this Agreement.

"Pre-Increase Revolver Lenders" has the meaning specified therefor in Section 2.14 of this Agreement.

"Preferred Equity Documents" means (i) that certain Certificate of Designation of Preferences, Rights, and Limitations of Series A Convertible Preferred Stock of Pandora, dated as of June 9, 2017, (ii) that certain Investment Agreement by and between Pandora and Sirius XM Radio Inc. dated as of June 9, 2017, or (iii) that certain Registration Rights Agreement by and between Pandora and the investors listed on the signature pages thereto, dated as of June 9, 2017, and (iv) all other documents entered into in connection with the Preferred Equity Investment (each as the same may be amended, modified, restated, or supplemented pursuant to the terms of this Agreement).

"Preferred Equity Investment" means the 480,000 shares of Pandora's Series A Convertible Preferred Stock held by Sirius XM Radio Inc. pursuant to the Preferred Equity Documents.

"Preferred Equity Investment Mandatory Redemption Date" means the Mandatory Redemption Date, as defined in the Preferred Equity Documents.

"Projections" means Borrowers' forecasted (a) balance sheets, (b) profit and loss statements, and (c) cash flow statements, all prepared on a basis consistent with Borrowers' historical financial statements, together with appropriate supporting details and a statement of underlying assumptions.

"Pro Rata Share" means, as of any date of determination:

(a) with respect to a Lender's obligation to make all or a portion of the Revolving Loans, with respect to such Lender's right to receive payments of interest, fees, and principal with respect to the Revolving Loans, and with respect to all other computations and other matters related to the Revolver Commitments or the Revolving Loans, the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender, by (ii) the aggregate Revolving Loan Exposure of all Lenders,

(b) with respect to a Lender's obligation to participate in the Letters of Credit, with respect to such Lender's obligation to reimburse Issuing Bank, and with respect to such Lender's right to receive payments of Letter of Credit Fees, and with respect to all other computations and other matters related to the Letters of Credit, the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender, by (ii) the aggregate Revolving Loan Exposure of all Lenders; <u>provided</u>, that if all of the Revolving Loans have been repaid in full and all Revolver Commitments have been terminated, but Letters of Credit remain outstanding, Pro Rata Share under this clause shall be the percentage obtained by dividing (A) the Letter of Credit Exposure of such Lender, by (B) the Letter of Credit Exposure of all Lenders, and

(c) with respect to all other matters and for all other matters as to a particular Lender (including the indemnification obligations arising under <u>Section 15.7</u> of this Agreement), the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender, by (ii) the aggregate Revolving Loan

-39-

Exposure of all Lenders, in any such case as the applicable percentage may be adjusted by assignments permitted pursuant to <u>Section 13.1</u>; provided, that if all of the Loans have been repaid in full and all Commitments have been terminated, Pro Rata Share under this clause shall be the percentage obtained by dividing (A) the Letter of Credit Exposure of such Lender, by (B) the Letter of Credit Exposure of all Lenders.

"Protective Advances" has the meaning specified therefor in Section 2.3(d)(i) of this Agreement.

"Public Lender" has the meaning specified therefor in Section 17.9(c) of this Agreement.

"<u>Purchase Price</u>" means, with respect to any Acquisition, an amount equal to the aggregate consideration, whether cash, property or securities (including the fair market value of any Equity Interests of Administrative Borrower issued in connection with such Acquisition and including the maximum amount of Earn-Outs), paid or delivered by a Loan Party or one of its Subsidiaries in connection with such Acquisition (whether paid at the closing thereof or payable thereafter and whether fixed or contingent), but excluding therefrom (a) any cash of the seller and its Affiliates used to fund any portion of such consideration, and (b) any cash or Cash Equivalents acquired in connection with such Acquisition.

"Qualified Cash" means, as of any date of determination, the amount of unrestricted cash and Cash Equivalents of the Loan Parties and their Subsidiaries that is in Deposit Accounts or in Securities Accounts, or any combination thereof, maintained by a branch office of any Lender and which is subject to a Control Agreement (subject to any extension of time to execute such Control Agreements as provided in <u>Schedule 3.6</u> hereof); provided that Qualified Cash may also include Deposit Accounts and Securities Accounts that were established with SVB Asset Management (which such account may be maintained by U.S. Bank, N.A.) or UBS Financial Services Inc. on or prior to the Closing Date so long as such Deposit Accounts and Securities Accounts are subject to a Control Agreement (subject to any extension of time to execute such Control Agreements as provided in <u>Schedule 3.6</u> hereof).

"Qualified Equity Interests" means and refers to any Equity Interests issued by Administrative Borrower (and not by one or more of its Subsidiaries) that is not a Disqualified Equity Interest.

"Real Property" means any estates or interests in real property now owned or hereafter acquired by any Loan Party or one of its Subsidiaries and the improvements thereto.

"Receivable Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to Section 2.1(c), to establish and maintain (including Landlord Reserves for books and records locations and reserves for rebates, discounts, warranty claims, and returns) with respect to the Eligible Accounts or the Maximum Revolver Amount.

"Recipient" means (a) the Agent, (b) any Lender or (c) any Participant, as applicable.

"Record" means information that is inscribed on a tangible medium or that is stored in an electronic or other medium and is retrievable in perceivable form.

"Reference Period" has the meaning set forth in the definition of EBITDA.

"Refinancing Indebtedness" means refinancings, renewals, or extensions of Indebtedness so long as:

(a) such refinancings, renewals, or extensions do not result in an increase in the principal amount of the Indebtedness so refinanced, renewed, or extended, other than by the amount of premiums paid

-40-

thereon and the fees and expenses incurred in connection therewith and by the amount of unfunded commitments with respect thereto,

(b) such refinancings, renewals, or extensions do not result in a shortening of the final stated maturity or the average weighted maturity (measured as of the refinancing, renewal, or extension) of the Indebtedness so refinanced, renewed, or extended, nor are they on terms or conditions that, taken as a whole, are or could reasonably be expected to be materially adverse to the interests of the Lenders,

(c) if the Indebtedness that is refinanced, renewed, or extended was subordinated in right of payment to the Obligations, then the terms and conditions of the refinancing, renewal, or extension must include subordination terms and conditions that are at least as favorable to the Lender Group as those that were applicable to the refinanced, renewed, or extended Indebtedness,

(d) the Indebtedness that is refinanced, renewed, or extended is not recourse to any Person that is liable on account of the Obligations other than those Persons which were obligated with respect to the Indebtedness that was refinanced, renewed, or extended,

(e) if the Indebtedness that is refinanced, renewed or extended was unsecured, such refinancing, renewal or extension shall be unsecured, and

(f) if the Indebtedness that is refinanced, renewed, or extended was secured (i) such refinancing, renewal, or extension shall be secured by substantially the same or less collateral as secured such refinanced, renewed or extended Indebtedness on terms no less favorable to Agent or the Lender Group and (ii) the Liens securing such refinancing, renewal or extension shall not have a priority more senior than the Liens securing such Indebtedness that is refinanced, renewed or extended.

"Register" has the meaning set forth in Section 13.1(h) of this Agreement.

"Registered Loan" has the meaning set forth in Section 13.1(h) of this Agreement.

"<u>Related Fund</u>" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course and that is administered, advised or managed by (a) a Lender, (b) an Affiliate of a Lender, or (c) an entity or an Affiliate of an entity that administers, advises or manages a Lender.

"Remedial Action" means all actions taken to (a) clean up, remove, remediate, contain, treat, monitor, assess, evaluate, or in any way address Hazardous Materials in the indoor or outdoor environment, (b) prevent or minimize a release or threatened release of Hazardous Materials so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (c) restore or reclaim natural resources or the environment, (d) perform any pre-remedial studies, investigations, or post-remedial operation and maintenance activities, or (e) conduct any other actions with respect to Hazardous Materials required by Environmental Laws.

"Replacement Lender" has the meaning specified therefor in Section 2.13(b) of this Agreement.

"Report" has the meaning specified therefor in Section 15.16 of this Agreement.

"<u>Required Lenders</u>" means, at any time, Lenders having or holding more than 50% of the aggregate Revolving Loan Exposure of all Lender; <u>provided</u>, that (i) the Revolving Loan Exposure of any Defaulting Lender shall be disregarded in the determination of the Required Lenders, and (ii) at any time there

-41-

are two or more Lenders (who are not Affiliates of one another or Defaulting Lenders), "Required Lenders" must include at least two Lenders (who are not Affiliates of one another).

"Reserves" means, as of any date of determination, Receivables Reserves, Bank Product Reserves, Landlord Reserves, and those other reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to Section 2.1(c), to establish and maintain (including reserves with respect to (a) sums that any Loan Party or its Subsidiaries are required to pay under any Section of this Agreement or any other Loan Document (such as taxes, assessments, insurance premiums, or, in the case of leased assets, rents or other amounts payable under such leases) and has failed to pay, and (b) amounts owing by any Loan Party or its Subsidiaries to any Person to the extent secured by a Lien on, or trust over, any of the Collateral (other than a Permitted Lien), which Lien or trust, in the Permitted Discretion of Agent likely would have a priority superior to the Agent's Liens (such as Liens or trusts in favor of landlords, warehousemen, carriers, mechanics, materialmen, laborers, or suppliers, or Liens or trusts for ad valorem, excise, sales, or other taxes where given priority under applicable law) in and to such item of the Collateral) with respect to the Borrowing Base or the Maximum Revolver Amount.

"Restricted Payment" means (a) any declaration or payment of any dividend or the making of any other payment or distribution, directly or indirectly, on account of Equity Interests issued by Administrative Borrower (including any payment in connection with any merger or consolidation involving Administrative Borrower) or to the direct or indirect holders of Equity Interests issued by Administrative Borrower or any of its Subsidiaries), or (b) any purchase, redemption, making of any sinking fund or similar payment, or other acquisition or retirement for value (including in connection with any merger or consolidation involving Administrative Borrower) any Equity Interests issued by Administrative Borrower or any of its Subsidiaries), or (b) any purchase, redemption, making of any sinking fund or similar payment, or other acquisition or retirement for value (including in connection with any merger or consolidation involving Administrative Borrower) any Equity Interests issued by Administrative Borrower or any of its Subsidiaries (other than amounts payable into Qualified Equity Interests issued by Administrative Borrower or any of its Subsidiaries), (c) any making of any payment to retire, or to obtain the surrender of, any outstanding warrants, options, or other rights to acquire Equity Interests of Administrative Borrower now or hereafter outstanding, or (d) any payment under the Permitted Convertible Notes, Permitted Subordinated Notes, 2020 Unsecured Convertible Notes, or the Preferred Equity Investment.

"<u>Revolver Commitment</u>" means, with respect to each Revolving Lender, its Revolver Commitment, and, with respect to all Revolving Lenders, their Revolver Commitments, in each case as such Dollar amounts are set forth beside such Revolving Lender's name under the applicable heading on <u>Schedule C-1</u> to this Agreement or in the Assignment and Acceptance or Increase Joinder pursuant to which such Revolving Lender became a Revolving Lender under this Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of <u>Section 13.1</u> of this Agreement, and as such amounts may be decreased by the amount of reductions in the Revolver Commitments made in accordance with <u>Section 2.4(c)</u> hereof.

"<u>Revolver Usage</u>" means, as of any date of determination, the sum of (a) the amount of outstanding Revolving Loans (inclusive of Swing Loans and Protective Advances), <u>plus</u> (b) the amount of the Letter of Credit Usage.

"Revolving Lender" means a Lender that has a Revolving Loan Exposure or Letter of Credit Exposure.

"<u>Revolving Loan Exposure</u>" means, with respect to any Revolving Lender, as of any date of determination (a) prior to the termination of the Revolver Commitments, the amount of such Lender's Revolver Commitment, and (b) after the termination of the Revolver Commitments, the aggregate outstanding principal amount of the Revolving Loans of such Lender.

"Revolving Loans" has the meaning specified therefor in Section 2.1(a) of this Agreement.



"Sanctioned Entity" means (a) a country or territory or a government of a country or territory, (b) an agency of the government of a country or territory, (c) an organization directly or indirectly controlled by a country or territory or its government, or (d) a Person resident in or determined to be resident in a country or territory, in each case of clauses (a) through (d) that is a target of Sanctions, including a target of any country sanctions program administered and enforced by OFAC.

"Sanctioned Person" means, at any time (a) any Person named on the list of Specially Designated Nationals and Blocked Persons maintained by OFAC, OFAC's consolidated Non-SDN list or any other Sanctions-related list maintained by any Governmental Authority, (b) a Person or legal entity that is a target of Sanctions, (c) any Person operating, organized or resident in a Sanctioned Entity, or (d) any Person directly or indirectly owned or controlled (individually or in the aggregate) by or acting on behalf of any such Person or Persons described in clauses (a) through (c) above.

"Sanctions," means individually and collectively, respectively, any and all economic sanctions, trade sanctions, financial sanctions, sectoral sanctions, secondary sanctions, trade embargoes anti-terrorism laws and other sanctions laws, regulations or embargoes, including those imposed, administered or enforced from time to time by: (a) the United States of America, including those administered by OFAC, the U.S. Department of State, the U.S. Department of Commerce, or through any existing or future executive order, (b) the United Nations Security Council, (c) the European Union or any European Union member state, (d) Her Majesty's Treasury of the United Kingdom, or (d) any other Governmental Authority with jurisdiction over any member of Lender Group or any Loan Party or any of their respective Subsidiaries or Affiliates.

"S&P" has the meaning specified therefor in the definition of Cash Equivalents.

"SEC" means the United States Securities and Exchange Commission and any successor thereto.

"Securities Account" means a securities account (as that term is defined in the Code).

"Securities Act" means the Securities Act of 1933, as amended from time to time, and any successor statute.

"Settlement" has the meaning specified therefor in Section 2.3(e)(i) of this Agreement.

"Settlement Date" has the meaning specified therefor in Section 2.3(e)(i) of this Agreement.

"Sirius Affiliates" means, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person; provided however, that (i) Pandora and its Subsidiaries shall not be deemed to be Sirius Affiliates and (ii) none of the Specified Persons will be treated as Sirius Affiliates for any purpose hereunder. For this purpose, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

"Solvent" means, with respect to any Person as of any date of determination, that (a) at fair valuations, the sum of such Person's debts (including contingent liabilities) is less than all of such Person's assets, (b) such Person is not engaged or about to engage in a business or transaction for which the remaining assets of such Person are unreasonably small in relation to the business or transaction or for which the property remaining with such Person is an unreasonably small capital, (c) such Person has not incurred and does not intend to incur, or reasonably believe that it will incur, debts beyond its ability to pay such debts as they become due (whether at maturity or otherwise), and (d) such Person is "solvent" or not "insolvent", as applicable within the meaning given those terms and similar terms under applicable laws relating to fraudulent transfers and conveyances. For purposes of this definition, the amount of any contingent liability at any time shall be computed as the amount

-43-

that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

"Specified Intercompany Indebtedness" means Indebtedness owing by any Subsidiary that is not a Loan Party to any Loan Party; provided that (a) such Indebtedness arises solely as a result of obligations of such Subsidiary in respect of (i) purchases, licenses and research and development cost sharing, in each case in respect of Intellectual Property, and (ii) shared services and overhead allocations and (b) no cash, cash equivalents or other assets (other than the rights in respect of Intellectual Property referred to in clause (a) above and the provision of shared services and other services giving rise to overhead allocations referred to in clause (a) above or indirectly, by any Loan Party to such Subsidiary in connection with such Indebtedness.

"Specified Persons" means John C. Malone, Liberty Interactive Corporation, Liberty Broadband Corporation, Live Nation Entertainment, Inc., Liberty Expedia Holdings, Inc., Liberty Global plc, Discovery Communications, Inc., Lions Gate Entertainment Corp., Ascent Capital Group, Inc., Charter Communications, Inc., CommerceHub, Inc., and, upon and following the Contribution Effective Time (as defined in the Agreement and Plan of Reorganization, dated as of April 4, 2017, by and among Liberty Interactive Corporation, Liberty Interactive LLC and General Communication Inc.), GCI Liberty, Inc.

"Specified Transaction" means, any Disposition, Investment, prepayment of Indebtedness or Restricted Payment (or declaration of any prepayment or Restricted Payment).

"Standard Letter of Credit Practice" means, for Issuing Bank, any domestic or foreign law or letter of credit practices applicable in the city in which Issuing Bank issued the applicable Letter of Credit or, for its branch or correspondent, such laws and practices applicable in the city in which it has advised, confirmed or negotiated such Letter of Credit, as the case may be, in each case, (a) which letter of credit practices are of banks that regularly issue letters of credit in the particular city, and (b) which laws or letter of credit practices are required or permitted under ISP or UCP, as chosen in the applicable Letter of Credit.

"Station" means, at any time and with respect to the radio broadcast stations of any Loan Party or its Subsidiaries as acquired, directly or indirectly, by any Loan Party or its Subsidiaries thereof after the Closing Date pursuant to a transaction permitted under the Loan Documents; provided that any such radio broadcast station that ceases to be owned, directly or indirectly, by a Loan Party or its Subsidiaries pursuant to a transaction permitted under the Loan Documents; provided that any Documents shall, upon consummation of such transaction, cease to be a "Station" hereunder. This definition of "Station" may be used with respect to any single radio station meeting any of the preceding requirements or all such radio stations, as the context requires.

"Subject Holder" has the meaning specified therefor in Section 2.4(e)(v) of this Agreement.

"Subsidiary" of a Person means a corporation, partnership, limited liability company, or other entity in which that Person directly or indirectly owns or controls the Equity Interests having ordinary voting power to elect a majority of the Board of Directors of such corporation, partnership, limited liability company, or other entity.

"Supermajority Lenders" means, at any time, Revolving Lenders having or holding more than 66 2/3% of the aggregate Revolving Loan Exposure of all Revolving Lenders; provided, that (i) the Revolving Loan Exposure of any Defaulting Lender shall be disregarded in the determination of the Supermajority Lenders, and (ii) at any time there are two or more Revolving Lenders (who are not Affiliates of one another), "Supermajority Lenders" must include at least two Revolving Lenders (who are not Affiliates of one another).



"Swap Obligation" means, with respect to any Loan Party, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of section 1a(47) of the Commodity Exchange Act.

"Swing Lender" means Wells Fargo or any other Lender that, at the request of Borrowers and with the consent of Agent agrees, in such Lender's sole discretion, to become the Swing Lender under <u>Section 2.3(b)</u> of this Agreement.

"Swing Loan" has the meaning specified therefor in Section 2.3(b) of this Agreement.

"Swing Loan Exposure" means, as of any date of determination with respect to any Lender, such Lender's Pro Rata Share of the Swing Loans on

such date.

"Taxes" means any taxes, levies, imposts, duties, fees, assessments or other charges imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein, and all interest and penalties with respect thereto.

"Tax Lender" has the meaning specified therefor in Section 14.2(a) of this Agreement.

"Trademark Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"<u>UCP</u>" means, with respect to any Letter of Credit, the Uniform Customs and Practice for Documentary Credits 2007 Revision, International Chamber of Commerce Publication No. 600 and any version or revision thereof accepted by Issuing Bank for use.

"Unfinanced Capital Expenditures" means Capital Expenditures (a) not financed with the proceeds of any incurrence of Indebtedness (other than the incurrence of any Revolving Loans), the proceeds of any sale or issuance of Equity Interests or equity contributions, the proceeds of any asset sale (other than the sale of Inventory in the ordinary course of business) or any insurance proceeds, and (b) that are not reimbursed by a third person (excluding any Loan Party or any of its Affiliates) in the period such expenditures are made pursuant to a written agreement.

"United States" means the United States of America.

"Unused Line Fee" has the meaning specified therefor in Section 2.10(b) of this Agreement.

"U.S. Tax Compliance Certificate" has the meaning specified therefor in Section 16.2(a)(i) of this Agreement.

"Voidable Transfer" has the meaning specified therefor in Section 17.8 of this Agreement.

"Wells Fargo" means Wells Fargo Bank, National Association, a national banking association.

"Withholding Agent" means any Loan Party and the Agent.

"<u>Write-Down and Conversion Powers</u>" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.



"2020 Unsecured Convertible Notes," means those certain 1.75% Convertible Senior Notes due 2020, issued by Pandora.

"2020 Unsecured Convertible Notes Initial Maturity Date" means December 1, 2020.

Accounting Terms. All accounting terms not specifically defined herein shall be construed in accordance with GAAP; provided, that if 1.2 Administrative Borrower notifies Agent that Borrowers request an amendment to any provision hereof to eliminate the effect of any Accounting Change occurring after the Closing Date or in the application thereof on the operation of such provision (or if Agent notifies Administrative Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such Accounting Change or in the application thereof, then Agent and Borrowers agree that they will negotiate in good faith amendments to the provisions of this Agreement that are directly affected by such Accounting Change with the intent of having the respective positions of the Lenders and Borrowers after such Accounting Change conform as nearly as possible to their respective positions immediately before such Accounting Change took effect and, until any such amendments have been agreed upon and agreed to by the Required Lenders, the provisions in this Agreement shall be calculated as if no such Accounting Change had occurred. When used herein, the term "financial statements" shall include the notes and schedules thereto. Whenever the term "Borrowers" is used in respect of a financial covenant or a related definition, it shall be understood to mean the Loan Parties and their Subsidiaries on a consolidated basis, unless the context clearly requires otherwise. Notwithstanding anything to the contrary contained herein, (a) all financial statements delivered hereunder shall be prepared, and all financial covenants contained herein shall be calculated, without giving effect to any election under the Statement of Financial Accounting Standards Board's Accounting Standards Codification Topic 825 (or any similar accounting principle) permitting a Person to value its financial liabilities or Indebtedness at the fair value thereof, and (b) the term "unqualified opinion" as used herein to refer to opinions or reports provided by accountants shall mean an opinion or report that is (i) unqualified, and (ii) does not include any explanation, supplemental comment, or other comment concerning the ability of the applicable Person to continue as a going concern or concerning the scope of the audit.

1.3 <u>Code</u>. Any terms used in this Agreement that are defined in the Code shall be construed and defined as set forth in the Code unless otherwise defined herein; <u>provided</u>, that to the extent that the Code is used to define any term herein and such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 of the Code shall govern.

1.4 <u>Construction</u>. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, references to the plural include the singular, references to the singular include the plural, the terms "includes" and "including" are not limiting, and the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or." The words "hereof," "herein," "hereby," "hereunder," and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be, scetcion, subsection, clause, schedule, and exhibit references herein are to this Agreement unless otherwise specified. Any reference in this Agreement or in any other Loan Document to any agreement, instrument, or document shall include all alterations, amendments, changes, extensions, modifications, replacements, substitutions, joinders, and supplements, thereto and thereof, as applicable (subject to any restrictions on such alterations, amendments, changes, extensions, modifications, renewals, replacements, replacements, substitutions, joinders, and supplements and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties. Any reference herein or in any other Loan Document to the satisfaction, repayment, or payment in full of the Obligations shall mean (a) the payment or repayment in full in immediately available funds of (i) the principal amount of, and interest accrued and unpaid with respect to, all outstanding Loans, together with the payment of any premium applicable to the repayment of the Loans, (ii) all Lender Group Expenses that have accrued and are unpaid regardless of whether demand has been made therefor, and (iii) all fees or charges that have accrued hereunder

-46-

or under any other Loan Document (including the Letter of Credit Fee and the Unused Line Fee) and are unpaid, (b) in the case of contingent reimbursement obligations with respect to Letters of Credit, providing Letter of Credit Collateralization, (c) in the case of obligations with respect to Bank Products (other than Hedge Obligations), providing Bank Product Collateralization, (d) the receipt by Agent of cash collateral in order to secure any other contingent Obligations for which a claim or demand for payment has been made on or prior to such time or in respect of matters or circumstances known to Agent or a Lender at such time that are reasonably expected to result in any loss, cost, damage, or expense (including attorneys' fees and legal expenses), such cash collateral to be in such amount as Agent reasonably determines is appropriate to secure such contingent Obligations, (e) the payment or repayment in full in immediately available funds of all other outstanding Obligations (including the payment of any termination amount then applicable (or which would or could become applicable as a result of the repayment of the other Obligations) under Hedge Agreements provided by Hedge Providers) other than (i) unasserted contingent indemnification Obligations, (ii) any Bank Product Obligations (other than Hedge Obligations) that, at such time, are allowed by the applicable Bank Product Provider to remain outstanding without being required to be repaid or cash collateralized, and (ii) any Hedge Obligations that, at such time, are allowed by the applicable Hedge Provider to remain outstanding without being required to be repaid, and (f) the termination of all of the Commitments of the Lenders. Any reference herein to any Person shall be construed to include such Person's successors and assigns. Any requirement of a writing contained herein or in any other Loan Document shall be satisfied by the transmission of a Record.

1.5 <u>Time References</u>. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, all references to time of day refer to Pacific standard time or Pacific daylight saving time, as in effect in San Francisco, California on such day. For purposes of the computation of a period of time from a specified date to a later specified date, unless otherwise expressly provided, the word "from" means "from and including" and the words "to" and "until" each means "to and including"; <u>provided</u>, that with respect to a computation of fees or interest payable to Agent or any Lender, such period shall in any event consist of at least one full day.

1.6 Schedules and Exhibits. All of the schedules and exhibits attached to this Agreement shall be deemed incorporated herein by reference.

2. LOANS AND TERMS OF PAYMENT.

2.1 Revolving Loans.

(a) Subject to the terms and conditions of this Agreement, and during the term of this Agreement, each Revolving Lender agrees (severally, not jointly or jointly and severally) to make revolving loans ("<u>Revolving Loans</u>") to Borrowers in an amount at any one time outstanding not to exceed *the lesser of*:

- (i) such Lender's Revolver Commitment, or
- (ii) such Lender's Pro Rata Share of an amount equal to the lesser of:

(A) the amount equal to (1) the Maximum Revolver Amount, <u>less</u> (2) the sum of (y) the Letter of Credit Usage at such time, <u>plus</u> (z) the principal amount of Swing Loans outstanding at such time, and

(B) the amount equal to (1) the Borrowing Base as of such date (based upon the most recent Borrowing Base Certificate delivered by Borrowers to Agent, as adjusted for Reserves established by Agent in accordance with Section 2.1(c)), less (2) the sum of (x) the Letter of Credit Usage at such time, <u>plus</u> (y) the principal amount of Swing Loans outstanding at such time.

-47-

(b) Amounts borrowed pursuant to this <u>Section 2.1</u> may be repaid and, subject to the terms and conditions of this Agreement, reborrowed at any time during the term of this Agreement. The outstanding principal amount of the Revolving Loans, together with interest accrued and unpaid thereon, shall constitute Obligations and shall be due and payable on the Maturity Date or, if earlier, on the date on which they otherwise become due and payable pursuant to the terms of this Agreement.

(c) Anything to the contrary in this <u>Section 2.1</u> notwithstanding, Agent shall have the right (but not the obligation) at any time, in the exercise of its Permitted Discretion, to establish and increase or decrease Reserves against the Borrowing Base or the Maximum Revolver Amount. The amount of any Reserve established by Agent, and any changes to the eligibility criteria set forth in the definitions of Eligible Accounts shall have a reasonable relationship to the event, condition, other circumstance, or fact that is the basis for such reserve or change in eligibility and shall not be duplicative of any other reserve established and currently maintained or eligibility criteria. Upon establishment or increase in Reserves, Agent agrees to make itself available to discuss the Reserve or increase, and Borrowers may take such action as may be required so that the event, condition, circumstance, or fact that is the basis for such reserve or increase no longer exists, in a manner and to the extent reasonably satisfactory to Agent in the exercise of its Permitted Discretion. In no event shall such opportunity limit the right of Agent to establish or change such Reserve, unless Agent shall have determined, in its Permitted Discretion, that the event, condition, other circumstance, or fact that was the basis for such Reserve or such change no longer exists or has otherwise been adequately addressed by Borrowers.

2.2 [Intentionally Omitted].

2.3 Borrowing Procedures and Settlements.

(a) **Procedure for Borrowing Revolving Loans**. Each Borrowing shall be made by a written request by an Authorized Person delivered to Agent (which may be delivered through Agent's electronic platform or portal) and received by Agent no later than 11:00 a.m. (i) on the Business Day that is the requested Funding Date in the case of a request for a Swing Loan, (ii) on the Business Day that is one Business Day prior to the requested Funding Date in the case of a request for a Base Rate Loan, and (iii) on the Business Day that is three Business Days prior to the requested Funding Date in the case of all other requests, specifying (A) the amount of such Borrowing, and (B) the requested Funding Date (which shall be a Business Day); provided, that Agent may, in its sole discretion, elect to accept as timely requests that are received later than 11:00 a.m. on the applicable Business Day. All Borrowing requests which are not made on-line via Agent's electronic platform or portal shall be subject to (and unless Agent elects otherwise in the exercise of its sole discretion, such Borrowings shall not be made until the completion of) Agent's authentication process (with results satisfactory to Agent) prior to the funding of any such requested Revolving Loan.

(b) **Making of Swing Loans**. In the case of a Revolving Loan and so long as any of (i) the aggregate amount of Swing Loans made since the last Settlement Date, <u>minus</u> all payments or other amounts applied to Swing Loans since the last Settlement Date, <u>plus</u> the amount of the requested Swing Loan does not exceed \$10,000,000, or (ii) Swing Lender, in its sole discretion, agrees to make a Swing Loan notwithstanding the foregoing limitation, Swing Lender shall make a Revolving Loan (any such Revolving Loan made by Swing Lender pursuant to this <u>Section 2.3(b)</u> being referred to as a "<u>Swing Loan</u>" and all such Revolving Loans being referred to as "<u>Swing Loans</u>") available to Borrowers on the Funding Date applicable thereto by transferring immediately available funds in the amount of such Borrowing to the Designated Account. Each Swing Loan shall be deemed to be a Revolving Loan hereunder and shall be subject to all the terms and conditions (including <u>Section 3</u>) applicable to other Revolving Loans, except that all payments (including interest) on any Swing Loan shall be payable to Swing Lender solely for its

-48-

own account. Subject to the provisions of Section 2.3(d)(ii), Swing Lender shall not make and shall not be obligated to make any Swing Loan if Swing Lender has actual knowledge that (i) one or more of the applicable conditions precedent set forth in Section 3 will not be satisfied on the requested Funding Date for the applicable Borrowing, or (ii) the requested Borrowing would exceed the Availability on such Funding Date. Swing Lender shall not otherwise be required to determine whether the applicable conditions precedent set forth in Section 3 have been satisfied on the Funding Date applicable thereto prior to making any Swing Loan. The Swing Loans shall be secured by Agent's Liens, constitute Revolving Loans and Obligations, and bear interest at the rate applicable from time to time to Revolving Loans that are Base Rate Loans.

(c) Making of Revolving Loans.

(i) In the event that Swing Lender is not obligated to make a Swing Loan, then after receipt of a request for a Borrowing pursuant to Section 2.3(a)(i). Agent shall notify the Lenders by telecopy, telephone, email, or other electronic form of transmission, of the requested Borrowing; such notification to be sent on the Business Day that is (A) in the case of a Base Rate Loan, at least one Business Day prior to the requested Funding Date, or (B) in the case of a LIBOR Rate Loan, prior to 11:00 a.m. at least three Business Days prior to the requested Funding Date, or (B) in the case of a requested Borrowing on the Business Day that is one Business Day prior to the requested Funding Date, the each Lender shall make the amount of such Lender's Pro Rata Share of the requested Funding Date. After Agent's receipt of the proceeds of such Revolving Loans from the Lenders, Agent shall make the proceeds thereof available to Borrowers on the applicable Funding Date by transferring immediately available funds equal to such proceeds received by Agent to the Designated Account; provided, that subject to the provisions of Section 2.3(d)(ii), no Lender shall have an obligation to make any Revolving Loan, if (1) one or more of the applicable conditions precedent set forth in Section 3 will not be satisfied on the requested Funding Date.

(ii) Unless Agent receives notice from a Lender prior to 9:30 a.m. on the Business Day that is the requested Funding Date relative to a requested Borrowing as to which Agent has notified the Lenders of a requested Borrowing that such Lender will not make available as and when required hereunder to Agent for the account of Borrowers the amount of that Lender's Pro Rata Share of the Borrowing, Agent may assume that each Lender has made or will make such amount available to Agent in immediately available funds on the Funding Date and Agent may (but shall not be so required), in reliance upon such assumption, make available to Borrowers a corresponding amount. If, on the requested Funding Date, any Lender shall not have remitted the full amount that it is required to make available to Agent in immediately available funds and if Agent has made available to Borrowers such amount on the requested Funding Date, then such Lender shall make the amount of such Lender's Pro Rata Share of the requested Borrowing available to Agent in immediately available funds, to Agent's Account, no later than 10:00 a.m. on the Business Day that is the first Business Day after the requested Funding Date (in which case, the interest accrued on such Lender's portion of such Borrowing for the Funding Date shall be for Agent's separate account). If any Lender shall not remit the full amount that it is required to make available to Agent in immediately available funds as and when required hereby and if Agent has made available to Borrowers such amount, then that Lender shall be obligated to immediately remit such amount to Agent, together with interest at the Defaulting Lender Rate for each day until the date on which such amount is so remitted. A notice submitted by Agent to any Lender with respect to amounts owing under this <u>Section 2.3(c)(ii)</u> shall be conclusive, absent manifest error. If the amount that a Lender is required to remit is made available to Agent, then such payment to Agent shall constitute such Lender's Revolving Loan for a

-49-

Administrative Borrower of such failure to fund and, upon demand by Agent, Borrowers shall pay such amount to Agent for Agent's account, together with interest thereon for each day elapsed since the date of such Borrowing, at a rate per annum equal to the interest rate applicable at the time to the Revolving Loans composing such Borrowing.

(d) Protective Advances and Optional Overadvances.

(i) Any contrary provision of this Agreement or any other Loan Document notwithstanding (but subject to Section 2.3(d)(iv)), at any time (A) after the occurrence and during the continuance of a Default or an Event of Default, or (B) that any of the other applicable conditions precedent set forth in Section 3 are not satisfied, Agent hereby is authorized by Borrowers and the Lenders, from time to time, in Agent's sole discretion, to make Revolving Loans to, or for the benefit of, Borrowers, on behalf of the Revolving Lenders, that Agent, in its Permitted Discretion, deems necessary or desirable (1) to preserve or protect the Collateral, or any portion thereof, or (2) to enhance the likelihood of repayment of the Obligations (other than the Bank Product Obligations) (the Revolving Loans described in this Section 2.3(d)(i) shall be referred to as "Protective Advances"). In any event (x) if any Protective Advance remains outstanding for more than 30 days, unless otherwise agreed to by the Required Lenders, Borrowers shall immediately repay such Protective Advances are made pursuant to this Section 2.3(d)(i). Agent's authorization to make Protective Advances may be revoked at any time by the Required Lenders delivering written notice of such revocation to Agent. Any such revocation shall become effective prospectively upon Agent's receipt thereof. Notwithstanding the foregoing, the aggregate amount of all Protective Advances and Overadvances outstanding at any one time shall not exceed 10% of the Borrowing Base.

(ii) Any contrary provision of this Agreement or any other Loan Document notwithstanding, the Lenders hereby authorize Agent or Swing Lender, as applicable, and either Agent or Swing Lender, as applicable, may, but is not obligated to, knowingly and intentionally, continue to make Revolving Loans (including Swing Loans) to Borrowers notwithstanding that an Overadvance exists or would be created thereby, so long as (A) after giving effect to such Revolving Loans and any Protective Advances then outstanding, the outstanding Revolver Usage does not exceed the Borrowing Base by more than 10% of the Borrowing Base, and (B) subject to Section 2.3(d)(iv) below, after giving effect to such Revolving Loans, the outstanding Revolver Usage (except for and excluding amounts charged to the Loan Account for interest, fees, or Lender Group Expenses) does not exceed the Maximum Revolver Amount. In the event Agent obtains actual knowledge that the Revolver Usage exceeds the amounts permitted by this Section 2.3(d), regardless of the amount of, or reason for, such excess, Agent shall notify the Lenders as soon as practicable (and prior to making any (or any additional) intentional Overadvances (except for and excluding amounts charged to the Loan Account for interest, fees, or Lender Group Expenses) unless Agent determines that prior notice would result in imminent harm to the Collateral or its value, in which case Agent may make such Overadvances and provide notice as promptly as practicable thereafter), and the Lenders with Revolver Commitments thereupon shall, together with Agent, jointly determine the terms of arrangements that shall be implemented with Borrowers intended to reduce, within a reasonable time, the outstanding principal amount of the Revolving Loans to Borrowers to an amount permitted by the preceding sentence. In such circumstances, if any Lender with a Revolver Commitment objects to the proposed terms of reduction or repayment of any Overadvance, the terms of reduction or repayment thereof shall be implemented according to the determination of the Required Lenders. In any event (x) if any Overadvance not otherwise made or permitted pursuant to this Section 2.3(d) remains outstanding for more than 30 days, unless otherwise agreed to by the Required Lenders, Borrowers shall immediately repay Revolving Loans in an amount sufficient to eliminate all such Overadvances not otherwise made or permitted to this Section

-50-

2.3(d), and (y) after the date all such Overadvances have been eliminated, there must be at least five consecutive days before additional Revolving Loans are made pursuant to this <u>Section 2.3(d)(ii)</u>. The foregoing provisions are meant for the benefit of the Lenders and Agent and are not meant for the benefit of Borrowers, which shall continue to be bound by the provisions of <u>Section 2.4(e)(1)</u>. Agent's and Swing Lender's authorization to make intentional Overadvances may be revoked at any time by the Required Lenders delivering written notice of such revocation to Agent. Any such revocation shall become effective prospectively upon Agent's receipt thereof.

(iii) Each Protective Advance and each Overadvance (each, an "<u>Extraordinary Advance</u>") shall be deemed to be a Revolving Loan hereunder, except that no Extraordinary Advance shall be eligible to be a LIBOR Rate Loan. Prior to Settlement of any Extraordinary Advance, all payments with respect thereto, including interest thereon, shall be payable to Agent solely for its own account. Each Revolving Lender shall be obligated to settle with Agent as provided in <u>Section 2.3(e)</u> (or <u>Section 2.3(g)</u>, as applicable) for the amount of such Lender's Pro Rata Share of any Extraordinary Advance. The Extraordinary Advances shall be repayable on demand, secured by Agent's Liens, constitute Obligations hereunder, and bear interest at the rate applicable from time to time to Revolving Loans that are Base Rate Loans. The provisions of this <u>Section 2.3(d)</u> are for the exclusive benefit of Agent, Swing Lender, and the Lenders and are not intended to benefit Borrowers (or any other Loan Party) in any way.

(iv) Notwithstanding anything contained in this Agreement or any other Loan Document to the contrary, no Extraordinary Advance may be made by Agent if such Extraordinary Advance would cause the aggregate Revolver Usage to exceed the Maximum Revolver Amount or any Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Revolver Commitments; provided that Agent may make Extraordinary Advances in excess of the foregoing limitations so long as such Extraordinary Advances that cause the aggregate Revolver Usage to exceed the Maximum Revolver Amount or a Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Revolver Commitments are for Agent's sole and separate account and not for the account of any Lender. No Lender shall have an obligation to settle with Agent for such Extraordinary Advances that cause the aggregate Revolver Usage to exceed the Maximum Revolver Amount or a Lender's Pro Rata Share of the Maximum Revolver Amount or a Lender's Pro Rata Share of the Share of any Lender. No Lender shall have an obligation to settle with Agent for such Extraordinary Advances that cause the aggregate Revolver Usage to exceed the Maximum Revolver Amount or a Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Pro Rata Share of the Revolver Usage to exceed such Lender's Revolver Commitments as provided in Section 2.3(e) (or Section 2.3(g), as applicable).

(e) **Settlement**. It is agreed that each Lender's funded portion of the Revolving Loans is intended by the Lenders to equal, at all times, such Lender's Pro Rata Share of the outstanding Revolving Loans. Such agreement notwithstanding, Agent, Swing Lender, and the other Lenders agree (which agreement shall not be for the benefit of Borrowers) that in order to facilitate the administration of this Agreement and the other Loan Documents, settlement among the Lenders as to the Revolving Loans (including Swing Loans and Extraordinary Advances) shall take place on a periodic basis in accordance with the following provisions:

(i) Agent shall request settlement ("<u>Settlement</u>") with the Lenders on a weekly basis, or on a more frequent basis if so determined by Agent in its sole discretion (1) on behalf of Swing Lender, with respect to the outstanding Swing Loans, (2) for itself, with respect to the outstanding Extraordinary Advances, and (3) with respect to any Loan Party's or any of their Subsidiaries' payments or other amounts received, as to each by notifying the Lenders by telecopy, telephone, or other similar form of transmission, of such requested Settlement, no later than 2:00 p.m. on the Business Day immediately prior to the date of such requested Settlement (the date of such requested Settlement being the "<u>Settlement Date</u>"). Such notice of a Settlement Date shall include a summary statement of the amount of outstanding Revolving Loans (including Swing Loans and Extraordinary Advances) for the period since the prior Settlement Date. Subject to the terms and conditions contained herein (including <u>Section 2.3(g)</u>): (y) if the amount of the Revolving Loans (including Swing Loans and Extraordinary Advances) made by a Lender that is not a

-51-

Defaulting Lender exceeds such Lender's Pro Rata Share of the Revolving Loans (including Swing Loans and Extraordinary Advances) as of a Settlement Date, then Agent shall, by no later than 12:00 p.m. on the Settlement Date, transfer in immediately available funds to a Deposit Account of such Lender (as such Lender may designate), an amount such that each such Lender shall, upon receipt of such amount, have as of the Settlement Date, its Pro Rata Share of the Revolving Loans (including Swing Loans and Extraordinary Advances) made by a Lender is less than such Lender's Pro Rata Share of the Revolving Loans (including Swing Loans and Extraordinary Advances) as of a Settlement Date, such Lender shall no later than 12:00 p.m. on the Settlement Date transfer in immediately available funds to Agent's Account, an amount such that each such Lender shall no later than 12:00 p.m. on the Settlement Date transfer in immediately available funds to Agent's Account, an amount such that each such Lender shall, upon transfer of such amount, have as of the Settlement Date, its Pro Rata Share of the Revolving Loans (including Swing Loans and Extraordinary Advances). Such amount, have as of the Settlement Cause (z) of the immediately preceding sentence shall be applied against the amounts of the applicable Swing Loans or Extraordinary Advances and, together with the portion of such Swing Loans or Extraordinary Advances representing Swing Lender's Pro Rata Share thereof, shall constitute Revolving Loans of such Lenders. If any such amount is not made available to Agent by any Lender on the Settlement Date applicable thereto to the extent required by the terms hereof, Agent shall be entitled to recover for its account such amount on demand from such Lender together with interest thereon at the Defaulting Lender Rate.

(ii) In determining whether a Lender's balance of the Revolving Loans (including Swing Loans and Extraordinary Advances) is less than, equal to, or greater than such Lender's Pro Rata Share of the Revolving Loans (including Swing Loans and Extraordinary Advances) as of a Settlement Date, Agent shall, as part of the relevant Settlement, apply to such balance the portion of payments actually received in good funds by Agent with respect to principal, interest, fees payable by Borrowers and allocable to the Lenders hereunder, and proceeds of Collateral.

(iii) Between Settlement Dates, Agent, to the extent Extraordinary Advances or Swing Loans are outstanding, may pay over to Agent or Swing Lender, as applicable, any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the Revolving Loans, for application to the Extraordinary Advances or Swing Loans. Between Settlement Dates, Agent, to the extent no Extraordinary Advances or Swing Loans are outstanding, may pay over to Swing Lender any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the Revolving Loans, for application to Swing Lender's Pro Rata Share of the Revolving Loans. If, as of any Settlement Date, payments or other amounts of the Loan Parties or their Subsidiaries received since the then immediately preceding Settlement Date have been applied to Swing Lender's Pro Rata Share of the Revolving Loans other than to Swing Loans, as provided for in the previous sentence, Swing Lender shall pay to Agent for the accounts of the Lenders, and Agent shall pay to the Lenders (other than a Defaulting Lender if Agent has implemented the provisions of <u>Section 2.3(g)</u>), to be applied to the outstanding Revolving Loans of such Lenders, an amount such that each such Lender shall, upon receipt of such amount, have, as of such Settlement Date, its Pro Rata Share of the Revolving Loans of such Lenders, and each Lender with respect to the Revolving Loans other than Swing Loans and Extraordinary Advances, shall be entitled to interest at the applicable rate or rates payable under this Agreement on the daily amount of funds employed by Swing Lender, Agent, or the Lenders, as applicable.

(iv) Anything in this <u>Section 2.3(e)</u> to the contrary notwithstanding, in the event that a Lender is a Defaulting Lender, Agent shall be entitled to refrain from remitting settlement amounts to

-52-

the Defaulting Lender and, instead, shall be entitled to elect to implement the provisions set forth in Section 2.3(g).

(f) Notation. Consistent with <u>Section 13.1(h)</u>, Agent, as a non-fiduciary agent for Borrowers, shall maintain a register showing the principal amount and stated interest of the Revolving Loans owing to each Lender, including the Swing Loans owing to Swing Lender, and Extraordinary Advances owing to Agent, and the interests therein of each Lender, from time to time and such register shall, absent manifest error, conclusively be presumed to be correct and accurate.

(g) Defaulting Lenders.

(i) Notwithstanding the provisions of Section 2.4(b)(iii), Agent shall not be obligated to transfer to a Defaulting Lender any payments made by Borrowers to Agent for the Defaulting Lender's benefit or any proceeds of Collateral that would otherwise be remitted hereunder to the Defaulting Lender, and, in the absence of such transfer to the Defaulting Lender, Agent shall transfer any such payments (A) first, to Agent to the extent of any Extraordinary Advances that were made by Agent and that were required to be, but were not, paid by Defaulting Lender, (B) second, to Swing Lender to the extent of any Swing Loans that were made by Swing Lender and that were required to be, but were not, paid by the Defaulting Lender, (C) third, to Issuing Bank, to the extent of the portion of a Letter of Credit Disbursement that was required to be, but was not, paid by the Defaulting Lender, (D) fourth, to each Non-Defaulting Lender ratably in accordance with their Commitments (but, in each case, only to the extent that such Defaulting Lender's portion of a Revolving Loan (or other funding obligation) was funded by such other Non-Defaulting Lender), (E) fifth, in Agent's sole discretion, to a suspense account maintained by Agent, the proceeds of which shall be retained by Agent and may be made available to be re-advanced to or for the benefit of Borrowers (upon the request of Borrowers and subject to the conditions set forth in Section 3.2) as if such Defaulting Lender had made its portion of Revolving Loans (or other funding obligations) hereunder, and (F) sixth, from and after the date on which all other Obligations have been paid in full, to such Defaulting Lender in accordance with tier (L) of Section 2.4(b)(iii). Subject to the foregoing, Agent may hold and, in its discretion, re-lend to Borrowers for the account of such Defaulting Lender the amount of all such payments received and retained by Agent for the account of such Defaulting Lender. Solely for the purposes of voting or consenting to matters with respect to the Loan Documents (including the calculation of Pro Rata Share in connection therewith) and for the purpose of calculating the fee payable under Section 2.10(b), such Defaulting Lender shall be deemed not to be a "Lender" and such Lender's Commitment shall be deemed to be zero; provided, that the foregoing shall not apply to any of the matters governed by Section 14.1(a)(i) through (iii). The provisions of this Section 2.3(g) shall remain effective with respect to such Defaulting Lender until the earlier of (y) the date on which all of the Non-Defaulting Lenders, Agent, Issuing Bank, and Borrowers shall have waived, in writing, the application of this Section 2.3(g) to such Defaulting Lender, or (z) the date on which such Defaulting Lender makes payment of all amounts that it was obligated to fund hereunder, pays to Agent all amounts owing by Defaulting Lender in respect of the amounts that it was obligated to fund hereunder, and, if requested by Agent, provides adequate assurance of its ability to perform its future obligations hereunder (on which earlier date, so long as no Event of Default has occurred and is continuing, any remaining cash collateral held by Agent pursuant to Section 2.3(g)(ii) shall be released to Borrowers). The operation of this Section 2.3(g) shall not be construed to increase or otherwise affect the Commitment of any Lender, to relieve or excuse the performance by such Defaulting Lender or any other Lender of its duties and obligations hereunder, or to relieve or excuse the performance by any Borrower of its duties and obligations hereunder to Agent, Issuing Bank, or to the Lenders other than such Defaulting Lender. Any failure by a Defaulting Lender to fund amounts that it was obligated to fund hereunder shall constitute a material breach by such Defaulting Lender of this Agreement and shall entitle Borrowers, at their option, upon written notice to Agent, to arrange for a substitute Lender to assume the Commitment of

-53-

such Defaulting Lender, such substitute Lender to be reasonably acceptable to Agent. In connection with the arrangement of such a substitute Lender, the Defaulting Lender shall have no right to refuse to be replaced hereunder, and agrees to execute and deliver a completed form of Assignment and Acceptance in favor of the substitute Lender (and agrees that it shall be deemed to have executed and delivered such document if it fails to do so) subject only to being paid its share of the outstanding Obligations (other than Bank Product Obligations, but including (1) all interest, fees, and other amounts that may be due and payable in respect thereof, and (2) an assumption of its Pro Rata Share of its participation in the Letters of Credit); provided, that any such assumption of the Commitment of such Defaulting Lender shall not be deemed to constitute a waiver of any of the Lender Groups' or Borrowers' rights or remedies against any such Defaulting Lender arising out of or in relation to such failure to fund. In the event of a direct conflict between the priority provisions of this Section 2.3(g) and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this Section 2.3(g) shall control and govern.

(ii) If any Swing Loan or Letter of Credit is outstanding at the time that a Lender becomes a Defaulting Lender then:

(A) such Defaulting Lender's Swing Loan Exposure and Letter of Credit Exposure shall be reallocated among the Non-Defaulting Lenders in accordance with their respective Pro Rata Shares but only to the extent (x) the sum of all Non-Defaulting Lenders' Pro Rata Share of Revolver Usage plus such Defaulting Lender's Swing Loan Exposure and Letter of Credit Exposure does not exceed the total of all Non-Defaulting Lenders' Revolver Commitments and (y) the conditions set forth in Section 3.2 are satisfied at such time;

(B) if the reallocation described in clause (A) above cannot, or can only partially, be effected, Borrowers shall within one Business Day following notice by the Agent (x) first, prepay such Defaulting Lender's Swing Loan Exposure (after giving effect to any partial reallocation pursuant to clause (A) above), and (y) second, cash collateralize such Defaulting Lender's Letter of Credit Exposure (after giving effect to any partial reallocation pursuant to clause (A) above), pursuant to a cash collateral agreement to be entered into in form and substance reasonably satisfactory to the Agent, for so long as such Letter of Credit Exposure is outstanding; <u>provided</u>, that Borrowers shall not be obligated to cash collateralize any Defaulting Lender's Letter of Credit Exposure if such Defaulting Lender is also Issuing Bank;

(C) if Borrowers cash collateralize any portion of such Defaulting Lender's Letter of Credit Exposure pursuant to this <u>Section 2.3(g)(ii)</u>, Borrowers shall not be required to pay any Letter of Credit Fees to Agent for the account of such Defaulting Lender pursuant to <u>Section 2.6(b)</u> with respect to such cash collateralized portion of such Defaulting Lender's Letter of Credit Exposure during the period such Letter of Credit Exposure is cash collateralized;

(D) to the extent the Letter of Credit Exposure of the Non-Defaulting Lenders is reallocated pursuant to this <u>Section</u> 2.3(g)(ii), then the Letter of Credit Fees payable to the Non-Defaulting Lenders pursuant to <u>Section 2.6(b)</u> shall be adjusted in accordance with such Non-Defaulting Lenders' Letter of Credit Exposure;

(E) to the extent any Defaulting Lender's Letter of Credit Exposure is neither cash collateralized nor reallocated pursuant to this $\underline{\text{Section 2.3(g)(ii)}}$, then, without prejudice to any rights or remedies of Issuing Bank or any Lender hereunder, all Letter of Credit Fees that would have otherwise been payable to such Defaulting Lender under $\underline{\text{Section 2.6(b)}}$ with respect to such portion of such



Letter of Credit Exposure shall instead be payable to Issuing Bank until such portion of such Defaulting Lender's Letter of Credit Exposure is cash collateralized or reallocated;

(F) so long as any Lender is a Defaulting Lender, the Swing Lender shall not be required to make any Swing Loan and Issuing Bank shall not be required to issue, amend, or increase any Letter of Credit, in each case, to the extent (x) the Defaulting Lender's Pro Rata Share of such Swing Loans or Letter of Credit cannot be reallocated pursuant to this Section 2.3(g)(ii), or (y) the Swing Lender or Issuing Bank, as applicable, has not otherwise entered into arrangements reasonably satisfactory to the Swing Lender or Issuing Bank, as applicable, and Borrowers to eliminate the Swing Lender's or Issuing Bank's risk with respect to the Defaulting Lender's participation in Swing Loans or Letters of Credit; and

(G) Agent may release any cash collateral provided by Borrowers pursuant to this <u>Section 2.3(g)(ii)</u> to Issuing Bank and Issuing Bank may apply any such cash collateral to the payment of such Defaulting Lender's Pro Rata Share of any Letter of Credit Disbursement that is not reimbursed by Borrowers pursuant to <u>Section 2.11(d)</u>. Subject to <u>Section 17.14</u>, no reallocation hereunder shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender arising from that Lender having become a Defaulting Lender, including any claim of a Non-Defaulting Lender's increased exposure following such reallocation.

(h) **Independent Obligations**. All Revolving Loans (other than Swing Loans and Extraordinary Advances) shall be made by the Lenders contemporaneously and in accordance with their Pro Rata Shares. It is understood that (i) no Lender shall be responsible for any failure by any other Lender to perform its obligation to make any Revolving Loan (or other extension of credit) hereunder, nor shall any Commitment of any Lender be increased or decreased as a result of any failure by any other Lender to perform its obligations hereunder, and (ii) no failure by any Lender to perform its obligations hereunder shall excuse any other Lender from its obligations hereunder.

2.4 Payments; Reductions of Commitments; Prepayments.

(a) **Payments by Borrowers**.

(i) Except as otherwise expressly provided herein, all payments by Borrowers shall be made to Agent's Account for the account of the Lender Group and shall be made in immediately available funds, no later than 1:30 p.m. on the date specified herein. Any payment received by Agent later than 1:30 p.m. shall be deemed to have been received (unless Agent, in its sole discretion, elects to credit it on the date received) on the following Business Day and any applicable interest or fee shall continue to accrue until such following Business Day.

(ii) Unless Agent receives notice from Borrowers prior to the date on which any payment is due to the Lenders that Borrowers will not make such payment in full as and when required, Agent may assume that Borrowers have made (or will make) such payment in full to Agent on such date in immediately available funds and Agent may (but shall not be so required), in reliance upon such assumption, distribute to each Lender on such due date an amount equal to the amount then due such Lender. If and to the extent Borrowers do not make such payment in full to Agent on the date when due, each Lender severally shall repay to Agent on demand such amount distributed to such Lender, together with interest thereon at the Defaulting Lender Rate for each day from the date such amount is distributed to such Lender until the date repaid.

(b) Apportionment and Application.

-55-

(i) So long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all principal and interest payments received by Agent shall be apportioned ratably among the Lenders (according to the unpaid principal balance of the Obligations to which such payments relate held by each Lender) and all payments of fees and expenses received by Agent (other than fees or expenses that are for Agent's separate account or for the separate account of Issuing Bank) shall be apportioned ratably among the Lenders having a Pro Rata Share of the type of Commitment or Obligation to which a particular fee or expense relates.

(ii) Subject to <u>Section 2.4(b)(v)</u>, <u>Section 2.4(d)(ii)</u>, and <u>Section 2.4(e)</u>, all payments to be made hereunder by Borrowers shall be remitted to Agent and all such payments, and all proceeds of Collateral received by Agent, shall be applied, so long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, to reduce the balance of the Revolving Loans outstanding and, thereafter, to Borrowers (to be wired to the Designated Account) or such other Person entitled thereto under applicable law.

(iii) At any time that an Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all payments remitted to Agent and all proceeds of Collateral received by Agent shall be applied as follows:

(A) <u>first</u>, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to Agent under the Loan Documents and to pay interest and principal on Extraordinary Advances that are held solely by Agent pursuant to the terms of <u>Section 2.4(d)(iv)</u>, until paid in full,

(B) second, to pay any fees or premiums then due to Agent under the Loan Documents, until paid in full,

- (C) third, to pay interest due in respect of all Protective Advances, until paid in full,
- (D) fourth, to pay the principal of all Protective Advances, until paid in full,

(E) <u>fifth</u>, ratably, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to any of the Lenders under the Loan Documents, until paid in full,

full,

- (F) sixth, ratably, to pay any fees or premiums then due to any of the Lenders under the Loan Documents, until paid in
- (G) <u>seventh</u>, to pay interest accrued in respect of the Swing Loans, until paid in full,
- (H) eighth, to pay the principal of all Swing Loans, until paid in full,
- (I) <u>ninth</u>, ratably, to pay interest accrued in respect of the Revolving Loans (other than Protective Advances), until paid

in full,

(J) tenth, ratably

-56-

i. ratably, to pay the principal of all Revolving Loans, until paid in full,

ii. to Agent, to be held by Agent, for the benefit of Issuing Bank (and for the ratable benefit of each of the Lenders that have an obligation to pay to Agent, for the account of Issuing Bank, a share of each Letter of Credit Disbursement), as cash collateral in an amount up to 105% of the Letter of Credit Usage (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any Letter of Credit Disbursement as and when such disbursement occurs and, if a Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this <u>Section 2.4(b)(iii)</u>, beginning with tier (A) hereof),

iii. ratably, up to the amount (after taking into account any amounts previously paid pursuant to this clause iii. during the continuation of the applicable Application Event) of the most recently established Bank Product Reserve, which amount was established prior to the occurrence of, and not in contemplation of, the subject Application Event, to (y) the Bank Product Providers based upon amounts then certified by each applicable Bank Product Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such Bank Product Provider on account of Bank Product Obligations (but not in excess of the Bank Product Reserve established for the Bank Product Obligations of such Bank Product Provider), and (z) with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Bank Product Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Bank Product Obligations owed to the applicable Bank Product Provider to the payment or reimbursement of any amounts due and payable with respect to Bank Product Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Bank Product Obligations shall be reapplied pursuant to this <u>Section 2.4(b)(iii)</u>, beginning with tier (A) hereof,

(K) <u>eleventh</u>, to pay any other Obligations other than Obligations owed to Defaulting Lenders (including being paid, ratably, to the Bank Product Providers on account of all amounts then due and payable in respect of Bank Product Obligations, with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Bank Product Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Bank Product Provider and applied by such Bank Product Provider to the payment or reimbursement of any amounts due and payable with respect to Bank Product Obligations owed to the applicable Bank Product Provider as and when such amounts first become due and payable and, if and at such time as all such Bank Product Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Bank Product Obligations shall be reapplied pursuant to this Section 2.4(b)(iii), beginning with tier (A) hereof)

- (L) twelfth, ratably to pay any Obligations owed to Defaulting Lenders; and
- (M) thirteenth, to Borrowers (to be wired to the Designated Account) or such other Person entitled thereto under

applicable law.

(iv) Agent promptly shall distribute to each Lender, pursuant to the applicable wire instructions received from each Lender in writing, such funds as it may be entitled to receive, subject to a Settlement delay as provided in <u>Section 2.3(e)</u>.

-57-

(v) In each instance, so long as no Application Event has occurred and is continuing, <u>Section 2.4(b)(ii)</u> shall not apply to any payment made by Borrowers to Agent and specified by Borrowers to be for the payment of specific Obligations then due and payable (or prepayable) under any provision of this Agreement or any other Loan Document.

(vi) For purposes of <u>Section 2.4(b)(iii)</u>, "paid in full" of a type of Obligation means payment in cash or immediately available funds of all amounts owing on account of such type of Obligation, including interest accrued after the commencement of any Insolvency Proceeding, default interest, interest on interest, and expense reimbursements, irrespective of whether any of the foregoing would be or is allowed or disallowed in whole or in part in any Insolvency Proceeding.

(vii) In the event of a direct conflict between the priority provisions of this Section 2.4 and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, if the conflict relates to the provisions of Section 2.3(g) and this Section 2.4, then the provisions of Section 2.3(g) shall control and govern, and if otherwise, then the terms and provisions of this Section 2.4 shall control and govern.

(c) **Reduction of Commitments**. The Revolver Commitments shall terminate on the Maturity Date or earlier termination thereof pursuant to the terms of this Agreement. Borrowers may reduce the Revolver Commitments to an amount not less than the sum of (A) the Revolver Usage as of such date, <u>plus</u> (B) the principal amount of all Revolving Loans not yet made as to which a request has been given by Borrowers under <u>Section 2.3(a)</u>, <u>plus</u> (C) the amount of all Letters of Credit not yet issued as to which a request has been given by Borrowers pursuant to <u>Section 2.11(a)</u>. Each such reduction shall be in an amount which is not less than \$5,000,000 (unless the Revolver Commitments are being reduced to zero and the amount of the Revolver Commitments in effect immediately prior to such reduction are less than \$5,000,000), shall be made by providing not less than ten Business Days prior written notice to Agent, and shall be irrevocable. The Revolver Commitments, once reduced, may not be increased. Each such reduction of the Revolver Commitments shall reduce the Revolver Commitments of each Lender proportionately in accordance with its ratable share thereof. In connection with any reduction in the Revolver Commitments prior to the Maturity Date, if any Loan Party or any of its Subsidiaries owns any Margin Stock, Borrowers shall deliver to Agent an updated Form U-1 (with sufficient additional originals thereof for each Lender), duly executed and delivered by the Borrowers, together with such other documentation as Agent shall reasonably request, in order to enable Agent and the Lenders to comply with any of the requirements under Regulations T, U or X of the Federal Reserve Board.

(d) **Optional Prepayments**. Borrowers may prepay the principal of any Revolving Loan at any time in whole or in part, without premium or penalty.

(e) **Mandatory Prepayments.** If, at any time, (A) the Revolver Usage on such date exceeds (B) the lesser of (x) the Borrowing Base reflected in the Borrowing Base Certificate most recently delivered by Borrowers to Agent, or (y) the Maximum Revolver Amount, in all cases as adjusted for Reserves established by Agent in accordance with <u>Section 2.1(c)</u>, then Borrowers shall immediately prepay the Obligations in accordance with <u>Section 2.4(f)</u> in an aggregate amount equal to the amount of such excess.

(f) Application of Payments. Each prepayment pursuant to <u>Section 2.4(e)</u> shall, (1) so long as no Application Event shall have occurred and be continuing, be applied, *first*, to the outstanding principal amount of the Revolving Loans until paid in full, and *second*, to cash collateralize the Letters of

-58-

Credit in an amount equal to 105% of the then outstanding Letter of Credit Usage, and (2) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in Section 2.4(b)(iii).

2.5 Promise to Pay; Promissory Notes.

(a) Borrowers agree to pay the Lender Group Expenses on the earlier of (i) the first day of the month following the date on which the applicable Lender Group Expenses were first incurred, or (ii) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the Loan Account pursuant to the provisions of <u>Section 2.6(d)</u> shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (ii)). Borrowers promise to pay all of the Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses)) in full on the Maturity Date or, if earlier, on the date on which the Obligations (other than the Bank Product Obligations) become due and payable pursuant to the terms of this Agreement. Borrowers agree that their obligations contained in the first sentence of this <u>Section 2.5(a)</u> shall survive payment or satisfaction in full of all other Obligations.

(b) Any Lender may request that any portion of its Commitments or the Loans made by it be evidenced by one or more promissory notes. In such event, Borrowers shall execute and deliver to such Lender the requested promissory notes payable to the order of such Lender in a form furnished by Agent and reasonably satisfactory to Borrowers. Thereafter, the portion of the Commitments and Loans evidenced by such promissory notes and interest thereon shall at all times be represented by one or more promissory notes in such form payable to the order of the payee named therein.

2.6 Interest Rates and Letter of Credit Fee: Rates, Payments, and Calculations.

(a) Interest Rates. Except as provided in <u>Section 2.6(c)</u>, all Obligations (except for undrawn Letters of Credit) that have been charged to the Loan Account pursuant to the terms hereof shall bear interest as follows:

-59-

Margin, and

- (i) if the relevant Obligation is a LIBOR Rate Loan, at a *per annum* rate equal to the LIBOR Rate *plus* the LIBOR Rate
- (ii) otherwise, at a *per annum* rate equal to the Base Rate *plus* the Base Rate Margin.

(b) Letter of Credit Fee. Borrowers shall pay Agent (for the ratable benefit of the Revolving Lenders), a Letter of Credit fee (the "Letter of Credit Fee") (which fee shall be in addition to the fronting fees and commissions, other fees, charges and expenses set forth in <u>Section 2.11(k)</u>) that shall accrue at a *per annum* rate equal to the LIBOR Rate Margin times the average amount of the Letter of Credit Usage during the immediately preceding month.

(c) **Default Rate**. (i) Automatically upon the occurrence and during the continuation of an Event of Default under <u>Section 8.4</u> or <u>8.5</u> and (ii) upon the occurrence and during the continuation of any other Event of Default (other than an Event of Default under <u>Section 8.4</u> or <u>8.5</u>), at the direction of Agent or the Required Lenders, and upon written notice by Agent to Borrowers of such direction (<u>provided</u>, that such notice shall not be required for any Event of Default under <u>Section 8.1</u>), (A) all Loans and all Obligations (except for undrawn Letters of Credit) that have been charged to the Loan Account pursuant to the terms hereof shall bear interest at a *per annum* rate equal to two percentage points above the per annum rate otherwise applicable thereunder, and (B) the Letter of Credit Fee shall be increased to two percentage points above the *per annum* rate otherwise applicable hereunder.

(d) **Payment**. Except to the extent provided to the contrary in Section 2.10, Section 2.11(k) or Section 2.12(a), (i) all interest and all other fees payable hereunder or under any of the other Loan Documents (other than Letter of Credit Fees) shall be due and payable, in arrears, on the first day of each month, (ii) all Letter of Credit Fees payable hereunder, and all fronting fees and all commissions, other fees, charges and expenses provided for in Section 2.11(k) shall be due and payable, in arrears, on the first Business Day of each month, and (iii) all costs and expenses payable hereunder or under any of the other Loan Documents, and all other Lender Group Expenses shall be due and payable on the earlier of (x) the first day of the month following the date on which the applicable costs, expenses, or Lender Group Expenses were first incurred, or (y) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the Loan Account pursuant to the provisions of the following sentence shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (y)). Borrowers hereby authorize Agent, from time to time without prior notice to Borrowers, to charge to the Loan Account (A) on the first day of each month, all interest accrued during the prior month on the Revolving Loans hereunder, (B) on the first Business Day of each month, all Letter of Credit Fees accrued or chargeable hereunder during the prior month, (C) as and when incurred or accrued, all fees and costs provided for in Section 2.10(a) or (c), (D) on the first day of each month, the Unused Line Fee accrued during the prior month pursuant to Section 2.10(b), (E) as and when due and payable, all other fees payable hereunder or under any of the other Loan Documents, (F) as and when incurred or accrued, all other Lender Group Expenses, and (G) as and when due and payable all other payment obligations payable under any Loan Document or any Bank Product Agreement (including any amounts due and payable to the Bank Product Providers in respect of Bank Products). All amounts (including interest, fees, costs, expenses, Lender Group Expenses, or other amounts payable hereunder or under any other Loan Document or under any Bank Product Agreement) charged to the Loan Account shall thereupon constitute Revolving Loans hereunder, shall constitute Obligations hereunder, and shall initially accrue interest at the rate then applicable to Revolving Loans that are Base Rate Loans (unless and until converted into LIBOR Rate Loans in accordance with the terms of this Agreement).

-60-

(e) **Computation**. All interest and fees chargeable under the Loan Documents shall be computed on the basis of a 360 day year, in each case, for the actual number of days elapsed in the period during which the interest or fees accrue. In the event the Base Rate is changed from time to time hereafter, the rates of interest hereunder based upon the Base Rate automatically and immediately shall be increased or decreased by an amount equal to such change in the Base Rate.

(f) Intent to Limit Charges to Maximum Lawful Rate. In no event shall the interest rate or rates payable under this Agreement, <u>plus</u> any other amounts paid in connection herewith, exceed the highest rate permissible under any law that a court of competent jurisdiction shall, in a final determination, deem applicable. Borrowers and the Lender Group, in executing and delivering this Agreement, intend legally to agree upon the rate or rates of interest and manner of payment stated within it; <u>provided</u>, that anything contained herein to the contrary notwithstanding, if such rate or rates of interest or manner of payment exceeds the maximum allowable under applicable law, then, *ipso facto*, as of the date of this Agreement, Borrowers are and shall be liable only for the payment of such maximum amount as is allowed by law, and payment received from Borrowers in excess of such legal maximum, whenever received, shall be applied to reduce the principal balance of the Obligations to the extent of such excess.

2.7 <u>Crediting Payments</u>. The receipt of any payment item by Agent shall not be required to be considered a payment on account unless such payment item is a wire transfer of immediately available funds made to Agent's Account or unless and until such payment item is honored when presented for payment. Should any payment item not be honored when presented for payment, then Borrowers shall be deemed not to have made such payment. Anything to the contrary contained herein notwithstanding, any payment item shall be deemed received by Agent only if it is received into Agent's Account on a Business Day on or before 1:30 p.m. If any payment item is received into Agent's Account on a non-Business Day or after 1:30 p.m. on a Business Day (unless Agent, in its sole discretion, elects to credit it on the date received), it shall be deemed to have been received by Agent as of the opening of business on the immediately following Business Day.

2.8 **Designated Account**. Agent is authorized to make the Revolving Loans, and Issuing Bank is authorized to issue the Letters of Credit, under this Agreement based upon telephonic or other instructions received from anyone purporting to be an Authorized Person or, without instructions, if pursuant to <u>Section</u> <u>2.6(d)</u>. Borrowers agree to establish and maintain the Designated Account with the Designated Account Bank for the purpose of receiving the proceeds of the Revolving Loans requested by Borrowers and made by Agent or the Lenders hereunder. Unless otherwise agreed by Agent and Borrowers, any Revolving Loan or Swing Loan requested by Borrowers and made by Agent or the Lenders hereunder shall be made to the Designated Account.

2.9 <u>Maintenance of Loan Account; Statements of Obligations</u>. Agent shall maintain an account on its books in the name of Borrowers (the "Loan <u>Account</u>") on which Borrowers will be charged with all Revolving Loans (including Extraordinary Advances and Swing Loans) made by Agent, Swing Lender, or the Lenders to Borrowers or for Borrowers' account, the Letters of Credit issued or arranged by Issuing Bank for Borrowers' account, and with all other payment Obligations hereunder or under the other Loan Documents, including, accrued interest, fees and expenses, and Lender Group Expenses. In accordance with <u>Section</u> <u>2.7</u>, the Loan Account will be credited with all payments received by Agent from Borrowers or for Borrowers' account. Agent shall make available to Borrowers monthly statements regarding the Loan Account, including the principal amount of the Revolving Loans, interest accrued hereunder, fees accrued or charged hereunder or under the other Loan Documents, and a summary itemization of all charges and expenses constituting Lender Group Expenses accrued hereunder or under the other Loan Documents, absent manifest error, shall be conclusively presumed to be correct and accurate and constitute an account stated between Borrowers and the Lender Group unless, within 30 days after Agent first makes such a statement available to Borrowers shall deliver to Agent written objection thereto describing the error or errors contained in such statement.

2.10 Fees.

-61-

(a) Agent Fees. Borrowers shall pay to Agent, for the account of Agent, as and when due and payable under the terms of the Fee Letter, the fees set forth in the Fee Letter.

(b) **Unused Line Fee**. Borrowers shall pay to Agent, for the ratable account of the Revolving Lenders, an unused line fee (the "<u>Unused Line Fee</u>") in an amount equal to the Applicable Unused Line Fee Percentage *per annum* times the result of (i) the aggregate amount of the Revolver Commitments, less (ii) the Average Revolver Usage during the immediately preceding month (or portion thereof), which Unused Line Fee shall be due and payable, in arrears, on the first day of each month from and after the Closing Date up to the first day of the month prior to the date on which the Obligations are paid in full and on the date on which the Obligations are paid in full.

(c) **Field Examination and Other Fees.** Borrowers shall pay to Agent, field examination, appraisal, and valuation fees and charges, as and when incurred or chargeable, as follows (i) a fee of \$1,000 per day, per examiner, *plus* out-of-pocket expenses (including travel, meals, and lodging) for each field examination of any Loan Party or its Subsidiaries performed by or on behalf of Agent, (ii) if implemented, a fee of \$1,000 per day, per person plus out-of-pocket expenses for the establishment of electronic collateral reporting, and (iii) the fees, charges or expenses paid or incurred by Agent (but in no event less than a charge of \$1,000 per day, per person) if it elects to employ the services of one or more third Persons to appraise the Collateral, or any portion thereof, or to assess any Loan Party's or its Subsidiaries' business valuation.

2.11 Letters of Credit.

(a) Subject to the terms and conditions of this Agreement, upon the request of Borrowers made in accordance herewith, and prior to the Maturity Date, Issuing Bank agrees to issue a requested standby Letter of Credit or a sight commercial Letter of Credit for the account of Borrowers. By submitting a request to Issuing Bank for the issuance of a Letter of Credit, Borrowers shall be deemed to have requested that Issuing Bank issue the requested Letter of Credit. Each request for the issuance of a Letter of Credit, or the amendment, renewal, or extension of any outstanding Letter of Credit, shall be (i) irrevocable and made in writing by an Authorized Person, (ii) delivered to Agent and Issuing Bank via telefacsimile or other electronic method of transmission reasonably acceptable to Agent and Issuing Bank and reasonably in advance of the requested date of issuance, amendment, renewal, or extension, and (iii) subject to Issuing Bank's authentication procedures with results satisfactory to Issuing Bank. Each such request shall be in form and substance reasonably satisfactory to Agent and Issuing Bank and (i) shall specify (A) the amount of such Letter of Credit, (B) the date of issuance, amendment, renewal, or extension of such Letter of Credit, (C) the proposed expiration date of such Letter of Credit, (D) the name and address of the beneficiary of the Letter of Credit, and (E) such other information (including, the conditions to drawing, and, in the case of an amendment, renewal, or extension, identification of the Letter of Credit to be so amended, renewed, or extended) as shall be necessary to prepare, amend, renew, or extend such Letter of Credit, and (ii) shall be accompanied by such Issuer Documents as Agent or Issuing Bank may request or require, to the extent that such requests or requirements are consistent with the Issuer Documents that Issuing Bank generally requests for Letters of Credit in similar circumstances. Issuing Bank's records of the content of any such request will be conclusive. Anything contained herein to the contrary notwithstanding, Issuing Bank may, but shall not be obligated to, issue a Letter of Credit that supports the obligations of a Loan Party or one of its Subsidiaries in respect of (x) a lease of real property, or (y) an employment contract.

(b) Issuing Bank shall have no obligation to issue a Letter of Credit if any of the following would result after giving effect to the requested issuance:

(i) the Letter of Credit Usage would exceed the Letter of Credit Sublimit, or

-62-

(ii) [reserved],

(iii) the Letter of Credit Usage would exceed the Maximum Revolver Amount <u>less</u> the outstanding amount of Revolving Loans (including Swing Loans), or

(iv) the Letter of Credit Usage would exceed the Borrowing Base at such time <u>less</u> the outstanding principal balance of the Revolving Loans (inclusive of Swing Loans) at such time.

(c) In the event there is a Defaulting Lender as of the date of any request for the issuance of a Letter of Credit, Issuing Bank shall not be required to issue or arrange for such Letter of Credit to the extent (i) the Defaulting Lender's Letter of Credit Exposure with respect to such Letter of Credit may not be reallocated pursuant to Section 2.3(g)(ii), or (ii) Issuing Bank has not otherwise entered into arrangements reasonably satisfactory to it and Borrowers to eliminate Issuing Bank's risk with respect to the participation in such Letter of Credit of the Defaulting Lender, which arrangements may include Borrowers cash collateralizing such Defaulting Lender's Letter of Credit Exposure in accordance with Section 2.3(g)(ii). Additionally, Issuing Bank shall have no obligation to issue or extend a Letter of Credit if (A) any order, judgment, or decree of any Governmental Authority or arbitrator shall, by its terms, purport to enjoin or restrain Issuing Bank from issuing such Letter of Credit, or any law applicable to Issuing Bank or any request that Issuing Bank refrain from the issuance of letters of credit generally or such Letter of Credit in particular, or (B) the issuance of such Letter of Credit would violate one or more policies of Issuing Bank applicable to letters of credit generally.

(d) Any Issuing Bank (other than Wells Fargo or any of its Affiliates) shall notify Agent in writing no later than the Business Day prior to the Business Day on which such Issuing Bank issues any Letter of Credit. In addition, each Issuing Bank (other than Wells Fargo or any of its Affiliates) shall, on the first Business Day of each week, submit to Agent a report detailing the daily undrawn amount of each Letter of Credit issued by such Issuing Bank during the prior calendar week. Each Letter of Credit shall be in form and substance reasonably acceptable to Issuing Bank, including the requirement that the amounts payable thereunder must be payable in Dollars. If Issuing Bank makes a payment under a Letter of Credit, Borrowers shall pay to Agent an amount equal to the applicable Letter of Credit Disbursement on the Business Day such Letter of Credit Disbursement is made and, in the absence of such payment, the amount of the Letter of Credit Disbursement immediately and automatically shall be deemed to be a Revolving Loan hereunder (notwithstanding any failure to satisfy any condition precedent set forth in Section 3) and, initially, shall bear interest at the rate then applicable to Revolving Loans that are Base Rate Loans. If a Letter of Credit Disbursement is deemed to be a Revolving Loan hereunder, Borrowers' obligation to pay the amount of such Letter of Credit Disbursement to Issuing Bank shall be automatically converted into an obligation to pay the resulting Revolving Loan. Promptly following receipt by Agent of any payment from Borrowers pursuant to this paragraph, Agent shall distribute such payment to Issuing Bank or, to the extent that Revolving Lenders have made payments pursuant to <u>Section 2.11(e)</u> to reimburse Issuing Bank, then to such Revolving Lenders and Issuing Bank as their interests may appear.

(e) Promptly following receipt of a notice of a Letter of Credit Disbursement pursuant to <u>Section 2.11(d)</u>, each Revolving Lender agrees to fund its Pro Rata Share of any Revolving Loan deemed made pursuant to <u>Section 2.11(d)</u> on the same terms and conditions as if Borrowers had requested the amount thereof as a Revolving Loan and Agent shall promptly pay to Issuing Bank the amounts so received by it from the Revolving Lenders. By the issuance of a Letter of Credit (or an amendment, renewal, or extension of a Letter of Credit) and without any further action on the part of Issuing Bank or the Revolving Lenders, Issuing Bank shall be deemed to have granted to each Revolving Lender, and each Revolving Lender shall be deemed to have purchased, a participation in each Letter of Credit issued by Issuing Bank, in an amount

-63-

equal to its Pro Rata Share of such Letter of Credit, and each such Revolving Lender agrees to pay to Agent, for the account of Issuing Bank, such Revolving Lender's Pro Rata Share of any Letter of Credit Disbursement made by Issuing Bank under the applicable Letter of Credit. In consideration and in furtherance of the foregoing, each Revolving Lender hereby absolutely and unconditionally agrees to pay to Agent, for the account of Issuing Bank, such Revolving Lender's Pro Rata Share of each Letter of Credit Disbursement made by Issuing Bank and not reimbursed by Borrowers on the date due as provided in <u>Section 2.11(d)</u>, or of any reimbursement payment that is required to be refunded (or that Agent or Issuing Bank elects, based upon the advice of counsel, to refund) to Borrowers for any reason. Each Revolving Lender acknowledges and agrees that its obligation to deliver to Agent, for the account of Issuing Bank, an amount equal to its respective Pro Rata Share of each Letter of Credit Disbursement pursuant to this <u>Section 2.11(e)</u> shall be absolute and unconditional and such remittance shall be made notwithstanding the occurrence or continuation of an Event of Default or Default or the failure to satisfy any condition set forth in <u>Section 3</u>. If any such Revolving Lender fails to make available to Agent the amount of such Revolving Lender and Agent (for the account of Issuing Bank) shall be entitled to recover such amount on demand from such Revolving Lender together with interest thereon at the Defaulting Lender Rate until paid in full.

(f) Each Borrower agrees to indemnify, defend and hold harmless each member of the Lender Group (including Issuing Bank and its branches, Affiliates, and correspondents) and each such Person's respective directors, officers, employees, attorneys and agents (each, including Issuing Bank, a "Letter of Credit Related Person") (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all reasonable fees and disbursements of attorneys, experts, or consultants and all other costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), which may be incurred by or awarded against any such Letter of Credit Related Person (other than Taxes, which shall be governed by Section 16) (the "Letter of Credit Indemnified Costs"), and which arise out of or in connection with, or as a result of:

(i) any Letter of Credit or any pre-advice of its issuance;

(ii) any transfer, sale, delivery, surrender or endorsement (or lack thereof) of any Drawing Document at any time(s) held by any such Letter of Credit Related Person in connection with any Letter of Credit;

(iii) any action or proceeding arising out of, or in connection with, any Letter of Credit (whether administrative, judicial or in connection with arbitration), including any action or proceeding to compel or restrain any presentation or payment under any Letter of Credit, or for the wrongful dishonor of, or honoring a presentation under, any Letter of Credit;

(iv) any independent undertakings issued by the beneficiary of any Letter of Credit;

(v) any unauthorized instruction or request made to Issuing Bank in connection with any Letter of Credit or requested Letter of Credit, or any error, omission, interruption or delay in such instruction or request, whether transmitted by mail, courier, electronic transmission, SWIFT, or any other telecommunication including communications through a correspondent;

(vi) an adviser, confirmer or other nominated person seeking to be reimbursed, indemnified or compensated;

-64-

(vii) any third party seeking to enforce the rights of an applicant, beneficiary, nominated person, transferee, assignee of Letter of Credit proceeds or holder of an instrument or document;

(viii) the fraud, forgery or illegal action of parties other than the Letter of Credit Related Person;

(ix) any prohibition on payment or delay in payment of any amount payable by Issuing Bank to a beneficiary or transferee beneficiary of a Letter of Credit arising out of Anti-Corruption Laws, Anti-Money Laundering Laws, or Sanctions;

confirmation;

(x) Issuing Bank's performance of the obligations of a confirming institution or entity that wrongfully dishonors a

(xi) any foreign language translation provided to Issuing Bank in connection with any Letter of Credit;

(xii) any foreign law or usage as it relates to Issuing Bank's issuance of a Letter of Credit in support of a foreign guaranty including the expiration of such guaranty after the related Letter of Credit expiration date and any resulting drawing paid by Issuing Bank in connection therewith; or

(xiii) the acts or omissions, whether rightful or wrongful, of any present or future de jure or de facto governmental or regulatory authority or cause or event beyond the control of the Letter of Credit Related Person;

provided, that such indemnity shall not be available to any Letter of Credit Related Person claiming indemnification under clauses (i) through (x) above to the extent that such Letter of Credit Indemnified Costs may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction to have resulted directly from the gross negligence or willful misconduct of the Letter of Credit Related Person claiming indemnity. Borrowers hereby agree to pay the Letter of Credit Related Person claiming indemnity on demand from time to time all amounts owing under this Section 2.11(f). If and to the extent that the obligations of Borrowers under this Section 2.11(f) are unenforceable for any reason, Borrowers agree to make the maximum contribution to the Letter of Credit Indemnified Costs permissible under applicable law. This indemnification provision shall survive termination of this Agreement and all Letters of Credit.

(g) The liability of Issuing Bank (or any other Letter of Credit Related Person) under, in connection with or arising out of any Letter of Credit (or pre-advice), regardless of the form or legal grounds of the action or proceeding, shall be limited to direct damages suffered by Borrowers that are caused directly by Issuing Bank's gross negligence or willful misconduct in (i) honoring a presentation under a Letter of Credit that on its face does not at least substantially comply with the terms and conditions of such Letter of Credit, (ii) failing to honor a presentation under a Letter of Credit that strictly complies with the terms and conditions of such Letter of Credit, or (iii) retaining Drawing Documents presented under a Letter of Credit. Borrowers' aggregate remedies against Issuing Bank and any Letter of Credit Related Person for wrongfully honoring a presentation under any Letter of Credit or wrongfully retaining honored Drawing Documents shall in no event exceed the aggregate amount paid by Borrowers to Issuing Bank in respect of the honored presentation in connection with such Letter of Credit under <u>Section 2.11(d)</u>, *plus* interest at the rate then applicable to Base Rate Loans hereunder. Borrowers shall take action to avoid and mitigate the amount of any damages claimed against Issuing Bank or any other Letter of Credit Related Person, including by enforcing its rights against the beneficiaries of the Letters of Credit. Any claim by Borrowers under or in connection with any Letter of Credit shall be reduced by an amount equal to the sum of (x) the amount (if any) saved by Borrowers as a result of the breach or alleged wrongful conduct complained of, and (y)

-65-

the amount (if any) of the loss that would have been avoided had Borrowers taken all reasonable steps to mitigate any loss, and in case of a claim of wrongful dishonor, by specifically and timely authorizing Issuing Bank to effect a cure.

(h) Borrowers are responsible for the final text of the Letter of Credit as issued by Issuing Bank, irrespective of any assistance Issuing Bank may provide such as drafting or recommending text or by Issuing Bank's use or refusal to use text submitted by Borrowers. Borrowers understand that the final form of any Letter of Credit may be subject to such revisions and changes as are deemed necessary or appropriate by Issuing Bank, and Borrowers hereby consent to such revisions and changes not materially different from the application executed in connection therewith. Borrowers are solely responsible for the suitability of the Letter of Credit for Borrowers' purposes. If Borrowers request Issuing Bank to issue a Letter of Credit for an affiliated or unaffiliated third party (an "Account Party"), (i) such Account Party shall have no rights against Issuing Bank; (ii) Borrowers shall be responsible for the application and obligations under this Agreement; and (iii) communications (including notices) related to the respective Letter of Credit shall be among Issuing Bank and Borrowers. Borrowers will examine the copy of the Letter of Credit and any other documents sent by Issuing Bank in connection therewith and shall promptly notify Issuing Bank (not later than three (3) Business Days following Borrowers' receipt of documents from Issuing Bank) of any non-compliance with Borrowers' instructions and of any discrepancy in any document under any presentment or other irregularity. Borrowers understand and agree that Issuing Bank is not required to extend the expiration date of any Letter of Credit for any reason. With respect to any Letter of Credit containing an "automatic amendment" to extend the expiration date of such Letter of Credit shall be of nonrenewal of such Letter of Credit and, if Borrowers do not at any time want the then current expiration date of such Letter of Credit to notify Agent and Issuing Bank at least 30 calendar days before Issuing Bank is required to notify the beneficiary of such Letter of Credit

(i) Borrowers' reimbursement and payment obligations under this <u>Section 2.11</u> are absolute, unconditional and irrevocable and shall be performed strictly in accordance with the terms of this Agreement under any and all circumstances whatsoever, including:

(i) any lack of validity, enforceability or legal effect of any Letter of Credit, any Issuer Document, this Agreement, or any Loan Document, or any term or provision therein;

(ii) payment against presentation of any draft, demand or claim for payment under any Drawing Document that does not comply in whole or in part with the terms of the applicable Letter of Credit or which proves to be fraudulent, forged or invalid in any respect or any statement therein being untrue or inaccurate in any respect, or which is signed, issued or presented by a Person or a transferee of such Person purporting to be a successor or transferee of the beneficiary of such Letter of Credit;

(iii) Issuing Bank or any of its branches or Affiliates being the beneficiary of any Letter of Credit;

(iv) Issuing Bank or any correspondent honoring a drawing against a Drawing Document up to the amount available under any Letter of Credit even if such Drawing Document claims an amount in excess of the amount available under the Letter of Credit;

(v) the existence of any claim, set-off, defense or other right that any Loan Party or any of its Subsidiaries may have at any time against any beneficiary or transferee beneficiary, any assignee of proceeds, Issuing Bank or any other Person;

(vi) Issuing Bank or any correspondent honoring a drawing upon receipt of an electronic presentation under a Letter of Credit requiring the same, regardless of whether the original Drawing Documents arrive at Issuing Bank's counters or are different from the electronic presentation;

(vii) any other event, circumstance or conduct whatsoever, whether or not similar to any of the foregoing that might, but for this <u>Section 2.11(i)</u>, constitute a legal or equitable defense to or discharge of, or provide a right of set-off against, any Borrower's or any of its Subsidiaries' reimbursement and other payment obligations and liabilities, arising under, or in connection with, any Letter of Credit, whether against Issuing Bank, the beneficiary or any other Person; or

(viii) the fact that any Default or Event of Default shall have occurred and be continuing;

<u>provided</u>, that subject to <u>Section 2.11(g)</u> above, the foregoing shall not release Issuing Bank from such liability to Borrowers as may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction against Issuing Bank following reimbursement or payment of the obligations and liabilities, including reimbursement and other payment obligations, of Borrowers to Issuing Bank arising under, or in connection with, this <u>Section 2.11</u> or any Letter of Credit.

(j) Without limiting any other provision of this Agreement, Issuing Bank and each other Letter of Credit Related Person (if applicable) shall not be responsible to Borrowers for, and Issuing Bank's rights and remedies against Borrowers and the obligation of Borrowers to reimburse Issuing Bank for each drawing under each Letter of Credit shall not be impaired by:

(i) honor of a presentation under any Letter of Credit that on its face substantially complies with the terms and conditions of such Letter of Credit, even if the Letter of Credit requires strict compliance by the beneficiary;

(ii) honor of a presentation of any Drawing Document that appears on its face to have been signed, presented or issued (A) by any purported successor or transferee of any beneficiary or other Person required to sign, present or issue such Drawing Document or (B) under a new name of the beneficiary;

(iii) acceptance as a draft of any written or electronic demand or request for payment under a Letter of Credit, even if nonnegotiable or not in the form of a draft or notwithstanding any requirement that such draft, demand or request bear any or adequate reference to the Letter of Credit;

(iv) the identity or authority of any presenter or signer of any Drawing Document or the form, accuracy, genuineness or legal effect of any Drawing Document (other than Issuing Bank's determination that such Drawing Document appears on its face substantially to comply with the terms and conditions of the Letter of Credit);

(v) acting upon any instruction or request relative to a Letter of Credit or requested Letter of Credit that Issuing Bank in good faith believes to have been given by a Person authorized to give such instruction or request;

(vi) any errors, omissions, interruptions or delays in transmission or delivery of any message, advice or document (regardless of how sent or transmitted) or for errors in interpretation of technical terms or in translation or any delay in giving or failing to give notice to any Borrower;

-67-

(vii) any acts, omissions or fraud by, or the insolvency of, any beneficiary, any nominated person or entity or any other Person or any breach of contract between any beneficiary and any Borrower or any of the parties to the underlying transaction to which the Letter of Credit relates;

(viii) assertion or waiver of any provision of the ISP or UCP that primarily benefits an issuer of a letter of credit, including any requirement that any Drawing Document be presented to it at a particular hour or place;

(ix) payment to any presenting bank (designated or permitted by the terms of the applicable Letter of Credit) claiming that it rightfully honored or is entitled to reimbursement or indemnity under Standard Letter of Credit Practice applicable to it;

(x) acting or failing to act as required or permitted under Standard Letter of Credit Practice applicable to where Issuing Bank has issued, confirmed, advised or negotiated such Letter of Credit, as the case may be;

(xi) honor of a presentation after the expiration date of any Letter of Credit notwithstanding that a presentation was made prior to such expiration date and dishonored by Issuing Bank if subsequently Issuing Bank or any court or other finder of fact determines such presentation should have been honored;

(xii) dishonor of any presentation that does not strictly comply or that is fraudulent, forged or otherwise not entitled to honor;

or

(xiii) honor of a presentation that is subsequently determined by Issuing Bank to have been made in violation of international, federal, state or local restrictions on the transaction of business with certain prohibited Persons.

(k) Borrowers shall pay immediately upon demand to Agent for the account of Issuing Bank as non-refundable fees, commissions, and charges (it being acknowledged and agreed that any charging of such fees, commissions, and charges to the Loan Account pursuant to the provisions of Section 2.6(d) shall be deemed to constitute a demand for payment thereof for the purposes of this Section 2.11(k)): (i) a fronting fee which shall be imposed by Issuing Bank equal to 0.125% per annum times the average amount of the Letter of Credit Usage during the immediately preceding month, *plus* (ii) any and all other customary commissions, fees and charges then in effect imposed by, and any and all expenses incurred by, Issuing Bank, or by any adviser, confirming institution or entity or other nominated person, relating to Letters of Credit, at the time of issuance of any Letter of Credit and upon the occurrence of any other activity with respect to any Letter of Credit (including transfers, assignments of proceeds, amendments, drawings, renewals or cancellations).

(1) If by reason of (x) any Change in Law, or (y) compliance by Issuing Bank or any other member of the Lender Group with any direction, request, or requirement (irrespective of whether having the force of law) of any Governmental Authority or monetary authority including, Regulation D of the Board of Governors as from time to time in effect (and any successor thereto):

(i) any reserve, deposit, or similar requirement is or shall be imposed or modified in respect of any Letter of Credit issued or caused to be issued hereunder or hereby, or any Loans or obligations to make Loans hereunder or hereby, or

(ii) there shall be imposed on Issuing Bank or any other member of the Lender Group any other condition regarding any Letter of Credit, Loans, or obligations to make Loans hereunder,

-68-

and the result of the foregoing is to increase, directly or indirectly, the cost to Issuing Bank or any other member of the Lender Group of issuing, making, participating in, or maintaining any Letter of Credit or to reduce the amount receivable in respect thereof, then, and in any such case, Agent may, at any time within a reasonable period after the additional cost is incurred or the amount receivable is reduced, notify Borrowers, and Borrowers shall pay within 30 days after demand therefor, such amounts as Agent may specify to be necessary to compensate Issuing Bank or any other member of the Lender Group for such additional cost or reduced receipt, together with interest on such amount from the date of such demand until payment in full thereof at the rate then applicable to Base Rate Loans hereunder; provided, that (A) Borrowers shall not be required to provide any compensation pursuant to this Section 2.11(1) for any such amounts incurred more than 180 days prior to the date on which the demand for payment of such amounts is first made to Borrowers, and (B) if an event or circumstance giving rise to such amounts is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof. The determination by Agent of any amount due pursuant to this Section 2.11(1), as set forth in a certificate setting forth the calculation thereof in reasonable detail, shall, in the absence of manifest or demonstrable error, be final and conclusive and binding on all of the parties hereto.

(m) Each standby Letter of Credit shall expire not later than the date that is 12 months after the date of the issuance of such Letter of Credit; <u>provided</u>, that any standby Letter of Credit may provide for the automatic extension thereof for any number of additional periods each of up to one year in duration; <u>provided further</u>, that with respect to any Letter of Credit which extends beyond the Maturity Date, Letter of Credit Collateralization shall be provided therefor on or before the date that is five Business Days prior to the Maturity Date. Each commercial Letter of Credit shall expire on the earlier of (i) 120 days after the date of the issuance of such commercial Letter of Credit and (ii) five Business Days prior to the Maturity Date.

(n) If (i) any Event of Default shall occur and be continuing, or (ii) Availability shall at any time be less than zero, then on the Business Day following the date when the Administrative Borrower receives notice from Agent or the Required Lenders (or, if the maturity of the Obligations has been accelerated, Revolving Lenders with Letter of Credit Exposure representing greater than 50% of the total Letter Credit Exposure) demanding Letter of Credit Collateralization pursuant to this <u>Section 2.11(n)</u> upon such demand, Borrowers shall provide Letter of Credit Collateralization with respect to the then existing Letter of Credit Usage. If Borrowers fail to provide Letter of Credit Collateralization as required by this <u>Section 2.11(n)</u>, the Revolving Lenders may (and, upon direction of Agent, shall) advance, as Revolving Loans the amount of the cash collateral required pursuant to the Letter of Credit Collateralization provision so that the then existing Letter of Credit Usage is cash collateralized in accordance with the Letter of Credit Collateralization provision (whether or not the Revolver Commitments have terminated, an Overadvance exists or the conditions in <u>Section 3</u> are satisfied).

(o) Unless otherwise expressly agreed by Issuing Bank and Borrowers when a Letter of Credit is issued, (i) the rules of the ISP shall apply to each standby Letter of Credit, and (ii) the rules of the UCP shall apply to each commercial Letter of Credit.

(p) Issuing Bank shall be deemed to have acted with due diligence and reasonable care if Issuing Bank's conduct is in accordance with Standard Letter of Credit Practice or in accordance with this Agreement.

(q) In the event of a direct conflict between the provisions of this <u>Section 2.11</u> and any provision contained in any Issuer Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of

-69-

any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this Section 2.11 shall control and govern.

(r) The provisions of this <u>Section 2.11</u> shall survive the termination of this Agreement and the repayment in full of the Obligations with respect to any Letters of Credit that remain outstanding.

(s) At Borrowers' costs and expense, Borrowers shall execute and deliver to Issuing Bank such additional certificates, instruments and/or documents and take such additional action as may be reasonably requested by Issuing Bank to enable Issuing Bank to issue any Letter of Credit pursuant to this Agreement and related Issuer Document, to protect, exercise and/or enforce Issuing Banks' rights and interests under this Agreement or to give effect to the terms and provisions of this Agreement or any Issuer Document. Each Borrower irrevocably appoints Issuing Bank as its attorney-in-fact and authorizes Issuing Bank, without notice to Borrowers, to execute and deliver ancillary documents and letters customary in the letter of credit business that may include but are not limited to advisements, indemnities, checks, bills of exchange and issuance documents. The power of attorney granted by the Borrowers is limited solely to such actions related to the issuance, confirmation or amendment of any Letter of Credit and to ancillary documents or letters customary in the letter of credit business. This appointment is coupled with an interest.

2.12 LIBOR Option.

(a) Interest and Interest Payment Dates. In lieu of having interest charged at the rate based upon the Base Rate, Borrowers shall have the option, subject to <u>Section 2.12(b)</u> below (the "<u>LIBOR Option</u>") to have interest on all or a portion of the Revolving Loans be charged (whether at the time when made (unless otherwise provided herein), upon conversion from a Base Rate Loan to a LIBOR Rate Loan, or upon continuation of a LIBOR Rate Loan as a LIBOR Rate Loan) at a rate of interest based upon the LIBOR Rate. Interest on LIBOR Rate Loans shall be payable on the earliest of (i) the last day of the Interest Period applicable thereto; provided, that subject to the following clauses (ii) and (iii), in the case of any Interest Period and on the last day of such Interest Period), (ii) the date on which all or any portion of the Obligations are accelerated pursuant to the terms hereof, or (iii) the date on which this Agreement is terminated pursuant to the terms hereof. On the last day of each applicable Interest Period, unless Borrowers have properly exercised the LIBOR Option with respect thereto, the interest rate applicable to such LIBOR Rate Loan automatically shall convert to the rate of interest then applicable to Base Rate Loans of the same type hereunder. At any time that an Event of Default has occurred and is continuing, Borrowers no longer shall have the option to request that Revolving Loans bear interest at a rate based upon the LIBOR Rate.

(b) LIBOR Election.

(i) Borrowers may, at any time and from time to time, so long as Borrowers have not received a notice from Agent (which notice Agent may elect to give or not give in its discretion unless Agent is directed to give such notice by the Required Lenders, in which case, it shall give the notice to Borrowers), after the occurrence and during the continuance of an Event of Default, to terminate the right of Borrowers to exercise the LIBOR Option during the continuance of such Event of Default, elect to exercise the LIBOR Option by notifying Agent prior to 11:00 a.m. at least three Business Days prior to the commencement of the proposed Interest Period (the "<u>LIBOR Deadline</u>"). Notice of Borrowers' election of the LIBOR Option for a permitted portion of the Revolving Loans and an Interest Period pursuant to this Section shall be made by delivery to Agent of a LIBOR Notice received by Agent before the LIBOR Deadline. Promptly upon its receipt of each such LIBOR Notice, Agent shall provide a copy thereof to each of the affected Lenders.

(ii) Each LIBOR Notice shall be irrevocable and binding on Borrowers. In connection with each LIBOR Rate Loan, each Borrower shall indemnify, defend, and hold Agent and the Lenders harmless against any loss, cost, or expense actually incurred by Agent or any Lender as a result of (A) the payment or required assignment of any principal of any LIBOR Rate Loan other than on the last day of an Interest Period applicable thereto (including as a result of an Event of Default), (B) the conversion of any LIBOR Rate Loan other than on the last day of the Interest Period applicable thereto, or (C) the failure to borrow, convert, continue or prepay any LIBOR Rate Loan on the date specified in any LIBOR Notice delivered pursuant hereto (such losses, costs, or expenses, "Funding Losses"). A certificate of Agent or a Lender delivered to Borrowers setting forth in reasonable detail any amount or amounts that Agent or such Lender is entitled to receive pursuant to this Section 2.12 shall be conclusive absent manifest error. Borrowers shall pay such amount to Agent or the Lender, as applicable Interest Period would result in a Funding Loss, Agent may, in its sole discretion at the request of Borrowers, hold the amount of such payment as cash collateral in support of the Obligations until the last day of such Interest Period and apply such amounts to the payment of the applicable LIBOR Rate Loan on such last day, it being agreed that Agent has no obligation to so defer the application of payments to any LIBOR Rate Loan and that, in the event that Agent does not defer such application, Borrowers shall be obligated to pay any resulting Funding Losses.

(iii) Unless Agent, in its sole discretion, agrees otherwise, Borrowers shall have not more than five LIBOR Rate Loans in effect at any given time. Borrowers may only exercise the LIBOR Option for proposed LIBOR Rate Loans of at <u>least</u> \$1,000,000.

(c) **Conversion; Prepayment**. Borrowers may convert LIBOR Rate Loans to Base Rate Loans or prepay LIBOR Rate Loans at any time; <u>provided</u>, that in the event that LIBOR Rate Loans are converted or prepaid on any date that is not the last day of the Interest Period applicable thereto, including as a result of any prepayment through the required application by Agent of any payments or proceeds of Collateral in accordance with <u>Section 2.4(b)</u> or for any other reason, including early termination of the term of this Agreement or acceleration of all or any portion of the Obligations pursuant to the terms hereof, each Borrower shall indemnify, defend, and hold Agent and the Lenders and their Participants harmless against any and all Funding Losses in accordance with <u>Section 2.12 (b)(ii)</u>.

(d) Special Provisions Applicable to LIBOR Rate.

(i) The LIBOR Rate may be adjusted by Agent with respect to any Lender on a prospective basis to take into account any additional or increased costs to such Lender of maintaining or obtaining any eurodollar deposits or increased costs (other than Taxes which shall be governed by <u>Section 16</u>), in each case, due to changes in applicable law occurring subsequent to the commencement of the then applicable Interest Period, including any Changes in Law and changes in the reserve requirements imposed by the Board of Governors, which additional or increased costs would increase the cost of funding or maintaining loans bearing interest at the LIBOR Rate. In any such event, the affected Lender shall give Borrowers and Agent notice of such a determination and adjustment and Agent promptly shall transmit the notice to each other Lender and, upon its receipt of the notice from the affected Lender, Borrowers may, by notice to such affected Lender (A) require such Lender to furnish to Borrowers a statement setting forth in reasonable detail the basis for adjusting such LIBOR Rate and the method for determining the amount of such adjustment, or (B) repay the LIBOR Rate Loans of such Lender with respect to which such adjustment is made (together with any amounts due under <u>Section 2.12(b)(ii)</u>).

(ii) In the event that any change in market conditions or any Change in Law shall at any time after the date hereof, in the reasonable opinion of any Lender, make it unlawful or impractical

-71-

for such Lender to fund or maintain LIBOR Rate Loans or to continue such funding or maintaining, or to determine or charge interest rates at the LIBOR Rate, such Lender shall give notice of such changed circumstances to Agent and Borrowers and Agent promptly shall transmit the notice to each other Lender and (y) in the case of any LIBOR Rate Loans of such Lender that are outstanding, the date specified in such Lender's notice shall be deemed to be the last day of the Interest Period of such LIBOR Rate Loans, and interest upon the LIBOR Rate Loans of such Lender thereafter shall accrue interest at the rate then applicable to Base Rate Loans, and (z) Borrowers shall not be entitled to elect the LIBOR Option until such Lender determines that it would no longer be unlawful or impractical to do so.

(e) No Requirement of Matched Funding. Anything to the contrary contained herein notwithstanding, neither Agent, nor any Lender, nor any of their Participants, is required actually to acquire eurodollar deposits to fund or otherwise match fund any Obligation as to which interest accrues at the LIBOR Rate.

2.13 Capital Requirements.

(a) If, after the date hereof, Issuing Bank or any Lender determines that (i) any Change in Law regarding capital, liquidity or reserve requirements for banks or bank holding companies, or (ii) compliance by Issuing Bank or such Lender, or their respective parent bank holding companies, with any guideline, request or directive of any Governmental Authority regarding capital adequacy or liquidity requirements (whether or not having the force of law), has the effect of reducing the return on Issuing Bank's, such Lender's, or such holding companies' capital or liquidity as a consequence of Issuing Bank's or such Lender's commitments, Loans, participations or other obligations hereunder to a level below that which Issuing Bank, such Lender, or such holding companies could have achieved but for such Change in Law or compliance (taking into consideration Issuing Bank's, such Lender's, or such holding companies' then existing policies with respect to capital adequacy or liquidity requirements and assuming the full utilization of such entity's capital) by any amount deemed by Issuing Bank or such Lender to be material, then Issuing Bank or such Lender may notify Borrowers and Agent thereof. Following receipt of such notice, Borrowers agree to pay Issuing Bank or such Lender on demand the amount of such reduction of return of capital as and when such reduction is determined, payable within 30 days after presentation by Issuing Bank or such Lender of a statement in the amount and setting forth in reasonable detail Issuing Bank's or such Lender's calculation thereof and the assumptions upon which such calculation was based (which statement shall be deemed true and correct absent manifest error). In determining such amount, Issuing Bank or such Lender may use any reasonable averaging and attribution methods. Failure or delay on the part of Issuing Bank or any Lender to demand compensation pursuant to this Section shall not constitute a waiver of Issuing Bank's or such Lender's right to demand such compensation; provided, that Borrowers shall not be required to compensate Issuing Bank or a Lender pursuant to this Section for any reductions in return incurred more than 180 days prior to the date that Issuing Bank or such Lender notifies Borrowers of such Change in Law giving rise to such reductions and of such Lender's intention to claim compensation therefor; provided further, that if such claim arises by reason of the Change in Law that is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

(b) If Issuing Bank or any Lender requests additional or increased costs referred to in <u>Section 2.11(l)</u> or <u>Section 2.12(d)(i)</u> or amounts under <u>Section 2.13(a)</u> or sends a notice under <u>Section 2.12(d)(ii)</u> relative to changed circumstances (such Issuing Bank or Lender, an "<u>Affected Lender</u>"), then, at the request of Administrative Borrower, such Affected Lender shall use reasonable efforts to promptly designate a different one of its lending offices or to assign its rights and obligations hereunder to another of its offices or branches, if (i) in the reasonable judgment of such Affected Lender, such designation or assignment would eliminate or reduce amounts payable pursuant to <u>Section 2.11(l)</u>, <u>Section 2.12(d)(i)</u> or <u>Section 2.13(a)</u>, as

-72-

applicable, or would eliminate the illegality or impracticality of funding or maintaining LIBOR Rate Loans, and (ii) in the reasonable judgment of such Affected Lender, such designation or assignment would not subject it to any material unreimbursed cost or expense and would not otherwise be materially disadvantageous to it. Borrowers agree to pay all reasonable out-of-pocket costs and expenses incurred by such Affected Lender in connection with any such designation or assignment. If, after such reasonable efforts, such Affected Lender does not so designate a different one of its lending offices or assign its rights to another of its offices or branches so as to eliminate Borrowers' obligation to pay any future amounts to such Affected Lender pursuant to <u>Section 2.11(1)</u>, <u>Section 2.12(d)(i)</u> or <u>Section 2.13(a)</u>, as applicable, or to enable Borrowers to obtain LIBOR Rate Loans, then Borrowers (without prejudice to any amounts then due to such Affected Lender under <u>Section 2.11(1)</u>, <u>Section 2.12(d)(i)</u> or <u>Section 2.13(a)</u>, as applicable) may, unless prior to the effective date of any such assignment the Affected Lender withdraws its request for such additional amounts under <u>Section 2.12(d)(i)</u> or <u>Section 2.13(a)</u>, as applicable, or indicates that it is no longer unlawful or impractical to fund or maintain LIBOR Rate Loans, may designate a different Issuing Bank or substitute a Lender or prospective Lender, in each case, reasonably acceptable to Agent to purchase the Obligations owed to such Affected Lender shall assign to the Replacement Lender"), and if such Replacement Lender, which such Replacement Lender shall be deemed to be "Issuing Bank" or a "Lender" (as the case may be) for purposes of this Agreement.

(c) Notwithstanding anything herein to the contrary, the protection of <u>Sections 2.11(l)</u>, <u>2.12(d)</u>, and <u>2.13</u> shall be available to Issuing Bank and each Lender (as applicable) regardless of any possible contention of the invalidity or inapplicability of the law, rule, regulation, judicial ruling, judgment, guideline, treaty or other change or condition which shall have occurred or been imposed, so long as it shall be customary for issuing banks or lenders affected thereby to comply therewith. Notwithstanding any other provision herein, neither Issuing Bank nor any Lender shall demand compensation pursuant to this <u>Section 2.13</u> if it shall not at the time be the general policy or practice of Issuing Bank or such Lender (as the case may be) to demand such compensation in similar circumstances under comparable provisions of other credit agreements, if any.

2.14 Incremental Facilities.

Amount:

(a) At the option of Borrowers (but subject to the conditions set forth in clause (b) below), the Revolver Commitments and the Maximum Revolver Amount may be increased by an amount in the aggregate for all such increases of the Revolver Commitments and the Maximum Revolver Amount not to exceed the Available Increase Amount (each such increase, an "<u>Increase</u>"). Agent shall invite each Lender to increase its Revolver Commitments (it being understood that no Lender shall be obligated to increase its Revolver Commitments) in connection with a proposed Increase at the interest margin proposed by Borrowers, and if sufficient Lenders do not agree to increase their Revolver Commitments in connection with such proposed Increase, then Agent or Borrowers may invite any prospective lender who is reasonably satisfactory to Agent and Borrowers to become a Lender in connection with a proposed Increase. Any Increase shall be in an amount of at least \$5,000,000 and integral multiples of \$5,000,000 in excess thereof. Additionally, for the avoidance of doubt, it is understood and agreed that in no event shall the aggregate amount of the Increases to the Revolver Commitments exceed \$50,000,000.

(b) Each of the following shall be conditions precedent to any Increase of the Revolver Commitments and the Maximum Revolver

-73-

(i) Agent or Borrowers have obtained the commitment of one or more Lenders (or other prospective lenders) reasonably satisfactory to Agent and Borrowers to provide the applicable Increase and any such Lenders (or prospective lenders), Borrowers, and Agent have signed a joinder agreement to this Agreement (an "Increase Joinder"), in form and substance reasonably satisfactory to Agent, to which such Lenders (or prospective lenders), Borrowers, and Agent are party,

(ii) each of the conditions precedent set forth in Section 3.2 are satisfied,

(iii) in connection with any Increase, if any Loan Party or any of its Subsidiaries owns or will acquire any Margin Stock, Borrowers shall deliver to Agent an updated Form U-1 (with sufficient additional originals thereof for each Lender), duly executed and delivered by the Borrowers, together with such other documentation as Agent shall reasonably request, in order to enable Agent and the Lenders to comply with any of the requirements under Regulations T, U or X of the Federal Reserve Board,

(iv) Borrowers have delivered to Agent updated pro forma Projections (after giving effect to the applicable Increase) for the Loan Parties and their Subsidiaries evidencing compliance on a pro forma basis with <u>Section 7</u> for the twelve months (on a month-by-month basis) immediately following the proposed date of the applicable Increase (calculated as if a Covenant Testing Period was in effect during the entire twelve month period), and

(v) The interest rate margins with respect to the Revolving Loans to be made pursuant to the increased Revolver Commitments shall be the same as the interest rate margin applicable to Revolving Loans hereunder immediately prior to the applicable Increase Date (as defined below). The date of the effectiveness of the increased Revolver Commitments shall be referred to herein as the "<u>Increase Date</u>". Any Increase Joinder may, with the consent of Agent, Borrowers and the Lenders or prospective lenders agreeing to the proposed Increase, effect such amendments to this Agreement and the other Loan Documents as may be necessary to effectuate the provisions of this <u>Section 2.14</u>.

(c) Unless otherwise specifically provided herein, all references in this Agreement and any other Loan Document to Revolving Loans shall be deemed, unless the context otherwise requires, to include Revolving Loans made pursuant to the increased Revolver Commitments and Maximum Revolver Amount pursuant to this <u>Section 2.14</u>.

(d) Each of the Lenders having a Revolver Commitment prior to the Increase Date (the "<u>Pre-Increase Revolver Lenders</u>") shall assign to any Lender which is acquiring a new or additional Revolver Commitment on the Increase Date (the "<u>Post-Increase Revolver Lenders</u>"), and such Post-Increase Revolver Lenders shall purchase from each Pre-Increase Revolver Lender, at the principal amount thereof, such interests in the Revolving Loans and participation interests in Letters of Credit on such Increase Date as shall be necessary in order that, after giving effect to all such assignments and purchases, such Revolving Loans and participation interests in Letters of Credit will be held by Pre-Increase Revolver Lenders and Post-Increase Revolver Lenders and Post-Increase Revolver Lenders ratably in accordance with their Pro Rata Share after giving effect to such increased Revolver Commitments.

(e) The Revolving Loans, Revolver Commitments, and Maximum Revolver Amount established pursuant to this <u>Section 2.14</u> shall constitute Revolving Loans, Revolver Commitments, and Maximum Revolver Amount under, and shall be entitled to all the benefits afforded by, this Agreement and the other Loan Documents, and shall, without limiting the foregoing, benefit equally and ratably from any guarantees and the security interests created by the Loan Documents. Borrowers shall take any actions reasonably required by Agent to ensure and demonstrate that the Liens and security interests granted by the

-74-

Loan Documents continue to be perfected under the Code or otherwise after giving effect to the establishment of any such new Revolver Commitments and Maximum Revolver Amount.

2.15 Joint and Several Liability of Borrowers.

(a) Each Borrower is accepting joint and several liability hereunder and under the other Loan Documents in consideration of the financial accommodations to be provided by the Lender Group under this Agreement, for the mutual benefit, directly and indirectly, of each Borrower and in consideration of the undertakings of the other Borrowers to accept joint and several liability for the Obligations.

(b) Each Borrower, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other Borrowers, with respect to the payment and performance of all of the Obligations (including any Obligations arising under this <u>Section 2.15</u>), it being the intention of the parties hereto that all the Obligations shall be the joint and several obligations of each Borrower without preferences or distinction among them. Accordingly, each Borrower hereby waives any and all suretyship defenses that would otherwise be available to such Borrower under applicable law.

(c) If and to the extent that any Borrower shall fail to make any payment with respect to any of the Obligations as and when due, whether upon maturity, acceleration, or otherwise, or to perform any of the Obligations in accordance with the terms thereof, then in each such event the other Borrowers will make such payment with respect to, or perform, such Obligations until such time as all of the Obligations are paid in full, and without the need for demand, protest, or any other notice or formality.

(d) The Obligations of each Borrower under the provisions of this <u>Section 2.15</u> constitute the absolute and unconditional, full recourse Obligations of each Borrower enforceable against each Borrower to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of the provisions of this Agreement (other than this <u>Section 2.15(d)</u>) or any other circumstances whatsoever.

(e) Without limiting the generality of the foregoing and except as otherwise expressly provided in this Agreement, each Borrower hereby waives presentments, demands for performance, protests and notices, including notices of acceptance of its joint and several liability, notice of any Revolving Loans or any Letters of Credit issued under or pursuant to this Agreement, notice of the occurrence of any Default, Event of Default, notices of nonperformance, notices of protest, notices of dishonor, notices of acceptance of this Agreement, notices of the existence, creation, or incurring of new or additional Obligations or other financial accommodations or of any demand for any payment under this Agreement, notice of any other Person, to proceed against or exhaust any security held from any other Borrower or any other Person, to protect, secure, perfect, or insure any security interest or Lien on any property subject thereto or exhaust any right to take any action against any other Person, or many other remedy in any member of the Lender Group's or any Bank Product Provider's power whatsoever, any requirement of diligence or to mitigate damages and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Agreement (except as otherwise provided in this Agreement), any right to assert against any member of the Lender Group or any Bank Product Provider, any defense (legal or equitable), set-off, counterclaim, or claim which each Borrower may now or at any time hereafter have against any other Borrower or any other party liable to any member of the Lender Group or any Bank Product Provider, any defense, set-off, counterclaim, or claim, of any kind or nature, arising directly or indirectly from the present or future lack of perfection, sufficiency, validity, or enforceability of the Obligations or any security therefor,

-75-

and any right or defense arising by reason of any claim or defense based upon an election of remedies by any member of the Lender Group or any Bank Product Provider including any defense based upon an impairment or elimination of such Borrower's rights of subrogation, reimbursement, contribution, or indemnity of such Borrower against any other Borrower. Without limiting the generality of the foregoing, each Borrower hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Obligations, the acceptance of any payment of any of the Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by Agent or Lenders at any time or times in respect of any default by any Borrower in the performance or satisfaction of any term, covenant, condition or provision of this Agreement, any and all other indulgences whatsoever by Agent or Lenders in respect of any of the Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Obligations or the addition, substitution or release, in whole or in part, of any Borrower. Without limiting the generality of the foregoing, each Borrower assents to any other action or delay in acting or failure to act on the part of any Agent or Lender with respect to the failure by any Borrower to comply with any of its respective Obligations, including any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this Section 2.15 afford grounds for terminating, discharging or relieving any Borrower, in whole or in part, from any of its Obligations under this Section 2.15, it being the intention of each Borrower that, so long as any of the Obligations hereunder remain unsatisfied, the Obligations of each Borrower under this Section 2.15 shall not be discharged except by performance and then only to the extent of such performance. The Obligations of each Borrower under this Section 2.15 shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to any other Borrower or any Agent or Lender. Each of the Borrowers waives, to the fullest extent permitted by law, the benefit of any statute of limitations affecting its liability hereunder or the enforcement hereof. Any payment by any Borrower or other circumstance which operates to toll any statute of limitations as to any Borrower shall operate to toll the statute of limitations as to each of the Borrowers. Each of the Borrowers waives any defense based on or arising out of any defense of any Borrower or any other Person, other than payment of the Obligations to the extent of such payment, based on or arising out of the disability of any Borrower or any other Person, or the validity, legality, or unenforceability of the Obligations or any part thereof from any cause, or the cessation from any cause of the liability of any Borrower other than payment of the Obligations to the extent of such payment. Agent may, at the election of the Required Lenders, foreclose upon any Collateral held by Agent by one or more judicial or nonjudicial sales or other dispositions, whether or not every aspect of any such sale is commercially reasonable or otherwise fails to comply with applicable law or may exercise any other right or remedy Agent, any other member of the Lender Group, or any Bank Product Provider may have against any Borrower or any other Person, or any security, in each case, without affecting or impairing in any way the liability of any of the Borrowers hereunder except to the extent the Obligations have been paid.

(f) Each Borrower represents and warrants to Agent and Lenders that such Borrower is currently informed of the financial condition of Borrowers and of all other circumstances which a diligent inquiry would reveal and which bear upon the risk of nonpayment of the Obligations. Each Borrower further represents and warrants to Agent and Lenders that such Borrower has read and understands the terms and conditions of the Loan Documents. Each Borrower hereby covenants that such Borrower will continue to keep informed of Borrowers' financial condition and of all other circumstances which bear upon the risk of nonpayment or nonperformance of the Obligations.

(g) The provisions of this <u>Section 2.15</u> are made for the benefit of Agent, each member of the Lender Group, each Bank Product Provider, and their respective successors and assigns, and may be enforced by it or them from time to time against any or all Borrowers as often as occasion therefor may arise

-76-

and without requirement on the part of Agent, any member of the Lender Group, any Bank Product Provider, or any of their successors or assigns first to marshal any of its or their claims or to exercise any of its or their rights against any Borrower or to exhaust any remedies available to it or them against any Borrower or to resort to any other source or means of obtaining payment of any of the Obligations hereunder or to elect any other remedy. The provisions of this <u>Section 2.15</u> shall remain in effect until all of the Obligations shall have been paid in full or otherwise fully satisfied. If at any time, any payment, or any part thereof, made in respect of any of the Obligations, is rescinded or must otherwise be restored or returned by Agent or any Lender upon the insolvency, bankruptcy or reorganization of any Borrower, or otherwise, the provisions of this <u>Section 2.15</u> will forthwith be reinstated in effect, as though such payment had not been made.

(h) Each Borrower hereby agrees that it will not enforce any of its rights that arise from the existence, payment, performance or enforcement of the provisions of this Section 2.15, including rights of subrogation, reimbursement, exoneration, contribution or indemnification and any right to participate in any claim or remedy of Agent, any other member of the Lender Group, or any Bank Product Provider against any Borrower, whether or not such claim, remedy or right arises in equity or under contract, statute or common law, including the right to take or receive from any Borrower, directly or indirectly, in cash or other property or by set-off or in any other manner, payment or security solely on account of such claim, remedy or right, unless and until such time as all of the Obligations have been paid in full in cash. Any claim which any Borrower may have against any other Borrower with respect to any payments to any Agent or any member of the Lender Group hereunder or under any of the Bank Product Agreements are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full in cash of the Obligations and, in the event of any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction relating to any Borrower, its debts or its assets, whether voluntary or involuntary, all such Obligations shall be paid in full in cash before any payment or distribution of any character, whether in cash, securities or other property, shall be made to any other Borrower therefor. If any amount shall be paid to any Borrower in violation of the immediately preceding sentence, such amount shall be held in trust for the benefit of Agent, for the benefit of the Lender Group and the Bank Product Providers, and shall forthwith be paid to Agent to be credited and applied to the Obligations and all other amounts payable under this Agreement, whether matured or unmatured, in accordance with the terms of this Agreement, or to be held as Collateral for any Obligations or other amounts payable under this Agreement thereafter arising. Notwithstanding anything to the contrary contained in this Agreement, no Borrower may exercise any rights of subrogation, contribution, indemnity, reimbursement or other similar rights against, and may not proceed or seek recourse against or with respect to any property or asset of, any other Borrower (the "Foreclosed Borrower"), including after payment in full of the Obligations, if all or any portion of the Obligations have been satisfied in connection with an exercise of remedies in respect of the Equity Interests of such Foreclosed Borrower whether pursuant to this Agreement or otherwise.

(i) Each of the Borrowers hereby acknowledges and affirms that it understands that to the extent the Obligations are secured by Real Property located in California, the Borrowers shall be liable for the full amount of the liability hereunder notwithstanding the foreclosure on such Real Property by trustee sale or any other reason impairing such Borrower's right to proceed against any other Loan Party. In accordance with Section 2856 of the California Civil Code or any similar laws of any other applicable jurisdiction, each of the Borrowers hereby waives until such time as the Obligations have been paid in full:

(i) all rights of subrogation, reimbursement, indemnification, and contribution and any other rights and defenses that are or may become available to the Borrowers by reason of Sections 2787 to 2855, inclusive, 2899, and 3433 of the California Civil Code or any similar laws of any other applicable jurisdiction;

-77-

(ii) all rights and defenses that the Borrowers may have because the Obligations are secured by Real Property located in California, meaning, among other things, that: (A) Agent, the other members of the Lender Group, and the Bank Product Providers may collect from the Borrowers without first foreclosing on any real or personal property collateral pledged by any Loan Party, and (B) if Agent, on behalf of the Lender Group, forecloses on any Real Property Collateral pledged by any Loan Party, (1) the amount of the Obligations may be reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is worth more than the sale price, and (2) the Lender Group may collect from the Loan Parties even if, by foreclosing on the Real Property Collateral, Agent or the other members of the Lender Group have destroyed or impaired any right the Borrowers may have to collect from any other Loan Party, it being understood that this is an unconditional and irrevocable waiver of any rights and defenses the Borrowers may have because the Obligations are secured by Real Property (including any rights or defenses based upon Sections 580a, 580d, or 726 of the California Code of Civil Procedure or any similar laws of any other applicable jurisdiction); and

(iii) all rights and defenses arising out of an election of remedies by Agent, the other members of the Lender Group, and the Bank Product Providers, even though that election of remedies, such as a nonjudicial foreclosure with respect to security for the Obligations, has destroyed the Borrowers' rights of subrogation and reimbursement against any other Loan Party by the operation of Section 580d of the California Code of Civil Procedure or any similar laws of any other applicable jurisdiction or otherwise.

3. CONDITIONS; TERM OF AGREEMENT.

3.1 <u>Conditions Precedent to the Initial Extension of Credit</u>. The obligation of each Lender to make the initial extensions of credit provided for hereunder is subject to the fulfillment, to the satisfaction of Agent and each Lender, of each of the conditions precedent set forth on <u>Schedule 3.1</u> to this Agreement (the making of such initial extensions of credit by a Lender being conclusively deemed to be its satisfaction or waiver of the conditions precedent).

3.2 <u>Conditions Precedent to all Extensions of Credit</u>. The obligation of the Lender Group (or any member thereof) to make any Revolving Loans hereunder (or to extend any other credit hereunder) at any time shall be subject to the following conditions precedent:

(a) the representations and warranties of each Loan Party or its Subsidiaries contained in this Agreement or in the other Loan Documents shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date); and

(b) no Default or Event of Default shall have occurred and be continuing on the date of such extension of credit, nor shall either result from the making thereof.

3.3 <u>Maturity</u>. The Commitments shall continue in full force and effect for a term ending on the Maturity Date (unless terminated earlier in accordance with the terms hereof).

3.4 <u>Effect of Maturity</u>. On the Maturity Date, all commitments of the Lender Group to provide additional credit hereunder shall automatically be terminated and all of the Obligations (other than Hedge



Obligations) immediately shall become due and payable without notice or demand and Borrowers shall be required to repay all of the Obligations (other than Hedge Obligations) in full. No termination of the obligations of the Lender Group (other than payment in full of the Obligations and termination of the Commitments) shall relieve or discharge any Loan Party of its duties, obligations, or covenants hereunder or under any other Loan Document and Agent's Liens in the Collateral shall continue to secure the Obligations and shall remain in effect until all Obligations have been paid in full. When all of the Obligations have been paid in full, Agent will, at Borrowers' sole expense, execute and deliver any termination statements, lien releases, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, Agent's Liens and all notices of security interests and liens previously filed by Agent.

3.5 <u>Early Termination by Borrowers</u>. Borrowers have the option, at any time upon ten Business Days prior written notice to Agent, to repay all of the Obligations in full and terminate the Commitments. The foregoing notwithstanding, (a) Borrowers may rescind termination notices relative to proposed payments in full of the Obligations with the proceeds of third party Indebtedness if the closing for such issuance or incurrence does not happen on or before the date of the proposed termination (in which case, a new notice shall be required to be sent in connection with any subsequent termination), and (b) Borrowers may extend the date of termination at any time with the consent of Agent (which consent shall not be unreasonably withheld or delayed).

3.6 <u>Conditions Subsequent</u>. The obligation of the Lender Group (or any member thereof) to continue to make Revolving Loans (or otherwise extend credit hereunder) is subject to the fulfillment, on or before the date applicable thereto, of the conditions subsequent set forth on <u>Schedule 3.6</u> to this Agreement (the failure by Borrowers to so perform or cause to be performed such conditions subsequent as and when required by the terms thereof (unless such date is extended, in writing, by Agent, which Agent may do without obtaining the consent of the other members of the Lender Group), shall constitute an Event of Default).

4. REPRESENTATIONS AND WARRANTIES.

In order to induce the Lender Group to enter into this Agreement, each Borrower makes the following representations and warranties to the Lender Group which shall be true, correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the Closing Date, and shall be true, correct, and complete, in all material respects (except that such materiality are qualified or modified by materiality in the text thereof), as of the Closing Date, and shall be true, correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the date of the making of each Revolving Loan (or other extension of credit) made thereafter, as though made on and as of the date of such Revolving Loan (or other extension of credit) (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date), and such representations and warranties shall survive the execution and delivery of this Agreement:

4.1 Due Organization and Qualification; Subsidiaries.

(a) Each Loan Party (i) is duly organized and existing and in good standing under the laws of the jurisdiction of its organization, (ii) is qualified to do business in any state where the failure to be so qualified could reasonably be expected to result in a Material Adverse Effect, and (iii) has all requisite power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to enter into the Loan Documents to which it is a party and to carry out the transactions contemplated thereby.

(b) Set forth on <u>Schedule 4.1(b)</u> to this Agreement (as such Schedule may be updated from time to time to reflect changes resulting from transactions permitted under this Agreement) is a complete

-79-

and accurate description of the authorized Equity Interests of each Loan Party, by class, and, as of the Closing Date, a description of the number of shares of each such class that are issued and outstanding.

(c) Set forth on <u>Schedule 4.1(c)</u> to this Agreement (as such Schedule may be updated from time to time to reflect changes resulting from transactions permitted under this Agreement), is a complete and accurate list of the Loan Parties' direct and indirect Subsidiaries, showing: (i) the number of shares of each class of common and preferred Equity Interests authorized for each of such Subsidiaries, and (ii) the number and the percentage of the outstanding shares of each such class owned directly or indirectly by any Loan Party. All of the outstanding Equity Interests of each such Subsidiary has been validly issued and is fully paid and non-assessable.

(d) Except as set forth on <u>Schedule 4.1(d)</u> to this Agreement, there are no subscriptions, options, warrants, or calls relating to any shares of any Loan Party's or any of its Subsidiaries' Equity Interests, including any right of conversion or exchange under any outstanding security or other instrument. No Loan Party is subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any shares of its Equity Interests or any security convertible into or exchangeable for any of its Equity Interests.

4.2 Due Authorization; No Conflict.

(a) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of such Loan Party.

(b) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party do not and will not (i) violate any material provision of federal, state, or local law or regulation applicable to any Loan Party or its Subsidiaries, the Governing Documents of any Loan Party or its Subsidiaries, or any order, judgment, or decree of any court or other Governmental Authority binding on any Loan Party or its Subsidiaries, the Preferred Equity Documents, the Permitted Convertible Notes, or the Permitted Subordinated Notes, where any such conflict, breach or default could individually or in the aggregate reasonably be expected to have a Material Adverse Effect, (iii) require any approval of any holder of Equity Interests of a Loan Party or any approval or consent of any Person under any Material Contract of any Loan Party, the Permitted Convertible Notes, or the Permitted Subordinated Notes, the Preferred Equity Documents, the Permitted Subordinated Notes, other than consents or approvals of a Loan Party or any approval or consent of any Person under any Material Contract of any Loan Party, the Preferred Equity Documents, the Permitted Subordinated Notes, other than consents or approvals that have been obtained and that are still in force and effect and except, in the case of Material Contracts, the Preferred Equity Documents, the Permitted Convertible Notes, and the Permitted Subordinated Notes, for consents or approvals, the failure to obtain could not individually or in the aggregate reasonably be expected to cause a Material Adverse Effect.

4.3 **Governmental Consents**. The execution, delivery, and performance by each Loan Party of the Loan Documents to which such Loan Party is a party and the consummation of the transactions contemplated by the Loan Documents do not and will not require any registration with, consent, or approval of, or notice to, or other action with or by, any Governmental Authority, other than (i) registrations, consents, approvals, notices, or other actions that have been obtained and that are still in force and effect, and (ii) certain filings and recordings with respect to the Collateral to be made, or otherwise delivered to Agent for filing or recordation, as of the Closing Date.

4.4 Binding Obligations; Perfected Liens.

-80-

(a) Each Loan Document has been duly executed and delivered by each Loan Party that is a party thereto and is the legally valid and binding obligation of such Loan Party, enforceable against such Loan Party in accordance with its respective terms, except as enforcement may be limited by equitable principles or by bankruptcy, insolvency, reorganization, moratorium, or similar laws relating to or limiting creditors' rights generally.

(b) Agent's Liens are validly created, perfected (other than (i) in respect of motor vehicles that are subject to a certificate of title, (ii) money, (iii) letter-of-credit rights (other than supporting obligations), (iv) commercial tort claims (other than those that, by the terms of the Guaranty and Security Agreement, are required to be perfected), and (v) any Deposit Accounts and Securities Accounts not subject to a Control Agreement as permitted by Section 7(k)(iv) of the Guaranty and Security Agreement, and subject only to the filing of financing statements and the recordation of any Copyright Security Agreement, in each case, in the appropriate filing offices), and first priority Liens, subject only to Permitted Liens which are non-consensual Permitted Liens, permitted purchase money Liens, or the interests of lessors under Capital Leases.

4.5 <u>Title to Assets: No Encumbrances</u>. Each of the Loan Parties has (a) good, sufficient and legal title to (in the case of fee interests in Real Property), (b) valid leasehold interests in (in the case of leasehold interests in real or personal property), and (c) good and marketable title to (in the case of all other personal property), all of their respective assets reflected in their most recent financial statements delivered pursuant to <u>Section 5.1</u>, in each case except for assets disposed of since the date of such financial statements to the extent permitted hereby. All of such assets are free and clear of Liens except for Permitted Liens.

4.6 Litigation.

(a) There are no actions, suits, or proceedings pending or, to the knowledge of any Borrower, after due inquiry, threatened in writing against a Loan Party or any of its Subsidiaries that either individually or in the aggregate could reasonably be expected to result in a Material Adverse Effect.

(b) <u>Schedule 4.6(b)</u> to this Agreement sets forth a complete and accurate description of each of the actions, suits, or proceedings with asserted liabilities in excess of, or that could reasonably be expected to result in liabilities in excess of, \$5,000,000 that, as of the Closing Date, is pending or, to the knowledge of any Borrower, after due inquiry, threatened against a Loan Party or any of its Subsidiaries.

4.7 <u>Compliance with Laws</u>. No Loan Party nor any of its Subsidiaries (a) is in violation of any applicable laws, rules, regulations, executive orders, or codes (including Environmental Laws) that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect, or (b) is subject to or in default with respect to any final judgments, writs, injunctions, decrees, rules or regulations of any court or any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.

4.8 **No Material Adverse Effect.** All historical financial statements relating to the Loan Parties and their Subsidiaries that have been delivered by Borrowers to Agent have been prepared in accordance with GAAP (except, in the case of unaudited financial statements, for the lack of footnotes and being subject to year-end audit adjustments) and present fairly in all material respects, the Loan Parties' and their Subsidiaries' consolidated financial condition as of the date thereof and results of operations for the period then ended. Since December 31, 2016, no event, circumstance, or change has occurred that has or could reasonably be expected to result in a Material Adverse Effect.

4.9 Solvency.

-81-

(a) Each Loan Party is Solvent.

(b) No transfer of property is being made by any Loan Party and no obligation is being incurred by any Loan Party in connection with the transactions contemplated by this Agreement or the other Loan Documents with the intent to hinder, delay, or defraud either present or future creditors of such Loan Party.

4.10 **Employee Benefits**. No Loan Party, none of their Subsidiaries, nor any of their ERISA Affiliates maintains or contributes to any Benefit Plan.

4.11 Environmental Condition. Except as set forth on <u>Schedule 4.11</u> to this Agreement, (a) to each Borrower's knowledge, no Loan Party's nor any of its Subsidiaries' properties or assets has ever been used by a Loan Party, its Subsidiaries, or by previous owners or operators in the disposal of, or to produce, store, handle, treat, release, or transport, any Hazardous Materials, where such disposal, production, storage, handling, treatment, release or transport was in violation, in any material respect, of any applicable Environmental Law, (b) to each Borrower's knowledge, after due inquiry, no Loan Party's nor any of its Subsidiaries' properties or assets has ever been designated or identified in any manner pursuant to any environmental protection statute as a Hazardous Materials disposal site, (c) no Loan Party nor any of its Subsidiaries has received notice that a Lien arising under any Environmental Law has attached to any revenues or to any Real Property owned or operated by a Loan Party or its Subsidiaries, and (d) no Loan Party nor any of its Subsidiaries nor any of their respective facilities or operations is subject to any outstanding written order, consent decree, or settlement agreement with any Person relating to any Environmental Law or Environmental Liability that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.

4.12 **Complete Disclosure**. All factual information taken as a whole (other than forward-looking information and projections and information of a general economic nature and general information about the industry of any Loan Party or its Subsidiaries) furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender (including all information contained in the Schedules hereto or in the other Loan Documents) for purposes of or in connection with this Agreement or the other Loan Documents, and all other such factual information taken as a whole (other than forward-looking information and projections and information of a general economic nature and general information about the industry of any Loan Party or its Subsidiaries) hereafter furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender will be, true and accurate, in all material respects, on the date as of which such information is dated or certified and not incomplete by omitting to state any fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information all Projections represent, Borrowers' good faith estimate, on the date such Projections are delivered, of the Loan Parties' and their Subsidiaries' future performance for the periods covered thereby based upon assumptions believed by Borrowers to be reasonable at the time of the delivery thereof to Agent (it being understood that such Projections are subject to significant uncertainties and their Subsidiaries, and no assumptions which Borrowers' good faith estimate, projections or forecasts based on methods and assumptions which Borrowers' believed to be reasonable at the time such Projections are obtive the control of the Loan Parties and their Subsidiaries, and no assumptions which Borrowers' believed to be reasonable at the time such Projections were prepared, are not to be viewed as facts, and that actual results during the period

4.13 **Patriot Act**. To the extent applicable, each Loan Party is in compliance, in all material respects, with the (a) Trading with the Enemy Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, and (b) Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA Patriot Act of 2001, as amended) (the "Patriot Act").

-82-

4.14 **Indebtedness**. Set forth on <u>Schedule 4.14</u> to this Agreement is a true and complete list of all Indebtedness of each Loan Party and each of its Subsidiaries outstanding immediately prior to the Closing Date (other than unsecured Permitted Indebtedness outstanding immediately prior to the Closing Date that is to remain outstanding immediately after giving effect to the closing hereunder on the Closing Date and such Schedule accurately sets forth the aggregate principal amount of such Indebtedness as of the Closing Date.

4.15 **Payment of Taxes**. Except as otherwise permitted under <u>Section 5.5</u>, all U.S. federal income tax, and other material Tax returns and reports of each Loan Party and its Subsidiaries required to be filed by any of them have been timely filed, and all Taxes shown on such Tax returns to be due and payable and all other Taxes upon a Loan Party and its Subsidiaries and upon their respective assets, income, businesses and franchises that are due and payable have been paid when due and payable except such Tax returns or Taxes, as applicable, that are the subject of a Permitted Protest.

4.16 <u>Margin Stock</u>. Neither any Loan Party nor any of its Subsidiaries owns any Margin Stock or is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of the loans made to Borrowers will be used to purchase or carry any Margin Stock or to extend credit to others for the purpose of purchasing or carrying any Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors. Neither any Loan Party nor any of its Subsidiaries expects to acquire any Margin Stock.

4.17 <u>Governmental Regulation</u>. No Loan Party nor any of its Subsidiaries is subject to regulation under the Federal Power Act or the Investment Company Act of 1940 or under any other federal or state statute or regulation (other than the Communications Laws) which may limit its ability to incur Indebtedness or which may otherwise render all or any portion of the Obligations unenforceable. No Loan Party nor any of its Subsidiaries is a "registered investment company" or a company "controlled" by a "registered investment company" or a "principal underwriter" of a "registered investment company" as such terms are defined in the Investment Company Act of 1940.

4.18 **OFAC:** Sanctions: Anti-Corruption Laws: Anti-Money Laundering Laws. No Loan Party or any of its Subsidiaries is in violation of any Sanctions. No Loan Party nor any of its Subsidiaries nor, to the knowledge of such Loan Party, any director, officer, employee, agent or Affiliate of such Loan Party or such Subsidiary (a) is a Sanctioned Person or a Sanctioned Entity, (b) has any assets located in Sanctioned Entities, or (c) derives revenues from investments in, or to the knowledge of such Loan Party, transactions with Sanctioned Persons or Sanctioned Entities. Each of the Loan Parties and its Subsidiaries has implemented and maintains in effect policies and procedures designed to ensure compliance with all Sanctions, Anti-Corruption Laws and Anti-Money Laundering Laws. Each of the Loan Parties and its Subsidiaries, and to the knowledge of each such Loan Party, each director, officer, employee, agent and Affiliate of each such Loan Party and each such Subsidiary, is in compliance with all Sanctions, Anti-Corruption Laws. No proceeds of any Loan made or Letter of Credit issued hereunder will be used to fund any operations in, finance any investments or activities in, or make any payments to, a Sanctioned Person or a Sanctioned Entity, or otherwise used in any manner that would result in a violation of any Sanction, Anti-Corruption Law or Anti-Money Laundering Law by any Person (including any Lender, Bank Product Provider, or other individual or entity participating in any transaction).

4.19 <u>Employee and Labor Matters</u>. There is (i) no unfair labor practice complaint pending or, to the knowledge of any Borrower, threatened against any Loan Party or its Subsidiaries before any Governmental Authority and no grievance or arbitration proceeding pending or threatened against any Loan Party or its Subsidiaries which arises out of or under any collective bargaining agreement and that could reasonably be expected to result in a material liability, (ii) no strike, labor dispute, slowdown, stoppage or similar action or grievance pending or threatened in writing against any Loan Party or its Subsidiaries that could reasonably be expected to result in a material liability, (ii) to the knowledge of any Borrower, after due inquiry, no union representation question existing with respect to the employees of any Loan Party or its Subsidiaries and no union

-83-

organizing activity taking place with respect to any of the employees of any Loan Party or its Subsidiaries. None of any Loan Party or its Subsidiaries has incurred any liability or obligation under the Worker Adjustment and Retraining Notification Act or similar state law, which remains unpaid or unsatisfied. The hours worked and payments made to employees of each Loan Party and its Subsidiaries have not been in violation of the Fair Labor Standards Act or any other applicable legal requirements, except to the extent such violations could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. All material payments due from any Loan Party or its Subsidiaries on account of wages and employee health and welfare insurance and other benefits have been paid or accrued as a liability on the books of Borrowers, except where the failure to do so could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

4.20 <u>Material Contracts</u>. Set forth on <u>Schedule 4.20</u> (as such Schedule may be updated from time to time in accordance herewith) is a reasonably detailed description of Material Contracts of each Loan Party as if the most recent date on which Administrative Borrower provided the Compliance Certificate pursuant to <u>Section 5.1</u>; provided that Borrowers may amend <u>Schedule 4.20</u> to add additional Material Contracts so long as such amendment occurs by written notice to Agent on the date that Administrative Borrower provides the Compliance Certificate. Except for matters which, either individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, each Material Contract (other than those that have expired at the end of their normal terms) (a) is in full force and effect and is binding upon and enforceable against the applicable Loan Party and, to each Borrower's knowledge after due inquiry, each other Person that is a party thereto in accordance with its terms, (b) has not been otherwise amended or modified (other than amendments or modifications permitted by <u>Section 6.6(b)</u>, and (c) is not in default due to the action or inaction of the applicable Loan Party.

4.21 Leases. Each Loan Party enjoys peaceful and undisturbed possession under all leases material to their business and to which they are parties or under which they are operating, and, subject to Permitted Protests, all of such material leases are valid and subsisting and no material default by the applicable Loan Party exists under any of them.

4.22 Eligible Accounts. As to each Account that is identified by Borrowers as an Eligible Account in a Borrowing Base Certificate submitted to Agent, such Account is (a) a bona fide existing payment obligation of the applicable Account Debtor created by the sale and delivery of Inventory or the rendition of services to such Account Debtor in the ordinary course of a Borrower's business, (b) owed to a Borrower without any known defenses, disputes, offsets, counterclaims, or rights of return or cancellation, and (c) not excluded as ineligible by virtue of one or more of the excluding criteria (other than any Agent-discretionary criteria) set forth in the definition of Eligible Accounts.

4.23 <u>Intellectual Property</u>. Each Loan Party owns exclusively or holds licenses in all intellectual property that is necessary in or material to the conduct of the business of such Loan Party.

4.24 <u>Music Performance Rights</u>. The Loan Parties, individually and collectively, hold all Music Performance Rights necessary to the operation of the business of such Loan Parties except for the failure to hold such Music Performance Rights that individually or in the aggregate could be reasonably expected to result in a Material Adverse Effect.

4.25 Location of Books and Records. The books and records of the Loan Parties are not stored with a bailee, warehouseman, or similar party and is located only at the locations identified on <u>Schedule 4.25</u> to this Agreement (as such Schedule may be updated pursuant to <u>Section 5.14</u>).

4.26 **Inventory Records**. Each Loan Party keeps correct and accurate records itemizing and describing the type, quality, and quantity of its Inventory and the book value thereof.

-84-

4.27 **FCC Licenses**. No Loan Party or any of its Subsidiaries is required to maintain any FCC Licenses.

4.28 Material Subsidiaries. All Material Subsidiaries are Loan Parties.

4.29 <u>Hedge Agreements</u>. On each date that any Hedge Agreement is executed by any Hedge Provider, Borrower and each other Loan Party satisfy all eligibility, suitability and other requirements under the Commodity Exchange Act (7 U.S.C. § 1, et seq., as in effect from time to time) and the Commodity Futures Trading Commission regulations.

5. AFFIRMATIVE COVENANTS.

Each Borrower covenants and agrees that, until the termination of all of the Commitments and payment in full of the Obligations:

5.1 <u>Financial Statements, Reports, Certificates</u>. Borrowers (a) will deliver to Agent, with copies to each Lender, each of the financial statements, reports, and other items set forth on <u>Schedule 5.1</u> to this Agreement no later than the times specified therein, (b) agree that no Subsidiary of a Loan Party will have a fiscal year different from that of Administrative Borrower, (c) agree to maintain a system of accounting that enables Borrowers to produce financial statements in accordance with GAAP, and (d) agree that they will, and will cause each other Loan Party to, (i) keep a reporting system that shows all additions, sales, claims, returns, and allowances with respect to their and their Subsidiaries' sales, and (ii) maintain their billing systems and practices substantially as in effect as of the Closing Date and shall only make material modifications thereto with notice to, and with the consent of, Agent.

5.2 **Reporting**. Borrowers (a) will deliver to Agent (and if so requested by Agent, with copies for each Lender) each of the reports set forth on <u>Schedule</u> <u>5.2</u> to this Agreement at the times specified therein, and (b) agree to use commercially reasonable efforts in cooperation with Agent to facilitate and implement a system of electronic collateral reporting in order to provide electronic reporting of each of the items set forth on such Schedule.

5.3 **Existence**. Except as otherwise permitted under Section 6.3 or Section 6.4, each Loan Party will, and will cause each of its Subsidiaries to, at all times preserve and keep in full force and effect such Person's valid existence and good standing in its jurisdiction of organization and, except as could not reasonably be expected to result in a Material Adverse Effect, good standing with respect to all other jurisdictions in which it is qualified to do business and any rights, franchises, permits, licenses, accreditations, authorizations, or other approvals material to their businesses.

5.4 <u>Maintenance of Properties</u>. Each Loan Party will, and will cause each of its Subsidiaries to, maintain and preserve all of its assets that are necessary or useful in the proper conduct of its business in good working order and condition, ordinary wear, tear, casualty, and condemnation and Permitted Dispositions excepted.

5.5 <u>Taxes</u>. Each Loan Party will, and will cause each of its Subsidiaries to, pay in full before delinquency or before the expiration of any extension period all Taxes imposed, levied, or assessed against it, or any of its assets or in respect of any of its income, businesses, or franchises, other than (A) Taxes not in excess of \$1,000,000 outstanding at any time and to the extent that the failure to do so could not reasonably be expect to have a Material Adverse Effect; or (B) Taxes not in excess of \$5,000,000 outstanding at any time and to the extent that the validity of such Tax is the subject of a Permitted Protest, and to the extent that the failure to do so could not reasonably be expected to have a Material Adverse Effect.

5.6 Insurance. Each Loan Party will, and will cause each of its Subsidiaries to, at Borrowers' expense, maintain insurance respecting each of each Loan Party's and its Subsidiaries' assets wherever located,

-85-

covering liabilities, losses or damages as are customarily are insured against by other Persons engaged in same or similar businesses and similarly situated and located. All such policies of insurance shall be with financially sound and reputable insurance companies acceptable to Agent (it being agreed that, as of the Closing Date, Zurich Insurance Company is acceptable to Agent) and in such amounts as is carried generally in accordance with sound business practice by companies in similar businesses similarly situated and located and, in any event, in amount, adequacy, and scope reasonably satisfactory to Agent (it being agreed that the amount, adequacy, and scope of the policies of insurance of Borrowers in effect as of the Closing Date are acceptable to Agent). All property insurance policies are to be made payable to Agent for the benefit of Agent and the Lenders, as their interests may appear, in case of loss, pursuant to a standard lender's loss payable endorsement with a standard non-contributory "lender" or "secured party" clause and are to contain such other provisions as Agent may reasonably require to fully protect the Lenders' interest in the Collateral and to any payments to be made under such policies. All certificates of property and general liability insurance are to be delivered to Agent, with the lender's loss payable and additional insured endorsements in favor of Agent and shall provide for not less than thirty days (ten days in the case of non-payment) prior written notice to Agent of the exercise of any right of cancellation. If any Loan Party or its Subsidiaries fails to maintain such insurance, Agent may arrange for such insurance, but at Borrowers' expense and without any responsibility on Agent's part for obtaining the insurance, the solvency of the insurance companies, the adequacy of the coverage, or the collection of claims. Borrowers shall give Agent prompt notice of any loss exceeding \$2,000,000 covered by the casualty or business interruption insurance of any Loan Party or its Subsidiaries. Upon the occurrence and during the continuance of an Event of Default, Agent shall have the sole right to file claims under any property and general liability insurance policies in respect of the Collateral, to receive, receipt and give acquittance for any payments that may be payable thereunder, and to execute any and all endorsements, receipts, releases, assignments, reassignments or other documents that may be necessary to effect the collection, compromise or settlement of any claims under any such insurance policies.

5.7 Inspection.

(a) Each Loan Party will, and will cause each of its Subsidiaries to, permit Agent, any Lender, and each of their respective duly authorized representatives or agents to visit any of its properties and inspect any of its assets or books and records, to examine and make copies of its books and records, and to discuss its affairs, finances, and accounts with, and to be advised as to the same by, its officers and employees (provided, that an authorized representative of a Borrower shall be allowed to be present) at such reasonable times and intervals as Agent or any Lender, as applicable, may designate and, so long as no Default or Event of Default has occurred and is continuing, with reasonable prior notice to Borrowers and during regular business hours, at Borrowers' expense in accordance with the provisions of the Fee Letter, subject to the limitations set forth below in Section 5.7(c).

(b) Each Loan Party will, and will cause each of its Subsidiaries to, permit Agent and each of its duly authorized representatives or agents to conduct field examinations, appraisals or valuations at such reasonable times and intervals as Agent may designate, at Borrowers' expense in accordance with the provisions of the Fee Letter, subject to the limitations set forth below in Section 5.7(c).

(c) So long as (i) no Event of Default shall have occurred and be continuing during a calendar year, and (ii) Liquidity of the Loan Parties remains above the greater of (x) 12.5% of the Line Cap and (y) \$17,500,000, Borrowers shall not be obligated to reimburse Agent for more than one (1) field examinations in such calendar year, in each case, except for field examinations and appraisals conducted in connection with a proposed Permitted Acquisition (whether or not consummated).

5.8 <u>Compliance with Laws</u>. Each Loan Party will, and will cause each of its Subsidiaries to, comply with the requirements of all applicable laws, rules, regulations, and orders of any Governmental Authority, including the Communications Act, other than laws, rules, regulations, and orders the non-compliance with which, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect.

-86-

5.9 **Environmental**. Each Loan Party will, and will cause each of its Subsidiaries to,

(a) Keep any property either owned or operated by any Loan Party or its Subsidiaries free of any Environmental Liens or post bonds or other financial assurances sufficient to satisfy the obligations or liability evidenced by such Environmental Liens,

(b) Comply, in all material respects, with Environmental Laws and provide to Agent documentation of such compliance which Agent reasonably requests,

(c) Promptly notify Agent of any release of which any Loan Party has knowledge of a Hazardous Material in any reportable quantity from or onto property owned or operated by any Loan Party or its Subsidiaries and take any Remedial Actions required to abate said release or otherwise to come into compliance, in all material respects, with applicable Environmental Law, and

(d) Promptly, but in any event within five Business Days of its receipt thereof, provide Agent with written notice of any of the following: (i) notice that an Environmental Lien has been filed against any of the real or personal property of a Loan Party or its Subsidiaries, (ii) commencement of any Environmental Action or written notice that an Environmental Action will be filed against a Loan Party or its Subsidiaries, and (iii) written notice of a violation, citation, or other administrative order from a Governmental Authority.

5.10 **Disclosure Updates**. Each Loan Party will, promptly and in no event later than five Business Days after obtaining knowledge thereof, notify Agent if any written information, exhibit, or report furnished to Agent or the Lenders contained, at the time it was furnished, any untrue statement of a material fact or omitted to state any material fact necessary to make the statements contained therein not misleading in light of the circumstances in which made. The foregoing to the contrary notwithstanding, any notification pursuant to the foregoing provision will not cure or remedy the effect of the prior untrue statement of a material fact or omission of any material fact nor shall any such notification have the effect of amending or modifying this Agreement or any of the Schedules hereto.

5.11 **Formation of Subsidiaries**. Each Loan Party will, at the time that any Loan Party forms any direct or indirect Subsidiary, acquires any direct or indirect Subsidiary after the Closing Date, or at any time when any direct or indirect Subsidiary of a Loan Party that previously was an Immaterial Subsidiary becomes a Material Subsidiary, within ten days of such event (or such later date as permitted by Agent in its sole discretion) (a) unless such Subsidiary is an Excluded Subsidiary, cause such new Subsidiary (i) if Agent requests that such Subsidiary be joined as a Borrower hereunder, to provide to Agent a Joinder to this Agreement, and (ii) to provide to Agent a joinder to the Guaranty and Security Agreement, in each case, together with such other security agreements, as well as appropriate financing statements, all in form and substance reasonably satisfactory to Agent (including being sufficient to grant Agent a first priority Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Subsidiary); (b) provide, or cause the applicable Loan Party to provide, to Agent a pledge agreement (or an addendum to the Guaranty and Security Agreement) and appropriate certificates and powers or financing statements, pledging all of the direct or beneficial ownership interest in such new Subsidiary of a Loan Party that is a CFC or a Disregarded Domestic Person (and none of the Equity Interests of any Subsidiary of such CFC or Disregarded Domestic Person) shall be required to be pledged if pledging a greater amount would be reasonably likely to result in adverse tax consequences to Borrowers (which pledge, if reasonably requested by Agent, shall be governed by the laws of the jurisdiction of such Subsidiary), and (c) provide to Agent all other documentation, including the Governing Documents of such Subsidiary and one or more opinions of counsel reasonably satisfactory to Agent, which, in its opinion, is appropriate with respect to the execution and delivery of the applicable documentation r



document, agreement, or instrument executed or issued pursuant to this Section 5.11 shall constitute a Loan Document.

5.12 Further Assurances. Each Loan Party will, and will cause each of the other Loan Parties to, at any time upon the reasonable request of Agent, execute or deliver to Agent any and all financing statements, fixture filings, security agreements, pledges, assignments, mortgages, deeds of trust, opinions of counsel, and all other documents (the "Additional Documents") that Agent may reasonably request in form and substance reasonably satisfactory to Agent, to create, perfect, and continue perfected or to better perfect Agent's Liens in all of the assets of each of the Loan Parties (whether now owned or hereafter arising or acquired, tangible or intangible, real or personal) (other than any assets expressly excluded from the Collateral (as defined in the Guaranty and Security Agreement) pursuant to Section 3 of the Guaranty and Security Agreement), and in order to fully consummate all of the transactions contemplated hereby and under the other Loan Documents; provided, that the foregoing shall not apply to any Subsidiary of a Loan Party that is a CFC (or any Subsidiary thereof) or Disregarded Domestic Person if providing such documents would be reasonably likely to result in adverse tax consequences or the costs to the Loan Parties of providing such documents are unreasonably excessive (as determined by Agent in consultation with Borrowers) in relation to the benefits to Agent and the Lenders of the security afforded thereby. To the maximum extent permitted by applicable law, if any Borrower or any other Loan Party refuses or fails to execute or deliver any reasonably requested Additional Documents within a reasonable period of time not to exceed 5 Business Days following the request to do so, each Borrower and each other Loan Party hereby authorizes Agent to execute any such Additional Documents in the applicable Loan Party's name and authorizes Agent to file such executed Additional Documents in any appropriate filing office. In furtherance of, and not in limitation of, the foregoing, each Loan Party shall take such actions as Agent may reasonably request from time to time to ensure that the Obligations are guaranteed by the Guarantors and are secured by substantially all of the assets of the Loan Parties, including all of the outstanding capital Equity Interests of each Borrower (other than Administrative Borrower) and its Subsidiaries (in each case, other than with respect to any assets expressly excluded from the Collateral (as defined in the Guaranty and Security Agreement) pursuant to Section 3 of the Guaranty and Security Agreement).

5.13 <u>Material Contracts</u>. Each Loan Party will use commercially reasonable efforts to ensure that any Material Contract entered into after the Closing Date by any Loan Party or one of its Subsidiaries that generates or, by its terms, will generate revenue, permits the assignment of such agreement (and all rights of such Borrower or such Subsidiary, as applicable, thereunder) to such Borrower's or such Subsidiary's lenders or an agent for any lenders (and any transferees of such lenders or such agent, applicable).

5.14 Location of Books and Records; Chief Executive Office. Each Loan Party will, and will cause each of its Subsidiaries to, keep (a) their books and records only at the locations identified on <u>Schedule 4.25</u> to this Agreement (provided that Borrowers may amend <u>Schedule 4.25</u> to this Agreement so long as such amendment occurs by written notice to Agent not less than ten days prior to the date on which such Inventory is moved to such new location and so long as Agent has consented to such amendment and such new location is within the continental United States), and (b) their respective chief executive offices only at the locations identified on Schedule 7 to the Guaranty and Security Agreement. Each Loan Party will, and will cause each of its Subsidiaries to, use their commercially reasonable efforts to obtain Collateral Access Agreements for each of the locations identified on <u>Schedule 7</u> to the Guaranty and Security Agreement and <u>Schedule 4.25</u> to this Agreement.

5.15 **OFAC: Sanctions: Anti-Corruption Laws: Anti-Money Laundering Laws**. Each Loan Party will, and will cause each of its Subsidiaries to comply with all applicable Sanctions, Anti-Corruption Laws and Anti-Money Laundering Laws. Each of the Loan Parties and its Subsidiaries shall implement and maintain in effect policies and procedures designed to ensure compliance by the Loan Parties and their Subsidiaries and their respective directors, officers, employees, agents and Affiliates with all Sanctions, Anti-Corruption Laws and Anti-Money Laundering Laws.

6. NEGATIVE COVENANTS.

-88-

Each Borrower covenants and agrees that, until the termination of all of the Commitments and the payment in full of the Obligations:

6.1 <u>Indebtedness</u>. Each Loan Party will not, and will not permit any of its Subsidiaries to, create, incur, assume, suffer to exist, guarantee, or otherwise become or remain, directly or indirectly, liable with respect to any Indebtedness, except for Permitted Indebtedness.

6.2 Liens. Each Loan Party will not, and will not permit any of its Subsidiaries to, create, incur, assume, or suffer to exist, directly or indirectly, any Lien on or with respect to any of its assets, of any kind, whether now owned or hereafter acquired, or any income or profits therefrom, except for Permitted Liens.

6.3 **Restrictions on Fundamental Changes**. Each Loan Party will not, and will not permit any of its Subsidiaries to,

(a) Other than in order to consummate a Permitted Acquisition, enter into any merger, consolidation, reorganization, or recapitalization, or reclassify its Equity Interests, except for (i) any merger between Loan Parties; provided, that a Borrower must be the surviving entity of any such merger to which it is a party, (ii) any merger between a Loan Party and a Subsidiary of such Loan Party that is not a Loan Party so long as such Loan Party is the surviving entity of any such merger, (iii) any merger between Subsidiaries of any Loan Party that are not Loan Parties, and (iv) any Subsidiary may liquidate or dissolve if Administrative Borrower determines in good faith that such liquidation or dissolution is in the best interests of the Borrowers and is not materially disadvantageous to the Lenders; provided that any such merger or consolidation involving a Person that is not a wholly-owned Subsidiary immediately prior thereto shall not be permitted unless it is also permitted under Section 6.9.

(b) liquidate, wind up, or dissolve itself (or suffer any liquidation or dissolution), except for (i) the liquidation or dissolution of nonoperating Subsidiaries of any Loan Party with nominal assets and nominal liabilities, (ii) the liquidation or dissolution of a Loan Party (other than any Borrower) or any of its wholly-owned Subsidiaries so long as all of the assets (including any interest in any Equity Interests) of such liquidating or dissolving Loan Party or Subsidiary are transferred to a Loan Party that is not liquidating or dissolving, or (iii) the liquidation or dissolution of a Subsidiary of any Loan Party that is not a Loan Party (other than any such Subsidiary the Equity Interests of which (or any portion thereof) is subject to a Lien in favor of Agent) so long as all of the assets of such liquidating or dissolving Subsidiary are transferred to a Subsidiary of a Loan Party that is not liquidating or dissolving, or

(c) suspend or cease operating a substantial portion of its or their business, except as permitted pursuant to clauses (a) or (b) above or in connection with a transaction permitted under <u>Section 6.4</u>.

6.4 **Disposal of Assets**. Other than Permitted Dispositions or transactions expressly permitted by <u>Sections 6.3</u> or <u>6.9</u>, each Loan Party will not, and will not permit any of its Subsidiaries to, convey, sell, lease, license, assign, transfer, or otherwise dispose of any of its or their assets.

6.5 <u>Nature of Business</u>. Each Loan Party will not, and will not permit any of its Subsidiaries to, make any change in the nature of its or their business as described in <u>Schedule 6.5</u> to this Agreement or acquire any properties or assets that are not reasonably related to the conduct of such business activities; <u>provided</u>, that the foregoing shall not prevent any Loan Party and its Subsidiaries from engaging in any business that is reasonably related or ancillary to its or their business.

6.6 Prepayments and Amendments. Each Loan Party will not, and will not permit any of its Subsidiaries to,

-89-

(a) Except in connection with Refinancing Indebtedness permitted by Section 6.1,

(i) Optionally prepay, redeem, defease, purchase or otherwise acquire (A) any Indebtedness of any Loan Party or its Subsidiaries that is secured by a Lien on the Collateral that is junior to the Agent's Lien, (B) the Preferred Equity Investment, unless the Payment Conditions have been satisfied, (C) the Permitted Subordinated Notes; and (D) the Permitted Convertible Notes or the 2020 Unsecured Convertible Notes other than, solely with respect to this clause (D) any of the following: (1) payments of regularly scheduled interest as and when due in respect of any such Indebtedness, other than payments prohibited by the subordination provisions thereof, (2) refinancings of any such Indebtedness permitted by <u>Section 6.1</u> hereof, (3) payments upon conversion of any such Indebtedness into common stock of the Administrative Borrower made in common stock of the Administrative Borrower, together with cash payments in lieu of issuance of fractional shares and payments of accrued but unpaid interest, in each case in connection with such conversion, (4) other payments of or in respect of any such Indebtedness made solely with (or with the proceeds of a substantially concurrent issuance and sale of) Equity Interests (other than Disqualified Equity Interests) in the Administrative Borrower, (5) cash payments upon conversion of Permitted Convertible Notes or the 2020 Unsecured Convertible Notes pursuant to the terms thereof, and (6) other payments of the Permitted Convertible Notes or the 2020 Unsecured Convertible Notes so long as the Payment Conditions are satisfied, or

(ii) Make any payment on account of Indebtedness that has been contractually subordinated in right of payment to the Obligations if such payment is not permitted at such time under the subordination terms and conditions, or

(b) Directly or indirectly, amend, modify, or change any of the terms or provisions of:

(i) any agreement, instrument, document, indenture, or other writing evidencing or concerning Permitted Indebtedness other than (A) the Obligations in accordance with this Agreement, (B) Hedge Obligations, (C) Permitted Intercompany Advances, (D) the Preferred Equity Documents, but only to the extent permitted by subclause (b)(iii) below, and (E) Indebtedness permitted under <u>clauses (c)</u>, (j) and (k) of the definition of Permitted Indebtedness.

(ii) the Governing Documents of any Loan Party or any of its Subsidiaries if the effect thereof, either individually or in the aggregate, could reasonably be expected to be materially adverse to the interests of the Lenders, or

(iii) any Material Contract, the Preferred Equity Documents, the Permitted Subordinated Notes, and the Permitted Convertible Notes, each except to the extent that such amendment, modification, or change could not, individually or in the aggregate, reasonably be expected to be materially adverse to the interests of the Lenders.

6.7 <u>Restricted Payments</u>. Each Loan Party will not, and will not permit any of its Subsidiaries to, make any Restricted Payment; provided, that so long as it is permitted by law,

(a) Administrative Borrower may declare and pay dividends with respect to its Equity Interests payable solely in additional Equity Interests permitted hereunder,

(b) so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, Administrative Borrower may make distributions to former employees, officers, or directors of Administrative Borrower (or any spouses, ex-spouses, or estates of any of the foregoing) on account of redemptions of Equity Interests of Administrative Borrower held by such Persons; provided, that

-90-

the aggregate amount of such redemptions made by Administrative Borrower during the term of this Agreement <u>plus</u> the amount of Indebtedness outstanding under clause (1) of the definition of Permitted Indebtedness, does not exceed \$2,000,000 in the aggregate,

(c) so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, Administrative Borrower may make distributions to former employees, officers, or directors of Administrative Borrower (or any spouses, ex-spouses, or estates of any of the foregoing), solely in the form of forgiveness of Indebtedness of such Persons owing to Administrative Borrower on account of repurchases of the Equity Interests of Administrative Borrower held by such Persons; <u>provided</u>, that such Indebtedness was incurred by such Persons solely to acquire Equity Interests of Administrative Borrower,

(d) so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, Administrative Borrower's Subsidiaries may make distributions to Administrative Borrower (i) in an amount sufficient to pay franchise taxes and other fees required to maintain the legal existence of the Loan Parties and their Subsidiaries to the extent actually used by Administrative Borrower to pay such taxes, costs and expenses, and (ii) in an amount sufficient to pay out-of-pocket legal, accounting and filing costs and other expenses in the nature of overhead in the ordinary course of business of the Loan Parties and their Subsidiaries,

(e) any Subsidiary may declare and pay dividends or make other distributions with respect to its capital stock, partnership or membership interests or other similar Equity Interests, or make other Restricted Payments in respect of its Equity Interests, in each case ratably to the holders of such Equity Interests,

(f) Administrative Borrower may make cash payments in lieu of the issuance of fractional shares representing insignificant interests in Administrative Borrower in connection with the exercise of warrants, options or other securities convertible into or exchangeable for capital stock in Administrative Borrower,

(g) Administrative Borrower may make Restricted Payments, not exceeding \$2,000,000 in the aggregate for any fiscal year, pursuant to and in accordance with stock option plans or other benefit plans or agreements for directors, officers, or employees of Administrative Borrower and the Subsidiaries,

(h) Borrowers and their Subsidiaries may make dividends and distributions to Administrative Borrower to make payments permitted to be made pursuant to Section 6.10(f), or

(f) other Restricted Payments so long as the Payment Conditions are satisfied.

6.8 Accounting Methods. Each Loan Party will not, and will not permit any of its Subsidiaries to, modify or change its fiscal year or its method of accounting (other than as may be required to conform to GAAP).

6.9 **Investments**. Each Loan Party will not, and will not permit any of its Subsidiaries to, directly or indirectly, make or acquire any Investment or incur any liabilities (including contingent obligations) for or in connection with any Investment except for Permitted Investments.

6.10 <u>**Transactions with Affiliates**</u>. Each Loan Party will not, and will not permit any of its Subsidiaries to, directly or indirectly, enter into or permit to exist any transaction with any Affiliate of any Loan Party or any of its Subsidiaries except for:

-91-

(a) transactions (other than the payment of management, consulting, monitoring, or advisory fees) between such Loan Party or its Subsidiaries, on the one hand, and any Affiliate of such Loan Party or its Subsidiaries, on the other hand, so long as such transactions (i) are fully disclosed to Agent prior to the consummation thereof, if they involve one or more payments by such Loan Party or its Subsidiaries in excess of \$1,000,000 for any single transaction or series of related transactions, and (ii) are no less favorable, taken as a whole, to such Loan Party or its Subsidiaries, as applicable, than would be obtained in an arm's length transaction with a non-Affiliate,

(b) any indemnity provided for the benefit of directors (or comparable managers) of a Loan Party or one of its Subsidiaries so long as it has been approved by such Loan Party's or such Subsidiary's board of directors (or comparable governing body) in accordance with applicable law,

(c) the payment of reasonable compensation, severance, or employee benefit arrangements to employees, officers, and outside directors of a Loan Party or one of its Subsidiaries in the ordinary course of business and consistent with industry practice so long as it has been approved by such Loan Party's or such Subsidiary's board of directors (or comparable governing body) in accordance with applicable law,

(d) (i) transactions solely among the Loan Parties, and (ii) transactions solely among Subsidiaries of Loan Parties that are not Loan

Parties,

(e) transactions permitted by Section 6.3, Section 6.7, or Section 6.9, and

(f) agreements for the non-exclusive licensing of intellectual property, or distribution of products, in each case, among the Loan Parties and their Subsidiaries for the purpose of the counterparty thereof operating its business, and agreements for the assignment of intellectual property from any Loan Party or any of its Subsidiaries to any Loan Party.

6.11 **Use of Proceeds.** Each Loan Party will not, and will not permit any of its Subsidiaries to, use the proceeds of any Loan made hereunder for any purpose other than (a) on the Closing Date, (i) to repay, in full, the outstanding principal, accrued interest, and accrued fees and expenses owing under or in connection with the Existing Credit Facility, and (ii) to pay the fees, costs, and expenses incurred in connection with this Agreement, the other Loan Documents, and the transactions contemplated hereby and thereby, in each case, as set forth in the Flow of Funds Agreement, and (b) thereafter, consistent with the terms and conditions hereof, for their lawful and permitted purposes; provided that (x) no part of the proceeds of the Loans will be used to purchase or carry any such Margin Stock or to extend credit to others for the purpose of purchasing or carrying any such Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors, (y) no part of the proceeds of any Loan or Letter of Credit will be used, directly or indirectly, to make any payments to a Sanctioned Person, to fund any operations, activities or business of a Sanctioned Entity or a Sanctioned Person, to fund any operations, activities or business of a Sanctioned Entity or a Sanctioned Person, and (z) that no part of the proceeds of any Loan or Letter of Credit will be used, directly or indirectly, in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Sanctions, Anti-Corruption Laws.

6.12 <u>Limitation on Issuance of Equity Interests</u>. Except for the issuance or sale of Qualified Equity Interests by Administrative Borrower, each Loan Party will not, and will not permit any of its Subsidiaries to, issue or sell any of its Equity Interests.



6.13 **Inventory with Bailees**. Each Borrower will not, and will not permit any of its Subsidiaries to, store its Inventory at any time with a bailee, warehouseman, or similar party except as set forth on Schedule 4.25 (as such Schedule may be amended in accordance with <u>Section 5.14</u>).

6.14 **Intellectual Property.** No Subsidiary of a Loan Party which is not itself a Loan Party shall own intellectual property material to the conduct of the business of any Loan Party.

6.15 <u>Agency Relationships</u>. Each Loan Party will not, and will not permit any of its Subsidiaries to, enter into any, or renew any existing or expired, contract, agreement or order with any Account Debtor or other third party to a Material Contract, that contains any term that (x) establishes that a Loan Party is (i) holding any payment made by such Account Debtor or third party in trust, or similar language, by such Loan Party, for payment to any other Person or (ii) collecting any payment from such Account Debtor or third party as agent for such Account Debtor for payment to any other Person or (y) creates or establishes an agency or agency-like relationship between such Account Debtor or third party and a Loan Party or a Loan Party and another Person.

7. FINANCIAL COVENANTS.

Each Borrower covenants and agrees that, until the termination of all of the Commitments and the payment in full of the Obligations, Borrowers will maintain a Fixed Charge Coverage Ratio, calculated for each 12 month period ending on the first day of any Covenant Testing Period and the last day of each fiscal month occurring until the end of any Covenant Testing Period (including the last day thereof), in each case of at least 1.00 to 1.00.

8. EVENTS OF DEFAULT.

Any one or more of the following events shall constitute an event of default (each, an " Event of Default") under this Agreement:

8.1 **Payments**. If Borrowers fail to pay when due and payable, or when declared due and payable, (a) all or any portion of the Obligations consisting of interest, fees, or charges due the Lender Group, reimbursement of Lender Group Expenses, or other amounts (other than any portion thereof constituting principal) constituting Obligations (including any portion thereof that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), and such failure continues for a period of three Business Days, (b) all or any portion of the principal of the Loans, or (c) any amount payable to Issuing Bank in reimbursement of any drawing under a Letter of Credit;

8.2 **Covenants**. If any Loan Party or any of its Subsidiaries:

(a) fails to perform or observe any covenant or other agreement contained in any of (i) Sections 3.6, 5.1, 5.2, 5.3 (solely if any Borrower is not in good standing in its jurisdiction of organization), 5.6, 5.7 (solely if any Borrower refuses to allow Agent or its representatives or agents to visit any Borrower's properties, inspect its assets or books or records, examine and make copies of its books and records, or discuss Borrower's affairs, finances, and accounts with officers and employees of any Borrower), 5.10, 5.11, 5.13, or 5.14 of this Agreement, (ii) Section 6 of this Agreement, or (iv) Section 7 of the Guaranty and Security Agreement;

(b) fails to perform or observe any covenant or other agreement contained in any of Sections 5.3 (other than if any Borrower is not in good standing in its jurisdiction of organization), 5.4, 5.5, 5.8, and 5.12 of this Agreement and such failure continues for a period of ten days after the earlier of (i) the

date on which such failure shall first become known to any officer of any Borrower, or (ii) the date on which written notice thereof is given to Borrowers by Agent; or

(c) fails to perform or observe any covenant or other agreement contained in this Agreement, or in any of the other Loan Documents, in each case, other than any such covenant or agreement that is the subject of another provision of this <u>Section 8</u> (in which event such other provision of this <u>Section 8</u> shall govern), and such failure continues for a period of thirty days after the earlier of (i) the date on which such failure shall first become known to any officer of any Borrower, or (ii) the date on which written notice thereof is given to Borrowers by Agent;

8.3 **Judgments**. If one or more judgments, orders, or awards for the payment of money involving an aggregate amount of \$2,500,000, or more (except to the extent fully covered (other than to the extent of customary deductibles) by insurance pursuant to which the insurer has not denied coverage) is entered or filed against a Loan Party or any of its Subsidiaries, or with respect to any of their respective assets, and either (a) there is a period of thirty consecutive days at any time after the entry of any such judgment, order, or award during which (i) the same is not discharged, satisfied, vacated, or bonded pending appeal, or (ii) a stay of enforcement thereof is not in effect, or (b) enforcement proceedings are commenced upon such judgment, order, or award;

8.4 Voluntary Bankruptcy, etc. If an Insolvency Proceeding is commenced by a Loan Party or any of its Subsidiaries;

8.5 <u>Involuntary Bankruptcy, etc</u>. If an Insolvency Proceeding is commenced against a Loan Party or any of its Subsidiaries and any of the following events occur: (a) such Loan Party or such Subsidiary consents to the institution of such Insolvency Proceeding against it, (b) the petition commencing the Insolvency Proceeding is not timely controverted, (c) the petition commencing the Insolvency Proceeding is not dismissed within sixty calendar days of the date of the filing thereof, (d) an interim trustee is appointed to take possession of all or any substantial portion of the properties or assets of, or to operate all or any substantial portion of the business of, such Loan Party or its Subsidiary, or (e) an order for relief shall have been issued or entered therein;

8.6 **Default Under Other Agreements.** If there is (a) a default in one or more agreements to which a Loan Party or any of its Subsidiaries is a party with one or more third Persons relative to a Loan Party's or any of its Subsidiaries' Indebtedness involving an aggregate amount of \$2,500,000 or more, and such default (i) occurs at the final maturity of the obligations thereunder, or (ii) results in a right by such third Person, irrespective of whether exercised, to accelerate the maturity of such Loan Party's or its Subsidiary's obligations thereunder, or (b) a default under the 2020 Unsecured Convertible Notes, the Preferred Equity Documents, any Permitted Subordinated Notes;

8.7 **Representations, etc.** If any warranty, representation, certificate, statement, or Record made herein or in any other Loan Document or delivered in writing to Agent or any Lender in connection with this Agreement or any other Loan Document proves to be untrue in any material respect (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of the date of issuance or making or deemed making thereof;

8.8 **<u>Guaranty</u>**. If the obligation of any Guarantor under the guaranty contained in the Guaranty and Security Agreement is limited or terminated by operation of law or by such Guarantor (other than in accordance with the terms of this Agreement) or if any Guarantor repudiates or revokes or purports to repudiate or revoke any such guaranty;

8.9 <u>Security Documents</u>. If the Guaranty and Security Agreement or any other Loan Document that purports to create a Lien, shall, for any reason, fail or cease to create a valid and perfected and, (except to the extent of Permitted Liens which are non-consensual Permitted Liens, permitted purchase money Liens or the interests of lessors under Capital Leases) first priority Lien on the Collateral covered thereby, except (a) as a result

-94-

of a disposition of the applicable Collateral in a transaction permitted under this Agreement , or (b) with respect to Collateral the aggregate value of which, for all such Collateral, does not exceed at any time, \$1,000,000;

8.10 **Loan Documents**. The validity or enforceability of any Loan Document shall at any time for any reason (other than solely as the result of an action or failure to act on the part of Agent) be declared to be null and void, or a proceeding shall be commenced by a Loan Party or its Subsidiaries, or by any Governmental Authority having jurisdiction over a Loan Party or its Subsidiaries, seeking to establish the invalidity or unenforceability thereof, or a Loan Party or its Subsidiaries shall deny that such Loan Party or its Subsidiaries has any liability or obligation purported to be created under any Loan Document;

8.11 **Change of Control**. A Change of Control shall occur, whether directly or indirectly; or

8.12 Material Cessation of Programming.

(a) The transmission to end-users of the music programming of the Borrowers or the availability of such music programming to endusers shall be interrupted at any time for more than 48 hours, whether or not consecutive, during any period of 14 consecutive days, and such interruption could reasonably be expected to have a Material Adverse Effect.

(b) With respect to a Station, if any, the over the air operations of any such Station shall be interrupted at any time for more than 48 hours, whether or not consecutive, during any period of 14 consecutive days, and such interruption could reasonably be expected to have a Material Adverse Effect.

8.13 <u>Termination of Material Licenses</u>. Any (a) material License or (b) material Music Performance Right shall be revoked or cancelled or expired by its terms and not be renewed, or shall be modified, in each case in a manner which could, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

9. RIGHTS AND REMEDIES.

9.1 <u>**Rights and Remedies**</u>. Upon the occurrence and during the continuation of an Event of Default, Agent may, and, at the instruction of the Required Lenders, shall, in addition to any other rights or remedies provided for hereunder or under any other Loan Document or by applicable law, do any one or more of the following:

(a) by written notice to Borrowers, (i) declare the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents to be immediately due and payable, whereupon the same shall become and be immediately due and payable and Borrowers shall be obligated to repay all of such Obligations in full, without presentment, demand, protest, or further notice or other requirements of any kind, all of which are hereby expressly waived by each Borrower, and (ii) direct Borrowers to provide (and Borrowers agree that upon receipt of such notice Borrowers will provide) Letter of Credit Collateralization to Agent to be held as security for Borrowers' reimbursement obligations for drawings that may subsequently occur under issued and outstanding Letters of Credit;

(b) by written notice to Borrowers, declare the Commitments terminated, whereupon the Commitments shall immediately be terminated together with (i) any obligation of any Revolving Lender to make Revolving Loans, (ii) the obligation of the Swing Lender to make Swing Loans, and (iii) the obligation of Issuing Bank to issue Letters of Credit; and

-95-

(c) exercise all other rights and remedies available to Agent or the Lenders under the Loan Documents, under applicable law, or in

equity.

The foregoing to the contrary notwithstanding, upon the occurrence of any Event of Default described in <u>Section 8.4</u> or <u>Section 8.5</u>, in addition to the remedies set forth above, without any notice to Borrowers or any other Person or any act by the Lender Group, the Commitments shall automatically terminate and the Obligations (other than the Bank Product Obligations), inclusive of the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents, shall automatically become and be immediately due and payable and Borrowers shall automatically be obligated to repay all of such Obligations in full (including Borrowers being obligated to provide (and Borrowers agree that they will provide) (1) Letter of Credit Collateralization to Agent to be held as security for Borrowers' reimbursement obligations in respect of drawings that may subsequently occur under issued and outstanding Letters of Credit and (2) Bank Product Collateralization to be held as security for Borrowers, or notice or other requirements of any kind, all of which are expressly waived by Borrowers.

9.2 **Remedies Cumulative**. The rights and remedies of the Lender Group under this Agreement, the other Loan Documents, and all other agreements shall be cumulative. The Lender Group shall have all other rights and remedies not inconsistent herewith as provided under the Code, by law, or in equity. No exercise by the Lender Group of one right or remedy shall be deemed an election, and no waiver by the Lender Group of any Default or Event of Default shall be deemed a continuing waiver. No delay by the Lender Group shall constitute a waiver, election, or acquiescence by it.

10. WAIVERS; INDEMNIFICATION.

10.1 **Demand: Protest; etc.** Each Borrower waives demand, protest, notice of protest, notice of default or dishonor, notice of payment and nonpayment, nonpayment at maturity, release, compromise, settlement, extension, or renewal of documents, instruments, chattel paper, and guarantees at any time held by the Lender Group on which any Borrower may in any way be liable.

10.2 <u>The Lender Group's Liability for Collateral</u>. Each Borrower hereby agrees that: (a) so long as Agent complies with its obligations, if any, under the Code, the Lender Group shall not in any way or manner be liable or responsible for: (i) the safekeeping of the Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof, or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency, or other Person, and (b) all risk of loss, damage, or destruction of the Collateral shall be borne by the Loan Parties.

10.3 **Indemnification**. Each Borrower shall pay, indemnify, defend, and hold the Agent-Related Persons, the Lender-Related Persons, the Issuing Bank, and each Participant (each, an "<u>Indemnified Person</u>") harmless (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all reasonable fees and disbursements of attorneys, experts, or consultants and all other costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), at any time asserted against, imposed upon, or incurred by any of them (a) in connection with or as a result of or related to the execution and delivery (<u>provided</u>, that Borrowers shall not be liable for costs and expenses (including attorneys' fees) of any Lender (other than Wells Fargo) incurred in advising, structuring, drafting, reviewing, administering or syndicating the Loan Documents), enforcement, performance, or administration (including any restructuring or workout with respect hereto) of this Agreement, any of the other Loan Documents, or the transactions contemplated hereby or thereby or the monitoring of the Loan Parties and their Subsidiaries' compliance with the terms of the Loan Documents (<u>provided</u>, that the

-96-

indemnification in this clause (a) shall not extend to (i) disputes solely between or among the Lenders that do not involve any acts or omissions of any Loan Party, or (ii) disputes solely between or among the Lenders and their respective Affiliates that do not involve any acts or omissions of any Loan Party; it being understood and agreed that the indemnification in this clause (a) shall extend to Agent (but not the Lenders unless the dispute involves an act or omission of a Loan Party) relative to disputes between or among Agent on the one hand, and one or more Lenders, or one or more of their Affiliates, on the other hand, or (iii) any claims for Taxes, which shall be governed by Section 16, except for Taxes which arise from a non-Tax claim), (b) with respect to any actual or prospective investigation, litigation, or proceeding related to this Agreement, any other Loan Document, the making of any Loans or issuance of any Letters of Credit hereunder, or the use of the proceeds of the Loans or the Letters of Credit provided hereunder (irrespective of whether any Indemnified Person is a party thereto), or any act, omission, event, or circumstance in any manner related thereto, and (c) in connection with or arising out of any presence or release of Hazardous Materials at, on, under, to or from any assets or properties owned, leased or operated by any Loan Party or any of its Subsidiaries or any Environmental Actions, Environmental Liabilities or Remedial Actions related in any way to any such assets or properties of any Loan Party or any of its Subsidiaries (each and all of the foregoing, the "Indemnified Liabilities"). The foregoing to the contrary notwithstanding, no Borrower shall have any obligation to any Indemnified Person under this Section 10.3 with respect to any Indemnified Liability that a court of competent jurisdiction finally determines to have resulted from the gross negligence or willful misconduct of such Indemnified Person or its officers, directors, employees, attorneys, or agents. This provision shall survive the termination of this Agreement and the repayment in full of the Obligations. If any Indemnified Person makes any payment to any other Indemnified Person with respect to an Indemnified Liability as to which Borrowers were required to indemnify the Indemnified Person receiving such payment, the Indemnified Person making such payment is entitled to be indemnified and reimbursed by Borrowers with respect thereto. WITHOUT LIMITATION, THE FOREGOING INDEMNITY SHALL APPLY TO EACH INDEMNIFIED PERSON WITH RESPECT TO INDEMNIFIED LIABILITIES WHICH IN WHOLE OR IN PART ARE CAUSED BY OR ARISE OUT OF ANY NEGLIGENT ACT OR OMISSION OF SUCH INDEMNIFIED PERSON OR OF ANY OTHER PERSON.

11. NOTICES.

Unless otherwise provided in this Agreement, all notices or demands relating to this Agreement or any other Loan Document shall be in writing and (except for financial statements and other informational documents which may be sent by first-class mail, postage prepaid) shall be personally delivered or sent by registered or certified mail (postage prepaid, return receipt requested), overnight courier, electronic mail (at such email addresses as a party may designate in accordance herewith), or telefacsimile. In the case of notices or demands to any Loan Party or Agent, as the case may be, they shall be sent to the respective address set forth below:

-97-

If to any Loan Party:	PANDORA MEDIA, INC. 2101 Webster Street, Suite 1650
	Oakland, CA 94612
	Attn: Chief Financial Officer
with copies to:	PANDORA MEDIA, INC.
	2101 Webster Street, Suite 1650
	Oakland, CA 94612
	Attn: General Counsel Email: legal@pandora.com
	Email: legal@paildota.com
	and
	SIDLEY AUSTIN LLP
	1001 Page Mill, Road Building 1
	Palo Alto, CA 94304
	Attn: Pamela Martinson, Esq.
	Fax No.: (650) 565-7100
If to Agent:	WELLS FARGO BANK, NATIONAL ASSOCIATION
	2450 Colorado Avenue, Suite 3000 West
	Santa Monica, CA 90404
	Attn: Loan Portfolio Manager
	Fax No.: 310-453-7413
with copies to:	PAUL HASTINGS LLP
	200 Park Avenue
	New York, NY 10166
	Attn: Jennifer St. John Yount, Esq.
	Fax No.: (212) 230-7008

Any party hereto may change the address at which they are to receive notices hereunder, by notice in writing in the foregoing manner given to the other party. All notices or demands sent in accordance with this <u>Section 11</u>, shall be deemed received on the earlier of the date of actual receipt or three Business Days after the deposit thereof in the mail; <u>provided</u>, that (a) notices sent by overnight courier service shall be deemed to have been given when received, (b) notices by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next Business Day for the recipient) and (c) notices by electronic mail shall be deemed received upon the sender's receipt of an acknowledgment from the intended recipient (such as by the "return receipt requested" function, as available, return email or other written acknowledgment).

12. CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION.

(a) THE VALIDITY OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANOTHER LOAN

-98-

DOCUMENT IN RESPECT OF SUCH OTHER LOAN DOCUMENT), THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT HEREOF AND THEREOF, THE RIGHTS OF THE PARTIES HERETO AND THERETO WITH RESPECT TO ALL MATTERS ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO, AND ANY CLAIMS, CONTROVERSIES OR DISPUTES ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF CALIFORNIA.

(b) THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS SHALL BE TRIED AND LITIGATED ONLY IN THE STATE AND, TO THE EXTENT PERMITTED BY APPLICABLE LAW, FEDERAL COURTS LOCATED IN THE COUNTY OF LOS ANGELES, STATE OF CALIFORNIA; <u>PROVIDED</u>, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE AGENT ELECTS TO BRING SUCH ACTION OR WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF <u>FORUM NON CONVENIENS</u> OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS <u>SECTION 12(b)</u>.

(c) TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP HEREBY WAIVE THEIR RESPECTIVE RIGHTS, IF ANY, TO A JURY TRIAL OF ANY CLAIM, CONTROVERSY, DISPUTE OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS (EACH A "CLAIM"). EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP REPRESENT THAT EACH HAS REVIEWED THIS WAIVER AND EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, A COPY OF THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

(d) EACH BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE COUNTY OF LOS ANGELES AND THE STATE OF CALIFORNIA, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN DOCUMENTS, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT AGENT MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST ANY LOAN PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.

(e) NO CLAIM MAY BE MADE BY ANY LOAN PARTY AGAINST THE AGENT, THE SWING LENDER, ANY OTHER LENDER, ISSUING BANK, OR ANY AFFILIATE, DIRECTOR, OFFICER, EMPLOYEE, COUNSEL, REPRESENTATIVE, AGENT, OR

-99-

ATTORNEY-IN-FACT OF ANY OF THEM FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES OR LOSSES IN RESPECT OF ANY CLAIM FOR BREACH OF CONTRACT OR ANY OTHER THEORY OF LIABILITY ARISING OUT OF OR RELATED TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY ACT, OMISSION, OR EVENT OCCURRING IN CONNECTION THEREWITH, AND EACH LOAN PARTY HEREBY WAIVES, RELEASES, AND AGREES NOT TO SUE UPON ANY CLAIM FOR SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST IN ITS FAVOR.

(f) IN THE EVENT ANY LEGAL PROCEEDING IS FILED IN A COURT OF THE STATE OF CALIFORNIA (THE "COURT") BY OR AGAINST ANY PARTY HERETO IN CONNECTION WITH ANY CLAIM AND THE WAIVER SET FORTH IN CLAUSE (C) ABOVE IS NOT ENFORCEABLE IN SUCH PROCEEDING, THE PARTIES HERETO AGREE AS FOLLOWS:

(i) WITH THE EXCEPTION OF THE MATTERS SPECIFIED IN SUBCLAUSE (ii) BELOW, ANY CLAIM SHALL BE DETERMINED BY A GENERAL REFERENCE PROCEEDING IN ACCORDANCE WITH THE PROVISIONS OF CALIFORNIA CODE OF CIVIL PROCEDURE SECTIONS 638 THROUGH 645.1. THE PARTIES INTEND THIS GENERAL REFERENCE AGREEMENT TO BE SPECIFICALLY ENFORCEABLE. VENUE FOR THE REFERENCE PROCEEDING SHALL BE IN THE COUNTY OF LOS ANGELES, CALIFORNIA.

(ii) THE FOLLOWING MATTERS SHALL NOT BE SUBJECT TO A GENERAL REFERENCE PROCEEDING: (A) NON-JUDICIAL FORECLOSURE OF ANY SECURITY INTERESTS IN REAL OR PERSONAL PROPERTY, (B) EXERCISE OF SELF-HELP REMEDIES (INCLUDING SET-OFF OR RECOUPMENT), (C) APPOINTMENT OF A RECEIVER, AND (D) TEMPORARY, PROVISIONAL, OR ANCILLARY REMEDIES (INCLUDING WRITS OF ATTACHMENT, WRITS OF POSSESSION, TEMPORARY RESTRAINING ORDERS, OR PRELIMINARY INJUNCTIONS). THIS AGREEMENT DOES NOT LIMIT THE RIGHT OF ANY PARTY TO EXERCISE OR OPPOSE ANY OF THE RIGHTS AND REMEDIES DESCRIBED IN CLAUSES (A) - (D) AND ANY SUCH EXERCISE OR OPPOSITION DOES NOT WAIVE THE RIGHT OF ANY PARTY TO PARTICIPATE IN A REFERENCE PROCEEDING PURSUANT TO THIS AGREEMENT WITH RESPECT TO ANY OTHER MATTER.

(iii) UPON THE WRITTEN REQUEST OF ANY PARTY, THE PARTIES SHALL SELECT A SINGLE REFEREE, WHO SHALL BE A RETIRED JUDGE OR JUSTICE. IF THE PARTIES DO NOT AGREE UPON A REFEREE WITHIN TEN DAYS OF SUCH WRITTEN REQUEST, THEN, ANY PARTY SHALL HAVE THE RIGHT TO REQUEST THE COURT TO APPOINT A REFEREE PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE SECTION 640(B). THE REFEREE SHALL BE APPOINTED TO SIT WITH ALL OF THE POWERS PROVIDED BY LAW. PENDING APPOINTMENT OF THE REFEREE, THE COURT SHALL HAVE THE POWER TO ISSUE TEMPORARY OR PROVISIONAL REMEDIES.

(iv) EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, THE REFEREE SHALL DETERMINE THE MANNER IN WHICH THE REFERENCE PROCEEDING IS CONDUCTED INCLUDING THE TIME AND PLACE OF HEARINGS, THE ORDER OF PRESENTATION OF EVIDENCE, AND ALL OTHER QUESTIONS THAT ARISE WITH RESPECT TO THE COURSE OF THE REFERENCE PROCEEDING. ALL PROCEEDINGS AND HEARINGS CONDUCTED BEFORE THE REFEREE, EXCEPT FOR TRIAL, SHALL BE

-100-

CONDUCTED WITHOUT A COURT REPORTER, EXCEPT WHEN ANY PARTY SO REQUESTS A COURT REPORTER AND A TRANSCRIPT IS ORDERED, A COURT REPORTER SHALL BE USED AND THE REFEREE SHALL BE PROVIDED A COURTESY COPY OF THE TRANSCRIPT. THE PARTY MAKING SUCH REQUEST SHALL HAVE THE OBLIGATION TO ARRANGE FOR AND PAY THE COSTS OF THE COURT REPORTER; <u>PROVIDED</u>, THAT SUCH COSTS, ALONG WITH THE REFEREE'S FEES, SHALL ULTIMATELY BE BORNE BY THE PARTY WHO DOES NOT PREVAIL, AS DETERMINED BY THE REFEREE.

(v) THE REFEREE MAY REQUIRE ONE OR MORE PREHEARING CONFERENCES. THE PARTIES HERETO SHALL BE ENTITLED TO DISCOVERY, AND THE REFEREE SHALL OVERSEE DISCOVERY IN ACCORDANCE WITH THE RULES OF DISCOVERY, AND SHALL ENFORCE ALL DISCOVERY ORDERS IN THE SAME MANNER AS ANY TRIAL COURT JUDGE IN PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA.

(vi) THE REFEREE SHALL APPLY THE RULES OF EVIDENCE APPLICABLE TO PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA AND SHALL DETERMINE ALL ISSUES IN ACCORDANCE WITH CALIFORNIA SUBSTANTIVE AND PROCEDURAL LAW. THE REFEREE SHALL BE EMPOWERED TO ENTER EQUITABLE AS WELL AS LEGAL RELIEF AND RULE ON ANY MOTION WHICH WOULD BE AUTHORIZED IN A TRIAL, INCLUDING MOTIONS FOR DEFAULT JUDGMENT OR SUMMARY JUDGMENT. THE REFEREE SHALL REPORT HIS OR HER DECISION, WHICH REPORT SHALL ALSO INCLUDE FINDINGS OF FACT AND CONCLUSIONS OF LAW. THE REFEREE SHALL ISSUE A DECISION AND PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE, SECTION 644, THE REFEREE'S DECISION SHALL BE ENTERED BY THE COURT AS A JUDGMENT IN THE SAME MANNER AS IF THE ACTION HAD BEEN TRIED BY THE COURT. THE FINAL JUDGMENT OR ORDER FROM ANY APPEALABLE DECISION OR ORDER ENTERED BY THE REFEREE SHALL BE FULLY APPEALABLE AS IF IT HAS BEEN ENTERED BY THE COURT.

(vii) THE PARTIES RECOGNIZE AND AGREE THAT ALL CLAIMS RESOLVED IN A GENERAL REFERENCE PROCEEDING PURSUANT HERETO WILL BE DECIDED BY A REFEREE AND NOT BY A JURY. AFTER CONSULTING (OR HAVING HAD THE OPPORTUNITY TO CONSULT) WITH COUNSEL OF THEIR OWN CHOICE, EACH PARTY HERETO KNOWINGLY AND VOLUNTARILY AND FOR THEIR MUTUAL BENEFIT AGREES THAT THIS REFERENCE PROVISION SHALL APPLY TO ANY DISPUTE BETWEEN THEM THAT ARISES OUT OF OR IS RELATED TO THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS.

13. ASSIGNMENTS AND PARTICIPATIONS; SUCCESSORS.

13.1 Assignments and Participations.

(a) (i) Subject to the conditions set forth in clause (a)(ii) below, any Lender may assign and delegate all or any portion of its rights and duties under the Loan Documents (including the Obligations owed to it and its Commitments) to one or more assignees so long as such prospective assignee is an Eligible Transferee (each, an "Assignee"), with the prior written consent (such consent not be unreasonably withheld or delayed) of:

(A) Borrowers; <u>provided</u>, that no consent of Borrowers shall be required (1) if a Default or Event of Default has occurred and is continuing, (2) in connection with the Primary Syndication of the Commitments and the Obligations by Wells Fargo (<u>provided</u>, that Wells Fargo shall consult with Borrowers

-101-

in connection with such Primary Syndication (it being understood that in no event shall Wells Fargo be required to obtain Borrowers' consent with respect to any assignment made in connection with such Primary Syndication)), or (3) in connection with an assignment to a Person that is a Lender or an Affiliate (other than natural persons) of a Lender; provided further, that Borrowers shall be deemed to have consented to a proposed assignment unless they object thereto by written notice to Agent within five Business Days after having received notice thereof; and

(B) Agent, Swing Lender, and Issuing Bank.

- (ii) Assignments shall be subject to the following additional conditions:
- (A) no assignment may be made (i) so long as no Event of Default has occurred and is continuing, to a Disqualified Institution, or (ii) to a natural person,
- (B) no assignment may be made to a Loan Party or an Affiliate of a Loan Party,

(C) the amount of the Commitments and the other rights and obligations of the assigning Lender hereunder and under the other Loan Documents subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to Agent) shall be in a minimum amount (unless waived by Agent) of \$5,000,000 (except such minimum amount shall not apply to (I) an assignment or delegation by any Lender to any other Lender, an Affiliate of any Lender, or a Related Fund of such Lender, or (II) a group of new Lenders, each of which is an Affiliate of each other or a Related Fund of such new Lender to the extent that the aggregate amount to be assigned to all such new Lenders is at least \$5,000,000),

(D) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement,

(E) the parties to each assignment shall execute and deliver to Agent an Assignment and Acceptance; <u>provided</u>, that Borrowers and Agent may continue to deal solely and directly with the assigning Lender in connection with the interest so assigned to an Assignee until written notice of such assignment, together with payment instructions, addresses, and related information with respect to the Assignee, have been given to Borrowers and Agent by such Lender and the Assignee,

(F) unless waived by Agent, the assigning Lender or Assignee has paid to Agent, for Agent's separate account, a processing fee in the amount of \$3,500, and

(G) the assignee, if it is not a Lender, shall deliver to Agent an Administrative Questionnaire in a form approved by Agent (the "<u>Administrative</u> <u>Questionnaire</u>").

(b) From and after the date that Agent receives the executed Assignment and Acceptance and, if applicable, payment of the required processing fee, (i) the Assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, shall be a "Lender" and shall have the rights and obligations of a Lender under the Loan Documents, and (ii) the assignment and Acceptance, relinquish its rights (except with respect to <u>Section 10.3</u>) and be released from any future obligations under this Agreement (and in the case of an Assignment and Acceptance covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement and the other Loan Documents, such Lender shall cease to be a party hereto and thereto); <u>provided</u>, that nothing contained herein shall release any assigning Lender from obligations that survive the termination of this Agreement, including such assigning Lender's obligations under <u>Section 15</u> and <u>Section 17.9(a)</u>.

-102-

(c) By executing and delivering an Assignment and Acceptance, the assigning Lender thereunder and the Assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than as provided in such Assignment and Acceptance, such assigning Lender makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement or any other Loan Document furnished pursuant hereto, (ii) such assigning Lender makes no representation or warranty and assumes no responsibility with respect to the financial condition of any Loan Party or the performance or observance by any Loan Party of any of its obligations under this Agreement or any other Loan Document furnished pursuant hereto, (iii) such Assignee confirms that it has received a copy of this Agreement, together with such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance, (iv) such Assignee will, independently and without reliance upon Agent, such assigning Lender or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement, (v) such Assignee appoints and authorizes Agent to take such actions and to exercise such powers under this Agreement and the other Loan Documents as are delegated to Agent, by the terms hereof and thereof, together with such powers as are reasonably incidental thereto, and (vi) such Assignee agrees that it will perform all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

(d) Immediately upon Agent's receipt of the required processing fee, if applicable, and delivery of notice to the assigning Lender pursuant to <u>Section 13.1(b)</u>, this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to reflect the addition of the Assignee and the resulting adjustment of the Commitments arising therefrom. The Commitment allocated to each Assignee shall reduce such Commitments of the assigning Lender *pro tanto*.

(e) Any Lender may at any time sell to one or more commercial banks, financial institutions, or other Persons (a "Participant") participating interests in all or any portion of its Obligations, its Commitment, and the other rights and interests of that Lender (the "Originating Lender") hereunder and under the other Loan Documents; provided, that (i) the Originating Lender shall remain a "Lender" for all purposes of this Agreement and the other Loan Documents and the Participant receiving the participating interest in the Obligations, the Commitments, and the other rights and interests of the Originating Lender hereunder shall not constitute a "Lender" hereunder or under the other Loan Documents and the Originating Lender's obligations under this Agreement shall remain unchanged, (ii) the Originating Lender shall remain solely responsible for the performance of such obligations, (iii) Borrowers, Agent, and the Lenders shall continue to deal solely and directly with the Originating Lender in connection with the Originating Lender's rights and obligations under this Agreement and the other Loan Documents, (iv) no Lender shall transfer or grant any participating interest under which the Participant has the right to approve any amendment to, or any consent or waiver with respect to, this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or of any other Loan Document would (A) extend the final maturity date of the Obligations hereunder in which such Participant is participating, (B) reduce the interest rate applicable to the Obligations hereunder in which such Participant is participating, (C) release all or substantially all of the Collateral or guaranties (except to the extent expressly provided herein or in any of the Loan Documents) supporting the Obligations hereunder in which such Participant is participating, (D) postpone the payment of, or reduce the amount of, the interest or fees payable to such Participant through such Lender (other than a waiver of default interest), or (E) decreases the amount or postpones the due dates of scheduled principal repayments or prepayments or premiums payable to such Participant through such Lender, (v) no participation shall be sold to a natural person, (vi) no participation shall be sold to a Loan Party or an Affiliate of a Loan Party, and (vii) all amounts payable by Borrowers



hereunder shall be determined as if such Lender had not sold such participation, except that, if amounts outstanding under this Agreement are due and unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall be deemed to have the right of set off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement. No Participant shall have any rights under this Agreement or the other Loan Documents other than those rights of the Originating Lender with whom such Participant participates or any direct rights as to the other Lenders, Agent, Borrowers, the Collateral, or otherwise in respect of the Obligations. No Participant shall have the right to participate directly in the making of decisions by the Lenders among themselves.

(f) In connection with any such assignment or participation or proposed assignment or participation or any grant of a security interest in, or pledge of, its rights under and interest in this Agreement, a Lender may, subject to the provisions of <u>Section 17.9</u>, disclose all documents and information which it now or hereafter may have relating to any Loan Party and its Subsidiaries and their respective businesses.

(g) Any other provision in this Agreement notwithstanding, any Lender may at any time create a security interest in, or pledge, all or any portion of its rights under and interest in this Agreement to secure obligations of such Lender, including any pledge in favor of any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Bank or U.S. Treasury Regulation 31 CFR §203.24, and such Federal Reserve Bank may enforce such pledge or security interest in any manner permitted under applicable law; provided, that no such pledge shall release such Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto.

(h) Agent (solely for this purpose as a non-fiduciary agent on behalf of Borrowers) shall maintain, or cause to be maintained, at one of its offices a register (the "<u>Register</u>") on which it enters the name and address of each Lender as the registered owner of the Revolving Loans (and the principal amount thereof and stated interest thereon) held by such Lender pursuant to the terms hereof (each, a "<u>Registered Loan</u>"). A Registered Loan (and the registered note, if any, evidencing the same) may be assigned or sold in whole or in part only by registration of such assignment or sale on the Register (and each registered note shall expressly so provide) and any assignment or sale of all or part of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by registration of such assignment or sale on the Register. The entries in the Register shall be conclusive absent manifest error and prior to the registration of assignment or sale of any Registered Loan (and the registered note, if any, evidencing the same), Borrowers, the Agent and the Lenders shall treat the Person in whose name such Registered Loan (and the registered note, if any, evidencing the same) is registered in the Register as the owner thereof for the purpose of receiving all payments thereon and for all other purposes, notwithstanding notice to the contrary.

(i) In the event that a Lender sells participations in the Registered Loan, such Lender, as a non-fiduciary agent on behalf of Borrowers, shall maintain or cause to be maintained a register on which it enters the name of all Participants in the Registered Loans held by it (and the principal amount (and stated interest thereon) of the portion of such Registered Loans that is subject to such participations) (the "<u>Participant Register</u>"). A Registered Loan (and the Registered Note, if any, evidencing the same) may be participated in whole or in part only by registration of such participation on the Participant Register (and each registered note shall expressly so provide). Any participation of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by the registration of such participation on the Participant Register. No Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that

-104-

such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Agent (in its capacity as Agent) shall have no responsibility for maintaining a Participant Register.

(j) Agent shall promptly make a copy of the Register (and each Lender that has sold a participation in accordance with Section 13.1(i) shall make a copy of its Participant Register to the extent it has one) available for review by Borrowers from time to time as Borrowers may reasonably request. Upon the request of any Lender, Agent will either post the identities of the Disqualified Institutions to its designated agency site or otherwise make available to such Lender the identity of the Disqualified Institutions.

13.2 **Successors.** This Agreement shall bind and inure to the benefit of the respective successors and assigns of each of the parties; provided, that no Borrower may assign this Agreement or any rights or duties hereunder without the Lenders' prior written consent and any prohibited assignment shall be absolutely void *ab initio*. No consent to assignment by the Lenders shall release any Borrower from its Obligations. A Lender may assign this Agreement and the other Loan Documents and its rights and duties hereunder and thereunder pursuant to Section 13.1 and, except as expressly required pursuant to Section 13.1, no consent or approval by any Borrower is required in connection with any such assignment.

14. AMENDMENTS; WAIVERS.

14.1 Amendments and Waivers.

(a) No amendment, waiver or other modification of any provision of this Agreement or any other Loan Document (other than the Fee Letter), and no consent with respect to any departure by any Borrower therefrom, shall be effective unless the same shall be in writing and signed by the Required Lenders (or by Agent at the written request of the Required Lenders) and the Loan Parties that are party thereto and then any such waiver or consent shall be effective, but only in the specific instance and for the specific purpose for which given; provided, that no such waiver, amendment, or consent shall, unless in writing and signed by all of the Lenders directly affected thereby and all of the Loan Parties that are party thereto, do any of the following:

(i) increase the amount of or extend the expiration date of any Commitment of any Lender or amend, modify, or eliminate the last sentence of $\underline{\text{Section } 2.4(c)(i)}$,

(ii) postpone or delay any date fixed by this Agreement or any other Loan Document for any payment of principal, interest, fees, or other amounts due hereunder or under any other Loan Document,

(iii) reduce the principal of, or the rate of interest on, any loan or other extension of credit hereunder, or reduce any fees or other amounts payable hereunder or under any other Loan Document (except (y) in connection with the waiver of applicability of Section 2.6(c) (which waiver shall be effective with the written consent of the Required Lenders), and (z) that any amendment or modification of defined terms used in the financial covenants in this Agreement shall not constitute a reduction in the rate of interest or a reduction of fees for purposes of this clause (iii)),

Lenders,

(iv) amend, modify, or eliminate this Section or any provision of this Agreement providing for consent or other action by all

-105-

- (v) amend, modify, or eliminate Section 3.1 or 3.2,
- (vi) amend, modify, or eliminate Section 15.11,

(vii) other than as permitted by Section 15.11, release or contractually subordinated Agent's Lien in and to any of the

Collateral,

(viii) amend, modify, or eliminate the definitions of "Required Lenders", Supermajority Lenders or "Pro Rata Share",

(ix) other than in connection with a merger, liquidation, dissolution or sale of such Person expressly permitted by the terms hereof or the other Loan Documents, release any Borrower or any Guarantor from any obligation for the payment of money or consent to the assignment or transfer by any Borrower or any Guarantor of any of its rights or duties under this Agreement or the other Loan Documents,

(x) amend, modify, or eliminate any of the provisions of Section 2.4(b)(i), (ii) or (iii), or

(xi) amend, modify, or eliminate any of the provisions of <u>Section 13.1</u> with respect to assignments to, or participations with, Persons who are Loan Parties or Affiliates of a Loan Party;

(b) No amendment, waiver, modification, or consent shall amend, modify, waive, or eliminate,

(i) the definition of, or any of the terms or provisions of, the Fee Letter, without the written consent of Agent and Borrowers (and shall not require the written consent of any of the Lenders),

(ii) any provision of <u>Section 15</u> pertaining to Agent, or any other rights or duties of Agent under this Agreement or the other Loan Documents, without the written consent of Agent, Borrowers, and the Required Lenders;

(c) No amendment, waiver, modification, elimination, or consent shall amend, without written consent of Agent, Borrowers and the Supermajority Lenders, modify, or eliminate the definition of Borrowing Base or any of the defined terms (including the definitions of Eligible Accounts, Eligible Advertising Accounts, and Eligible Subscription Accounts) that are used in such definition to the extent that any such change results in more credit being made available to Borrowers based upon the Borrowing Base, but not otherwise, or the definition of Maximum Revolver Amount, or change Section 2.1(c);

(d) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to Issuing Bank, or any other rights or duties of Issuing Bank under this Agreement or the other Loan Documents, without the written consent of Issuing Bank, Agent, Borrowers, and the Required Lenders;

(e) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to Swing Lender, or any other rights or duties of Swing Lender under this Agreement or the other Loan Documents, without the written consent of Swing Lender, Agent, Borrowers, and the Required Lenders; and

(f) Anything in this <u>Section 14.1</u> to the contrary notwithstanding, (i) any amendment, modification, elimination, waiver, consent, termination, or release of, or with respect to, any provision of

-106-

this Agreement or any other Loan Document that relates only to the relationship of the Lender Group among themselves, and that does not affect the rights or obligations of any Loan Party, shall not require consent by or the agreement of any Loan Party, and (ii) any amendment, waiver, modification, elimination, or consent of or with respect to any provision of this Agreement or any other Loan Document may be entered into without the consent of, or over the objection of, any Defaulting Lender.

14.2 Replacement of Certain Lenders.

(a) If (i) any action to be taken by the Lender Group or Agent hereunder requires the consent, authorization, or agreement of all Lenders or all Lenders affected thereby and if such action has received the consent, authorization, or agreement of the Required Lenders but not of all Lenders or all Lenders affected thereby, or (ii) any Lender makes a claim for compensation under <u>Section 16</u>, then Borrowers or Agent, upon at least five Business Days prior irrevocable notice, may permanently replace any Lender that failed to give its consent, authorization, or agreement (a "<u>Non-Consenting Lender</u>") or any Lender that made a claim for compensation (a "<u>Tax Lender</u>") with one or more Replacement Lenders, and the Non-Consenting Lender or Tax Lender, as applicable, shall have no right to refuse to be replaced hereunder. Such notice to replace the Non-Consenting Lender or Tax Lender, as applicable, shall specify an effective date for such replacement, which date shall not be later than 15 Business Days after the date such notice is given.

(b) Prior to the effective date of such replacement, the Non-Consenting Lender or Tax Lender, as applicable, and each Replacement Lender shall execute and deliver an Assignment and Acceptance, subject only to the Non-Consenting Lender or Tax Lender, as applicable, being repaid in full its share of the outstanding Obligations (without any premium or penalty of any kind whatsoever, but including (i) all interest, fees and other amounts that may be due in payable in respect thereof, (ii) an assumption of its Pro Rata Share of participations in the Letters of Credit, and (iii) Funding Losses). If the Non-Consenting Lender or Tax Lender, as applicable, shall refuse or fail to execute and deliver any such Assignment and Acceptance prior to the effective date of such replacement, Agent may, but shall not be required to, execute and deliver such Assignment and Acceptance in the name or and on behalf of the Non-Consenting Lender or Tax Lender, as applicable, and irrespective of whether Agent executes and delivers such Assignment and Acceptance. The replacement of any Non-Consenting Lender or Tax Lender, as applicable, shall be deemed to have executed and delivered such Assignment and Acceptance. The replacement of any Non-Consenting Lender or Tax Lender, as applicable, shall be made in accordance with the terms of <u>Section 13.1</u>. Until such time as one or more Replacement Lenders shall have acquired all of the Obligations, the Commitments, and the other rights and obligations of the Non-Consenting Lender or Tax Lender, as applicable, hereunder and under the other Loan Documents, the Non-Consenting Lender or Tax Lender, as applicable, shall set executes applicable, Pro Rata Share of Revolving Loans and to purchase a participation in each Letter of Credit, in an amount equal to its Pro Rata Share of participations in such Letters of Credit.

14.3 <u>No Waivers; Cumulative Remedies</u>. No failure by Agent or any Lender to exercise any right, remedy, or option under this Agreement or any other Loan Document, or delay by Agent or any Lender in exercising the same, will operate as a waiver thereof. No waiver by Agent or any Lender will be effective unless it is in writing, and then only to the extent specifically stated. No waiver by Agent or any Lender on any occasion shall affect or diminish Agent's and each Lender's rights thereafter to require strict performance by Borrowers of any provision of this Agreement. Agent's and each Lender's rights under this Agreement and the other Loan Documents will be cumulative and not exclusive of any other right or remedy that Agent or any Lender may have.

15. AGENT; THE LENDER GROUP.

15.1 **Appointment and Authorization of Agent**. Each Lender hereby designates and appoints Wells Fargo as its agent under this Agreement and the other Loan Documents and each Lender hereby irrevocably

-107-

authorizes (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to designate, appoint, and authorize) Agent to execute and deliver each of the other Loan Documents on its behalf and to take such other action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to Agent by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Agent agrees to act as agent for and on behalf of the Lenders (and the Bank Product Providers) on the conditions contained in this Section 15. Any provision to the contrary contained elsewhere in this Agreement or in any other Loan Document notwithstanding, Agent shall not have any duties or responsibilities, except those expressly set forth herein or in the other Loan Documents, nor shall Agent have or be deemed to have any fiduciary relationship with any Lender (or Bank Product Provider), and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent. Without limiting the generality of the foregoing, the use of the term "agent" in this Agreement or the other Loan Documents with reference to Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only a representative relationship between independent contracting parties. Each Lender hereby further authorizes (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent to act as the secured party under each of the Loan Documents that create a Lien on any item of Collateral. Except as expressly otherwise provided in this Agreement, Agent shall have and may use its sole discretion with respect to exercising or refraining from exercising any discretionary rights or taking or refraining from taking any actions that Agent expressly is entitled to take or assert under or pursuant to this Agreement and the other Loan Documents. Without limiting the generality of the foregoing, or of any other provision of the Loan Documents that provides rights or powers to Agent, Lenders agree that Agent shall have the right to exercise the following powers as long as this Agreement remains in effect: (a) maintain, in accordance with its customary business practices, ledgers and records reflecting the status of the Obligations, the Collateral, payments and proceeds of Collateral, and related matters, (b) execute or file any and all financing or similar statements or notices, amendments, renewals, supplements, documents, instruments, proofs of claim, notices and other written agreements with respect to the Loan Documents, or to take any other action with respect to any Collateral or Loan Documents which may be necessary to perfect, and maintain perfected, the security interests and Liens upon Collateral pursuant to the Loan Documents, (c) make Revolving Loans, for itself or on behalf of Lenders, as provided in the Loan Documents, (d) exclusively receive, apply, and distribute payments and proceeds of the Collateral as provided in the Loan Documents, (e) open and maintain such bank accounts and cash management arrangements as Agent deems necessary and appropriate in accordance with the Loan Documents for the foregoing purposes, (f) perform, exercise, and enforce any and all other rights and remedies of the Lender Group with respect to any Loan Party or its Subsidiaries, the Obligations, the Collateral, or otherwise related to any of same as provided in the Loan Documents, and (g) incur and pay such Lender Group Expenses as Agent may deem necessary or appropriate for the performance and fulfillment of its functions and powers pursuant to the Loan Documents.

15.2 **Delegation of Duties**. Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys in fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. Agent shall not be responsible for the negligence or misconduct of any agent or attorney in fact that it selects as long as such selection was made without gross negligence or willful misconduct.

15.3 **Liability of Agent**. None of the Agent-Related Persons shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct), or (b) be responsible in any manner to any of the Lenders (or Bank Product Providers) for any recital, statement, representation or warranty made by any Loan Party or any of its Subsidiaries or Affiliates, or any officer or director thereof, contained in this Agreement or in any other Loan Document, or in any certificate, report, statement or other document referred to or provided for in, or received by Agent under or in connection with, this Agreement

-108-

or any other Loan Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document, or for any failure of any Loan Party or its Subsidiaries or any other party to any Loan Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lenders (or Bank Product Providers) to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the books and records or properties of any Loan Party or its Subsidiaries. No Agent-Related Person shall have any liability to any Lender, and Loan Party or any of their respective Affiliates if any request for a Loan, Letter of Credit or other extension of credit was not authorized by the applicable Borrower. Agent shall not be required to take any action that, in its opinion or in the opinion of its counsel, may expose it to liability or that is contrary to any Loan Document or applicable law or regulation.

15.4 **Reliance by Agent**. Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telegram, telefacsimile or other electronic method of transmission, telex or telephone message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent, or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to Borrowers or counsel to any Lender), independent accountants and other experts selected by Agent. Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless Agent shall first receive such advice or concurrence of the Lenders as it deems appropriate and until such instructions are received, Agent shall act, or refrain from acting, as it deems advisable. If Agent so requests, it shall first be indemnified to its reasonable satisfaction by the Lenders (and, if it so elects, the Bank Product Providers) against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action. Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of the Required Lenders and such request and any action taken or failure to act pursuant thereto shall be binding upon all of the Lenders (and Bank Product Providers).

15.5 Notice of Default or Event of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest, fees, and expenses required to be paid to Agent for the account of the Lenders and, except with respect to Events of Default of which Agent has actual knowledge, unless Agent shall have received written notice from a Lender or Borrowers referring to this Agreement, describing such Default or Event of Default, and stating that such notice is a "notice of default." Agent promptly will notify the Lenders of its receipt of any such notice or of any Event of Default of which Agent has actual knowledge. If any Lender obtains actual knowledge of any Event of Default, such Lender promptly shall notify the other Lenders and Agent of such Event of Default. Each Lender shall be solely responsible for giving any notices to its Participants, if any. Subject to Section 15.4, Agent shall take such action with respect to such Default or Event of Default as may be requested by the Required Lenders in accordance with Section 9; provided, that unless and until Agent has received any such request, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable.

15.6 <u>Credit Decision</u>. Each Lender (and Bank Product Provider) acknowledges that none of the Agent-Related Persons has made any representation or warranty to it, and that no act by Agent hereinafter taken, including any review of the affairs of any Loan Party and its Subsidiaries or Affiliates, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender (or Bank Product Provider). Each Lender represents (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to represent) to Agent that it has, independently and without reliance upon any Agent-Related Person and based on such due diligence, documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of each Borrower or any other Person party to a Loan Document, and all applicable bank regulatory laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to Borrowers. Each Lender also represents (and by entering into a Bank Product Agreement, each Bank

-109-

Product Provider shall be deemed to represent) that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of each Borrower or any other Person party to a Loan Document. Except for notices, reports, and other documents expressly herein required to be furnished to the Lenders by Agent, Agent shall not have any duty or responsibility to provide any Lender (or Bank Product Provider) with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any Borrower or any other Person party to a Loan Document that may come into the possession of any of the Agent-Related Persons. Each Lender acknowledges (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that Agent does not have any duty or responsibility, either initially or on a continuing basis (except to the extent, if any, that is expressly specified herein) to provide such Lender (or Bank Product Provider) with nespect to any Borrower, its Affiliates or any of their respective business, legal, financial or other affairs, and irrespective of whether such information came into Agent's or its Affiliates' or representatives' possession before or after the date on which such Lender became a party to this Agreement (or such Bank Product Provider entered into a Bank Product Agreement).

15.7 Costs and Expenses; Indemnification. Agent may incur and pay Lender Group Expenses to the extent Agent reasonably deems necessary or appropriate for the performance and fulfillment of its functions, powers, and obligations pursuant to the Loan Documents, including court costs, attorneys' fees and expenses, fees and expenses of financial accountants, advisors, consultants, and appraisers, costs of collection by outside collection agencies, auctioneer fees and expenses, and costs of security guards or insurance premiums paid to maintain the Collateral, whether or not Borrowers are obligated to reimburse Agent or Lenders for such expenses pursuant to this Agreement or otherwise. Agent is authorized and directed to deduct and retain sufficient amounts from payments or proceeds of the Collateral received by Agent to reimburse Agent for such out-of-pocket costs and expenses prior to the distribution of any amounts to Lenders (or Bank Product Providers). In the event Agent is not reimbursed for such costs and expenses by the Loan Parties and their Subsidiaries, each Lender hereby agrees that it is and shall be obligated to pay to Agent such Lender's ratable thereof. Whether or not the transactions contemplated hereby are consummated, each of the Lenders, on a ratable basis, shall indemnify and defend the Agent-Related Persons (to the extent not reimbursed by or on behalf of Borrowers and without limiting the obligation of Borrowers to do so) from and against any and all Indemnified Liabilities; provided, that no Lender shall be liable for the payment to any Agent-Related Person of any portion of such Indemnified Liabilities resulting solely from such Person's gross negligence or willful misconduct nor shall any Lender be liable for the obligations of any Defaulting Lender in failing to make a Revolving Loan or other extension of credit hereunder. Without limitation of the foregoing, each Lender shall reimburse Agent upon demand for such Lender's ratable share of any costs or out of pocket expenses (including attorneys, accountants, advisors, and consultants fees and expenses) incurred by Agent in connection with the preparation, execution, delivery, administration, modification, amendment, or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement or any other Loan Document to the extent that Agent is not reimbursed for such expenses by or on behalf of Borrowers. The undertaking in this Section shall survive the payment of all Obligations hereunder and the resignation or replacement of Agent.

15.8 <u>Agent in Individual Capacity</u>. Wells Fargo and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, provide Bank Products to, acquire Equity Interests in, and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with any Loan Party and its Subsidiaries and Affiliates and any other Person party to any Loan Document as though Wells Fargo were not Agent hereunder, and, in each case, without notice to or consent of the other members of the Lender Group. The other members of the Lender Group acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, pursuant to such activities, Wells Fargo or its Affiliates may receive information regarding a Loan Party or its Affiliates or any other Person party to any Loan

-110-

Documents that is subject to confidentiality obligations in favor of such Loan Party or such other Person and that prohibit the disclosure of such information to the Lenders (or Bank Product Providers), and the Lenders acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver Agent will use its reasonable best efforts to obtain), Agent shall not be under any obligation to provide such information to them. The terms "Lender" and "Lenders" include Wells Fargo in its individual capacity.

15.9 Successor Agent. Agent may resign as Agent upon 30 days (ten days if an Event of Default has occurred and is continuing) prior written notice to the Lenders (unless such notice is waived by the Required Lenders) and Borrowers (unless such notice is waived by Borrowers or a Default or Event of Default has occurred and is continuing) and without any notice to the Bank Product Providers. If Agent resigns under this Agreement, the Required Lenders shall be entitled, with (so long as no Event of Default has occurred and is continuing) the consent of Borrowers (such consent not to be unreasonably withheld, delayed, or conditioned), appoint a successor Agent for the Lenders (and the Bank Product Providers). If, at the time that Agent's resignation is effective, it is acting as Issuing Bank or the Swing Lender, such resignation shall also operate to effectuate its resignation as Issuing Bank or the Swing Lender, as applicable, and it shall automatically be relieved of any further obligation to issue Letters of Credit, or to make Swing Loans. If no successor Agent is appointed prior to the effective date of the resignation of Agent, Agent may appoint, after consulting with the Lenders and Borrowers, a successor Agent. If Agent has materially breached or failed to perform any material provision of this Agreement or of applicable law, the Required Lenders may agree in writing to remove and replace Agent with a successor Agent from among the Lenders with (so long as no Event of Default has occurred and is continuing) the consent of Borrowers (such consent not to be unreasonably withheld, delayed, or conditioned). In any such event, upon the acceptance of its appointment as successor Agent hereunder, such successor Agent shall succeed to all the rights, powers, and duties of the retiring Agent and the term "Agent" shall mean such successor Agent and the retiring Agent's appointment, powers, and duties as Agent shall be terminated. After any retiring Agent's resignation hereunder as Agent, the provisions of this Section 15 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement. If no successor Agent has accepted appointment as Agent by the date which is 30 days following a retiring Agent's notice of resignation, the retiring Agent's resignation shall nevertheless thereupon become effective and the Lenders shall perform all of the duties of Agent hereunder until such time, if any, as the Lenders appoint a successor Agent as provided for above.

15.10 Lender in Individual Capacity. Any Lender and its respective Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, provide Bank Products to, acquire Equity Interests in and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with any Loan Party and its Subsidiaries and Affiliates and any other Person party to any Loan Documents as though such Lender were not a Lender hereunder without notice to or consent of the other members of the Lender Group (or the Bank Product Providers). The other members of the Lender Group acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, pursuant to such activities, such Lender and its respective Affiliates may receive information regarding a Loan Party or its Affiliates or any other Person party to any Loan Documents that is subject to confidentiality obligations in favor of such Loan Party or such other Person and that prohibit the disclosure of such information to the Lenders, and the Lenders acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver such Lender will use its reasonable best efforts to obtain), such Lender shall not be under any obligation to provide such information to them.

15.11 Collateral Matters.

(a) The Lenders hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent to release any Lien on any

-111-

Collateral (i) upon the termination of the Commitments and payment and satisfaction in full by the Loan Parties and their Subsidiaries of all of the Obligations, (ii) constituting property being sold or disposed of if a release is required or desirable in connection therewith and if Borrowers certify to Agent that the sale or disposition is permitted under Section 6.4 (and Agent may rely conclusively on any such certificate, without further inquiry), (iii) constituting property in which no Loan Party or any of its Subsidiaries owned any interest at the time Agent's Lien was granted nor at any time thereafter, (iv) constituting property leased or licensed to a Loan Party or its Subsidiaries under a lease or license that has expired or is terminated in a transaction permitted under this Agreement, or (v) in connection with a credit bid or purchase authorized under this Section 15.11. The Loan Parties and the Lenders hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent, based upon the instruction of the Required Lenders, to (a) consent to the sale of, credit bid, or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any sale thereof conducted under the provisions of the Bankruptcy Code, including Section 363 of the Bankruptcy Code, (b) credit bid or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any sale or other disposition thereof conducted under the provisions of the Code, including pursuant to Sections 9-610 or 9-620 of the Code, or (c) credit bid or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any other sale or foreclosure conducted or consented to by Agent in accordance with applicable law in any judicial action or proceeding or by the exercise of any legal or equitable remedy. In connection with any such credit bid or purchase, (i) the Obligations owed to the Lenders and the Bank Product Providers shall be entitled to be, and shall be, credit bid on a ratable basis (with Obligations with respect to contingent or unliquidated claims being estimated for such purpose if the fixing or liquidation thereof would not impair or unduly delay the ability of Agent to credit bid or purchase at such sale or other disposition of the Collateral and, if such contingent or unliquidated claims cannot be estimated without impairing or unduly delaying the ability of Agent to credit bid at such sale or other disposition, then such claims shall be disregarded, not credit bid, and not entitled to any interest in the Collateral that is the subject of such credit bid or purchase) and the Lenders and the Bank Product Providers whose Obligations are credit bid shall be entitled to receive interests (ratably based upon the proportion of their Obligations credit bid in relation to the aggregate amount of Obligations so credit bid) in the Collateral that is the subject of such credit bid or purchase (or in the Equity Interests of the any entities that are used to consummate such credit bid or purchase), and (ii) Agent, based upon the instruction of the Required Lenders, may accept non-cash consideration, including debt and equity securities issued by any entities used to consummate such credit bid or purchase and in connection therewith Agent may reduce the Obligations owed to the Lenders and the Bank Product Providers (ratably based upon the proportion of their Obligations credit bid in relation to the aggregate amount of Obligations so credit bid) based upon the value of such non-cash consideration; provided, that Bank Product Obligations not entitled to the application set forth in Section 2.4(b)(iii)(J) shall not be entitled to be, and shall not be, credit bid, or used in the calculation of the ratable interest of the Lenders and Bank Product Providers in the Obligations which are credit bid. Except as provided above, Agent will not execute and deliver a release of any Lien on any Collateral without the prior written authorization of (y) if the release is of all or substantially all of the Collateral, all of the Lenders (without requiring the authorization of the Bank Product Providers), or (z) otherwise, the Required Lenders (without requiring the authorization of the Bank Product Providers). Upon request by Agent or Borrowers at any time, the Lenders will (and if so requested, the Bank Product Providers will) confirm in writing Agent's authority to release any such Liens on particular types or items of Collateral pursuant to this Section 15.11; provided, that (1) anything to the contrary contained in any of the Loan Documents notwithstanding, Agent shall not be required to execute any document or take any action necessary to evidence such release on terms that, in Agent's opinion, could expose Agent to liability or create any obligation or entail any consequence other than the release of such Lien without recourse, representation, or warranty, and (2) such release shall not in any manner discharge, affect, or impair the Obligations or any Liens (other than those expressly released) upon (or obligations of Borrowers in respect of) any and all interests retained by

-112-

any Borrower, including, the proceeds of any sale, all of which shall continue to constitute part of the Collateral. Each Lender further hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to irrevocably authorize) Agent, at its option and in its sole discretion, to subordinate (by contract or otherwise) any Lien granted to or held by Agent on any property under any Loan Document (a) to the holder of any Permitted Lien on such property if such Permitted Lien secures purchase money Indebtedness (including Capitalized Lease Obligations) which constitute Permitted Indebtedness and (b) to the extent Agent has the authority under this <u>Section 15.11</u> to release its Lien on such property.

(b) Agent shall have no obligation whatsoever to any of the Lenders (or the Bank Product Providers) (i) to verify or assure that the Collateral exists or is owned by a Loan Party or any of its Subsidiaries or is cared for, protected, or insured or has been encumbered, (ii) to verify or assure that Agent's Liens have been properly or sufficiently or lawfully created, perfected, protected, or enforced or are entitled to any particular priority, (iii) to verify or assure that any particular items of Collateral meet the eligibility criteria applicable in respect thereof, (iv) to impose, maintain, increase, reduce, implement, or eliminate any particular reserve hereunder or to determine whether the amount of any reserve is appropriate or not, or (v) to exercise at all or in any particular manner or under any duty of care, disclosure or fidelity, or to continue exercising, any of the rights, authorities and powers granted or available to Agent pursuant to any of the Loan Documents, it being understood and agreed that in respect of the Collateral, or any act, omission, or event related thereto, subject to the terms and conditions contained herein, Agent may act in any manner it may deem appropriate, in its sole discretion given Agent's own interest in the Collateral in its capacity as one of the Lenders and that Agent shall have no other duty or liability whatsoever to any Lender (or Bank Product Provider) as to any of the foregoing, except as otherwise expressly provided herein.

15.12 <u>Restrictions on Actions by Lenders; Sharing of Payments</u>.

(a) Each of the Lenders agrees that it shall not, without the express written consent of Agent, and that it shall, to the extent it is lawfully entitled to do so, upon the written request of Agent, set off against the Obligations, any amounts owing by such Lender to any Loan Party or its Subsidiaries or any deposit accounts of any Loan Party or its Subsidiaries now or hereafter maintained with such Lender. Each of the Lenders further agrees that it shall not, unless specifically requested to do so in writing by Agent, take or cause to be taken any action, including, the commencement of any legal or equitable proceedings to enforce any Loan Document against any Borrower or any Guarantor or to foreclose any Lien on, or otherwise enforce any security interest in, any of the Collateral.

(b) If, at any time or times any Lender shall receive (i) by payment, foreclosure, setoff, or otherwise, any proceeds of Collateral or any payments with respect to the Obligations, except for any such proceeds or payments received by such Lender from Agent pursuant to the terms of this Agreement, or (ii) payments from Agent in excess of such Lender's Pro Rata Share of all such distributions by Agent, such Lender promptly shall (A) turn the same over to Agent, in kind, and with such endorsements as may be required to negotiate the same to Agent, or in immediately available funds, as applicable, for the account of all of the Lenders and for application to the Obligations in accordance with the applicable provisions of this Agreement, or (B) purchase, without recourse or warranty, an undivided interest and participation in the Obligations owed to the other Lenders so that such excess payment received shall be applied ratably as among the Lenders in accordance with their Pro Rata Shares; <u>provided</u>, that to the extent that such excess payment received by the purchasing party is thereafter recovered from it, those purchases of participations shall be rescinded in whole or in part, as applicable, and the applicable portion of the purchase price paid therefor shall be returned to such purchasing party, but without interest except to the extent that such purchasing party is required to pay interest in connection with the recovery of the excess payment.



15.13 <u>Agency for Perfection</u>. Agent hereby appoints each other Lender (and each Bank Product Provider) as its agent (and each Lender hereby accepts (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to accept) such appointment) for the purpose of perfecting Agent's Liens in assets which, in accordance with Article 8 or Article 9, as applicable, of the Code can be perfected by possession or control. Should any Lender obtain possession or control of any such Collateral, such Lender shall notify Agent thereof, and, promptly upon Agent's request therefor shall deliver possession or control of such Collateral to Agent or in accordance with Agent's instructions.

15.14 **Payments by Agent to the Lenders.** All payments to be made by Agent to the Lenders (or Bank Product Providers) shall be made by bank wire transfer of immediately available funds pursuant to such wire transfer instructions as each party may designate for itself by written notice to Agent. Concurrently with each such payment, Agent shall identify whether such payment (or any portion thereof) represents principal, premium, fees, or interest of the Obligations.

15.15 **Concerning the Collateral and Related Loan Documents**. Each member of the Lender Group authorizes and directs Agent to enter into this Agreement and the other Loan Documents. Each member of the Lender Group agrees (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to agree) that any action taken by Agent in accordance with the terms of this Agreement or the other Loan Documents relating to the Collateral and the exercise by Agent of its powers set forth therein or herein, together with such other powers that are reasonably incidental thereto, shall be binding upon all of the Lenders (and such Bank Product Provider).

15.16 **Field Examination Reports; Confidentiality; Disclaimers by Lenders; Other Reports and Information**. By becoming a party to this Agreement, each Lender:

(a) is deemed to have requested that Agent furnish such Lender, promptly after it becomes available, a copy of each field examination report respecting any Loan Party or its Subsidiaries (each, a "<u>Report</u>") prepared by or at the request of Agent, and Agent shall so furnish each Lender with such Reports,

(b) expressly agrees and acknowledges that Agent does not (i) make any representation or warranty as to the accuracy of any Report, and (ii) shall not be liable for any information contained in any Report,

(c) expressly agrees and acknowledges that the Reports are not comprehensive audits or examinations, that Agent or other party performing any field examination will inspect only specific information regarding the Loan Parties and their Subsidiaries and will rely significantly upon Borrowers' and their Subsidiaries' books and records, as well as on representations of Borrowers' personnel,

(d) agrees to keep all Reports and other material, non-public information regarding the Loan Parties and their Subsidiaries and their operations, assets, and existing and contemplated business plans in a confidential manner in accordance with <u>Section 17.9</u>, and

(e) without limiting the generality of any other indemnification provision contained in this Agreement, agrees: (i) to hold Agent and any other Lender preparing a Report harmless from any action the indemnifying Lender may take or fail to take or any conclusion the indemnifying Lender may reach or draw from any Report in connection with any loans or other credit accommodations that the indemnifying Lender has made or may make to Borrowers, or the indemnifying Lender's participation in, or the indemnifying Lender's purchase of, a loan or loans of Borrowers, and (ii) to pay and protect, and indemnify, defend and hold Agent, and any such other Lender preparing a Report harmless from and against, the claims, actions, proceedings, damages, costs, expenses, and other amounts (including, attorneys' fees and costs)

-114-

incurred by Agent and any such other Lender preparing a Report as the direct or indirect result of any third parties who might obtain all or part of any Report through the indemnifying Lender.

In addition to the foregoing, (x) any Lender may from time to time request of Agent in writing that Agent provide to such Lender a copy of any report or document provided by any Loan Party or its Subsidiaries to Agent that has not been contemporaneously provided by such Loan Party or such Subsidiary to such Lender, and, upon receipt of such request, Agent promptly shall provide a copy of same to such Lender, (y) to the extent that Agent is entitled, under any provision of the Loan Documents, to request additional reports or information from any Loan Party or its Subsidiaries, any Lender may, from time to time, reasonably request Agent to exercise such right as specified in such Lender's notice to Agent, whereupon Agent promptly shall request of Borrowers the additional reports or information reasonably specified by such Lender, and, upon receipt thereof from such Loan Party or such Subsidiary, Agent promptly shall provide a copy of same to such Lender, and (z) any time that Agent renders to Borrowers a statement regarding the Loan Account, Agent shall send a copy of such statement to each Lender.

15.17 Several Obligations: No Liability. Notwithstanding that certain of the Loan Documents now or hereafter may have been or will be executed only by or in favor of Agent in its capacity as such, and not by or in favor of the Lenders, any and all obligations on the part of Agent (if any) to make any credit available hereunder shall constitute the several (and not joint) obligations of the respective Lenders on a ratable basis, according to their respective Commitments, to make an amount of such credit not to exceed, in principal amount, at any one time outstanding, the amount of their respective Commitments. Nothing contained herein shall confer upon any Lender any interest in, or subject any Lender to any liability for, or in respect of, the business, assets, profits, losses, or liabilities of any other Lender. Each Lender shall be solely responsible for notifying its Participants of any matters relating to the Loan Documents to the extent any such notice may be required, and no Lender shall have any obligation, duty, or liability to any Participant of any other Lender. Except as provided in Section 15.7, no member of the Lender Group shall have any liability for the acts of any other member of the Lender Group. No Lender shall be responsible to any Borrower or any other Person for any failure by any other Lender (or Bank Product Provider) to fulfill its obligations to make credit available hereunder, nor to advance for such Lender (or Bank Product Provider) or on its behalf, nor to take any other action on behalf of such Lender (or Bank Product Provider) hereunder or in connection with the financing contemplated herein.

15.18 Joint Lead Arrangers and Joint Book Runners. Each of the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall not have any right, power, obligation, liability, responsibility, or duty under this Agreement other than those applicable to it in its capacity as a Lender, as Agent, as Swing Lender, or as Issuing Bank. Without limiting the foregoing, each of the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall not have or be deemed to have any fiduciary relationship with any Lender or any Loan Party. Each Lender, Agent, Swing Lender, Issuing Bank, and each Loan Party acknowledges that it has not relied, and will not rely, on the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall be entitled to resign at any time by giving notice to Agent and Borrowers.

16. WITHHOLDING TAXES.

16.1 **Payments**. All payments made by any Loan Party under any Loan Document will be made free and clear of, and without deduction or withholding for, any Taxes, except as otherwise required by applicable law, and in the event any deduction or withholding of Taxes is required by applicable law, the applicable Withholding Agent shall make the requisite withholding, promptly pay over to the applicable Governmental Authority the withheld tax, and, in the case of deduction or withholding by any Loan Party, furnish to Agent as promptly as possible after the date the payment of any such Tax is due pursuant to applicable law, certified copies of tax receipts evidencing such payment by such Loan Party. Furthermore, if any such Tax is an Indemnified

-115-

Tax or an Indemnified Tax is so levied or imposed, the applicable Loan Party agrees to pay the full amount of such Indemnified Taxes and such additional amounts as may be necessary so that every payment of all amounts due under this Agreement or any Loan Document, including any amount paid pursuant to this <u>Section</u> <u>16.1</u>, after withholding or deduction for or on account of any Indemnified Taxes, will not be less than the amount provided for herein. The Loan Parties will promptly pay any Other Taxes in accordance with applicable law, or reimburse Agent for such Other Taxes upon Agent's demand. The Loan Parties shall jointly and severally indemnify each Recipient (in the case of a Participant, subject to <u>Section 16.2(e)</u>) for the full amount of Indemnified Taxes arising in connection with this Agreement or any other Loan Document (including any Indemnified Taxes imposed or asserted on, or attributable to, amounts payable under this <u>Section 16.1</u>) imposed on, or paid by, such Recipient and all reasonable costs and expenses related thereto within 10 days after demand therefor and irrespective of whether suit is brought, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority (other than Indemnified Taxes and additional amounts that a court of competent jurisdiction finally determines to have resulted from the gross negligence or willful misconduct of such Tax Indemnifiee).

16.2 Exemptions.

(a) If a Lender is entitled to claim an exemption or reduction from withholding tax with respect to payments made under any Loan Document, such Lender agrees to deliver to Agent and the Administrative Borrower on behalf of all Borrowers, at the time or times reasonably requested by the Agent or the Administrative Borrower, such properly completed and executed documentation reasonably requested by the Agent or the Administrative Borrower, such properly completed and executed documentation reasonably requested by the Agent or the Administrative Borrower, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Agent or the Administrative Borrower as will enable the Agent or the Administrative Borrower to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation est forth in paragraphs (a)(i) through (a)(v) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender. Without limiting the generality of the foregoing, if a Lender or Participant is entitled to claim an exemption or reduction from United States withholding tax, such Lender or Participant agrees with and in favor of Agent, to deliver to Agent and the Administrative Borrower on behalf of all Borrowers (or, in the case of a Participant, to the Lender or Participant under this Agreement (and from time to time thereafter upon the reasonable requested by the recipient) promptly upon becoming a Lender or Participant (and from time to time thereafter upon the reasonable requested by the Regent, Administrative Borrower or Lender, as applicable):

(i) if such Lender or Participant is entitled to claim an exemption from United States withholding tax pursuant to the portfolio interest exception, (A) a statement of the Lender or Participant, in a form reasonably acceptable to the recipient and signed under penalty of perjury, that it is not (I) a "bank" as described in Section 881(c)(3)(A) of the IRC, (II) a 10% shareholder of any Borrower (within the meaning of Section 871(h)(3) (B) of the IRC), or (III) a controlled foreign corporation related to any Borrower described in Section 881(c)(3)(C) of the IRC (a "U.S. Tax Compliance Certificate"), and (B) a properly completed and executed IRS Form W-8BEN, Form W-8BEN-E or Form W-8IMY (with proper attachments as applicable);

(ii) if such Lender or Participant is entitled to claim an exemption from, or a reduction of, withholding tax under a United States tax treaty, a properly completed and executed copy of IRS Form W-8BEN or Form W-8BEN-E, as applicable;

-116-

(iii) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding tax because it is effectively connected with a United States trade or business of such Lender, a properly completed and executed copy of IRS Form W-8ECI;

(iv) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding tax because such Lender or Participant serves as an intermediary, a properly completed and executed copy of IRS Form W-8IMY (including IRS Form W-8ECI, IRS Form W-8BEN or W-8BEN-E, as applicable, a U.S. Tax Compliance Certificate, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; <u>provided</u> that if such Lender or Participant is a partnership and one or more direct or indirect partners of such Lender or Participant are claiming the portfolio interest exemption, such Lender or Participant may provide a U.S. Tax Compliance Certificate on behalf of each such direct and indirect partner); or

(v) if such Lender or Participant is a U.S. Person (as defined in Section 7701(a)(30) of the IRC), a properly completed and executed copy of any other form or forms, including IRS Form W-9, as may be required under the IRC or other laws of the United States as a condition to exemption from, or reduction of, United States withholding or backup withholding tax.

(b) Each Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Agent and the Administrative Borrower (in such number of copies as shall be requested by the recipient) promptly upon becoming a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Agent or the Administrative Borrower), executed copies of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit the Agent or the Administrative Borrower to determine the withholding or deduction required to be made.

(c) Each Lender or Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to promptly notify Agent and Administrative Borrower (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.

(d) If a Lender or Participant claims an exemption from withholding tax in a jurisdiction other than the United States, such Lender or such Participant agrees with and in favor of Agent and Borrowers, to deliver to Agent and Administrative Borrower (or, in the case of a Participant, to the Lender granting the participation only) any such form or forms, as may be required under the laws of such jurisdiction as a condition to exemption from, or reduction of, foreign withholding or backup withholding tax promptly upon becoming a Lender or Participant under this Agreement, but only if such Lender or such Participant is legally able to deliver such forms, or the providing of or delivery of such forms in the Lender's reasonable judgment would not subject such Lender to any material unreimbursed cost or expense or materially prejudice the legal or commercial position of such Lender (or its Affiliates); provided, further, that nothing in this Section 16.2(d) shall require a Lender or Participant to disclose any information that it deems to be confidential (including its tax returns). Each Lender and each Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to promptly notify Agent and Administrative Borrower (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.

(e) If a Lender or Participant claims exemption from, or reduction of, withholding tax and such Lender or Participant sells, assigns, grants a participation in, or otherwise transfers all or part of the Obligations of Borrowers to such Lender or Participant, such Lender or Participant agrees to notify Agent

-117-

and Administrative Borrower (or, in the case of a sale of a participation interest, to the Lender granting the participation only) of the percentage amount in which it is no longer the beneficial owner of Obligations of Borrowers to such Lender or Participant. To the extent of such percentage amount, Agent and Administrative Borrower will treat such Lender's or such Participant's documentation provided pursuant to <u>Section 16.2(a)</u>, <u>16.2(b)</u> or <u>16.2(d)</u> as no longer valid. With respect to such percentage amount, such Participant or Assignee may provide new documentation, pursuant to <u>Section 16.2(a)</u>, <u>16.2(b)</u> or <u>16.2(d)</u>, if applicable. Borrowers agree that each Participant shall be entitled to the benefits of this <u>Section 16</u> with respect to its participation in any portion of the Commitments and the Obligations so long as such Participant complies with the obligations set forth in this <u>Section 16</u> with respect thereto; <u>provided</u> that such Participant shall not be entitled to receive any greater payment under this <u>Section 16</u> with respect to any participation, than its participating Lender would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation.

(a) If a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding tax imposed by FATCA if such Lender were to fail to comply with the applicable due diligence and reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the IRC, as applicable), such Lender shall deliver to Agent and the Administrative Borrower (or, in the case of a Participant, to the Lender granting the participation only) at the time or times prescribed by law and at such time or times reasonably requested by Agent and the Administrative Borrower (or, in the case of a Participant, the Lender granting the participation) such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the IRC) and such additional documentation reasonably requested by Agent and the Administrative Borrower (or, in the case of a Participant, the Lender granting the participation) as may be necessary for Agent and Borrowers to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (f), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

(b) Agent and any successor Agent shall deliver to the Administrative Borrower on or prior to the date on which such Agent or successor Agent becomes an Agent or successor Agent under this Agreement (and from time to time thereafter upon the reasonable request of the Administrative Borrower), executed copies of IRS Form W-9 certifying that such Agent or successor Agent is exempt from U.S. federal backup withholding tax.

16.3 **<u>Reductions</u>**.

(a) If a Lender or a Participant is subject to an applicable withholding tax, Agent (or, in the case of a Participant, the Lender granting the participation) may withhold from any payment to such Lender or such Participant an amount equivalent to the applicable withholding tax. If the forms or other documentation required by Section 16.2(a) or 16.2(c) are not delivered to Agent (or, in the case of a Participant, to the Lender granting the participation), then Agent (or, in the case of a Participant, to the Lender granting the participation), then Agent (or, in the case of a Participant, to the Lender granting the participant not providing such forms or other documentation an amount equivalent to the applicable withholding tax.

(b) If the IRS or any other Governmental Authority of the United States or other jurisdiction asserts a claim that Agent (or, in the case of a Participant, to the Lender granting the participation) did not properly withhold tax from amounts paid to or for the account of any Lender or any Participant due to a failure on the part of the Lender or any Participant (because the appropriate form was not delivered, was not properly executed, or because such Lender failed to notify Agent (or such Participant failed to notify the Lender granting the participation) of a change in circumstances which rendered the exemption from, or reduction of, withholding tax ineffective, or for any other reason) such Lender shall indemnify and hold

-118-

Agent harmless (or, in the case of a Participant, such Participant shall indemnify and hold the Lender granting the participation harmless) for all amounts paid, directly or indirectly, by Agent (or, in the case of a Participant, to the Lender granting the participation), as tax or otherwise, including penalties and interest, and including any taxes imposed by any jurisdiction on the amounts payable to Agent (or, in the case of a Participant, to the Lender granting the participant, to the Lender granting the participation only) under this <u>Section 16.3(b)</u>, together with all costs and expenses (including attorneys' fees and expenses). The obligation of the Lenders and the Participants under this subsection shall survive the payment of all Obligations and the resignation or replacement of Agent.

16.4 **Refunds**. If Agent or a Lender determines, in its sole discretion exercised in good faith, that it has received a refund of any Indemnified Taxes to which the Loan Parties have paid additional amounts pursuant to <u>Section 16</u> (including by the payment of additional amounts pursuant to <u>Section 16</u>), so long as no Default or Event of Default has occurred and is continuing, it shall pay over such refund to the Administrative Borrower on behalf of the Loan Parties (but only to the extent of payments made, or additional amounts paid, by the Loan Parties under <u>Section 16</u> with respect to Indemnified Taxes giving rise to such a refund), net of all out-of-pocket expenses of Agent or such Lender and without interest (other than any interest paid by the applicable Governmental Authority with respect to such a refund); <u>provided</u>, that the Loan Parties, upon the request of Agent or such Lender, agrees to repay the amount paid over to the Loan Parties (plus any penalties, interest or other charges, imposed by the applicable Governmental Authority, other than such penalties, interest or other charges imposed as a result of the willful misconduct or gross negligence of Agent or Lender hereunder as finally determined by a court of competent jurisdiction) to Agent or such Lender in the event Agent or such Lender is required to repay such refund to such Governmental Authority. Notwithstanding anything in this Agreement to the contrary, this <u>Section 16.4</u> shall not be construed to require Agent or any Lender to make available its tax returns (or any other information which it deems confidential) to Loan Parties or any other Person or require Agent or any Lender to pay any amount to an indemnifying party pursuant to this <u>Section 16.4</u>, the payment of which would place Agent or such Lender in a less favorable net after-Tax position than such Person would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments o

17. GENERAL PROVISIONS.

17.1 **Effectiveness**. This Agreement shall be binding and deemed effective when executed by each Borrower, Agent, and each Lender whose signature is provided for on the signature pages hereof.

17.2 <u>Section Headings</u>. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Agreement.

17.3 **Interpretation**. Neither this Agreement nor any uncertainty or ambiguity herein shall be construed against the Lender Group or any Borrower, whether under any rule of construction or otherwise. On the contrary, this Agreement has been reviewed by all parties and shall be construed and interpreted according to the ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.

17.4 <u>Severability of Provisions</u>. Each provision of this Agreement shall be severable from every other provision of this Agreement for the purpose of determining the legal enforceability of any specific provision.

17.5 **Bank Product Providers**. Each Bank Product Provider in its capacity as such shall be deemed a third party beneficiary hereof and of the provisions of the other Loan Documents for purposes of any reference in a Loan Document to the parties for whom Agent is acting. Agent hereby agrees to act as agent for such Bank Product Providers and, by virtue of entering into a Bank Product Agreement, the applicable Bank Product Provider shall be automatically deemed to have appointed Agent as its agent and to have accepted the benefits of the Loan Documents. It is understood and agreed that the rights and benefits of each Bank Product Provider under the

-119-

Loan Documents consist exclusively of such Bank Product Provider's being a beneficiary of the Liens and security interests (and, if applicable, guarantees) granted to Agent and the right to share in payments and collections out of the Collateral as more fully set forth herein. In addition, each Bank Product Provider, by virtue of entering into a Bank Product Agreement, shall be automatically deemed to have agreed that Agent shall have the right, but shall have no obligation, to establish, maintain, relax, or release reserves in respect of the Bank Product Obligations and that if reserves are established there is no obligation on the part of Agent to determine or insure whether the amount of any such reserve is appropriate or not. In connection with any such distribution of payments or proceeds of Collateral, Agent shall be entitled to assume no amounts are due or owing to any Bank Product Provider unless such Bank Product Provider has provided a written certification (setting forth a reasonably detailed calculation) to Agent as to the amounts that are due and owing to it and such written certification is received by Agent a reasonable period of time prior to the making of such distribution. Agent shall have no obligation to calculate the amount due and payable with respect to any Bank Products, but may rely upon the written certification of the amount due and payable from the applicable Bank Product Provider. In the absence of an updated certification. Agent shall be entitled to assume that the amount due and payable to the applicable Bank Product Provider is the amount last certified to Agent by such Bank Product Provider as being due and payable (less any distributions made to such Bank Product Provider on account thereof). Borrowers may obtain Bank Products from any Bank Product Provider, although Borrowers are not required to do so. Each Borrower acknowledges and agrees that no Bank Product Provider has committed to provide any Bank Products and that the providing of Bank Products by any Bank Product Provider is in the sole and absolute discretion of such Bank Product Provider. Notwithstanding anything to the contrary in this Agreement or any other Loan Document, no provider or holder of any Bank Product shall have any voting or approval rights hereunder (or be deemed a Lender) solely by virtue of its status as the provider or holder of such agreements or products or the Obligations owing thereunder, nor shall the consent of any such provider or holder be required (other than in their capacities as Lenders, to the extent applicable) for any matter hereunder or under any of the other Loan Documents, including as to any matter relating to the Collateral or the release of Collateral or Guarantors.

17.6 **Debtor-Creditor Relationship**. The relationship between the Lenders and Agent, on the one hand, and the Loan Parties, on the other hand, is solely that of creditor and debtor. No member of the Lender Group has (or shall be deemed to have) any fiduciary relationship or duty to any Loan Party arising out of or in connection with the Loan Documents or the transactions contemplated thereby, and there is no agency or joint venture relationship between the members of the Lender Group, on the one hand, and the Loan Parties, on the other hand, by virtue of any Loan Document or any transaction contemplated therein.

17.7 **Counterparts: Electronic Execution**. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together, shall constitute but one and the same Agreement. Delivery of an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Agreement. Any party delivering an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission also shall deliver an original executed counterpart of this Agreement. The foregoing shall apply to each other Loan Document *mutatis mutandis*.

17.8 Revival and Reinstatement of Obligations; Certain Waivers.

(a) If any member of the Lender Group or any Bank Product Provider repays, refunds, restores, or returns in whole or in part, any payment or property (including any proceeds of Collateral) previously paid or transferred to such member of the Lender Group or such Bank Product Provider in full or partial satisfaction of any Obligation or on account of any other obligation of any Loan Party under any Loan Document or any Bank Product Agreement, because the payment, transfer, or the incurrence of the obligation so satisfied is asserted or declared to be void, voidable, or otherwise recoverable under any

-120-

law relating to creditors' rights, including provisions of the Bankruptcy Code relating to fraudulent transfers, preferences, or other voidable or recoverable obligations or transfers (each, a "<u>Voidable Transfer</u>"), or because such member of the Lender Group or Bank Product Provider elects to do so on the reasonable advice of its counsel in connection with a claim that the payment, transfer, or incurrence is or may be a Voidable Transfer, then, as to any such Voidable Transfer, or the amount thereof that such member of the Lender Group or Bank Product Provider elects to repay, restore, or return (including pursuant to a settlement of any claim in respect thereof), and as to all reasonable costs, expenses, and attorneys' fees of such member of the Lender Group or Bank Product Provider related thereto, (i) the liability of the Loan Parties with respect to the amount or property paid, refunded, restored, or returned will automatically and immediately be revived, reinstated, and restored and will exist, and (ii) Agent's Liens securing such liability shall be effective, revived, and remain in full force and effect, in each case, as fully as if such Voidable Transfer had never been made. If, prior to any of the foregoing, (A) Agent's Liens shall have been released or terminated, or (B) any provision of this Agreement shall have been terminated or cancelled, Agent's Liens, or such provision of this Agreement, shall be reinstated in full force and effect and such prior release, termination, cancellation or surrender shall not diminish, release, discharge, impair or otherwise affect the obligation of any Loan Party in respect of such liability or any Collateral securing such liability. This provision shall survive the termination of this Agreement and the repayment in full of the Obligations.

(b) Anything to the contrary contained herein notwithstanding, if Agent or any Lender accepts a guaranty of only a portion of the Obligations pursuant to any guaranty, each Borrower hereby waive its right under Section 2822(a) of the California Civil Code or any similar laws of any other applicable jurisdiction to designate the portion of the Obligations satisfied by the applicable guarantor's partial payment.

17.9 Confidentiality.

(a) Agent and Lenders each individually (and not jointly or jointly and severally) agree that material, non-public information regarding the Loan Parties and their Subsidiaries, their operations, assets, and existing and contemplated business plans ("Confidential Information") shall be treated by Agent and the Lenders in a confidential manner, and shall not be disclosed by Agent and the Lenders to Persons who are not parties to this Agreement, except: (i) to attorneys for and other advisors, accountants, auditors, and consultants to any member of the Lender Group and to employees, directors and officers of any member of the Lender Group (the Persons in this clause (i), "Lender Group Representatives") on a "need to know" basis in connection with this Agreement and the transactions contemplated hereby and on a confidential basis, (ii) to Subsidiaries and Affiliates of any member of the Lender Group (including the Bank Product Providers); provided, that any such Subsidiary or Affiliate shall have agreed to receive such information hereunder subject to the terms of this Section 17.9, (iii) as may be required by regulatory authorities so long as such authorities are informed of the confidential nature of such information, (iv) as may be required by statute, decision, or judicial or administrative order, rule, or regulation; provided, that (x) prior to any disclosure under this clause (iv), the disclosing party agrees to provide Borrowers with prior notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior notice to Borrowers pursuant to the terms of the applicable statute, decision, or judicial or administrative order, rule, or regulation and (y) any disclosure under this clause (iv) shall be limited to the portion of the Confidential Information as may be required by such statute, decision, or judicial or administrative order, rule, or regulation, (v) as may be agreed to in advance in writing by Borrowers, (vi) as requested or required by any Governmental Authority pursuant to any subpoena or other legal process; provided, that (x) prior to any disclosure under this clause (vi) the disclosing party agrees to provide Borrowers with prior written notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior written notice to Borrowers pursuant to the terms of the subpoena or other legal process and (y) any disclosure under this clause (vi) shall be limited to the portion of the

-121-

Confidential Information as may be required by such Governmental Authority pursuant to such subpoena or other legal process, (vii) as to any such information that is or becomes generally available to the public (other than as a result of prohibited disclosure by Agent or the Lenders or the Lender Group Representatives), (viii) (x) in connection with any assignment, participation or pledge of any Lender's interest under this Agreement, or (y) to any actual or prospective party to any swap, derivative or other transaction under which payments are to be made by reference to any Borrower and its Obligations, this Agreement or payments hereunder; provided, that prior to receipt of Confidential Information any such assignee, participant, pledgee, or actual or prospective party shall have agreed in writing to receive such Confidential Information either subject to the terms of this <u>Section 17.9</u> or pursuant to confidentiality requirements substantially similar to those contained in this <u>Section 17.9</u> (and such Person may disclose such Confidential Information to Persons employed or engaged by them as described in clause (i) above), (ix) in connection with any litigation or other adversary proceeding involves claims related to the rights or duties of such parties under this Agreement or the other Loan Documents; <u>provided</u>, that prior to any disclosure to any Person (other than any Lender, any of their respective counsel) under this clause (ix) with respect to litigation involving any Person (other than any Borrower, Agent, any Lender, any of their respective Affiliates, or their respective counsel) under this clause (ix) with respect to litigation involving any Person (other than any Borrower, Agent, any Lender, any of their respective Affiliates, or their respective counsel), the disclosing party agrees to provide Borrowers with prior written notice thereof, and (x) in connection with, and to the extent reasonably necessary for, the exercise of any secured creditor remedy under this Agreement or under any other Lo

(b) Anything in this Agreement to the contrary notwithstanding, Agent may disclose information concerning the terms and conditions of this Agreement and the other Loan Documents to loan syndication and pricing reporting services or in its marketing or promotional materials, with such information to consist of deal terms and other information customarily found in such publications or marketing or promotional materials and may otherwise use the name, logos, and other insignia of any Borrower or the other Loan Parties and the Commitments provided hereunder in any "tombstone" or other advertisements, on its website or in other marketing materials of the Agent.

Each Loan Party agrees that Agent may make materials or information provided by or on behalf of Borrowers hereunder (c) (collectively, "Borrower Materials") available to the Lenders by posting the Communications on IntraLinks, SyndTrak or a substantially similar secure electronic transmission system (the "Platform"). The Platform is provided "as is" and "as available." Agent does not warrant the accuracy or completeness of the Borrower Materials, or the adequacy of the Platform and expressly disclaim liability for errors or omissions in the communications. No warranty of any kind, express, implied or statutory, including any warranty of merchantability, fitness for a particular purpose, non-infringement of third party rights or freedom from viruses or other code defects, is made by Agent in connection with the Borrower Materials or the Platform. In no event shall Agent or any of the Agent-Related Persons have any liability to the Loan Parties, any Lender or any other person for damages of any kind, including direct or indirect, special, incidental or consequential damages, losses or expenses (whether in tort, contract or otherwise) arising out of any Loan Party's or Agent's transmission of communications through the Internet, except to the extent the liability of such person is found in a final non-appealable judgment by a court of competent jurisdiction to have resulted from such person's gross negligence or willful misconduct. Each Loan Party further agrees that certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Loan Parties or their securities) (each, a "Public Lender"). The Loan Parties shall be deemed to have authorized Agent and its Affiliates and the Lenders to treat Borrower Materials marked "PUBLIC" or otherwise at any time filed with the SEC as not containing any material non-public information with respect to the Loan Parties or their securities for purposes of United States federal and state securities laws. All Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated as "Public Investor" (or another similar term). Agent

-122-

and its Affiliates and the Lenders shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" or that are not at any time filed with the SEC as being suitable only for posting on a portion of the Platform not marked as "Public Investor" (or such other similar term).

17.10 **Survival**. All representations and warranties made by the Loan Parties in the Loan Documents and in the certificates or other instruments delivered in connection with or pursuant to this Agreement or any other Loan Document shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of the Loan Documents and the making of any Loans and issuance of any Letters of Credit, regardless of any investigation made by any such other party or on its behalf and notwithstanding that Agent, Issuing Bank, or any Lender may have had notice or knowledge of any Default or Event of Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of, or any accrued interest on, any Loan or any fee or any other amount payable under this Agreement is outstanding or unpaid or any Letter of Credit is outstanding and so long as the Commitments have not expired or been terminated.

17.11 **Patriot Act; Due Diligence**. Each Lender that is subject to the requirements of the Patriot Act hereby notifies the Loan Parties that pursuant to the requirements of the Patriot Act, it is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender to identify each Loan Party in accordance with the Patriot Act. In addition, Agent and each Lender shall have the right to periodically conduct due diligence on all Loan Parties, their senior management and key principals and legal and beneficial owners. Each Loan Party agrees to cooperate in respect of the conduct of such due diligence and further agrees that the reasonable costs and charges for any such due diligence by Agent shall constitute Lender Group Expenses hereunder and be for the account of Borrowers.

17.12 **Integration**. This Agreement, together with the other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the date hereof. The foregoing to the contrary notwithstanding, all Bank Product Agreements, if any, are independent agreements governed by the written provisions of such Bank Product Agreements, which will remain in full force and effect, unaffected by any repayment, prepayments, acceleration, reduction, increase, or change in the terms of any credit extended hereunder, except as otherwise expressly provided in such Bank Product Agreement.

17.13 **Pandora_as Agent for Borrowers**. Each Borrower hereby irrevocably appoints Pandora as the borrowing agent and attorney-in-fact for all Borrowers (the "Administrative Borrower") which appointment shall remain in full force and effect unless and until Agent shall have received prior written notice signed by each Borrower that such appointment has been revoked and that another Borrower has been appointed Administrative Borrower. Each Borrower hereby irrevocably appoints and authorizes the Administrative Borrower (a) to provide Agent with all notices with respect to Revolving Loans and Letters of Credit obtained for the benefit of any Borrower and all other notices and instructions under this Agreement and the other Loan Documents (and any notice or instruction provided by Administrative Borrower shall be deemed to be given by Borrowers hereunder and shall bind each Borrower), (b) to receive notices and instructions from members of the Lender Group (and any notice or instruction provided by any member of the Lender Group to the Administrative Borrower in accordance with the terms hereof shall be deemed to have been given to each Borrower), (c) to enter into Bank Product Provider Agreements on behalf of Borrowers and their Subsidiaries, and (d) to take such action as the Administrative Borrower deems appropriate on its behalf to obtain Revolving Loans and Letters of Credit and to exercise such other powers as are reasonably incidental thereto to carry out the purposes of this Agreement. It is understood that the handling of the Loan Account and Collateral in a combined fashion, as more fully set forth herein, is done solely as an accommodation to Borrowers in order to utilize the collective borrowing powers of Borrowers in the most efficient and economical manner and at their request, and that Lender Group shall not incur liability to any Borrower as a result hereof. Each Borrower expects to derive benefit, directly or indirectly, from

-123-

the handling of the Loan Account and the Collateral in a combined fashion since the successful operation of each Borrower is dependent on the continued successful performance of the integrated group. To induce the Lender Group to do so, and in consideration thereof, each Borrower hereby jointly and severally agrees to indemnify each member of the Lender Group and hold each member of the Lender Group harmless against any and all liability, expense, loss or claim of damage or injury, made against the Lender Group by any Borrower or by any third party whosoever, arising from or incurred by reason of (i) the handling of the Loan Account and Collateral of Borrowers as herein provided, or (ii) the Lender Group's relying on any instructions of the Administrative Borrower, except that Borrowers will have no liability to the relevant Agent-Related Person or Lender-Related Person under this <u>Section 17.13</u> with respect to any liability that has been finally determined by a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of such Agent-Related Person or Lender-Related Person, as the case may be.

17.14 Acknowledgement and Consent to Bail-In of EEA Financial Institutions. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an EEA Financial Institution; and

- (b) the effects of any Bail-in Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;

(ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or

(iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of any EEA Resolution Authority.

[Signature pages to follow.]

-124-

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed and delivered as of the date first above written.

BORROWERS:

PANDORA MEDIA, INC., a Delaware corporation

By:/s/ Naveen ChopraName:Naveen ChopraTitle:Chief Financial Officer

PANDORA MEDIA CALIFORNIA, LLC,

a California limited liability company

By: <u>/s/ Karen Walker</u> Name: Karen Walker Title: President

[SIGNATURE PAGE TO CREDIT AGREEMENT]

WELLS FARGO BANK, NATIONAL ASSOCIATION,

a national banking association, as Agent, as a Joint Lead Arranger, as a Joint Book Runner, and as a Lender

By: <u>/s/ Minna Lee</u> Name: Minna Lee Its Authorized Signatory

[SIGNATURE PAGE TO CREDIT AGREEMENT]

JP MORGAN CHASE BANK, N.A.,

as a Joint Lead Arranger, as a Joint Book Runner, and as a Lender

By:/s/ Bruce BordenName:Bruce BordenTitle:Executive Director

MORGAN STANLEY SENIOR FUNDING, INC., as a Joint Lead Arranger, as a Joint Book Runner, and as a Lender

By:/s/ Julie LilienfeldName:Julie LilienfeldTitle:Vice President

[SIGNATURE PAGE TO CREDIT AGREEMENT]

Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the following Registration Statements:

- Registration Statement (Form S-8 No. 333-175378) pertaining to the 2000 Stock Incentive Plan, 2004 Stock Plan, and 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (2) Registration Statement (Form S-8 No. 333-182212) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- Registration Statement (Form S-8 No. 333-187340) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (4) Registration Statement (Form S-8 No. 333-193612) pertaining to the 2011 Equity Incentive Plan and 2014 Employee Stock Purchase Plan of Pandora Media, Inc.,
- (5) Registration Statement (Form S-8 No. 333-202029) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (6) Registration Statement (Form S-8 No. 333-208005) pertaining to the Options to purchase stock granted under the Ticketfly, Inc. 2008 Stock Plan and restricted stock units granted under the Ticketfly, Inc. 2008 Stock Plan, and assumed by Pandora Media, Inc.,
- (7) Registration Statement (Form S-8 No. 333-208006) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (8) Registration Statement (Form S-8 No. 333-209609) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (9) Registration Statement (Form S-8 No. 333- 216087) pertaining to the 2011 Equity Incentive Plan of Pandora Media, Inc.,
- (10) Registration Statement (Form S-8 No. 333- 221287) pertaining to the 2014 Employee Stock Purchase Plan of Pandora Media, Inc., and
- (11) Registration Statement (Form S-3ASR No. 333-223224) of Pandora Media, Inc.;

of our reports dated February 26, 2018, with respect to the consolidated financial statements of Pandora Media, Inc. and the effectiveness of internal control over financial reporting of Pandora Media, Inc. included in this Annual Report (Form 10-K) of Pandora Media, Inc. for the year ended December 31, 2017.

/s/ Ernst & Young LLP

San Francisco, California February 26, 2018

Certification of Principal Executive Officer Pursuant to Section 302 of Sarbanes-Oxley Act of 2002

I, Roger Lynch, certify that:

1. I have reviewed this Annual Report on Form 10-K of Pandora Media, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report, based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 26, 2018

/s/ Roger Lynch

 Name:
 Roger Lynch

 Title:
 President and Chief Executive Officer (Principal Executive Officer) and Director

Certification of Principal Financial Officer Pursuant to Section 302 of Sarbanes-Oxley Act of 2002

I, Naveen Chopra, certify that:

1. I have reviewed this Annual Report on Form 10-K of Pandora Media, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report, based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 26, 2018

/s/ Naveen Chopra

 Name:
 Naveen Chopra

 Title:
 Chief Financial Officer (Duly Authorized Officer and Principal Financial Officer)

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER AND PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF SARBANES-OXLEY ACT OF 2002

The certification set forth below is being submitted in connection with this Annual Report on Form 10-K for theyear ended December 31, 2017 (the "Report") for the purpose of complying with Rule 13a-14(b) or Rule 15d-14(b) of the Securities Exchange Act of 1934 (the "Exchange Act") and Section 1350 of Chapter 63 of Title 18 of the United States Code.

Each of the undersigned certifies that, to his knowledge:

- 1. the Report fully complies with the requirements of Section 13(a) or 15(d) of the Exchange Act; and
- 2. the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Pandora Media, Inc.

February 26, 2018

/s/ Roger Lynch	
Name:	Roger Lynch
Title:	President and Chief Executive Officer (Principal Executive Officer) and Director
/s/ Naveer	n Chopra
Name:	Naveen Chopra

Title: Chief Financial Officer (Duly Authorized Officer and Principal Financial Officer)

This certification accompanying the Report is not deemed filed with the Securities and Exchange Commission for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, or otherwise subject to the liabilities such Section, and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before, on or after the date of the Report), irrespective of any general incorporation language contained in such filing.